Analysis of Dynamic System Identification using Rational Orthonormal Bases

Juan Carlos Gómez

A thesis submitted in partial fulfilment of the requirements for the degree of

Doctor of Philosophy

The Department of Electrical and Computer Engineering



The University of NewCastle New South Wales, 2308 Australia

February 1998

Copyright ©by Juan Carlos Gómez Department of Electrical and Computer Engineering The University of Newcastle NSW 2308 Australia I hereby certify that the work embodied in this thesis is the result of original research and has not been submitted for a higher degree to any other University or Institution.

Juan Carlos Gómez

Acknowledgments

I want to thank my Supervisor, Dr. Brett Ninness, for his support both technical and financial, during these three and a half years we have worked together. He is responsible for most of the ideas, and many of the words, presented in this thesis, and it is him the other person the word 'we' refers to when appearing in the text (the exception being when there is a mistake, in which case 'we' refers exclusively to me). He has always been very encouraging and patient with my different pace in understanding things. I also want to thank him for giving me the oportunity of doing some teaching at this University.

I want to thank Prof. Graham Goodwin and the Centre for Industrial Control Science for their financial support during the first part of my studies, and to all the Faculty and staff of the Department of Electrical and Computer Engineering.

Very special thanks go to my friends Maria and Julio. I owe to them all this Australian experience. Thanks for your friendship and good advise. Special thanks also to Carmen Vechi and Josi and Raimundo.

A big hug goes to all the postgrads and visitors who shared with me many moments at this Department, specially: Jose and Sandra, Misko, Guillermo, Li Xi, Mohammed, Hari, Turker, Woody, Thomas, Wei Zhou, Chaman, Gjerrit (the netherlander), Juan, Lin Xie, Matthew, Andrew, the swedish Bengt, Sven and Markus, and Reza.

Special thanks go for the people of the Newcastle University Choir, and the International Choir with whom I shared many musical moments.

A la gente del Departamento de Electrónica, y especialmente a mis amigos Sergio y Daniel, les agradezco la amistad y todo el entusiasmo que me dieron para emprender esta aventura. Especial agradecimiento para la gente del Coro Pablo Casals.

Finalmente, todo mi amor y agradecimiento van para mamá y papá, y mis queridas hermanas Ale y Marce, mi sobrina Macu y mi sobrino Tomi.

Contents

1	Intr	oduction		1		
2	Background					
	2.1 Signals, Systems, and related Hilbert Spaces					
	2.1.1 Signals and Systems					
		2.1.1.1	Convergence of random variables	8		
		2.1.1.2	Persistency of Excitation	9		
		2.1.1.3	State-Space Descriptions	11		
		2.1.2 Some H	Iilbert (Hardy) spaces related to signals and systems	14		
	2.2 Prediction Error Methods					
		2.2.1 Problem	m Formulation	20		
		2.2.2 Asymp	totic Analysis	22		
		2.2.2.1	Asymptotic Estimate	23		
		2.2.2.2	Asymptotic distribution of the parameter estimate	24		
		2.2.3 Compu	tational aspects	24		
3	Ortl	onormal Bases	on the Unit Circle	27		
	3.1	Introduction		27		
	3.2	Reproducing F	Cernels	28		
3.3 FIR. Laguerre and Kautz Basis		and Kautz Basis	31			
		3.3.1 FIR Ba	asis	31		
		3.3.2 Laguer:	re Basis	32		
			Basis	33		
		3.3.3 Kautz	Orthonormal Basis Generated by Inner Functions			
3.5 Orthonormal Basis with Fixed Poles			Basis Generated by Inner Functions	34		
	3.4 3.5	Orthonormal I Orthonormal I	Basis Generated by Inner FunctionsBasis with Fixed Poles	34 36		
	3.4 3.5	Orthonormal H Orthonormal H 3.5.1 A mini	Basis Generated by Inner Functions	34 36 39		
	3.4 3.5 3.6	3.3.3 Kautz Orthonormal I Orthonormal I 3.5.1 A mini Orthonormal I	Basis Generated by Inner Functions	34 36 39 43		
	3.4 3.5 3.6	Orthonormal I Orthonormal I 3.5.1 A mini Orthonormal I 3.6.1 A Mini	Basis Generated by Inner Functions \dots Basis with Fixed Poles \dots mal state-space realization for the OBFP \dots Bases on $H_2^{m \times n}(\mathbb{T})$ \dots \dots mal State-Space Realization for the MIMO-OBFP	34 36 39 43 44		
	3.4 3.5 3.6 3.7	3.3.3 Kautz Orthonormal I Orthonormal I 3.5.1 A mini Orthonormal I 3.6.1 A Mini Conclusions .	Basis Generated by Inner Functions \dots Basis with Fixed Poles \dots mal state-space realization for the OBFP \dots Bases on $H_2^{m \times n}(\mathbb{T})$ \dots mal State-Space Realization for the MIMO-OBFP \dots	34 36 39 43 44 48		
	 3.4 3.5 3.6 3.7 	3.3.3 Kautz Orthonormal I Orthonormal I 3.5.1 A mini Orthonormal I 3.6.1 A Mini Conclusions . APPENDICES	Basis Generated by Inner FunctionsBasis With Fixed PolesBasis with Fixed Polesmal state-space realization for the OBFPBases on $H_2^{m \times n}(\mathbb{T})$ Bases on State-Space Realization for the MIMO-OBFP	34 36 39 43 44 48 49		

4.1Introduction554.2Problem Formulation574.3Undermodelling Error594.4Noise Induced Error634.4.1The Fixed Denominator case664.4.2The Fixed Denominator case664.4.3Variance error using OBFP674.5Bias/Variance Trade-off714.6Conclusions72APPENDICES734.AProofs for Chapter 4734.BConvergence of Toeplitz-like Matrices764.CTechnical Lemmas795MIMO Identification using Orthonormal Bases815.1Introduction815.2Problem Formulation845.3An equivalent MIMO parameterization885.4Parameter Estimation895.5Numerical properties915.6Undermodelling Error1005.7.1Convergence of Block Toeplitz-like Matrices1015.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1095.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1115.9.3.1Deterministic Case1145.9.3.2Combined Deterministic-Stochastic Case1155.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1295.4Proofs for	4	SISO	Identification using Orthonormal Bases	55					
4.2 Problem Formulation 57 4.3 Undermodelling Error 59 4.4 Noise Induced Error 63 4.4.1 The FiR case 65 4.4.2 The Fixed Denominator case 66 4.4.3 Variance error using OBFP 67 4.5 Bias/Variance Trade-off 71 4.6 Conclusions 72 APPENDICES 73 4.A Proofs for Chapter 4 73 4.B Convergence of Toeplitz-like Matrices 76 4.C Technical Lemmas 79 5 MIMO Identification using Orthonormal Bases 81 5.1 Introduction 81 5.2 Problem Formulation 88 5.4 Parameter Estimation 89 5.5 Numerical properties 91 5.6 Undermodelling Error 96 5.7 Noise Induced Error 100 5.7.1 Convergence of Block Toeplitz-like Matrices 101 5.7.2 Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate 103 5.8		4.1	Introduction	55					
4.3Undermodelling Error594.4Noise Induced Error634.4.1The FIR case654.4.2The Fixed Denominator case664.4.3Variance error using OBFP674.5Bias/Variance Trade-off714.6Conclusions72APPENDICES734.AProofs for Chapter 4734.BConvergence of Toeplitz-like Matrices764.CTechnical Lemmas795MIMO Identification using Orthonormal Bases815.1Introduction815.2Problem Formulation845.3An equivalent MIMO parameterization885.4Parameter Estimation895.5Numerical properties915.6Undermodelling Error1005.7.1Convergence of Block Toeplitz-like Matrices1015.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1095.9.1Introduction1105.9.2Realization-based 4SID Methods1115.9.3.1Deterministic Case1155.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1295.AProofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.DProof of Convergence of Block Toepli		4.2	Problem Formulation	57					
4.4 Noise Induced Error 63 4.4.1 The FIR case 65 4.4.2 The Fixed Denominator case 66 4.4.3 Variance error using OBFP 67 4.5 Bias/Variance Trade-off 71 4.6 Conclusions 72 APPENDICES 73 4.A Proofs for Chapter 4 73 4.B Convergence of Toeplitz-like Matrices 76 4.C Technical Lemmas 79 5 MIMO Identification using Orthonormal Bases 81 5.1 Introduction 84 5.2 Problem Formulation 84 5.3 An equivalent MIMO parameterization 88 5.4 Parameter Estimation 89 5.5 Numerical properties 91 5.6 Undermodelling Error 100 5.7.1 Convergence of Block Toeplitz-like Matrices 101 5.7.2 Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate 103 5.8 Bias/Variance Trade-off 109 5.9.1 Introduction 110		4.3	Undermodelling Error	59					
4.4.1The FIR case654.4.2The Fixed Denominator case664.4.3Variance error using OBFP674.5Bias/Variance Trade-off714.6Conclusions72APPENDICES734.AProofs for Chapter 4734.BConvergence of Toeplitz-like Matrices764.CTechnical Lemmas795MIMO Identification using Orthonormal Bases815.1Introduction815.2Problem Formulation845.3An equivalent MIMO parameterization885.4Parameter Estimation895.5Numerical properties915.6Undermodelling Error965.7Noise Induced Error1005.7.1Convergence of Block Toeplitz-like Matrices1015.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1095.9.1Introduction1105.9.2Combined Deterministic Case1155.9.3.1Deterministic Case1155.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1295.4Proofs for Lemmas 5.6.1 and 5.6.21305.7Relation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.0Proof of Convergence of Block Toeplitz-like Matrices134		4.4	Noise Induced Error	63					
4.4.2The Fixed Denominator case664.4.3Variance error using OBFP674.5Bias/Variance Trade-off714.6Conclusions72APPENDICES734.AProofs for Chapter 4734.BConvergence of Toeplitz-like Matrices764.CTechnical Lemmas795MIMO Identification using Orthonormal Bases815.1Introduction815.2Problem Formulation845.3An equivalent MIMO parameterization885.4Parameter Estimation895.5Numerical properties915.6Undermodelling Error965.7Noise Induced Error1005.7.1Convergence of Block Toeplitz-like Matrices1015.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1095.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1095.9.1Introduction1115.9.3.2Combined Deterministic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1295.4Proofs for Lemmas 5.6.1 and 5.6.21305.7Broofs for Lemmas 5.6.1 and 5.6.21305.8Proof of Convergence of Block Toeplitz-like Matrices134			4.4.1 The FIR case	65					
4.4.3 Variance error using OBFP 67 4.5 Bias/Variance Trade-off 71 4.6 Conclusions 72 APPENDICES 73 4.A Proofs for Chapter 4 73 4.B Convergence of Toeplitz-like Matrices 76 4.C Technical Lemmas 79 5 MIMO Identification using Orthonormal Bases 81 5.1 Introduction 81 5.2 Problem Formulation 84 5.3 An equivalent MIMO parameterization 88 5.4 Parameter Estimation 89 5.5 Numerical properties 91 5.6 Undermodelling Error 96 5.7 Noise Induced Error 100 5.7.1 Convergence of Block Toeplitz-like Matrices 101 5.7.2 Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate 103 5.8 Bias/Variance Trade-off 109 5.9 Interdezo: Brief Overview of Subspace-Based Identification Methods 111 5.9.3.2 Combined Deterministic Case 111 5.9.3.1			4.4.2 The Fixed Denominator case	66					
4.5Bias/Variance Trade-off714.6Conclusions72APPENDICES734.AProofs for Chapter 4734.BConvergence of Toeplitz-like Matrices764.CTechnical Lemmas795MIMO Identification using Orthonormal Bases815.1Introduction815.2Problem Formulation845.3An equivalent MIMO parameterization885.4Parameter Estimation895.5Numerical properties915.6Undermodelling Error965.7Noise Induced Error1005.7.1Convergence of Block Toeplitz-like Matrices1015.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1095.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1105.9.2Combined Deterministic Case1155.9.3.1Deterministic Case1145.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1291245.12Aroofs for Lemmas 5.3.1 and 5.5.11295.3.2Froofs for Lemmas 5.3.1 and 5.6.21305.4Proof of Convergence of Block Toeplitz-like Matrices134			4.4.3 Variance error using OBFP	67					
4.6Conclusions72 APPENDICESAPPENDICES734.AProofs for Chapter 4734.BConvergence of Toeplitz-like Matrices764.CTechnical Lemmas795MIMO Identification using Orthonormal Bases815.1Introduction815.2Problem Formulation845.3An equivalent MIMO parameterization885.4Parameter Estimation895.5Numerical properties915.6Undermodelling Error965.7Noise Induced Error1005.7.1Convergence of Block Toeplitz-like Matrices1015.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1095.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1095.9.1Introduction1105.9.2Combined Deterministic-Stochastic Case1195.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES129245.12Proofs for Lemmas 5.3.1 and 5.5.11295.3Proofs for Lemmas 5.4.1 and 5.6.21305.0Proof of Convergence of Block Toeplitz-like Matrices134		4.5	Bias/Variance Trade-off	71					
APPENDICES734.AProofs for Chapter 4734.BConvergence of Toeplitz-like Matrices764.CTechnical Lemmas795MIMO Identification using Orthonormal Bases815.1Introduction815.2Problem Formulation845.3An equivalent MIMO parameterization885.4Parameter Estimation895.5Numerical properties915.6Undermodelling Error965.7Noise Induced Error1005.7.1Convergence of Block Toeplitz-like Matrices1015.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1095.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1115.9.2Realization-based 4SID Methods1115.9.3.1Deterministic Case1155.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1295.4Proofs for Lemmas 5.3.1 and 5.5.11295.8Proofs for Lemmas 5.3.1 and 5.5.11295.9Proofs for Lemmas 5.4.1 and 5.6.21305.0Proof of Convergence of Block Toeplitz-like Matrices134		4.6	Conclusions	72					
4.AProofs for Chapter 4734.BConvergence of Toeplitz-like Matrices764.CTechnical Lemmas795MIMO Identification using Orthonormal Bases815.1Introduction815.2Problem Formulation845.3An equivalent MIMO parameterization885.4Parameter Estimation895.5Numerical properties915.6Undermodelling Error965.7Noise Induced Error1005.7.1Convergence of Block Toeplitz-like Matrices1015.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1095.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1115.9.3Direct 4SID Algorithms1145.9.3.1Deterministic Case1155.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1291295.4Proofs for Lemmas 5.3.1 and 5.5.11295.5Proofs for Lemmas 5.4.1 and 5.6.21305.6Relation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.0Proof of Convergence of Block Toeplitz-like Matrices134			Appendices	73					
4.BConvergence of Toeplitz-like Matrices764.CTechnical Lemmas795MIMO Identification using Orthonormal Bases815.1Introduction815.2Problem Formulation845.3An equivalent MIMO parameterization885.4Parameter Estimation895.5Numerical properties915.6Undermodelling Error965.7Noise Induced Error1005.7.1Convergence of Block Toeplitz-like Matrices1015.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1095.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1115.9.3Direct 4SID Algorithms1145.9.3.1Deterministic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1291295.4Proofs for Lemmas 5.3.1 and 5.5.11295.5Proofs for Lemmas 5.6.1 and 5.6.21305.0Proof of Convergence of Block Toeplitz-like Matrices134		4.A	Proofs for Chapter 4	73					
4.C Technical Lemmas 79 5 MIMO Identification using Orthonormal Bases 81 5.1 Introduction 81 5.2 Problem Formulation 84 5.3 An equivalent MIMO parameterization 84 5.4 Parameter Estimation 89 5.5 Numerical properties 91 5.6 Undermodelling Error 96 5.7 Noise Induced Error 100 5.7.1 Convergence of Block Toeplitz-like Matrices 101 5.7.2 Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate 103 5.8 Bias/Variance Trade-off 109 5.9 Intermezzo: Brief Overview of Subspace-Based Identification Methods 101 5.9.2 Realization-based 4SID Methods 111 5.9.3.1 Deterministic Case 119 5.10 Simulation Examples 124 5.11 Conclusions 128 APPENDICES 129 128 5.10 Simulation Examples 124 5.11 Conclusions 128 APPENDICES 12		4.B	Convergence of Toeplitz-like Matrices	76					
5 MIMO Identification using Orthonormal Bases 81 5.1 Introduction 81 5.2 Problem Formulation 84 5.3 An equivalent MIMO parameterization 88 5.4 Parameter Estimation 89 5.5 Numerical properties 91 5.6 Undermodelling Error 96 5.7 Noise Induced Error 100 5.7.1 Convergence of Block Toeplitz-like Matrices 101 5.7.2 Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate 103 5.8 Bias/Variance Trade-off 109 5.9 Intermezzo: Brief Overview of Subspace-Based Identification Methods 109 5.9.1 Introduction 110 5.9.2 Realization-based 4SID Methods 111 5.9.3.1 Deterministic Case 119 5.10 Simulation Examples 124 5.11 Conclusions 128 APPENDICES 129 5.A Proofs for Lemmas 5.3.1 and 5.5.1 129 5.B Proofs for Lemmas 5.6.1 and 5.6.2 130 5.C		4.C	Technical Lemmas	79					
5 MIMO Identification using Orthonormal Bases 81 5.1 Introduction 81 5.2 Problem Formulation 84 5.3 An equivalent MIMO parameterization 84 5.4 Parameter Estimation 89 5.5 Numerical properties 91 5.6 Undermodelling Error 96 5.7 Noise Induced Error 100 5.7.1 Convergence of Block Toeplitz-like Matrices 101 5.7.2 Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate 103 5.8 Bias/Variance Trade-off 109 5.9.1 Intermezzo: Brief Overview of Subspace-Based Identification Methods 109 5.9.1 Introduction 110 5.9.2 Realization-based 4SID Methods 111 5.9.3.1 Deterministic Case 119 5.10 Simulation Examples 124 5.11 Conclusions 128 APPENDICES 129 5.A Proofs for Lemmas 5.3.1 and 5.5.1 129 5.B Proofs for Lemmas 5.6.1 and 5.6.2 130 5.C									
5.1Introduction815.2Problem Formulation845.3An equivalent MIMO parameterization885.4Parameter Estimation895.5Numerical properties915.6Undermodelling Error965.7Noise Induced Error1005.7.1Convergence of Block Toeplitz-like Matrices1015.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1095.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1095.9.1Introduction1105.9.2Realization-based 4SID Methods1115.9.3.1Deterministic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1291295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.DProof of Convergence of Block Toeplitz-like Matrices134	5	MIM	IO Identification using Orthonormal Bases	81					
5.2Problem Formulation845.3An equivalent MIMO parameterization885.4Parameter Estimation895.5Numerical properties915.6Undermodelling Error965.7Noise Induced Error1005.7.1Convergence of Block Toeplitz-like Matrices1015.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1095.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1095.9.1Introduction1105.9.2Realization-based 4SID Methods1115.9.3Direct 4SID Algorithms1145.9.3.1Deterministic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1291295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.DProof of Convergence of Block Toeplitz-like Matrices134		5.1	Introduction	81					
5.3An equivalent MIMO parameterization885.4Parameter Estimation895.5Numerical properties915.6Undermodelling Error965.7Noise Induced Error1005.7.1Convergence of Block Toeplitz-like Matrices1015.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1095.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1095.9.1Introduction1105.9.2Realization-based 4SID Methods1115.9.3.1Deterministic Case1155.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions1295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.DProof of Convergence of Block Toeplitz-like Matrices134		5.2	Problem Formulation	84					
5.4Parameter Estimation895.5Numerical properties915.6Undermodelling Error965.7Noise Induced Error1005.7.1Convergence of Block Toeplitz-like Matrices1015.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1095.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1005.9.2Realization-based 4SID Methods1115.9.3Direct 4SID Algorithms1145.9.3.1Deterministic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1291295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.DProof of Convergence of Block Toeplitz-like Matrices134		5.3	An equivalent MIMO parameterization	88					
5.5Numerical properties915.6Undermodelling Error965.7Noise Induced Error1005.7.1Convergence of Block Toeplitz-like Matrices1015.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1095.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1095.9.1Introduction1105.9.2Realization-based 4SID Methods1115.9.3Direct 4SID Algorithms1145.9.3.1Deterministic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1291295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.DProof of Convergence of Block Toeplitz-like Matrices134		5.4	Parameter Estimation	89					
5.6Undermodelling Error965.7Noise Induced Error1005.7.1Convergence of Block Toeplitz-like Matrices1015.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1095.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1095.9.1Introduction1105.9.2Realization-based 4SID Methods1115.9.3Direct 4SID Algorithms1145.9.3.1Deterministic Case1155.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions1295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.DProof of Convergence of Block Toeplitz-like Matrices134		5.5	Numerical properties	91					
5.7Noise Induced Error1005.7.1Convergence of Block Toeplitz-like Matrices1015.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1095.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1095.9.1Introduction1105.9.2Realization-based 4SID Methods1115.9.3Direct 4SID Algorithms1145.9.3.1Deterministic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1291295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.DProof of Convergence of Block Toeplitz-like Matrices134		5.6	Undermodelling Error						
5.7.1Convergence of Block Toeplitz-like Matrices1015.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1035.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1095.9.1Introduction1105.9.2Realization-based 4SID Methods1115.9.3Direct 4SID Algorithms1145.9.3.1Deterministic Case1155.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions1295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 134		5.7	Noise Induced Error	100					
5.7.2Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate1035.8Bias/Variance Trade-off1095.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1095.9.1Introduction1105.9.2Realization-based 4SID Methods1115.9.3Direct 4SID Algorithms1145.9.3.1Deterministic Case1155.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1291295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.DProof of Convergence of Block Toeplitz-like Matrices134			5.7.1 Convergence of Block Toeplitz-like Matrices	101					
trix Estimate1035.8Bias/Variance Trade-off1095.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1095.9.1Introduction1105.9.2Realization-based 4SID Methods1115.9.3Direct 4SID Algorithms1145.9.3.1Deterministic Case1155.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1295.8Proofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.DProof of Convergence of Block Toeplitz-like Matrices134			5.7.2 Asymptotic Distribution of the (Vectorized) Tranfer Ma-						
5.8Bias/Variance Trade-off1095.9Intermezzo: Brief Overview of Subspace-Based Identification Methods1095.9.1Introduction1105.9.2Realization-based 4SID Methods1115.9.3Direct 4SID Algorithms1145.9.3.1Deterministic Case1155.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.DProof of Convergence of Block Toeplitz-like Matrices134			$trix Estimate \ . \ . \ . \ . \ . \ . \ . \ . \ . \ $	103					
5.9Intermezzo: Brief Overview of Subspace-Based Identification Methodsthods1095.9.1Introduction1105.9.2Realization-based 4SID Methods1115.9.3Direct 4SID Algorithms1145.9.3.1Deterministic Case1155.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.DProof of Convergence of Block Toeplitz-like Matrices134		5.8	Bias/Variance Trade-off	109					
thods1095.9.1Introduction1105.9.2Realization-based 4SID Methods1115.9.3Direct 4SID Algorithms1145.9.3.1Deterministic Case1155.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.DProof of Convergence of Block Toeplitz-like Matrices134		5.9	Intermezzo: Brief Overview of Subspace-Based Identification Me-						
5.9.1Introduction1105.9.2Realization-based 4SID Methods1115.9.3Direct 4SID Algorithms1145.9.3.1Deterministic Case1155.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.DProof of Convergence of Block Toeplitz-like Matrices134			thods	109					
5.9.2Realization-based 4SID Methods1115.9.3Direct 4SID Algorithms1145.9.3.1Deterministic Case1155.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.DProof of Convergence of Block Toeplitz-like Matrices134			5.9.1 Introduction	110					
5.9.3Direct 4SID Algorithms1145.9.3.1Deterministic Case1155.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.DProof of Convergence of Block Toeplitz-like Matrices134			5.9.2 Realization-based 4SID Methods	111					
5.9.3.1Deterministic Case1155.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.DProof of Convergence of Block Toeplitz-like Matrices134			5.9.3 Direct 4SID Algorithms	114					
5.9.3.2Combined Deterministic-Stochastic Case1195.10Simulation Examples1245.11Conclusions128APPENDICES1295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.DProof of Convergence of Block Toeplitz-like Matrices134			5.9.3.1 Deterministic Case	115					
5.10Simulation Examples1245.11Conclusions128APPENDICES1295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.DProof of Convergence of Block Toeplitz-like Matrices134			5.9.3.2 Combined Deterministic-Stochastic Case	119					
5.11Conclusions128APPENDICES1295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.DProof of Convergence of Block Toeplitz-like Matrices134		5.10	Simulation Examples	124					
APPENDICES1295.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.DProof of Convergence of Block Toeplitz-like Matrices134		5.11	1 Conclusions						
5.AProofs for Lemmas 5.3.1 and 5.5.11295.BProofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.DProof of Convergence of Block Toeplitz-like Matrices134			Appendices	129					
5.BProofs for Lemmas 5.6.1 and 5.6.21305.CRelation between the estimates $\hat{\eta}$ and $\hat{\Theta}$ 1325.DProof of Convergence of Block Toeplitz-like Matrices134		5.A	Proofs for Lemmas 5.3.1 and 5.5.1	129					
5.C Relation between the estimates $\hat{\eta}$ and $\widehat{\Theta}$		5.B	Proofs for Lemmas 5.6.1 and 5.6.2	130					
5.D Proof of Convergence of Block Toeplitz-like Matrices 134		5.C	Relation between the estimates \hat{n} and $\hat{\Theta}$ 132						
0		5.D	Proof of Convergence of Block Toeplitz-like Matrices						
5.E Technical Lemmas		5.E	Technical Lemmas	137					

6	Adap	tive Tracking Performance A	Analysis	using	Orthonormal	Bases	147				
	6.1	Introduction					147				
	6.2	Motivation					149				
	6.3	Problem Formulation					151				
	6.4	Model Structures					156				
	6.5	Transient analysis					159				
	6.6	Steady State Analysis					162				
	6.7	Simulation Examples					167				
	6.8	Conclusions					172				
		Appendices					173				
	6.A	Proof for Chapter 6					173				
	6.B	Technical Results					177				
7	Conclusions										
A	Kron	ecker Product and vec-Oper	rator				183				
В	3 Notation										
	Bibliography										

Abstract

This thesis illustrates the use of a particular class of rational orthonormal bases for the purposes of analyzing the performance of least-squares dynamic system estimates that involve model structures which are linear in the parameters. The genesis of this work comes from seminal results on approximating variance error of estimated frequency responses, and work on the use of restricted classes of orthonormal bases (Laguerre and two-parameter Kautz for example) as a model structure parameterization option. A key original perspective of this thesis is that by generalizing the bases involved to the case of arbitrary pole locations, these bases can be viewed as more than an implementational option, but also as an analysis tool of great utility, since it can be applied regardless of whether the bases are used for model structure parameterization or not. This utility is illustrated by deriving approximations of estimate variability that are extensions of pre-existing ones, in that for scenarios where poles are not all fixed at the origin, they can provide improved accuracy. The key tools in this analysis involve the development of new results of generalized Fourier series convergence and generalization of the asymptotic properties of Toeplitz matrices. Initially these results are derived for time invariant and single-input, single-output scenarios, but subsequently extensions to multipleinput, multiple-output and time varying situations are also provided.

1

Introduction

This thesis deals with the use of rational orthonormal bases in identification of discrete-time linear systems from input-output data.

System Identification can be defined as the area of System Theory that deals with the study of methodologies for constructing a mathematical model of a dynamical system based on measurements from the system. The area has become an important tool in many branches of engineering (and in other fields such as biology and economy) that rely on accurate models of the systems for the purposes of analysis and design. In the Control and Signal Processing Areas, mathematical models are important for several reasons. Among them we can mention the following:

- Most of control design methodologies are based on the assumption that a model for the system is available.
- For the purposes of simulation and prediction, a parameterized model of the system is required.
- Mathematical models are required for the software and hardware implementation of digital filters.

A typical identification procedure can be summarized as follows:

- Design of the experiments to be performed on the system, and the way the data has to be collected.
- Selection of the model set. This is usually done based on physical laws, which allows the incorporation of prior information, but can also be done in a *black-box* fashion [Lju87]. Here the parameters to be estimated do not have necessarily a physical meaning.
- Selection of the model in the set that best match with the measured data.

• Model validation. This usually implies testing the model quality with a different set of data.

For the case of time invariant systems (that is, systems whose parameters do not change in time) the identification can be performed *off-line*, which means that the parameter estimation is performed once a whole set of data is collected from the system. Most of this thesis will deal with time-invariant systems, the exception being Chapter 6 where time-varying systems are considered. In these last cases the identification can not be performed off-line but has to be carried out *on-line* or *recursively*, which means that the parameter estimates are updated each time a new data is collected from the system.

A unified perspective of System Identification Theory is given in the textbooks [Lju87, SS89].

In this thesis, systems are modeled using rational orthonormal basis functions with fixed poles. These basis functions are generalizations of the well known trigonometric bases of classical Fourier analysis, but with arbitrary poles not restricted to be all at the origin. The transfer function G(z) describing the input-output properties of the system is represented as a series expansion in terms of the basis functions. This leads to a model which is linear in the unknown parameters. The estimation can then be performed in the framework of Prediction Error Methods [Lju87] that for this case provide estimates in closed form.

A dominating theme of this thesis will be the analysis of the accuracy of obtained estimates. For systems modeled with arbitrary fixed denominator structures, the quantification of the estimation error, consisting of both bias error and variance error, has proven to be a very difficult problem. In this thesis it is shown that the analysis can be made more tractable by re-parameterizing the system using orthonormal model structures with the same fixed poles. In consideration of this, orthonormal structures are also viewed in this thesis as an analysis tool that can be used to facilitate the study of estimation accuracy for a class of models. In fact, this perspective that orthonormal bases are equally (perhaps more) useful as an analysis tool, than as an implementational option in the formation of a model structure, is one of the original contributions. Expanding on this a little more, work on orthonormal bases pre-dating that of this thesis (which will be reviewed in the sequel) concentrated solely on the utility of orthonormal bases as an implementational tool for parameterising a fixed denominator model structure. In this thesis we establish a different perspective, that regardless of how the fixed denominator model is implemented, there is great utility for analysis purposes of re-parameterising the structure in orthonormal form, whether or not it was actually implemented in this form. Key to this strategy is that by virtue of linearity, the estimated frequency response is *invariant* to this parameterisation, so the most convenient one may as well be chosen for analysis purposes.

The quantification of the estimation accuracy for the proposed identification schemes is performed by establishing new results on the convergence of (block) Toeplitz-like matrices. These matrices appear in the computation of the component of the frequency response estimation error that is induced by the measurement noise. The Toeplitz-like convergence results allow the generalization to the arbitrary fixed denominator case, of corresponding results that exist in the literature specific to FIR (Finite Impulse Response) model structures.

A second theme we want to emphasize in this thesis is the fact that, in contradiction to what one would expect by appealing to existent frequency domain results on the variability of the estimates for FIR structures [LY85] and on linear filtering, the variability of the estimates with fixed denominator model structures is affected by the location of the poles in the model. The way the variability is affected by the pole locations is also an important issue studied in this thesis.

We now present a brief overview of the rest of the thesis.

Chapter 2: In this chapter, most of our notation is introduced. The classes of signals and systems this thesis deal with, and the associated Hilbert spaces in which these signals and systems can be embedded are presented. A succinct description of the identification methods employed in this thesis, viz. Prediction Error Methods, is also given.

Chapter 3: The construction of several families of (scalar) rational orthonormal bases on the unit circle is reviewed in this chapter. The concept of reproducing kernel associated with the bases is introduced, and its importance for the analysis of the approximating properties of the bases is pointed out. We focus on a particular family of orthonormal bases, namely the Orthonormal Bases with Fixed Poles, which will be used in the context of system identification in the following chapters. For this family, a closed form expression for the associated reproducing kernel is derived, and a minimal state-space realization is presented. In addition, it is shown how families of matrix orthonormal bases for the space $H_2^{m \times n}(\mathbb{T})$ can be constructed from orthonormal bases on $H_2(\mathbb{T})$.

Chapter 4: This chapter deals with the Least Squares identification (from input-output data) of Discrete-Time (DT), Linear Time-Invariant (LTI), Single-Input Single-Output (SISO) systems represented using Orthonormal Bases with Fixed Poles. The accuracy of the estimation is analyzed and the two components of the estimation error (namely, the bias and variance errors) are quantified. The derivation of new results on convergence of some Toeplitz-like matrices will prove to be fundamental for the analysis in this chapter.

Chapter 5: In this chapter the idea of using orthonormal bases and PEM (least squares techniques) for the identification of discrete-time linear time-invariant systems is extended from the SISO to the MIMO (Multiple-Input Multiple-Output) setting. It will be shown how the rational orthonormal bases with fixed poles introduced in Section 3.5 (or the corresponding MIMO bases generated as in Section 3.6) can be used to linearly parameterize any multivariable fixed denominator model structure. As in the SISO case, the accuracy of the estimation is quantified by deriving expressions for an upper bound on the undermodelling error and for the asymptotic (in data-length and model order) covariance of the transfer matrix estimate (noise induced error). The obtained results generalize to the MIMO case and to general orthonormal bases with fixed poles the preexisting FIR results in [Lju85, YL84]. Fundamental in this derivation are some new convergence properties of generalized block Toeplitz-like matrices. Subspace Identification methods are briefly reviewed for the purposes of comparison with the orthonormal basis-based methods proposed in this thesis.

Chapter 6: In this chapter, a frequency domain analysis of the tracking performance of several adaptive algorithms for the recursive identification of time-varying linear systems is carried out for the case in which the system is represented by a fixed denominator model structure. The focus is on the study of the trade-off between disturbance rejection and tracking ability, and how these properties are influenced by input and noise spectral densities, step size of the adaptive algorithms, and the choice of the fixed pole locations in the model structure.

Chapter 7: In this chapter we summarized the contributions of this thesis, and present some concluding remarks.

2

Background

In this chapter, some elementary definitions regarding the classes of signals and systems this thesis deals with, and the related Hilbert spaces in which these signals and systems can be embedded are given. We also give a succinct description of Prediction Error Methods for System Identification, since these are the identification methods we will employ throughout the thesis.

2.1 Signals, Systems, and related Hilbert Spaces

2.1.1 Signals and Systems

To begin with, some standard material concerning the representation of signals and systems will be presented. A more detailed and complete treatment of these topics can be found (for example) in the books [Lju87, SS89].

Most of this thesis will deal with discrete-time (DT), linear time-invariant (LTI), multiple-input-multiple-output (MIMO) systems whose input-output relationship can be described by

$$y_k = \sum_{\ell=0}^{\infty} g(\ell) u_{k-\ell} + \nu_k = G(q) u_k + \nu_k,$$
(2.1)

where $y_k \in \mathbb{R}^m$, $u_k \in \mathbb{R}^n$, and $\nu_k \in \mathbb{R}^m$ denote respectively the vectors of output, input, and disturbance (or measurement noise) signals at time k, where $\{g(\ell)\}$ are the so-called $(m \times n)$ 'Markov parameters', and where

$$G(q) \triangleq \sum_{\ell=0}^{\infty} g(\ell) q^{-\ell}$$

denotes the $(m \times n)$ input to output transfer matrix operator (provided the

infinite sum exists), with q standing for the 'forward shift operator' defined as¹

$$q \ u_k \triangleq u_{k+1}$$

for any sequence $\{u_k\}$, and for all k.

Sometimes, a z-domain representation of system (2.1) will be preferred. This complex-plane representation can be obtained by taking the Z-transform on both sides of equation (2.1). The Z-transform of a signal $\{x_k\}$ is defined as

$$X(z) \triangleq \sum_{k=-\infty}^{\infty} x_k z^{-k}.$$
 (2.2)

Assuming that the system is initially relaxed, the representation (2.1) can be written in the z-domain as

$$Y(z) = G(z)U(z) + V(z),$$

where G(z) is the transfer matrix of the system, and Y(z), U(z) and V(z) are the Z-transforms of the sequences $\{y_k\}, \{u_k\}$ and $\{\nu_k\}$ respectively. The definition of the Z-transform in (2.2) requires that the signals be specified for the entire time range $-\infty < k < \infty$. In practical problems the signals are known for $k \ge 0$, but by no means are zero for k < 0. It is then useful to define the 'one-sided' or 'unilateral' Z-transform as

Definition 2.1.1. [OS89, PM92] The one-sided or unilateral Z-transform of a signal $\{x_k\}$ is defined by

$$X(z) \triangleq \sum_{k=0}^{\infty} x_k z^{-k}$$

 \diamond

It will be assumed that the input sequence $\{u_k\}$ is a 'quasi-stationary' (*n*-dimensional) process, which means that the following limits defining the mean (m_u) and covariance $(R_u(\tau))$ functions exist [Lju87]

$$m_u \triangleq \lim_{N \to \infty} \frac{1}{N} \sum_{k=1}^{N} \mathbf{E} \{u_k\},$$

$$R_u(\tau) \triangleq \lim_{N \to \infty} \frac{1}{N} \sum_{k=1}^{N} \mathbf{E} \{(u_k - m_u)(u_{k-\tau} - m_u)^T\},$$

¹The 'backward shift operator' is defined similarly as

$$q^{-1} u_k \triangleq u_{k-1}$$

where $E\{\cdot\}$ stands for the statistical expectation operator; the expectation being over the probability space that any random components are defined on. When deterministic inputs are considered, the expectation operator can be omitted in the previous definitions, and of course, in these situations it makes no difference to retain it. We will sometimes denote the operator

$$\lim_{N \to \infty} \frac{1}{N} \sum_{k=1}^{N} \mathsf{E}\left\{\cdot\right\}$$

with the more compact notation $\overline{\mathbf{E}} \{\cdot\}$ introduced by Ljung in [Lju87].

The disturbance sequence $\{\nu_k\}$ will usually be assumed to be a stationary² (*m*-dimensional) stochastic process with mean and covariance functions defined as

$$\begin{array}{rcl} m_{\nu} & \triangleq & \mathbf{\mathsf{E}} \left\{ \nu_{k} \right\}, \\ R_{\nu}(\tau) & \triangleq & \mathbf{\mathsf{E}} \left\{ (\nu_{k} - m_{\nu}) (\nu_{k-\tau} - m_{\nu})^{T} \right\}, \end{array}$$

respectively. At times, we will also assume that the disturbance $\{\nu_k\}$ is statistically independent of the input $\{u_k\}$.

Associated with a stochastic process $\{u_k\}$ (either stationary or quasi-stationary) is the '(Power) Spectral Density Function' defined as the Discrete Fourier Transform of the covariance function

$$\Phi_u(\omega) \triangleq \sum_{\tau=-\infty}^{\infty} R_u(\tau) e^{-j\tau\omega}.$$

The covariance function can then be recovered from the spectral density by Inverse Discrete Fourier transformation:

$$R_u(\tau) = \frac{1}{2\pi} \int_{-\pi}^{\pi} \Phi_u(\omega) e^{j\tau\omega} d\omega$$

In the case of a quasi-stationary process $\{u_k\}$, the last expression leads to

$$\overline{\mathbf{E}}\left\{u_{k}u_{k}^{T}\right\} = \frac{1}{2\pi} \int_{-\pi}^{\pi} \Phi_{u}(\omega) \mathrm{d}\omega, \qquad (2.3)$$

which is a form of *Parseval's identity* or *Parseval's Theorem* (see next subsection). We will frequently use this identity in the derivation of frequency domain expressions of estimation accuracy.

²By 'stationary' we actually mean 'weakly stationary'. A stochastic process $\{\nu_k\}$ is said to be weakly stationary if [Doo53, Ros85]

Sometimes we will be interested in the statistics of the output y_k of a linear system G(q) as a function of the statistics of the input u_k , in the absence of noise. In this case we have a linear filtering of the input u_k

$$y_k = G(q) \ u_k$$

where the filter is the transfer matrix operator G(q). As already mentioned, assuming that the filter (the system) is stable and causal, it can be represented as

$$G(q) = \sum_{\ell=0}^{\infty} g(\ell) \ q^{-\ell}.$$

Stability of G(q) implies that $||g(\ell)|| \to 0$ as $\ell \to \infty$, where $|| \cdot ||$ denotes a matrix norm (for instance the 2-norm). Causality means that the impulse response sequence $\{g(\ell)\}$ is one-sided, which implies that the output at time k doesn't depend on future values of the input u_{k+1}, u_{k+2}, \cdots . In this situation it is not difficult to prove (see [Lju87, SS89]) that the mean value of y_k is given by

$$m_y \triangleq \mathbf{E} \{y_k\} = \sum_{\ell=0}^{\infty} g(\ell) \ m_u = G(1) \ m_u,$$

and that the deviations from the mean value of inputs and outputs are related according to

$$y_k - m_y = G(q) \ (u_k - m_u).$$

Furthermore, the spectral density of the output signal is given by

$$\Phi_y(\omega) = G(e^{j\omega}) \ \Phi_u(\omega) \ G^{\star}(e^{j\omega}),$$

where $(\cdot)^*$ denotes the complex conjugate transposed.

2.1.1.1 Convergence of random variables

At several points in this thesis, different concepts of convergence of random variables will be used. These concepts are summarized in the following definition (see [Doo53, Pap84] for a more detailed treatment of these topics).

Definition 2.1.2. Let $\{x_k\}$ be an indexed sequence of random vectors, and let x_* be a random vector of the same dimension. Then

• x_k is said to converge to x_\star with probability 1, or almost surely, and denoted

$$x_k \xrightarrow{\text{a.s.}} x_\star, \qquad \text{as} \qquad k \to \infty,$$

if $P \{x_k \to x_\star\} = 1$. Here P is the underlying measure on the probability space $\{\Omega, \mathcal{F}, P\}$, that the random variables $\{x_k\}$ are defined on.

• x_k is said to converge to x_{\star} in probability, denoted

$$x_k \xrightarrow{\rho} x_\star, \quad \text{as} \quad k \to \infty$$

if for any $\epsilon > 0$,

$$P\{|x_k - x_\star| > \epsilon\} \to 0, \quad \text{as} \quad k \to \infty,$$

where $|\cdot|$ denotes the Euclidean norm of a vector.

• x_k is said to weakly converge to x_{\star} if the distribution function $F_{x_k}(x)$ of x_k converges to that of x_{\star} , say $F_{x_{\star}}(x)$, as $k \to \infty$. We also say that x_k converges in distribution to x_{\star} , and we denote

$$x_k \xrightarrow{\text{dist}} x_\star, \quad \text{as} \quad k \to \infty.$$

• x_k is said to converge to x_{\star} in the mean square sense if

$$\mathsf{E}\left\{|x_k - x_\star|^2\right\} \to 0, \quad \text{as} \quad k \to \infty.$$

It can be proved that a.s. convergence implies convergence in probability, which in turn implies weak convergence; also that convergence in mean square sense implies convergence in probability [Doo53, Pap84].

2.1.1.2 Persistency of Excitation

In Chapter 6, the convergence properties of adaptive identification algorithms are analyzed. A fundamental concept in this analysis is that of 'persistency of excitation' of a signal. This property is required of the input signal to ensure the stability of the algorithms. Persistency of excitation essentially means that the signal is rich enough to excite all the modes of the system. A more precise meaning of this concept is given in the following definition (see for instance [And82, GS84, SS89]).

Definition 2.1.3. The sequence $\{u_k\}$ (with $u_k \in \mathbb{R}^n$) is said to be 'persistently exciting' of order M if there exists some integer p, and positive constants $\alpha > 0$ and $\beta > 0$ such that³

$$0 < \alpha I < \sum_{k=\ell}^{\ell+p} \varphi_k \varphi_k^T < \beta I < \infty, \qquad \text{for all } \ell,$$
(2.4)

³Given a symmetric matrix R, the notation R > 0 means that R is positive definite, i.e.

$$x^T R x > 0,$$
 for all $x \neq 0,$

where x is a column vector of compatible dimension. Analogously, given R_1 and R_2 symmetric matrices, the expression $R_1 > R_2$ (respectively, $R_1 \ge R_2$) indicates that $R_1 - R_2$ is positive definite (respectively, positive semidefinite).

where I is the identity matrix (of dimensions $nM \times nM$ in this case), and the vector φ_k is defined as

$$\varphi_k \triangleq [u_{k-1}^T, u_{k-2}^T, \cdots, u_{k-M}^T]^T.$$

In the case that $\{u_k\}$ is a stationary stochastic process, condition (2.4) can be simplified to

$$\lim_{N \to \infty} \frac{1}{N} \sum_{k=1}^{N} \varphi_k \varphi_k^T > 0.$$

If in addition $\{u_k\}$ is ergodic⁴, the time average operator

$$\lim_{N \to \infty} \frac{1}{N} \sum_{k=1}^{N} (\cdot)$$

can be substituted by the ensemble average operator $E \{\cdot\}$, so that the condition for the signal $\{u_k\}$ to be persistently exciting of order M becomes

$$\mathbf{E}\left\{\varphi_k\varphi_k^T\right\} > 0.$$

Sometimes, a frequency domain interpretation of this concept will be more useful. In the frequency domain, persistent excitation of a signal requires the positive definiteness of its spectral density matrix for a number of frequencies. The precise statement is given in the following proposition (see for instance [SS89]).

Proposition 2.1.1. A stationary, ergodic, stochastic process $\{u_k\}$ is persistently exciting of order M provided its spectral density matrix $\Phi_u(\omega)$ is positive definite for at least M distinct values of ω in the interval $(-\pi, \pi)$.

$$\frac{1}{N} \sum_{k=1}^{N} u_k \quad \xrightarrow{\text{a.s.}} \quad \mathbf{E} \{ u_k \}$$
$$\frac{1}{N} \sum_{k=1}^{N} u_{k+\tau} u_k \quad \xrightarrow{\text{a.s.}} \quad \mathbf{E} \{ u_{k+\tau} u_k \}$$

 $^{^4 \}mathrm{A}$ stationary stochastic process $\{u_k\}$ is said to be ergodic up to second order statistics if [Ros85]

2.1.1.3 State-Space Descriptions

At times in this thesis we will consider internal representation of systems in the form of state-space realizations of the form

$$x_{k+1} = Ax_k + Bu_k + w_k, (2.5)$$

$$y_k = Cx_k + Du_k + v_k, (2.6)$$

$$\mathbf{E}\left\{\begin{bmatrix}w_k\\v_k\end{bmatrix}\begin{bmatrix}w_s^T & v_s^T\end{bmatrix}\right\} = \begin{bmatrix}Q & S\\S^T & R\end{bmatrix}\delta_{ks}$$
(2.7)

where $x_k \in \mathbb{R}^{n_x}$, $y_k \in \mathbb{R}^m$, $u_k \in \mathbb{R}^n$, are respectively the state, the output and the input vectors at time k, and $A \in \mathbb{R}^{n_x \times n_x}$, $B \in \mathbb{R}^{n_x \times n}$, $C \in \mathbb{R}^{m \times n_x}$, and $D \in \mathbb{R}^{m \times n}$ are the state-feedback, the input, the output, and the input feedthrough matrices respectively [Kai80]. The vectors $w_k \in \mathbb{R}^{n_x}$, and $v_k \in \mathbb{R}^m$, represent the process and output measurement noise vectors at time k respectively, while the matrices $Q \in \mathbb{R}^{n_x \times n_x}$, $S \in \mathbb{R}^{n_x \times m}$, and $R \in \mathbb{R}^{m \times m}$ are the covariance matrices of these noise sequences. In most of the cases it will be assumed that w_k and v_k are zero mean, stationary, white noise vector sequences.

For obvious reasons, equations (2.5) and (2.6) are called *state equation* and *output equation*, respectively.

For deterministic systems (where both the process noise and the output measurement noise are identically zero), the state space description (2.5)-(2.7) reduces to

$$x_{k+1} = Ax_k + Bu_k, (2.8)$$

$$y_k = Cx_k + Du_k. (2.9)$$

An $m \times n$ rational transfer matrix G(z) has an n_x -dimensional state-space realization (A, B, C, D), with $A \in \mathbb{R}^{n_x \times n_x}$, $B \in \mathbb{R}^{n_x \times n}$, $C \in \mathbb{R}^{m \times n_x}$, and $D \in \mathbb{R}^{m \times n}$, if

$$G(z) = C(zI - A)^{-1}B + D.$$

The above expression can be obtained from equations (2.8) and (2.9) by taking the Z-transform of both equations and eliminating the state variable.

The observability and controllability matrices associated with the system (2.8)-(2.9) are defined as

$$\mathcal{O} \triangleq [C^T, (CA)^T, \cdots, (CA^{n_x-1})^T]^T, \\ \mathcal{C} \triangleq [B, AB, \cdots, A^{n_x-1}B],$$

respectively. In a similar way the *extended* $(i, j > n_x)$ observability \mathcal{O}_i and controllability \mathcal{C}_j matrices are defined as

$$\mathcal{O}_i \triangleq [C^T, (CA)^T, \cdots, (CA^{i-1})^T]^T, \qquad (2.10)$$

$$\mathcal{C}_j \triangleq [B, AB, \cdots, A^{j-1}B]. \tag{2.11}$$

Definition 2.1.4. Controllability: [Kai80] The pair (A, B) is said to be *controllable* if there exists an input sequence $\{u_k\}$ that takes the state of the system from any initial state $x_0 = x^0$ to any desired final state $x_k = x^f$ in a finite number of steps (k is finite). This concept should more properly be called *Controllability from the origin* or *Reachability* [Kai80], but with some abuse of terminology we will still refer to it as *Controllability*. A necessary and sufficient condition for the controllability of the pair (A, B) is that the controllability matrix C has full rank n_x .

Definition 2.1.5. Observability: [Kai80] The pair (C, A) is said to be observable if for any initial state $x_0 = x^0$ there is a finite number of steps ℓ such that x_0 can be determined from the sequences u_k and y_k for $0 \le k \le \ell$. A necessary and sufficient condition for the observability of the pair (C, A) is that the observability matrix \mathcal{O} has full rank n_x .

Definition 2.1.6. Minimal Realization: A realization is said to be minimal if it has minimal dimension, that is if there exists no other realization with lower dimension. The dimension of a minimal realization is called the McMillan degree of the MIMO system. A necessary and sufficient condition for the minimality of a realization is that the pair (A, B) is controllable and the pair (C, A) is observable [Kai80]. The minimality of the realization (A, B, C, D) then implies that the matrices \mathcal{O}_i , \mathcal{C}_j (with $i, j > n_x$), \mathcal{O} and \mathcal{C} have all rank n_x [Kai80]. \diamondsuit

The realization (A, B, C, D) in (2.8)-(2.9) uniquely defines the input-output properties of the system via

$$y_k = \sum_{\ell=0}^{\infty} g(\ell) \, u_{k-\ell},$$

where, as already mentioned, $g(\ell) \in \mathbb{R}^{m \times n}$ are the impulse response coefficients or Markov (matrix) parameters given by

$$g(\ell) = \begin{cases} D, & \ell = 0\\ CA^{\ell-1}B, & \ell > 0 \end{cases}$$

Based on the impulse response coefficients, the following *Impulse Response* Block Hankel Matrix can be constructed [HK66]

$$\mathcal{H}_{ij} \triangleq \begin{bmatrix} g(1) & g(2) & \cdots & g(j) \\ g(2) & g(3) & \cdots & g(j+1) \\ \vdots & \vdots & \ddots & \vdots \\ g(i) & g(i+1) & \cdots & g(i+j-1) \end{bmatrix},$$
(2.12)

where $\mathcal{H}_{ij} \in \mathbb{R}^{im imes jn}$. It is not difficult to see that \mathcal{H}_{ij} can be factorized as [Kai80]

$$\mathcal{H}_{ij} = \mathcal{O}_i \mathcal{C}_j. \tag{2.13}$$

Definition 2.1.7. Internal Stability:⁵ [Kai80] We say that a realization (A, B, C)

⁵Internal stability is often called asymptotic stability.

is (internally) stable or stable in the sense of Lyapunov, if the solution of

$$x_{k+1} = Ax_k,$$

with initial condition x_0 , tends to zero as $k \to \infty$, for arbitrary x_0 . A necessary and sufficient condition for internal stability is that all the eigenvalues of Amust lie strictly inside the unit circle, that is

$$|\lambda_i(A)| < 1,$$

where $\lambda_i(A)$ are the eigenvalues of A.

If the system (2.8)-(2.9) is stable, the associated controllability Gramian \mathcal{P} and observability Gramian \mathcal{Q} are defined as the solution of the Lyapunov equations

$$\begin{aligned} A\mathcal{P}A^T + BB^T &= \mathcal{P}, \\ A^T\mathcal{Q}A + C^TC &= \mathcal{Q}, \end{aligned}$$

respectively.

Definition 2.1.8. Hankel Singular Values: [Glo84] Let (2.8)-(2.9) be a stable state-space realization of the transfer matrix G(z), so that G(z) can be written as

$$G(z) = C(zI - A)^{-1}B + D.$$

Then, the Hankel Singular values of G(z) are defined as

$$\sigma_i^{\mathcal{H}}(G(z)) \triangleq \sqrt{\lambda_i(\mathcal{PQ})},$$

where \mathcal{P} and \mathcal{Q} are the controllability and observability Gramians, respectively. It is not difficult to show that the Hankel singular values of G(z) are the singular values of the impulse response block Hankel matrix (2.12) [Glo84].

Definition 2.1.9. Balance Realizations: [PS82] A stable state-space realization is called *internally balanced*, or simply *balanced* if

$$\mathcal{P}=\mathcal{Q}=\Sigma,$$

where $\Sigma \triangleq \text{diag} \{ \sigma_1^{\mathcal{H}}, \sigma_2^{\mathcal{H}}, \cdots, \sigma_{n_x}^{\mathcal{H}} \}$, $\sigma_1^{\mathcal{H}} \ge \sigma_2^{\mathcal{H}} \ge \cdots \ge \sigma_{n_x}^{\mathcal{H}}$, is a diagonal matrix with the Hankel singular values as diagonal elements. A stable state-space realization is called *input balanced* if

$$\mathcal{P} = I, \qquad \mathcal{Q} = \Sigma^2,$$

and is called *output balanced* if

$$\mathcal{P} = \Sigma^2, \qquad \mathcal{Q} = I.$$

 \diamond

 \diamond

 \diamond

2.1.2 Some Hilbert (Hardy) spaces related to signals and systems

In this subsection, some basic concepts related to general Hilbert spaces are presented, and the particular Hilbert (Hardy) spaces associated with the classes of signals and systems used in this thesis are introduced. An advanced treatment of these topics can be found in the books [You88, Rud74].

Let us first quote the definition of a Hilbert space [You88, Rud74]:

"A Hilbert space is an inner product space which is complete with respect to the metric induced by (the norm induced by) the inner product."

Here 'inner product space' means a linear vector space H with an inner product operation defined between two of its elements. This inner product is a real-valued function usually denoted⁶ $\langle \cdot, \cdot \rangle$, that verifies the following properties:

1.
$$\langle x, y \rangle = \langle y, x \rangle$$
, for all $x, y \in H$.

2.
$$\langle \alpha x + \beta y, u \rangle = \alpha \langle x, u \rangle + \beta \langle y, u \rangle$$
, for all $x, y, u \in H$ and scalar α, β .

3. $\langle x, x \rangle > 0$, and $\langle x, x \rangle = 0$ if and only if $x = \mathbb{O}$.

Where $\overline{(\cdot)}$ indicates complex conjugate, and where \mathbb{O} is the null element in the space H. The inner product can be used to define the norm of an element of the space as follows:

$$||x|| \triangleq \langle x, x \rangle^{1/2}, \qquad x \in H.$$

Sometimes the notation $\|\cdot\|_H$ will also be used to emphasize that it is the norm induced by the inner product in the space H. This norm can be used now to define a metric in the space measuring the distance between two elements:

$$d(x, y) \triangleq ||x - y||, \qquad x, y \in H.$$

This concept of distance between two elements of the space can be used in turn to define 'convergence' of sequences of elements as follows:

Definition 2.1.10. A sequence $\{x(k)\}$ of elements of the Hilbert space H converges, with respect to the metric d(x, y), to an element $x \in H$ if

$$\lim_{k \to \infty} d(x(k), x) = 0.$$

The element x is called the 'limit of the sequence', and we write without distinction

$$\lim_{k \to \infty} x(k) = x, \quad \text{or} \quad x(k) \to x.$$

⁶Sometimes, the notation $\langle \cdot, \cdot \rangle_H$ will also be used to emphasize that it is the inner product on the space H.

Completeness of the space in the metric induced by (the norm induced by) the inner product means that any convergent Cauchy sequence⁷ of elements of the space converges to an element of the space.

Fundamental in this thesis will be the concepts of orthogonality and orthonormality of elements and subsets of a Hilbert space leading to the concept of 'orthonormal basis'. These concepts are summarized in the following definition:

Definition 2.1.11.

- i. Two elements x, y in a Hilbert space H are said to be orthogonal, written $x \perp y$, if $\langle x, y \rangle = 0$.
- ii. The element $x \in H$ is said to be orthogonal to the subset $S \subset H$, written $x \perp S$, if x is orthogonal to every $y \in S$.
- iii. Two subsets $S, Q \subset H$ are orthogonal, $S \perp Q$, if $\langle x, y \rangle = 0$ for all $x \in S$ and $y \in Q$.
- iv. For a given subset $S \subset H$, the set S^{\perp} defined as

$$S^{\perp} \triangleq \{ y \in H : y \perp x, \forall x \in S \}$$

is called the 'orthogonal complement of S'.

- v. An element $x \in H$ is said to be 'normal' if ||x|| = 1.
- vi. A set of elements $S = \{x_i\}$ in H is said to be an 'orthogonal set' if the elements of the set are orthogonal to each other. If in addition, for each $x_i \in S$, $||x_i|| = 1$, the set is said to be 'orthonormal'. An important property of an orthogonal set is that its elements are linearly independent.
- vii. An orthonormal set $S = \{x_i\}_{i=0}^{\infty}$ in an inner product space is said to be complete if its closed linear span⁸ is the whole space. For the case of a Hilbert space H, there is a theorem [You88] establishing that an orthonormal set is 'complete' if and only if the only element in H which is orthogonal to each of the x_i is the zero element.
- viii. A complete orthonormal set in a Hilbert space is called an 'orthonormal basis'.

 \diamond

⁷A sequence $\{x(k)\}$ in a Hilbert space is said to be a Cauchy sequence if $d(x(k), x(\ell)) \to 0$ as $k, \ell \to \infty$. This means that for any $\epsilon > 0$ there exists an integer N_{ϵ} such that $d(x(k), x(\ell)) \le \epsilon$ for any $k, \ell \ge N_{\epsilon}$.

⁸Let A be a set in the normed linear space $E, A \subseteq E$. The closed linear span of A, denoted Span $\{A\}$, is the intersection of all closed linear subspaces of E which contain A.

It is possible to show that every Hilbert space has an orthonormal basis. This fact is important because it allows a unique representation of any element of the space as an orthonormal series expansion in terms of the elements of the basis (Generalized Fourier Series). In other words, given an orthonormal basis $\{\mathcal{B}_k\}_{k=0}^{\infty}$ in a (separable ^{9,10}) Hilbert space H, every element $x \in H$ can be represented as

$$x = \sum_{k=0}^{\infty} \langle x, \mathcal{B}_k \rangle \mathcal{B}_k, \qquad (2.14)$$

where the equality has to be interpreted as convergence in the norm induced by the inner $product^{11}$, that is as

$$\lim_{n \to \infty} \left\| x - \sum_{k=0}^{n} \langle x, \mathcal{B}_k \rangle \mathcal{B}_k \right\| = 0.$$

It is clear that the representation in (2.14) is not useful in practical problems where only a finite number of terms can be handled. The solution is then to approximate x by the element \hat{x} ,

$$\widehat{x} = \min_{y \in X_p} \|x - y\|$$

which is the closest element to x belonging to the subspace X_p spanned by the first p elements of the orthonormal basis $\{\mathcal{B}_k\}_{k=0}^{p-1}$. By appealing to the Projection Theorem¹² it can be proved that \hat{x} is given by

$$\widehat{x} = \sum_{k=0}^{p-1} \langle x, \mathcal{B}_k \rangle \, \mathcal{B}_k$$

⁹An inner product space is said to be separable if it contains a countable subset which is everywhere dense (see next footnote). All Hilbert spaces considered in this thesis are separable.

$$\sum_{k=0}^{N} \left\langle x(t), \mathcal{B}_k \right\rangle \mathcal{B}_k$$

as converging to the element $x(t) \in H$ for all values of t. This is called 'pointwise convergence'. The two concepts are different, and, in general, convergence in the norm does not necessarily imply pointwise convergence.

¹²**Projection Theorem:** Let S be a linear closed subspace in the Hilbert space H, and let $x \in H$ be a vector not in S. Then there exists a unique vector $y_0 \in S$ such that the distance from x to the subspace S is given by

$$d(x,S) = ||x - y_0||.$$

Furthermore $(x - y_0) \perp S$.

¹⁰A subset S of an inner product space H is everywhere dense if for every $x \in S$ and $\epsilon > 0$ there is a $y \in S$ such that $||x - y|| < \epsilon$.

¹¹When dealing with spaces of functions $(x(t) \in H)$ a different interpretation of equation (2.14) is possible. Namely, one can interpret the partial sums

The approximation error for this case lies in the orthogonal complement of X_p .

Some important properties of orthonormal sets in Hilbert spaces are summarized in the following proposition [You88, Rud74].

Proposition 2.1.2.

i. Let $\{\mathcal{B}_k\}_{k=0}^p$ be an orthonormal set in the Hilbert space H. Let $\beta_0, \dots, \beta_p \in \mathbb{C}$, and let x be an element in H. Then

$$\left\|x - \sum_{k=0}^{p} \beta_k \mathcal{B}_k\right\|^2 = \left\|x\right\|^2 + \sum_{k=0}^{p} \left|\beta_k - \langle x, \mathcal{B}_k \rangle\right|^2 - \sum_{k=0}^{p} \left|\langle x, \mathcal{B}_k \rangle\right|^2.$$

ii. Let $\{\mathcal{B}_k\}_{k=0}^{\infty}$ be a countable orthonormal set in the Hilbert space H, and let x be an element in H. Then, the expansion coefficients, $\langle x, \mathcal{B}_k \rangle$, of $x \in H$, and the norm of x satisfy

$$||x||^2 \ge \sum_{k=0}^{\infty} |\langle x, \mathcal{B}_k \rangle|^2, \qquad (2.15)$$

which is known as *Bessel's inequality*. If in addition, the orthonormal set $\{\mathcal{B}_k\}_{k=1}^{\infty}$ is complete (i.e., an orthonormal basis), Bessel's inequality becomes an equality, namely

$$||x||^{2} = \sum_{k=0}^{\infty} |\langle x, \mathcal{B}_{k} \rangle|^{2}, \qquad (2.16)$$

which is known as *Parseval's identity* (or *Parseval's Theorem*). The condition (2.16) is a necessary and sufficient condition for the orthonormal set to be an orthonormal basis. An alternative formulation is given by

$$\langle x, y \rangle = \sum_{k=0}^{\infty} \langle x, \mathcal{B}_k \rangle \overline{\langle y, \mathcal{B}_k \rangle}, \qquad \forall x, y \in H.$$

iii. Let $\{\mathcal{B}_k\}_{k=0}^{\infty}$ be an orthonormal basis in the Hilbert space H. Then the closed linear span of $\{\mathcal{B}_k\}_{k=0}^{\infty}$ is the whole space H.

In the following, some particular Hilbert and Hardy spaces that will be used in this thesis are introduced.

Let us consider first a Hilbert space related to the impulse response sequence of stable, causal, discrete-time systems. Let $\{g(k)\}$ denote the impulse response sequence of a stable, causal, discrete-time scalar system with input-output representation

$$y_k = G(q)u_k \triangleq \sum_{\ell=0}^{\infty} g(\ell)q^{-\ell} u_k.$$
(2.17)

The absolute summability of the impulse response sequence

$$\sum_{k=0}^{\infty} |g(k)| < \infty$$

ensures the (BIBO¹³) stability of the system¹⁴ [Kai80] and also implies the square summability of the sequence¹⁵. It is then natural to define the Hilbert space $\ell_2(\mathbb{N}_0)$ of square summable sequences with support in \mathbb{N}_0 (the set of non-negative integer numbers), equipped with the inner product

$$\langle f,g\rangle = \sum_{k\in\mathbb{N}_0} f(k)\overline{g(k)},$$

so that the impulse response sequences of all stable, causal, discrete-time systems belong to this space. Of course, not every element of this space can be associated with the impulse response of a stable, causal, discrete-time system, since square summability of the sequence does not necessarily imply its absolute summability. The space $\ell_2(\mathbb{N}_0)$ is a proper subspace of the Hilbert space $\ell_2(\mathbb{Z})$ of (two-sided) square summable sequences with support in \mathbb{Z} (the set of integer numbers), with the same definition for the inner product (substituting the index set \mathbb{N}_0 by \mathbb{Z}).

Let us now turn our attention to a Hilbert space related to the transfer functions of stable, causal, discrete-time systems. First we recall the definition of the Discrete Fourier Transform (DFT) [OS89, PM92] of a sequence $\{f(k)\} \in \ell_2(\mathbb{Z})$

$$F(e^{j\omega}) \triangleq \sum_{k=-\infty}^{\infty} f(k)e^{-j\omega k}.$$
(2.18)

¹⁵Absolute Summability \Rightarrow Square Summability

$$\begin{split} \sum_{k=0}^{\infty} |g(k)| &< \infty \quad \Rightarrow \quad \sum_{k=0}^{\infty} |g(k)| \sum_{h=0}^{\infty} |g(h)| < \infty \Rightarrow \\ \sum_{k=0}^{\infty} \sum_{h=0}^{\infty} |g(k)| |g(h)| < \infty \quad \Rightarrow \quad \sum_{k=0}^{\infty} |g(k)|^2 + \sum_{k=0}^{\infty} \sum_{h=0}^{\infty} |g(k)| |g(h)| < \infty \Rightarrow \\ \Rightarrow \quad \sum_{k=0}^{\infty} |g(k)|^2 < \infty \end{split}$$

¹³Bounded-Input-Bounded Output

¹⁴**BIBO Stability:** A causal system is said to be externally stable or BIBO stable if a bounded input $u_k < M_u < \infty, 0 \le k < \infty$ produces a bounded output $y_k < M_y < \infty, 0 \le k < \infty$. A necessary and sufficient condition for BIBO stability is the absolute summability of the impulse response sequence [Kai80].

It can be proved that the Discrete Fourier transform of a sequence in $\ell_2(\mathbb{Z})$ belongs to the Hilbert space $L_2(\mathbb{T})$ of Lebesgue square-integrable functions on the unit circle in the complex plane $\mathbb{T} \triangleq \{z : |z| = 1\}$ (or equivalently $\mathbb{T} \triangleq \{e^{j\omega} : -\pi \le \omega \le \pi\}$), equipped with the inner product

$$\langle F, G \rangle \triangleq \frac{1}{2\pi} \int_{-\pi}^{\pi} F(e^{j\omega}) \overline{G(e^{j\omega})} d\omega.$$
 (2.19)

Recalling the definition of the 'two-sided' Z-transform of a sequence $\{f(k)\} \in \ell_2(\mathbb{Z})$ [OS89, PM92],

$$F(z) \triangleq \sum_{k=-\infty}^{\infty} f(k) z^{-k},$$
(2.20)

it can be seen that the discrete Fourier transform (equation (2.18)) can be interpreted as the Z-transform computed on the unit circle \mathbb{T} . We can then give the following alternative z-domain expression for the inner product in $L_2(\mathbb{T})^{16}$

$$\langle F, G \rangle \triangleq \frac{1}{2\pi j} \oint_{\mathbb{T}} F(z) \overline{G(1/z)} \frac{\mathrm{d}z}{z}.$$
 (2.21)

A proper subspace of $L_2(\mathbb{T})$ is the Hardy space¹⁷ $H_2(\mathbb{T})$ of Lebesgue squareintegrable functions on the unit circle \mathbb{T} that are analytic outside the unit disc $\mathbb{D} \triangleq \{z : |z| < 1\}$. It is clear that the discrete Fourier transforms of the impulse response sequences of all stable, causal, discrete-time systems belong to this space¹⁸. With some abuse of terminology we will refer to this Hardy space as 'the space of all stable, causal, discrete-time transfer functions', since the discrete Fourier transform (respectively, the Z-transform) of the impulse response sequence is nothing else but the transfer function $G(e^{j\omega})$ (respectively, G(z)) of the system¹⁹.

An important property of Hilbert spaces is that all separable Hilbert spaces with the same cardinality are isometrically isomorphic to each other. This means that between any two separable Hilbert spaces (with the same cardinality) there exists a one to one mapping preserving norms (an isometry). For the spaces $\ell_2(\mathbb{Z})$ and $L_2(\mathbb{T})$, the isometry is the discrete Fourier transform. Parseval's identity then allows us to write

$$\langle f, g \rangle = \langle F, G \rangle.$$

¹⁶Equation (2.19) follows from equation (2.21) by definition of contour integral, with $z(\omega) = e^{j\omega}$, $-\pi < \omega \leq \pi$, on the unit circle.

 $^{^{17}}H_2(\mathbb{T})$ is a Hilbert space with the inner product in $L_2(\mathbb{T})$.

¹⁸Of course, not every element of this space can be interpreted as the discrete Fourier transform of the impulse response sequence of a stable, causal, discrete-time system.

¹⁹See also the previous footnote.

When dealing with identification of multivariable systems in chapter 5 we will be interested in a Hilbert space related to the transfer matrices of stable, causal, discrete-time MIMO systems. We will denote with $H_2^{m \times n}(\mathbb{T})$ the Hardy space of $(m \times n)$ matrices whose elements are functions of the complex variable z, belonging to $H_2(\mathbb{T})$. With some abuse of terminology we will refer to this space as 'the space of all stable, causal, discrete-time, $(m \times n)$ transfer matrices'. $H_2^{m \times n}(\mathbb{T})$ is a Hilbert space with the following definition for the inner product

$$\langle \mathcal{B}_{\ell}, \mathcal{B}_{k} \rangle = \frac{1}{2\pi} \int_{-\pi}^{\pi} \operatorname{Tr} \left\{ \mathcal{B}_{\ell}(e^{j\omega}) \mathcal{B}_{k}(e^{j\omega})^{\star} \right\} \mathrm{d}\omega,$$

or equivalently

$$\langle \mathcal{B}_{\ell}, \mathcal{B}_{k} \rangle = \frac{1}{2\pi j} \oint_{\mathbb{T}} \operatorname{Tr} \left\{ \mathcal{B}_{\ell}(z) \mathcal{B}_{k}(1/z)^{\star} \right\} \frac{\mathrm{d}z}{z}$$

2.2 Prediction Error Methods

In this section, system identification methods based on minimization of prediction errors are reviewed. Usually these techniques are called Prediction Error Methods (PEM) (see for instance [Lju87, SS89] for a more detailed discussion).

2.2.1 Problem Formulation

To formalize the problem, let us consider a discrete-time linear time-invariant MIMO system with a general model structure given by

$$\mathcal{M}(\theta): \quad y_k = G(q,\theta) \, u_k + H(q,\theta) \, e_k, \tag{2.22}$$

where y_k is the *m*-dimensional output vector, u_k is the *n*-dimensional input vector, and e_k is a sequence of *m*-dimensional, independent and identically distributed (iid) random variables with zero mean (i.e. white noise), and co-variance matrix

$$\mathbf{E}\left\{e_k e_s^T\right\} = R_e \delta(k-s),$$

where $\delta(k - s)$ stands for the Kronecker delta²⁰. Furthermore, $G(q, \theta)$ and $H(q, \theta)$ are matrices of real, rational, stable, strictly proper²¹ transfer functions parameterized by the *p*-dimensional parameter vector θ lying in the parameter

 $^{{}^{20}\}delta(k-s) \triangleq \left\{ \begin{array}{cc} 1 & \text{for} & k=s \\ 0 & \text{for} & k\neq s \end{array} \right.$

²¹A rational transfer function is said to be strictly proper if the relative order (that is the difference between the degrees of the numerator and denominator polynomials) is strictly negative [Kai80].

space $D_{\mathcal{M}} \subset \mathbb{R}^p$. How the set $D_{\mathcal{M}}$ has to be defined will become clear later in the section.

We assume also that an N-point data set

$$Z^N = \{y_k, u_k : k = 1, \cdots, N\},\$$

consisting of an input sequence $\{u_k\}$ and an output sequence $\{y_k\}$ is available for the purposes of estimating $G(q, \theta)$ and $H(q, \theta)$. We will denote by $\hat{y}_k(\theta)$ a prediction of y_k given the data up to time k - 1, and based on the parameter vector θ . In general the predictor can be a linear or a nonlinear filter applied to the data, and it can be constructed in various ways for any given model. It can be determined from the underlying system description, or from other considerations. For instance, a frequently used predictor determined from the model structure (2.22) is the so-called mean square optimal one-step-ahead predictor

$$\widehat{y}_k(\theta) = \begin{bmatrix} 1 - H^{-1}(q,\theta) \end{bmatrix} y_k + H^{-1}(q,\theta)G(q,\theta) u_k$$

A more general linear predictor is given as

$$\widehat{y}_k(\theta) = F_1(q,\theta) \ y_k + F_2(q,\theta) \ u_k, \tag{2.23}$$

where the predictor filters $F_1(q,\theta)$ and $F_2(q,\theta)$ are such that $\hat{y}_k(\theta)$ is a function of past data only.

Given a model structure and a predictor, the *prediction errors* can then be defined as

$$\epsilon_k(\theta) = y_k - \widehat{y}_k(\theta) . \tag{2.24}$$

The objective is then to find an estimate $\hat{\theta}$ of θ that minimizes a given function $V_N(\theta)$ of the prediction errors. This function is called a criterion (loss function or cost function), and it is a scalar-valued function (typically positive) of all the prediction errors $\epsilon_1(\theta), \epsilon_2(\theta), \cdots, \epsilon_N(\theta)$, which will assess the performance of the predictor used. The criterion is minimized with respect to the parameter vector θ to choose the 'best' predictor among the class considered. A criterion which is often adopted is the quadratic one

$$V_N(\theta) = \frac{1}{N} \sum_{k=1}^{N} \operatorname{Tr} \left\{ \epsilon_k(\theta) \epsilon_k^T(\theta) \right\}, \qquad (2.25)$$

but the range of possibilities is wide. A more general expression for the criterion would be

$$V_N(\theta) = \frac{1}{N} \sum_{k=1}^N \upsilon\left(\epsilon_k^F(\theta)\right), \qquad (2.26)$$

where $v(\cdot)$ is a scalar-valued function (typically positive) of some filtered version $\epsilon_k^F(\theta)$ of the prediction errors $\epsilon_k(\theta)$.

The estimate $\hat{\theta}_N$ is then computed as the minimizing argument of the criterion (2.26), i.e.

$$\widehat{\theta}_{N} = \operatorname*{arg\,min}_{\theta \in D_{\mathcal{M}}} \left\{ V_{N}(\theta) \right\}.$$
(2.27)

It is now clear that the set $D_{\mathcal{M}}$ (the parameter space) has to be defined as those values of θ for which the predictor is asymptotically stable.

Summarizing, a Prediction Error Method (PEM) can be described as follows:

- Choice of the model structure: This concerns the parameterization of the transfer matrices $G(q, \theta)$ and $H(q, \theta)$ as a function of θ .
- Choice of the predictor: This concerns the definition of the predictor. For example, if a linear predictor is specified as in equation (2.23), the user has to choose the prediction filters $F_1(q, \theta)$ and $F_2(q, \theta)$.
- Choice of the criterion: This concerns the choice of the scalar-valued function v(·) in the general criterion (2.26) which will assess the performance of the predictor.
- Computation of the parameter estimate $\hat{\theta}_N$ that minimizes the criterion (2.26). Implicit in this step is the choice of the minimization technique.

Particular choices of the model structure, the predictor, the criterion, and the minimization technique, result in particular methods with specific names. For example, the well known Least-Squares, Maximum Likelihood, and Instrumental-Variables Methods can all be considered as Prediction Error Methods [Lju87, SS89].

2.2.2 Asymptotic Analysis

In this subsection, some results regarding the limiting properties of the estimated parameters as the number of data points tends to infinity are reviewed. These results are concerned with two main aspects; namely, consistency of the estimate, and estimation accuracy. Again, the interested reader is referred to [Lju87, SS89], for a thorough treatment of these topics.

We will denote by $\hat{\theta}_N$ the parameter estimate based on N data points, that is, $\hat{\theta}_N$ is the minimizing argument of the criterion $V_N(\theta)$.

For the analysis, the following basic assumptions will be made:

- A1. The data $\{u_k, y_k\}$ are stationary processes.
- A2. The input $\{u_k\}$ is persistently exciting.

- A3. The Hessian $V_N''(\theta)$ is nonsingular locally around the minimum points of $V_N(\theta)$.
- A4. The transfer matrices $G(q, \theta)$ and $H(q, \theta)$ are smooth (differentiable) functions of the parameter vector θ .

Part of the analysis will require the following additional assumption about the true system:

A5. The set²²

$$D_T \triangleq \{\theta : G(q, \theta) = G(q); H(q, \theta) = H(q)\}$$

consisting of those parameter vectors for which the model structure gives an exact description of the true system consists of precisely one point. This point will be denoted by θ_0 , which will be called the true parameter vector.

2.2.2.1 Asymptotic Estimate

We are interested here in the limit value to which the estimate $\hat{\theta}_N$ converges as the number of data N tends to infinity. This analysis is related to the consistency of the estimation method, that is, with the issue of whether the estimates converge to the 'true parameter' (provided that this value exists) when increasing number of data are considered. The main result is summarized in the following Theorem [Lju87, SS89].

Theorem 2.2.1. Let the estimate $\widehat{\theta}_N$ be defined as in equation (2.27), with the general criterion as in equation (2.26), and suppose that assumptions A1 to A4 hold. Then, the criterion function $V_N(\theta)$ converges uniformly in $\theta \in D_M$ to the limit function $V_{\infty}(\theta)$, i.e.

$$\sup_{\theta \in D_{\mathcal{M}}} |V_N(\theta) - V_{\infty}(\theta)| \xrightarrow{\text{a.s.}} 0 \qquad \text{as} \quad N \to \infty$$

where $V_{\infty}(\theta) = \lim_{N \to \infty} V_N(\theta)$. Moreover, the minimizing argument $\hat{\theta}_N$ of $V_N(\theta)$ converges to a value θ_{\star} in the set D_C of minimizing arguments of $V_{\infty}(\theta)$. That is

$$\widehat{\theta}_N \xrightarrow{\text{a.s.}} \theta_\star$$
 as $N \to \infty$, $\theta_\star \in D_C$,

where D_C is defined as

$$D_C = \underset{\theta \in D_{\mathcal{M}}}{\operatorname{arg\,min}} \left\{ V_{\infty}(\theta) \right\} = \left\{ \theta : \ \theta \in D_{\mathcal{M}}, V_{\infty}(\theta) = \underset{\theta' \in D_{\mathcal{M}}}{\operatorname{min}} V_{\infty}(\theta') \right\}.$$

 $^{^{22}}$ When the set D_T is not empty, the system is said to be 'system identifiable'. If in addition, D_T consists of only one element, the system is said to be 'parameter identifiable'.

The result implies that when the set D_T is empty, the asymptotic estimate will be 'biased' but it will give the best possible approximation of the system which is available in the model set. If the system is parameter identifiable (condition A5 holds), then the set D_T is not empty and consists of only one element. Under some weak assumptions on the data set, it is possible to show that in this case $D_C = D_T = \{\theta_0\}$, so that the estimate is strongly consistent.

2.2.2.2 Asymptotic distribution of the parameter estimate

We analyze here the limiting distribution of the parameter estimates. The following theorem shows that the distribution of the random variable

$$\sqrt{N}(\widehat{\theta}_N - \theta_\star),$$

where $\hat{\theta}_N$ is the parameter estimate and θ_{\star} is the asymptotic estimate (as defined in Theorem 2.2.1), converges to a Gaussian distribution under weak assumptions [Lju87, SS89].

Theorem 2.2.2. Under the conditions of Theorem 2.2.1

$$\sqrt{N}(\widehat{\theta}_N - \theta_\star) \xrightarrow{\text{dist}} \mathcal{N}(0, P),$$

where

$$P = [V_{\infty}''(\theta_{\star})]^{-1} \left[\lim_{N \to \infty} N \mathsf{E} \left\{ V_{N}'(\theta_{\star})^{T} V_{N}'(\theta_{\star}) \right\} \right] [V_{\infty}''(\theta_{\star})]^{-1},$$
(2.28)

with $V'_N(\theta_{\star})$ denoting the gradient of $V_N(\theta)$ computed at $\theta = \theta_{\star}$, and $V''_{\infty}(\theta_{\star})$ denoting the Hessian of $V_{\infty}(\theta)$ computed at $\theta = \theta_{\star}$.

This result is important because it gives an expression for the asymptotic covariance matrix P (equation (2.28)), that can be used to quantify the estimation accuracy. The asymptotic covariance expression can also be used to derive confidence intervals for each particular estimate $\hat{\theta}_N$ obtained from the data set [Lju87]. Unfortunately, the expression for the asymptotic covariance (2.28) requires the knowledge of the asymptotic estimate θ_{\star} which is unknown to the user. This problem can be solved by instead using an estimate of P. A simple estimate of P in equation (2.28) can be obtained by replacing θ_{\star} by $\hat{\theta}_N$, and the expectation operator by the sample average.

2.2.3 Computational aspects

In general, the minimization of the criterion $V_N(\theta)$ cannot be done analytically. Only for the special case where the prediction error depends linearly on θ (which corresponds to the case of linear regressor) an analytic expression can be found for the solution of the minimization problem. In most cases the minimization problem is nonlinear and nonconvex, and the solution must be obtained by using some numerical iterative search method. Some commonly used algorithms are the Newton-Raphson Algorithm and the Gauss-Newton Algorithm [SS89, Lju87]. The main problem with these iterative search algorithms is that convergence only to a local minimum of the criterion can be guaranteed. Usually, the way to find the global minimum is to run the algorithm from different initial conditions and then to compare the estimates [Lju87]. In general, this constitutes a computationally intensive procedure.

In this thesis, systems are parameterized using orthonormal bases that lead to linear regressor forms for which the problems of local minima are avoided.

3

Orthonormal Bases on the Unit Circle

In this chapter, the construction of several families of (scalar) rational orthonormal bases on the unit circle will be reviewed. The concept of reproducing kernel associated with the bases will be introduced, and its importance for the analysis of the approximating properties of the bases will be pointed out. The emphasis of the chapter will be on a particular family of orthonormal bases, namely the Orthonormal Bases with Fixed Poles, for which a minimal state space realization will be derived and a closed form expression for the reproducing kernel associated with them will be given. Furthermore, it is shown how families of (matrix) orthonormal bases for the space $H_2^{m \times n}(\mathbb{T})$ can be constructed from orthonormal bases on $H_2(\mathbb{T})$.

3.1 Introduction

In this chapter we study various families of rational orthonormal bases for the Hardy space $H_2(\mathbb{T})$ of functions analytic outside, and square integrable on the unit circle \mathbb{T} . The motivation for this is the practical utility of rational orthonormal bases for the approximation of elements of $H_2(\mathbb{T})$ by rational transfer functions [Wal35], even when the element is non rational. These approximating properties of rational orthonormal bases are exploited in this thesis in the context of identification of discrete-time, linear systems. As already mentioned, a main advantage of using orthonormal bases in an identification context is that prior information about the dominant dynamics of the system can be easily incorporated in the process of basis construction, and that the resulting model structures become linear in the parameters which simplifies the estimation problem.

It is not the intention of this chapter to give a complete survey on the area, which would be an overwhelming task. Instead, the purpose is to focus on what this author believes is relevant for the material in the subsequent chapters. In Section 3.2, the concept of *reproducing kernel* associated with the space spanned by an orthonormal set is introduced and some of its properties are studied. Closed form expressions for the reproducing kernels will be fundamental for the quantification of the accuracy of the estimation we perform in Chapter 4.

In Section 3.3, a brief description of some 'standard' orthonormal bases on the unit circle such as the trigonometric bases (corresponding to the socalled FIR (Finite Impulse Response) model structures), Laguerre [Wah91b] bases and Kautz [Wah94b] bases is given. A more detailed description of more general orthonormal bases that allow prior information about several dominant dynamics of the system to be included in the identification process, and that have the more common FIR, Laguerre and Kautz bases as special cases, is given in Sections 3.4 and 3.5 where the Orthonormal Bases Generated from Inner Functions (OBGIF) [HbVB95, VHB95], and the Orthonormal Bases with Fixed Poles (OBFP) [NG94a, NG97] are respectively considered.

Finally, in Section 3.6 it is shown how families of orthonormal bases for the space of stable $(m \times n)$ transfer matrices $H_2^{m \times n}(\mathbb{T})$ can be constructed from orthonormal bases on $H_2(\mathbb{T})$.

3.2 Reproducing Kernels

As mentioned in the previous chapter, $H_2(\mathbb{T})$ is a Hilbert space when it is endowed with the inner product in $L_2(\mathbb{T})$, as defined in equations (2.19) or (2.21). An orthonormal basis $\{\mathcal{B}_k(z)\}_{k=0}^{\infty}$ in $H_2(\mathbb{T})$ is a complete orthonormal set. The orthonormality is reflected by the property

$$\langle \mathcal{B}_k, \mathcal{B}_\ell \rangle = \begin{cases} 1 & \text{for } (k = \ell) \\ 0 & \text{for } (k \neq \ell) \end{cases}$$

while the completeness is characterized by the fact that the closed linear span of the set is the whole space.

In this thesis, rational orthonormal bases on the unit circle are used to represent discrete-time linear systems for the purposes of identification from input-output measurements. The transfer function of the unknown system is modeled as a linear combination of the rational basis functions, and the identification is carried out by estimating a finite number of coefficients in this orthonormal expansion, using least squares techniques. In this context, the accuracy of the estimation is affected by two causes: the noise corruption of the measured data that generates the so-called *variance error*, and the parsimony of the model structure which is too simple to describe the real system, that results in the so-called *bias error*.

In [LY85], Ljung and Yuan show that, for the case of the standard trigonometric bases (or FIR model structure) $\{z^{-k}\}$, the variance error in the frequency response estimate can be approximated (for large model order and length of the
available data) by the product of the noise-to-signal ratio and the model orderto-data length ratio. Essential in the derivation of this result has been the observation that, due to the algebraic structure

$$\mathcal{B}_n(z)\mathcal{B}_m(z) = \mathcal{B}_{m+n}(z), \tag{3.1}$$

enjoyed by the trigonometric bases, the covariance matrix of the transfer function estimate has a Toeplitz structure, and then some classical results on asymptotic properties of Toeplitz matrices [GS58] can be exploited to carry out the analysis. Unfortunately, the Orthonormal Bases with Fixed Poles we consider in this thesis do not have the algebraic structure (3.1) and consequently the covariance matrices do not have a Toeplitz structure. A key rôle in the analysis of the estimation accuracy is then played by the reproducing kernel associated with the bases [Aro50, Dav75], since it allows to generalize the convergence results of Toeplitz matrices for the case in which the orthonormal structure is not the trigonometric one [NHG97b, NHG97a].

In the following, we introduce the concept of reproducing kernel, and describe some of its properties [Aro50, Dav75].

Definition 3.2.1 (Reproducing Kernel). The reproducing kernel associated with a Hilbert space X of functions on a set S, is defined as the unique function $K_p(z,\mu)$ of the two variables $z, \mu \in S$, that satisfies the following two conditions [Aro50]

- i. For every $\mu \in S$, $K_p(z, \mu)$, as a function of z, belongs to X.
- ii. $K_p(z,\mu)$ has the *reproducing property*: for every function $G(z) \in X$ and every $\mu \in \mathbb{S}$

$$G(\mu) = \langle G(z), K_p(z, \mu) \rangle.$$

When the space X is spanned by a finite number of orthonormal basis functions $\{\mathcal{B}_k(z)\}_{k=0}^{p-1}$ with $z \in \mathbb{S}$, it is not difficult to prove that the reproducing kernel can be computed as

$$K_p(z,\mu) = \sum_{k=0}^{p-1} \mathcal{B}_k(z) \overline{\mathcal{B}_k(\mu)}.$$
(3.2)

To see this we have to check that conditions i. and ii. in Definition 3.2.1 are satisfied. That $K_p(z,\mu)$ in equation (3.2) belongs to X is obvious since X is spanned by the basis functions $\{\mathcal{B}_0(z), \mathcal{B}_1(z), \dots, \mathcal{B}_k(z)\}$. It remains to check

the reproducing property ii. Let $G(z) \in X$, and $\mu \in S$, then

$$\langle G(z), K_p(z, \mu) \rangle = \left\langle G(z), \sum_{k=0}^{p-1} \mathcal{B}_k(z) \overline{\mathcal{B}_k(\mu)} \right\rangle,$$

$$= \sum_{k=0}^{p-1} \left\langle G(z), \mathcal{B}_k(z) \overline{\mathcal{B}_k(\mu)} \right\rangle,$$

$$= \sum_{k=0}^{p-1} \left\langle G(z), \mathcal{B}_k(z) \right\rangle \mathcal{B}_k(\mu),$$

$$= G(\mu).$$
(3.3)

In passing to the last line we have used the fact that, since X is spanned by the functions $\{\mathcal{B}_0(z), \mathcal{B}_1(z), \cdots, \mathcal{B}_{p-1}(z)\}$, then any function $G(z) \in X$ has a unique representation of the form

$$G(z) = \sum_{k=0}^{p-1} \langle G(z), \mathcal{B}_k(z) \rangle \, \mathcal{B}_k(z).$$

The existence of the reproducing kernel (3.2) makes the associated Hilbert space a reproducing kernel Hilbert space (r.k.H.s.).

In this thesis, the space X is most commonly $H_2(\mathbb{T})$, and the set S is the open region outside the unit disk in the complex plane $\mathbb{E} \triangleq \{z \in \mathbb{C} : |z| > 1\}$. At times we will need to compute the reproducing kernel¹ on the unit circle \mathbb{T} , that is for $z = e^{j\omega}$, and $\mu = e^{j\sigma}$. In those cases we will use the shorthand $K_p(\omega, \sigma)$ to denote $K_p(e^{j\omega}, e^{j\sigma})$.

We will use equation (3.2) to derive closed form expressions for $K_p(z,\mu)$ for the particular bases. In the context of orthogonal polynomials [Sze59], these closed form expressions for $K_p(z,\mu)$ are known as 'Christoffel-Darboux' type formulas.

Let now $\{\mathcal{B}_k(z)\}_{k=0}^{\infty}$ be a complete orthonormal set in $H_2(\mathbb{T})$. By analogy with (3.2) we can define the function

$$K(z,\mu) \triangleq \sum_{k=0}^{\infty} \mathcal{B}_k(z) \overline{\mathcal{B}_k(\mu)}, \qquad (3.4)$$

for $z, \mu \in \mathbb{E}$. It is not difficult to prove [Reg95] that $K(z, \mu)$ has the reproducing property

$$G(\mu) = \langle G(z), K(z, \mu) \rangle, \qquad (3.5)$$

¹Properly speaking, $H_2(\mathbb{T})$ is not a function space, but a space of equivalence classes of functions, and then a reproducing kernel is not defined in $H_2(\mathbb{T})$. With some abuse of terminology we will still call a function in $H_2(\mathbb{T})$ satisfying conditions i. and ii. in Definition 3.2.1, a reproducing kernel.

and that as a function of $\mu \in \mathbb{E}$, it belongs to $H_2(\mathbb{T})$. A closed form expression for $K(z,\mu)$ will be very useful in some of the developments of the following chapters. The following Lemma shows that for any complete orthonormal basis in $H_2(\mathbb{T})$, and $\mu \in \mathbb{E}$, $K(z,\mu)$ can be computed as

$$K(z,\mu) = \frac{z\overline{\mu}}{z\overline{\mu} - 1}.$$

Lemma 3.2.1. Let $\{\mathcal{B}_k(z)\}_{k=0}^{\infty}$ be a complete orthonormal set in $H_2(\mathbb{T})$, and let $K(z,\mu)$ be defined as in equation (3.4). Then, independently of the particular basis, $K(z,\mu)$ is given by

$$K(z,\mu) = \frac{z\overline{\mu}}{z\overline{\mu} - 1}$$

for $|z| > 1, |\mu| > 1$.

Proof: See Appendix 3.A.

In the following sections various families of rational orthonormal bases on the unit circle are introduced. For some of them, closed form expressions of the associated reproducing kernels $K_p(z, \mu)$ are derived.

3.3 FIR, Laguerre and Kautz Basis

3.3.1 FIR Basis

The most common orthonormal basis on $L_2(\mathbb{T})$ are the well known trigonometric or FIR basis, that corresponds to the choice

$$\mathcal{B}_k(z) = z^{-k}; \quad k \ge 0. \tag{3.6}$$

The completeness of the basis in $L_2(\mathbb{T})$ is a standard result of classical Fourier series (the proof can be found for instance in [You88]). A direct calculation gives the following closed form expression for the reproducing kernel for this basis:

$$K_p(z,\mu) = \frac{(z\overline{\mu})^p - 1}{(z\overline{\mu})^p - (z\overline{\mu})^{p-1}}.$$
(3.7)

with |z| > 1 and $|\mu| > 1$.

As pointed out by several authors in different contexts (for instance in [GW90, Wah91b, WH93, LW93, NG97] in the context of system approximation and identification, or in [Wil93a, Pdd93, Oli94a, Oli95a, WZ96] in the context of signal processing) the use of FIR model structures to represent systems with long (possibly infinite) impulse responses has the disadvantage that the number of terms in the series expansion necessary to provide an acceptable approximation of the system is high, and this may lead to poor accuracy in the estimated model. As a counterpart, and as it has already been mentioned, the analysis of the accuracy of the least squares estimation using FIR structures is very tractable by exploiting the algebraic structure (3.1) of the bases, leading to Toeplitz structures of the covariance matrices of the estimates, and then using known results on asymptotics of Toeplitz matrices [GS58, HN77, HW89].

3.3.2 Laguerre Basis

The use of Laguerre series in engineering applications has a long history, that can be traced back to the thirties with the work of Wiener and Lee (see for instance [Lee60]) on synthesis of electrical networks. Since that work, Laguerre bases have been used in many different areas, such as system approximation [Nur87, Mäk90a, Mäk90b, Par91, WAH], system identification [KP79, GW90, GW91, Mäk91, Wah91b, Wah94b], filter design [KP77, den93b, den93a, FD93, den94, Oli94b, Oli95c, Oli95a, Oli95b], and control applications [ZD88, ZBD88, ZDP90].

In the z-domain, the Laguerre bases are given by

$$\mathcal{B}_k(z) = \left(\frac{\sqrt{1-\xi^2}}{z-\xi}\right) \left(\frac{1-\xi z}{z-\xi}\right)^k \qquad ; k \ge 0$$
(3.8)

where $\xi \in \mathbb{R}, |\xi| < 1$ is a free (real) parameter called the Laguerre coefficient, or Laguerre pole position. The orthonormality and completeness of the Laguerre basis in $H_2(\mathbb{T})$ follow from the fact that the bases are the Z-transform of the Laguerre sequences [Sze59] which are a complete orthonormal set in $\ell_2(\mathbb{N}_0)$ [Sze59], and the fact that $H_2(\mathbb{T})$ and $\ell_2(\mathbb{N}_0)$ are isometrically isomorphic (the isomorphism being the Z-transform).

As the reader can easily verify, the FIR model is a special case of the Laguerre structure corresponding to $\xi = 0$.

The following Lemma gives the closed form expression for the reproducing kernel of the Laguerre basis.

Lemma 3.3.1. Let $\{\mathcal{B}_k(z)\}_{k=0}^{\infty}$ be the Laguerre basis as defined in (3.8). Then the closed form expression for the reproducing kernel associated with the basis is given by

$$K_p(z,\mu) = \frac{1 - \varphi_p(z)\varphi_p(\mu)}{z\overline{\mu} - 1},$$
(3.9)

with |z| > 1 and $|\mu| > 1$, and where the definition

$$\varphi_p(z) \triangleq \left(\frac{1-\xi z}{z-\xi}\right)^p$$

has been used.

Proof: See Appendix 3.A.

If prior information about the dominant dynamics of the system to be approximated is available, then choosing the Laguerre coefficient close to the dominant pole will increase the rate of convergence of the Laguerre series expansion [Wah91b]. In this way, the number of terms needed to obtain an acceptable approximation will also be reduced. This property represents an advantage of the Laguerre bases when compared to the FIR structure where the possibility of incorporating 'a priori' information to accelerate the rate of convergence does not exist.

As pointed out in [WAH], highly resonant systems are very difficult to approximate with the Laguerre basis that only allows the incorporation of prior knowledge about non-resonant dominant dynamics. A more flexible structure that generalizes the Laguerre basis and is better suited for the approximation of systems with highly oscillatory impulse responses is the so-called 'two-parameter' Kautz basis [Kau52].

3.3.3 Kautz Basis

Since the work of Kautz [Kau52] on orthogonalization of a set of continuous exponentials, considerable research effort has been devoted to the study of applications of Kautz basis in system approximation [WAH], identification [Wah91a, LW93, Wah94b, Wah94a], and filter design [den93b, Oli94a, Oli95d, den96].

In the z-domain, the 'two-parameter' Kautz bases are given by [YH62, Bro65]

$$\mathcal{B}_{k}(z) = \begin{cases} \frac{\sqrt{(1-a^{2})(1-c^{2})}}{z^{2}-a(c+1)z+c} \left(\frac{cz^{2}-a(c+1)z+1}{z^{2}-a(c+1)z+c}\right)^{\frac{k-1}{2}} ; k \text{ odd} \\ \frac{\sqrt{(1-c^{2})}(z-a)}{z^{2}-a(c+1)z+c} \left(\frac{cz^{2}-a(c+1)z+1}{z^{2}-a(c+1)z+c}\right)^{\frac{k}{2}} ; k \text{ even} \end{cases}$$
(3.10)

with -1 < a < 1, -1 < c < 1, and $k \ge 0$. The Laguerre structure is a special case of the Kautz one when the poles are real and equal (i.e. for $a^2(c+1)^2 = 4c$).

The condition on the poles for the completeness of the Kautz bases in $H_2(\mathbb{T})$ has been derived in [DD81] (see also [Oli94a, Wah94b]).

For systems with several resonant dynamics, more general orthonormal bases allowing the incorporation of prior information about several modes would be

more desiderable. Examples of such more general basis are the Orthonormal Basis Generated by Inner Functions (OBGIF) introduced by Heuberger, Van den Hof and co-workers [HbVB95, VHB95], or the Orthonormal Basis with Fixed Poles (OBFP) studied by Ninness and co-workers in [NG94a, NG97]. The Kautz, Laguerre and FIR model structures are all special cases of these methods.

3.4 Orthonormal Basis Generated by Inner Functions

In a series of papers [HB90, HVB92, HVB93, VHB94a, VHB94b, HbVB95, VHB95], Van den Hof, Heuberger and co-workers show how an infinite set of orthonormal functions can be generated from a balanced realization of a square and inner transfer function. The bases are suited for the representation of systems with a wide range of dominant dynamics, since they allow the incorporation of prior information about a set of poles rather than one single pole. By choosing the poles of the bases closed to the actual poles, the speed of convergence of the orthonormal expansion can be increased.

Previous to the introduction of the orthonormal basis, we give the definition of an *inner function*.

Definition 3.4.1. Inner Function. A rational transfer function G(z) is called *inner* if it is stable and satisfies

$$G(z^{-1})G(z) = 1.$$

That is, if it is stable and *all-pass*.

The main result concerning the generation of orthonormal functions for the space $H_2(\mathbb{T})$ from an inner transfer function is summarized in the following theorem due to Van den Hof, Heuberger and co-workers [HbVB95].

Theorem 3.4.1. [HbVB95] Let $G_x(z)$ be a scalar inner function with McMillan degree $n_x > 0$, having a minimal balanced realization (A, B, C, D). Denote

$$V_k(z) = z(zI - A)^{-1} B G_x^k(z)$$
(3.11)

 \diamond

Then the sequence of scalar rational functions

$$\{\mathcal{B}_{i,k}(z)\} \triangleq \{e_i^T V_k(z)\}, \qquad (i = 1, \cdots, n_x); (k = 0, \cdots, \infty), \qquad (3.12)$$

where e_i stands for the *i*-th Euclidean basis vector in \mathbb{R}^{n_x} , forms an orthonormal basis for the Hilbert space $H_2(\mathbb{T})$. Moreover, these orthonormal bases induce associated bases for the signal space $\ell_2(\mathbb{N}_0)$ of squared summable sequences, through inverse z-transformation to the signal domain. Denoting

$$V_k(z) = \sum_{\ell=0}^{\infty} v_{\ell}^k z^{-\ell}$$

it follows that $\{e_i^T v_\ell^k\}$ with $(i = 1, \dots, n_x)$; $(k = 0, \dots, \infty)$ is an orthonormal basis for the signal space $\ell_2(\mathbb{N}_0)$.

Proof: See [HbVB95].

Since the sequence $\{\mathcal{B}_{i,k}(z)\}$ is an orthonormal basis in $H_2(\mathbb{T})$, then any transfer function $G(z) \in H_2(\mathbb{T})$ has a unique series representation

$$G(z) = \sum_{k=0}^{\infty} \sum_{i=1}^{n_x} \theta_k^i \mathcal{B}_{i,k}(z) = \sum_{k=0}^{\infty} \sum_{i=1}^{n_x} \theta_k^i e_i^T V_k(z) = \sum_{k=0}^{\infty} L_k V_k(z)$$
(3.13)

where $L_k \triangleq [\theta_k^1, \theta_k^2, \cdots, \theta_k^{n_x}] \in \ell_2^{1 \times n_x}(\mathbb{N}_0).$

This series representation is schematically depicted in the diagram of Figure 3.1, where we have defined

$$V_0(q) \triangleq q(qI - A)^{-1}B.$$

It can be seen that this orthonormal family for the space of stable systems $H_2(\mathbb{T})$



Figure 3.1: Schematic Representation of the Series Expansion in terms of Orthonormal Bases Generated by Inner Functions

can be generated by the cascade connection of identical balanced realizations of a stable all-pass filter $G_x(q)$, followed by low pass filters $L_k V_0(q)$. It should be noted that these basis functions can incorporate system dynamics of any complexity by the appropriate choice of the poles of the inner function $G_x(z)$. For specific choices of $G_x(z)$ the classical FIR, Laguerre and 'twoparameter' Kautz orthonormal basis can be generated (see [VHB95, HbVB95] for details).

3.5 Orthonormal Basis with Fixed Poles

The limitation with the orthonormal bases generated from inner functions is that they only allow the incorporation of prior information about one set of poles which is cyclically repeated. In [NG97] it was shown that the set

$$\mathcal{B}_k(z) = \left(\frac{\sqrt{1 - |\xi_k|^2}}{z - \xi_k}\right) \prod_{i=0}^{k-1} \left(\frac{1 - \overline{\xi_i}z}{z - \xi_i}\right)$$
(3.14)

is a complete orthonormal set in $H_2(\mathbb{T})$, but allowing prior knowledge about an arbitrary number of modes $\{\xi_0, \xi_1, \dots, \xi_{p-1}\} \in \mathbb{D}$ to be incorporated, without the restriction of periodic repetition. The reader can easily check that when all the poles are chosen at the origin ($\xi_k = 0$, for all k), then the construction (3.14) reduces to an FIR model structure, while for the choice $\xi_k = \xi \in \mathbb{R}, |\xi| < 1$, (3.14) reduces to the Laguerre basis.

As pointed out in [NG97], the construction (3.14) has to be modified to accomodate the case of resonant poles, since in this case the bases would have complex valued impulse responses, which would be inappropriate for their use in the representation of physical systems.

The idea in [NG97] is to still use the construction (3.14) for a complex pole, but including also the complex conjugate, and then replace these two basis functions by linear combinations of them in such a way that the resulting new basis functions are orthonormal to one another and to all the preceeding basis functions, and also have real-valued impulse responses.

To be more specific, suppose the *n*-th pole ξ_n is chosen as complex, and suppose that \mathcal{B}_n is the corresponding basis function (with complex-valued impulse response) computed as in (3.14). Then, the (n + 1)-th pole has to be chosen as the complex conjugate $\xi_{n+1} = \overline{\xi_n}$, leading to the basis function \mathcal{B}_{n+1} (also with complex-valued impulse response). Now the basis functions \mathcal{B}_n and \mathcal{B}_{n+1} are replaced by the linear combinations

$$\begin{aligned} \mathcal{B}'_n &= \alpha \mathcal{B}_n + \beta \mathcal{B}_{n+1}, \\ \mathcal{B}''_n &= \alpha' \mathcal{B}_n + \beta' \mathcal{B}_{n+1}, \end{aligned}$$

where in order to preserve orthonormality, and to have real-valued impulse

response, the coefficients α, β, α' and β' must satisfy²

$$\begin{bmatrix} \alpha'\\ \beta' \end{bmatrix} = \frac{1}{(\xi_n - \overline{\xi_n})\sqrt{1 - \mu^2}} \begin{bmatrix} \xi_n & -1\\ -\overline{\xi_n} & 1 \end{bmatrix} \begin{bmatrix} \mu & 1\\ -1 & -\mu \end{bmatrix} \begin{bmatrix} 1 & 1\\ \overline{\xi_n} & \xi_n \end{bmatrix} \begin{bmatrix} \alpha\\ \beta \end{bmatrix}$$
(3.15)

where we have defined

$$\mu \triangleq \frac{\xi_n + \overline{\xi_n}}{1 + |\xi_n|^2}.$$

When only one fixed complex mode $\xi_k = \xi$ is considered and the following choice for α and β , satisfying (3.15), is made

$$\alpha = -\beta = \frac{\sqrt{(1-\mu^2)(1+|\xi|^2)}}{\overline{\xi}-\xi},$$

then the Kautz basis are obtained.

The derivation of the basis (3.14) can be done in different ways. For instance, in [Wal32, NG94b] it is shown how the basis can be derived using the Gram-Schmidt orthonormalization procedure [You88] on the set of functions $\{\mathcal{A}_k(z)\}_{k=0}^{p-1}$ defined as

$$\mathcal{A}_k(z) = \frac{1}{z - \xi_k},$$

with the same fixed poles $\{\xi_0, \xi_1, \cdots, \xi_{p-1}\}$.

In [NG97] it is shown that, under certain conditions on the poles $\{\xi_k\}$, the set $\{z\mathcal{B}_k(z)\}_{p=0}^{\infty}$ is a complete orthonormal set in the space $H_2(\mathbb{T})$, and that under the same conditions the set $\{\mathcal{B}_k(z)\}_{p=0}^{\infty}$ spans the subset of $H_2(\mathbb{T})$ of all rational, causal, stable transfer functions so that we can have the following unique series representation

$$G(z) = \sum_{k=0}^{\infty} \theta_k \mathcal{B}_k(z).$$

The necessary and sufficient condition for the completeness of the basis functions in the space $H_2(\mathbb{T})$ is given in the following theorem due to Ninness and co-workers [NG97].

Theorem 3.5.1. [NG97] Consider the basis functions defined in (3.14). Then Span $\{\mathcal{B}_k(e^{j\omega})\}$ is dense in $H_2(\mathbb{T})$ if and only if

$$\sum_{k=0}^{\infty} (1 - |\xi_k|) = \infty.$$
 (3.16)

²For the derivation of this result see [NG97].

Proof: See [NG97]. The proof is based on the fact that a set is dense in a Hilbert space if and only if there does not exist a non-zero element of the space that is orthogonal to all the elements in the set. A completely different proof, based on some results by Szász ([Cle63]), is given in [Oli95e]. ■

The series expansion of the system in terms of the orthonormal bases is schematically represented in Figure 3.2, where we have defined

$$\begin{split} A_k(q) &\triangleq \quad \frac{1 - \overline{\xi_k}q}{q - \xi_k} \\ F_k(q) &\triangleq \quad \frac{\sqrt{1 - |\xi_k|^2}}{q - \xi_k} \\ \end{split} \qquad (\text{all-pass section}), \end{split}$$



Figure 3.2: Schematic Representation of the Series Expansion in terms of Orthonormal Bases with Fixed Poles

It follows that this family of orthonormal bases can be generated by the cascade connection of (different) first order all-pass sections $(A_k(q))$, followed by a first order low pass section $(F_k(q))$.

Remark 3.5.1. Notice the similarity between Figures 3.2 and 3.1. In [Bod95] it is shown how a generalized family of orthonormal bases in $H_2(\mathbb{T})$ can be generated by the cascade connection of stable all-pass filters with input balance realizations.

The Christoffel-Darboux formula for the reproducing kernel for these bases has been derived by Ninness and co-workers in [NHG97a], and is given in the following theorem.

Theorem 3.5.2. [NHG97a] Define the Blaschke product-like quantity

$$\varphi_p(z) \triangleq \prod_{k=0}^{p-1} \frac{1 - \overline{\xi_k} z}{z - \xi_k}.$$

Then the Reproducing Kernel associated to the orthonormal bases with fixed poles $\{B_k(z)\}$ as defined in (3.14) can be expressed as

$$K_p(z,\mu) = \frac{1 - \varphi_p(\mu)\varphi_p(z)}{z\overline{\mu} - 1},$$
(3.17)

with |z| > 1 and $|\mu| > 1$.

Proof: See [NHG97a]. An alternative proof, by induction, is given in Appendix 3.A.

3.5.1 A minimal state-space realization for the OBFP

In this subsection we derive a minimal state-space realization for the orthonormal expansion of the system in terms of the OBFP. The availability of a closed form expression for a minimal state-space realization will be important in the context of system identification since it will allow the implementation of reliable algorithms that can provide closed form estimates directly from input-output data. In addition, a state-space description of the identified system will be important for the purposes of simulation, control design, model order reduction [Glo84], or in the case of digital filters, for an actual hardware implementation of the system [RM87].

A state-space realization of the orthonormal expansion in terms of the OBFP can be obtained from the filter structure of Figure 3.2 by giving each (first order) all-pass section $A_k(q)$ and each (first order) low-pass section $F_k(q)$ a minimal state-space realization (which obviously will be one-dimensional) and then connecting these elemental blocks to obtain the state space realization of the compound system. It is clear that a realization obtained in this way will be in general non-minimal. Indeed, for the case of having p different poles, this procedure will yield a 2p-dimensional state space realization, while a minimal realization should be of dimension p, since only p different modes are present on the system. The reason for this is that in the filter structure of Figure 3.2 each pole ξ_k appears in both $A_k(q)$ and $F_k(q)$ sections.

An (input-output) equivalent filter structure where each pole appears in only one first order section is represented in Figure 3.3. In the following we will show that the procedure described in the previous paragraph, but applied to the filter structure of Figure 3.3, will yield a minimal state space realization for the OBFP.

Let us consider first the elemental first order section represented in Figure 3.4. An equivalent representation is given in Figure 3.5, where the associated state variable x_k^{ℓ} has been defined. From this diagram it is straightforward to write the following (minimal) state space realization

$$x_{k+1}^{\ell} = \xi_{\ell} x_{k}^{\ell} + (1 - \xi_{\ell} \xi_{\ell-1}) u_{k}^{\ell}, \qquad (3.18)$$

$$\widetilde{y}_k^\ell = x_k^\ell - \xi_{\ell-1} u_k^\ell. \tag{3.19}$$



Figure 3.3: An alternative filter structure for the Orthonormal Bases with Fixed Poles.



Figure 3.4: Elemental first order section in the filter structure of Figure 3.3.

The associated matrices of the state space description are $(A^{\ell}, B^{\ell}, C^{\ell}, D^{\ell}) = (\xi_{\ell}, (1 - \xi_{\ell}\xi_{\ell-1}), 1, -\xi_{\ell-1}).$

Let us consider now the cascade of, say, three of these elemental sections. This cascade connection is represented in Figure 3.6.

Defining

$$x_k \triangleq \left(x_k^{\ell-1}, x_k^{\ell}, x_k^{\ell+1}\right)^T, \qquad (3.20)$$

$$\widetilde{y}_{k} \triangleq \left(\widetilde{y}_{k}^{\ell-1}, \widetilde{y}_{k}^{\ell}, \widetilde{y}_{k}^{\ell+1}\right)^{T},$$
(3.21)

the state space realization of the compound system can be written as

$$\begin{aligned} x_{k+1} &= \begin{bmatrix} A^{\ell-1} & 0 & 0 \\ B^{\ell}C^{\ell-1} & A^{\ell} & 0 \\ B^{\ell+1}D^{\ell}C^{\ell-1} & B^{\ell+1}C^{\ell} & A^{\ell+1} \end{bmatrix} x_k + \begin{bmatrix} B^{\ell-1} \\ B^{\ell}D^{\ell-1} \\ B^{\ell+1}D^{\ell}D^{\ell-1} \end{bmatrix} u_k^{\ell-1}, \\ \widetilde{y}_k &= \begin{bmatrix} C^{\ell-1} & 0 & 0 \\ D^{\ell}C^{\ell-1} & C^{\ell} & 0 \\ D^{\ell+1}D^{\ell}C^{\ell-1} & D^{\ell+1}C^{\ell} & C^{\ell+1} \end{bmatrix} x_k + \begin{bmatrix} D^{\ell-1} \\ D^{\ell}D^{\ell-1} \\ D^{\ell+1}D^{\ell}D^{\ell-1} \end{bmatrix} u_k^{\ell-1}. \end{aligned}$$



Figure 3.5: An internal representation of the elemental first order section of Figure 3.4.



Figure 3.6: Cascade connection of elemental first order sections.

We are now able to write a state space realization for an arbitrary number of sections. For the case of p sections the minimal realization is given by

$$\begin{aligned} x_{k+1} &= Ax_k + Bu_k, \\ \widetilde{y}_k &= \widetilde{C}x_k + \widetilde{D}u_k, \end{aligned}$$

where now

$$\begin{aligned} x_k &\triangleq \left(x_k^0, x_k^1, \cdots, x_k^{p-1}\right)^T, \\ \widetilde{y}_k &\triangleq \left(\widetilde{y}_k^0, \widetilde{y}_k^1, \cdots, \widetilde{y}_k^{p-1}\right)^T, \end{aligned}$$

and the matrices A, B, \widetilde{C} and \widetilde{D} are given by

$$A = \begin{bmatrix} A^{0} & 0 & \cdots & 0 \\ B^{1}C^{0} & A^{1} & \cdots & 0 \\ B^{2}D^{1}C^{0} & B^{2}C^{1} & \cdots & 0 \\ B^{3}D^{2}D^{1}C^{0} & B^{3}D^{2}C^{1} & \cdots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ B^{p-1}D^{p-2}\cdots D^{1}C^{0} & B^{p-1}D^{p-2}\cdots D^{2}C^{1} & \cdots & A^{p-1} \end{bmatrix},$$
(3.22)

$$B = \begin{bmatrix} B^{0} \\ B^{1}D^{0} \\ B^{2}D^{1}D^{0} \\ \vdots \\ B^{p-1}D^{p-2}\cdots D^{1}D^{0} \end{bmatrix} ; \qquad \widetilde{D} = \begin{bmatrix} D^{0} \\ D^{1}D^{0} \\ D^{2}D^{1}D^{0} \\ \vdots \\ D^{p-1}D^{p-2}\cdots D^{1}D^{0} \end{bmatrix},$$

$$\widetilde{C} = \begin{bmatrix} C^{0} & 0 & \cdots & 0 \\ D^{1}C^{0} & C^{1} & \cdots & 0 \\ D^{2}D^{1}C^{0} & D^{2}C^{1} & \cdots & 0 \\ D^{3}D^{2}D^{1}C^{0} & D^{3}D^{2}C^{1} & \cdots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ D^{p-1}D^{p-2}\cdots D^{1}C^{0} & D^{p-1}D^{p-2}\cdots D^{2}C^{1} & \cdots & C^{p-1} \end{bmatrix},$$
(3.23)

and where for $\ell = 1, \cdots, p-1$

$$\begin{aligned}
A^{\ell} &= \xi_{\ell}, \\
B^{\ell} &= 1 - \xi_{\ell} \xi_{\ell-1}, \\
C^{\ell} &= 1, \\
D^{\ell} &= -\xi_{\ell-1}.
\end{aligned}$$

The case $\ell = 0$ (that corresponds to the first section in Figure 3.3) is different. It is not difficult to show that a minimal realization for this section is $(A^0, B^0, C^0, D^0) = (\xi_0, 1, 1, 0)$. This simplifies the expressions of matrices B and \widetilde{D} to

$$B = (1, 0, \cdots, 0)^T, \tag{3.24}$$

and $\widetilde{D} = 0$, respectively.

Considering that for $\ell = 0, \cdots, p-1$ we have

$$\widehat{y}_k^\ell = \sqrt{1-\xi_\ell^2} \ \widetilde{y}_k^\ell,$$

we can write

$$\widehat{y}_k = \Lambda \ \widetilde{y}_k = \Lambda \widetilde{C} x_k$$

where the matrix $\boldsymbol{\Lambda}$ is defined as

$$\Lambda \triangleq \begin{bmatrix} \sqrt{1-\xi_0^2} & 0 & \cdots & 0 \\ 0 & \sqrt{1-\xi_1^2} & \cdots & 0 \\ 0 & 0 & \ddots & 0 \\ 0 & 0 & \cdots & \sqrt{1-\xi_{p-1}^2} \end{bmatrix}.$$
 (3.25)

Finally, the output y_k can be computed as

$$y_k = (\theta_0, \theta_1, \cdots, \theta_{p-1}) \, \widehat{y}_k = \theta^T \Lambda \widehat{C} \, x_k.$$
(3.26)

Summarizing, a minimal state-space realization for the OBFP is given by the quadruplet

$$(A, B, \theta^T \Lambda \widetilde{C}, 0),$$

where matrices A, B, \tilde{C} , and Λ are defined in equations (3.22), (3.24), (3.23) and (3.25) respectively.

3.6 Orthonormal Bases on $H_2^{m \times n}(\mathbb{T})$

In this section we show how orthonormal bases for the space $H_2^{m \times n}(\mathbb{T})$ of stable and causal $(m \times n)$ transfer matrices, can be generated from orthonormal bases on $H_2(\mathbb{T})$. The result is summarized in the following theorem.

Theorem 3.6.1. Let $\{\mathcal{B}_{\ell}(z)\}_{\ell=0}^{\infty}$ be a complete orthonormal set (i.e., an orthonormal basis) on the Hilbert space $H_2(\mathbb{T})$ with the usual definition for the inner product:

$$\langle \mathcal{B}_{\ell}, \mathcal{B}_{k} \rangle = \frac{1}{2\pi} \int_{-\pi}^{\pi} \mathcal{B}_{\ell}(e^{j\omega}) \overline{\mathcal{B}_{k}(e^{j\omega})} d\omega = \frac{1}{2\pi j} \oint_{T} \mathcal{B}_{\ell}(z) \overline{\mathcal{B}_{k}(1/z)} \frac{dz}{z}$$
(3.27)

and let $\{\mathcal{B}_{\ell}^{ij}(z)\}_{\ell=0}^{\infty}$ $(i = 1, \ldots, m; j = 1, \ldots, n)$, be a set of transfer matrices whose elements belong to $H_2(\mathbb{T})$, and which are defined as:

$$\mathcal{B}_{\ell}^{ij}(z) = \begin{bmatrix} 0 & \dots & 0 & \dots & 0 \\ \vdots & \ddots & \vdots & \ddots & \vdots \\ 0 & \dots & \mathcal{B}_{\ell}(z) & \dots & 0 \\ \vdots & \ddots & \vdots & \ddots & \vdots \\ 0 & \dots & 0 & \dots & 0 \end{bmatrix} \leftarrow i$$
(3.28)

Then $\{\mathcal{B}_{\ell}^{ij}(z)\}_{\ell=0}^{\infty}$, $(i = 1, \ldots, m; j = 1, \ldots, n)$ is a complete orthonormal set in $H_2^{m \times n}(\mathbb{T})$, with the usual definition for an inner product on a space of matrix valued functions

$$\left\langle \mathcal{B}_{\ell}^{ij}, \mathcal{B}_{k}^{st} \right\rangle = \frac{1}{2\pi} \int_{-\pi}^{\pi} \operatorname{Tr} \left\{ \mathcal{B}_{\ell}^{ij}(e^{j\omega}) \mathcal{B}_{k}^{st}(e^{j\omega})^{*} \right\} d\omega.$$
(3.29)

Proof: See Appendix 3.A.

Assuming that $\{\mathcal{B}_{\ell}^{ij}(z)\}_{\ell=0}^{\infty}$, $(i = 1, \ldots, m; j = 1, \ldots, n)$ is an orthonormal basis for the space $H_2^{m \times n}(\mathbb{T})$ then any stable causal tranfer matrix G(z) in $H_2^{m \times n}(\mathbb{T})$ can be approximated by a linear combination of a finite number of elements of the orthonormal set, that is

$$G(z,\Theta) = \sum_{\ell=0}^{p-1} \sum_{i=1}^{m} \sum_{j=1}^{n} \theta_{\ell}^{ij} \mathcal{B}_{\ell}^{ij}(z).$$
(3.30)

By choosing $\{\mathcal{B}_{\ell}(z)\}_{\ell=0}^{\infty}$ as the orthonormal bases with fixed poles of Section 3.5, the previous Theorem allows us to construct a MIMO version of these OBFP. In the following subsection we derive a minimal state-space realization for these MIMO bases.

3.6.1 A Minimal State-Space Realization for the MIMO-OBFP

The corresponding MIMO version of the filter structure in Figure 3.3 is represented in Figure 3.7 for the case m = n = 2.

A minimal state-space realization for these MIMO-OBFP can be derived from the filter structure in Figure 3.7 by applying a procedure similar to the one employed in the derivation of the minimal realization for the scalar case in Subsection 3.5.1.

We consider first the elemental first order section represented in Figure 3.8, where the second superscript in the input and output variables indicates the corresponding input channel.

An equivalent representation is given in Figure 3.9, where the associated state variable $x_k^{\ell,i}$ has been defined. The following (minimal) state-space realization can be derived from the diagram in Figure 3.9

$$x_{k+1}^{\ell,i} = \xi_{\ell} x_k^{\ell,i} + (1 - \xi_{\ell} \xi_{\ell-1}) u_k^{\ell,i}, \qquad (3.31)$$

$$\widetilde{y}_{k}^{\ell,i} = x_{k}^{\ell,i} - \xi_{\ell-1} u_{k}^{\ell,i}.$$
(3.32)

The associated matrices of the state space description are $(A^{\ell,i}, B^{\ell,i}, C^{\ell,i}, D^{\ell,i}) = (\xi_{\ell}, (1 - \xi_{\ell}\xi_{\ell-1}), 1, -\xi_{\ell-1}).$

Defining

$$\begin{aligned} x_k^{\ell} &\triangleq \left[x_k^{\ell,1}, x_k^{\ell,2}, \cdots, x_k^{\ell,n} \right]^T, \\ \widetilde{y}_k^{\ell} &\triangleq \left[\widetilde{y}_k^{\ell,1}, \widetilde{y}_k^{\ell,2}, \cdots, \widetilde{y}_k^{\ell,n} \right]^T, \\ \widetilde{u}_k^{\ell} &\triangleq \left[\widetilde{u}_k^{\ell,1}, \widetilde{u}_k^{\ell,2}, \cdots, \widetilde{u}_k^{\ell,n} \right]^T, \end{aligned}$$

we can write in matrix form

$$\begin{array}{rcl} x^\ell_{k+1} &=& A^\ell x^\ell_k + B^\ell \widetilde{u}^\ell_k, \\ \widetilde{y}^\ell_k &=& C^\ell x^\ell_k + D^\ell \widetilde{u}^\ell_k, \end{array}$$



Figure 3.7: Filter structure for the Multivariable Orthonormal Bases with Fixed Poles (Case m = n = 2).

where

$$A^{\ell} \triangleq \begin{bmatrix} A^{\ell,1} & 0 & \cdots & 0 \\ 0 & A^{\ell,2} & \cdots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & \cdots & A^{\ell,n} \end{bmatrix} , \quad B^{\ell} \triangleq \begin{bmatrix} B^{\ell,1} & 0 & \cdots & 0 \\ 0 & B^{\ell,2} & \cdots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & \cdots & B^{\ell,n} \end{bmatrix},$$
$$C^{\ell} \triangleq \begin{bmatrix} C^{\ell,1} & 0 & \cdots & 0 \\ 0 & C^{\ell,2} & \cdots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & \cdots & C^{\ell,n} \end{bmatrix} , \quad D^{\ell} \triangleq \begin{bmatrix} D^{\ell,1} & 0 & \cdots & 0 \\ 0 & D^{\ell,2} & \cdots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & \cdots & D^{\ell,n} \end{bmatrix}.$$

Considering the definitions of $A^{\ell,i}, B^{\ell,i}, C^{\ell,i}$ and $D^{\ell,i}$, the expressions for A^ℓ, B^ℓ, C^ℓ



Figure 3.8: Elemental first order section in the filter structure of Figure 3.7.



Figure 3.9: An internal representation of the elemental first order section of Figure 3.8.

and D^ℓ reduce to

$$\begin{array}{rcl}
A^{\ell} &=& \xi_{\ell} I_{n}, \\
B^{\ell} &=& (1 - \xi_{\ell} \xi_{\ell-1}) I_{n}, \\
C^{\ell} &=& I_{n}, \\
D^{\ell} &=& -\xi_{\ell-1} I_{n}.
\end{array}$$

The filter structure of Figure 3.7 can now be considered as the cascade connection of p sections with minimal realizations given by the quadruplets $(A^{\ell}, B^{\ell}, C^{\ell}, D^{\ell})$ with $\ell = 0, \cdots, p-1$. Defining

$$\begin{aligned} x_k &\triangleq \left[x_k^0, x_k^{\ell,1}, \cdots, x_k^{p-1} \right]^T, \\ \widetilde{y}_k &\triangleq \left[\widetilde{y}_k^0, \widetilde{y}_k^1, \cdots, \widetilde{y}_k^{p-1} \right]^T, \end{aligned}$$

we can then write

$$\begin{aligned} x_{k+1} &= Ax_k + Bu_k, \\ \widetilde{y}_k &= \widetilde{C}x_k + \widetilde{D}u_k, \end{aligned}$$

where the matrices A,B,\widetilde{C} and \widetilde{D} are given by

$$A = \begin{bmatrix} A^{0} & 0 & \cdots & 0 \\ B^{1}C^{0} & A^{1} & \cdots & 0 \\ B^{2}D^{1}C^{0} & B^{2}C^{1} & \cdots & 0 \\ B^{3}D^{2}D^{1}C^{0} & B^{3}D^{2}C^{1} & \cdots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ B^{p-1}D^{p-2}\cdots D^{1}C^{0} & B^{p-1}D^{p-2}\cdots D^{2}C^{1} & \cdots & A^{p-1} \end{bmatrix},$$
 (3.33)

$$B = \begin{bmatrix} B^{0} \\ B^{1}D^{0} \\ B^{2}D^{1}D^{0} \\ \vdots \\ B^{p-1}D^{p-2}\cdots D^{1}D^{0} \end{bmatrix} ; \qquad \widetilde{D} = \begin{bmatrix} D^{0} \\ D^{1}D^{0} \\ D^{2}D^{1}D^{0} \\ \vdots \\ D^{p-1}D^{p-2}\cdots D^{1}D^{0} \end{bmatrix},$$

$$\widetilde{C} = \begin{bmatrix} C^{0} & 0 & \cdots & 0 \\ D^{1}C^{0} & C^{1} & \cdots & 0 \\ D^{2}D^{1}C^{0} & D^{2}C^{1} & \cdots & 0 \\ D^{3}D^{2}D^{1}C^{0} & D^{3}D^{2}C^{1} & \cdots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ D^{p-1}D^{p-2}\cdots D^{1}C^{0} & D^{p-1}D^{p-2}\cdots D^{2}C^{1} & \cdots & C^{p-1} \end{bmatrix}.$$
(3.34)

The expressions for B and D can be further simplified by noting that $B^0 = I_n$ and $D^0 = 0$, so that

$$B = \left[I_n, \mathbb{O}_n, \cdots, \mathbb{O}_n\right]^T, \qquad (3.35)$$

and D = 0, where \mathbb{O}_n is an $(n \times n)$ matrix of zeros.

Now, considering that

$$\widehat{y}_k^{\ell,i} = \sqrt{1 - \xi_\ell^2} \ \widetilde{y}_k^{\ell,i},$$

then

$$\widehat{y}_k^\ell = \sqrt{1 - \xi_\ell^2} \ \widetilde{y}_k^\ell,$$

so that

$$\widehat{y}_{k} \triangleq \begin{bmatrix} \widehat{y}_{k}^{0} \\ \widehat{y}_{k}^{1} \\ \vdots \\ \widehat{y}_{k}^{p-1} \end{bmatrix} = (\Lambda \otimes I_{n}) \ \widetilde{y}_{k},$$

where matrix Λ is given by equation (3.25), and \otimes stands for the Kronecker product (see Appendix A).

Finally, the j-th output is given by

$$y_k^j = \sum_{\ell=0}^{p-1} \left[\theta_\ell^{j1}, \theta_\ell^{j2}, \cdots, \theta_\ell^{jn} \right] \widehat{y}_k^\ell,$$
$$= \left[\theta_0^j, \theta_1^j, \cdots, \theta_{p-1}^j \right] \widehat{y}_k,$$

where we have defined

$$\theta_{\ell}^{j} \triangleq \left[\theta_{\ell}^{j1}, \theta_{\ell}^{j2}, \cdots, \theta_{\ell}^{jn}\right].$$

Vectorizing we then have that the output vector is given by

$$y_{k} \triangleq \begin{bmatrix} y_{k}^{1} \\ y_{k}^{2} \\ \vdots \\ y_{k}^{m} \end{bmatrix} = \begin{bmatrix} \theta_{0}^{1} & \theta_{1}^{1} & \cdots & \theta_{p-1}^{1} \\ \theta_{0}^{2} & \theta_{1}^{2} & \cdots & \theta_{p-1}^{2} \\ \vdots & \vdots & \ddots & \vdots \\ \theta_{0}^{m} & \theta_{1}^{m} & \cdots & \theta_{p-1}^{m} \end{bmatrix} \widehat{y}_{k} = \Theta^{T} \widehat{y}_{k} = \Theta^{T} (\Lambda \otimes I_{n}) \widetilde{C} x_{k}.$$

Summarizing, a minimal state-space realization for the MIMO-OBFP is given by the quadruplet

$$(A, B, \Theta^T (\Lambda \otimes I_n) \widetilde{C}, 0),$$

where matrices A, B, \tilde{C} , and Λ are defined in equations (3.33), (3.35), (3.34) and (3.25) respectively. The realization is of order np.

3.7 Conclusions

In this chapter, a review of rational orthonormal bases on the unit circle was presented. The concept of reproducing kernel associated with the bases was introduced, and the approximating properties of the more common orthonormal families in $H_2(\mathbb{T})$ were described. Emphasis was put on the Orthonormal Bases with Fixed Poles, since this family has more flexibility in the choice of the pole locations and encompasses the more common FIR, Laguerre and Kautz model structures in a unified formulation. For this family, a closed form expression for the reproducing kernel was given, and a minimal state-space realization was derived. This last issue is important in the context of system identification since it will allow the implementation of simple and reliable algorithms that will provide estimates in closed form (in state-space form) directly from inputoutput data.

APPENDICES

3.A Proofs for Chapter 3

Proof of Lemma 3.2.1 Let us compute first the inner product $\langle G(z), \frac{z\overline{\mu}}{z\overline{\mu}-1} \rangle$. We have

$$\begin{split} \left\langle G(z), \frac{z\overline{\mu}}{z\overline{\mu}-1} \right\rangle &= \frac{1}{2\pi j} \oint_{\mathbb{T}} G(z) \frac{\mu/z}{(\mu/z)-1} \frac{dz}{z}, \\ &= \frac{1}{2\pi j} \oint_{\mathbb{T}} G(z) \frac{1/z}{(1/z)-(1/\mu)} \frac{dz}{z} \\ &= \frac{1}{2\pi j} \oint_{\mathbb{T}} \frac{G(1/z)}{z-(1/\mu)} dz, \\ &= G(\mu), \end{split}$$

where in passing to the second last line the variable substitution $z \mapsto 1/z$ was used, and where in passing to the last line use was made of Cauchy Residue Theorem. Now, from equation (3.5) we also have

$$\langle G(z), K(z,\mu) \rangle = G(\mu),$$

so that for all $G(z) \in H_2(\mathbb{T})$, and for all $|\mu| > 1$, we can write

$$\left\langle G(z), \left(\frac{z\overline{\mu}}{z\overline{\mu}-1}-K(z,\mu)\right)\right\rangle = 0,$$

what concludes the proof, since G(z) is an arbitrary element of $H_2(\mathbb{T})$ and the only element of the space which is orthogonal to every other element of the space is the zero element.

Proof of Lemma 3.3.1 Considering the expression for the reproducing kernel in (3.2) we can write

$$\begin{split} K_{p}(z,\mu) &= \sum_{k=0}^{p-1} \mathcal{B}_{k}(z) \overline{\mathcal{B}_{k}(\mu)}, \\ &= \sum_{k=0}^{p-1} \frac{(1-\xi^{2})}{(z-\xi)(\overline{\mu}-\xi)} \left(\frac{1-\xi z}{z-\xi}\right)^{k} \left(\frac{1-\xi \overline{\mu}}{\overline{\mu}-\xi}\right)^{k}, \\ &= \frac{(1-\xi^{2})}{(z-\xi)(\overline{\mu}-\xi)} \sum_{k=0}^{p-1} \left(\frac{(1-\xi z)(1-\xi \overline{\mu})}{(z-\xi)(\overline{\mu}-\xi)}\right)^{k}, \\ &= \frac{(1-\xi^{2})}{(z-\xi)(\overline{\mu}-\xi)} \frac{1-\left(\frac{(1-\xi z)(1-\xi \overline{\mu})}{(z-\xi)(\overline{\mu}-\xi)}\right)^{p}}{1-\frac{(1-\xi z)(1-\xi \overline{\mu})}{(z-\xi)(\overline{\mu}-\xi)}}, \\ &= \frac{1-\left(\frac{(1-\xi z)(1-\xi \overline{\mu})}{(z-\xi)(\overline{\mu}-\xi)}\right)^{p}}{z\overline{\mu}-1}, \end{split}$$

where in passing to the fourth line use was made of the identity

$$\sum_{k=0}^{p-1} x^k = \frac{1-x^p}{1-x}.$$
(3.A.1)

Now the result follows by defining

$$\varphi_p(z) \triangleq \left(\frac{1-\xi z}{z-\xi}\right)^p.$$

Proof of Theorem 3.5.2 The proof proceeds by induction. We first prove that equation (3.17) holds for p = 1. Then, assuming that (3.17) holds for p = q - 1, we prove that it holds also for p = q, which concludes the proof.

Let p = 1, then

$$K_p(z,\mu) = \sum_{k=0}^{p-1} \overline{\mathcal{B}_k(\mu)} \mathcal{B}_k(z) = \overline{\mathcal{B}_0(\mu)} \mathcal{B}_0(z),$$
$$= \frac{1 - |\xi_0|^2}{(z - \xi_0)(\overline{\mu} - \overline{\xi_0})}.$$
(3.A.2)

On the other hand, for p = 1

$$\frac{1 - \overline{\varphi_p(\mu)}\varphi_p(z)}{z\overline{\mu} - 1} = \frac{1 - \overline{\varphi_1(\mu)}\varphi_1(z)}{z\overline{\mu} - 1},$$

$$= \frac{1 - \frac{(1 - \overline{\xi_0}z)(1 - \xi_0\overline{\mu})}{(z - \xi_0)(\overline{\mu} - \overline{\xi_0})},$$

$$= \frac{1 - |\xi_0|^2}{(z - \xi_0)(\overline{\mu} - \overline{\xi_0})}.$$
(3.A.3)

Therefor, the result holds for p = 1.

Let us now assume that equation (3.17) holds for p = q - 1, that is

$$\sum_{k=0}^{q-2} \overline{\mathcal{B}_k(\mu)} \mathcal{B}_k(z) = \frac{1 - \overline{\varphi_{q-1}(\mu)} \varphi_{q-1}(z)}{z\overline{\mu} - 1}$$

Then

$$\sum_{k=0}^{q-1} \overline{\mathcal{B}_{k}(\mu)} \mathcal{B}_{k}(z) = \overline{\mathcal{B}_{q-1}(\mu)} \mathcal{B}_{q-1}(z) + \sum_{k=0}^{q-2} \overline{\mathcal{B}_{k}(\mu)} \mathcal{B}_{k}(z),$$

$$= \overline{\mathcal{B}_{q-1}(\mu)} \mathcal{B}_{q-1}(z) + \frac{1 - \overline{\varphi_{q-1}(\mu)} \varphi_{q-1}(z)}{z\overline{\mu} - 1},$$

$$= \frac{(1 - |\xi_{q-1}|^{2})\overline{\varphi_{q-1}(\mu)} \varphi_{q-1}(z)}{(\overline{\mu} - \overline{\xi_{q-1}})(z - \xi_{q-1})} + \frac{1 - \overline{\varphi_{q-1}(\mu)} \varphi_{q-1}(z)}{z\overline{\mu} - 1},$$

$$= \frac{1}{z\overline{\mu} - 1} \left(1 - \frac{1 - \overline{\mu}\xi_{q-1}}{\overline{\mu} - \overline{\xi_{q-1}}} \overline{\varphi_{q-1}(\mu)} \frac{1 - z\overline{\xi_{q-1}}}{z - \xi_{q-1}} \varphi_{q-1}(z) \right),$$

$$= \frac{1 - \overline{\varphi_{q}(\mu)} \varphi_{q}(z)}{z\overline{\mu} - 1},$$
(3.A.4)

where in passing to the third line use was made of the identity

$$\mathcal{B}_q(z) \triangleq \frac{\sqrt{1-|\xi_q|^2}}{z-\xi_q} \prod_{k=0}^{q-1} \frac{1-\overline{\xi_k}z}{z-\xi_k} = \frac{\sqrt{1-|\xi_q|^2}}{z-\xi_q} \varphi_q(z).$$

Hence, equation (3.17) holds also for p = q. This completes the proof.

Proof of Theorem 3.6.1 The proof proceeds in two steps. First, it is shown that the set $\{\mathcal{B}_{\ell}^{ij}(z)\}$ is orthonormal with respect to the inner product in $H_2^{m \times n}(\mathbb{T})$, and then, that it is complete in that space.

i. Orthonormality:

It is clear that the matrices in the set $\{\mathcal{B}_{\ell}^{ij}(z)\}$ are linearly independent.

We will use the definition of the inner product in equation (3.29) to prove that $\{\mathcal{B}_{\ell}^{ij}(z)\}$ is an orthonormal set. Substituting $z = e^{j\omega}$, equation (3.29) can be written as:

$$\left\langle \mathcal{B}_{\ell}^{ij}, \mathcal{B}_{k}^{st} \right\rangle = \frac{1}{2\pi j} \oint_{T} \operatorname{Tr} \left\{ \mathcal{B}_{\ell}^{ij}(z) \mathcal{B}_{k}^{st}(1/z)^{\star} \right\} \frac{\mathrm{d}z}{z}.$$
 (3.A.5)

In order to prove that the matrices are orthonormal we have to check that the following conditions hold

$$\left\langle \mathcal{B}_{\ell}^{ij}, \mathcal{B}_{k}^{st} \right\rangle = \begin{cases} 0 & \text{if } \mathcal{B}_{\ell}^{ij} \neq \mathcal{B}_{k}^{st} \\ 1 & \text{if } \mathcal{B}_{\ell}^{ij} = \mathcal{B}_{k}^{st} \end{cases}$$
(3.A.6)

Let us consider first the inner product $\langle \mathcal{B}_{\ell}^{ij}, \mathcal{B}_{\ell}^{ij} \rangle$. We have

$$\begin{array}{lll} \left\langle \mathcal{B}_{\ell}^{ij}, \mathcal{B}_{\ell}^{ij} \right\rangle &=& \displaystyle \frac{1}{2\pi \mathrm{j}} \oint_{T} \mathrm{Tr} \left\{ \mathcal{B}_{\ell}^{ij}(z) \mathcal{B}_{\ell}^{ij}(1/z)^{\star} \right\} \frac{\mathrm{d}z}{z}, \\ (\text{by definition of } \mathcal{B}_{\ell}^{ij}) &=& \displaystyle \frac{1}{2\pi \mathrm{j}} \oint_{T} \mathcal{B}_{\ell}(z) \overline{\mathcal{B}_{\ell}(1/z)} \frac{\mathrm{d}z}{z}, \\ (\text{by the orthonormality of } \left\{ \mathcal{B}_{\ell} \right\}) &=& 1. \end{array}$$

$$(3.A.7)$$

Now, let us consider the inner product $\langle \mathcal{B}_{\ell}^{ij}, \mathcal{B}_{k}^{st} \rangle$ for the case $i \neq s$ for any j, t, ℓ, k , or $j \neq t$ for any i, s, ℓ, k . We have

$$\begin{array}{ll} \left\langle \mathcal{B}_{\ell}^{ij}, \mathcal{B}_{k}^{st} \right\rangle &=& \displaystyle \frac{1}{2\pi \mathrm{j}} \oint_{T} \mathrm{Tr} \left\{ \mathcal{B}_{\ell}^{ij}(z) \mathcal{B}_{k}^{st}(1/z)^{\star} \right\} \frac{\mathrm{d}z}{z}, \\ \text{by definition of } \mathcal{B}_{\ell}^{ij}) &=& 0. \end{array}$$
(3.A.8)

Finally, let us consider the case i = s, j = t and $\ell \neq k$. We have

$$\begin{array}{lll} \left\langle \mathcal{B}_{\ell}^{ij}, \mathcal{B}_{k}^{st} \right\rangle &=& \displaystyle \frac{1}{2\pi \mathrm{j}} \oint_{T} \mathrm{Tr} \left\{ \mathcal{B}_{\ell}^{ij}(z) \mathcal{B}_{k}^{st}(1/z)^{\star} \right\} \frac{\mathrm{d}z}{z}, \\ (\text{by definition of } \mathcal{B}_{\ell}^{ij}) &=& \displaystyle \frac{1}{2\pi \mathrm{j}} \oint_{T} \mathcal{B}_{\ell}(z) \overline{\mathcal{B}_{k}(1/z)} \frac{\mathrm{d}z}{z}, \quad (3.A.9) \\ (\text{by the orthonormality of } \left\{ \mathcal{B}_{\ell} \right\}) &=& 0. \end{array}$$

Hence, we can conclude that $\{\mathcal{B}_{\ell}^{ij}(z)\}$ is an orthonormal set in the space $H_2^{m \times n}(\mathbb{T})$.

ii. Completeness:

(

We use the idea that an orthonormal set in a Hilbert space H is complete if and only if the only element of H which is orthogonal to every element of the set is the null element. Suppose, in order to obtain a contradiction, that there exist a nonzero matrix $F(z) = (f_{ij}(z)) \in H_2^{m \times n}(\mathbb{T})$ which is orthonormal to every element of the set $\{\mathcal{B}_{\ell}^{ij}(z)\}$. Suppose also, without loss of generality, that the entry $f_{st}(z)$ of F(z) is a nonzero function. Then, the inner product of F with any one of the matrices in $\{\mathcal{B}_{\ell}^{ij}(z)\}$ is given by

$$\langle F, \mathcal{B}_{k}^{st} \rangle = \frac{1}{2\pi j} \oint_{T} \operatorname{Tr} \left\{ F(z) \mathcal{B}_{k}^{st} (1/z)^{\star} \right\} \frac{\mathrm{d}z}{z},$$
(by definition of \mathcal{B}_{ℓ}^{ij}) = $\frac{1}{2\pi j} \oint_{T} f_{st}(z) \overline{\mathcal{B}_{k}(1/z)} \frac{\mathrm{d}z}{z},$ (3.A.11)
= 0. (3.A.12)

where the last equality should hold for each k, which means that there exist a nonzero function, namely $f_{st}(z)$, which is orthogonal to every element in $\{\mathcal{B}_{\ell}(z)\}$, which represents a contradiction, since the set $\{\mathcal{B}_{\ell}(z)\}$ is complete by hypothesis. Hence, we can conclude that the orthonormal set $\{\mathcal{B}_{\ell}^{ij}(z)\}$ is complete in $H_2^{m \times n}(\mathbb{T})$.

-		-

4

SISO Identification using Orthonormal Bases

In this chapter, the problem of least squares identification (from input-output data in the time domain) of Discrete-Time (DT), Linear Time-Invariant (LTI), Single-Input Single-Output (SISO) systems represented using orthonormal model structures will be analyzed. We concentrate on identification using the orthonormal bases with fixed poles of Section 3.5, that have the most common FIR, Laguerre and Kautz bases as special cases. The estimation accuracy will be quantified by providing bounds on the undermodelling error and by deriving an asymptotic (in model order and data-length) expression for the noise induced error. Fundamental for the analysis of the noise induced error will be the extension, to the OBFP, of some known results on convergence of Toeplitz matrices available for FIR model structures.

4.1 Introduction

In the last years there has been significant interest in the use of orthonormal basis functions for approximation of dynamical systems [GKB89, WAH, Mäk90a, Mäk90b, Par91, Oli95a, Oli94a], system identification [KP79, Lju85, LY85, Nur87, Wah91b, Mäk91, Wah94b, NG97, VHB95, HbVB95, PT91], signal processing [KP77, Wil95, Pdd93, WZ96, MJM89, den93b], and control applications [ZDP90, ZBD88, ZD88]. Particularly in the area of system identification, several schemes have been proposed for the identification of linear systems from input-output data using orthonormal model structures and least squares techniques [Lju85, LY85, Wah91b, Wah94b, NG94a, NG97, VHB95, HbVB95]. In these methods, the transfer function of the system, say G(z) for the discrete time case, is represented as a series expansion in terms of orthonormal basis functions { $\mathcal{B}_k(z)$ }, which are stable-causal transfer functions, and then the identification is performed by estimating a finite number of expansion coefficients using least squares techniques. One of the main motivations for using orthonormal bases to represent the system is that the resulting model structure becomes linear in the parameters (a linear regressor form), with the regressors depending only on the observed input signal. It is well known that in this case the least squares estimate has a closed form solution which corresponds to a global minimum of the quadratic criterion [Lju87, SS89]. In this way, the need for costly iterative optimization procedures for the parameter estimation, and the associated problems of local minima are avoided.

A second factor that has been pointed out in the literature [Wah91b, Wah91a, Wah94b, Oli95a, Bod95] is the numerical robustness of these methods when compared to estimation using non-orthonormal structures. This is so since a worst case numerical conditioning of the least squares estimation problem can be guaranteed for the case of using orthonormal structures while in general this result can not be established for the non-orthonormal case. We defer the study of this issue until the next chapter, since it is in the context of multivariable systems, in which a large number of parameters need to be estimated, where the problem of numerical robustness is more relevant [Vd94a, Vib94].

Besides the above mentioned advantages, a third aspect that we want to emphasize in this thesis, following the lead of Ninness and co-workers [NHG97a], is the use of orthonormal bases as an analysis tool which is particularly suited for the study of estimation methods that employ fixed denominator model structures. This is so because fixed denominator model structures can be linearly re-parameterized using orthonormal bases with the same fixed poles. The analysis of estimation accuracy can then be carried out by extending convergence results of the well known trigonometric bases $\{e^{j\omega n}\}$ (i.e., results of classical Fourier analysis) to the more general orthonormal bases employed in this thesis.

In this chapter we concentrate on the study of identification of DT-LTI-SISO systems using orthonormal model structures and least squares techniques. The material in this chapter will be used as a paradigm to extend these results to the multivariable case in Chapter 5.

The remainder of the chapter is organized as follows. In Section 4.2 we consider the identification of DT-LTI-SISO systems from input-output data in the time domain using rational orthonormal bases and least squares techniques. We then particularized the study of the estimation accuracy for the case of using the OBFP introduced in Section 3.5. This analysis is carried out in Sections 4.3 and 4.4, where the bias error and the variance error are (respectively) considered. Fundamental in this analysis is the use of new results regarding the convergence of generalized Toeplitz-like matrices. These convergence results are also presented in this section. A new phenomenom of accuracy limitation that arises in the estimation using orthonormal bases with fixed poles is illustrated in Section 4.5. Specifically, it is shown that at a given frequency there is a trade-off between bias and variance errors regarding the choice of the poles of the bases.

4.2 **Problem Formulation**

It is assumed that the LTI-SISO system is described by the standard model

$$y_k = G(q) \ u_k + \nu_k, \tag{4.1}$$

and that an N point data record of input and output sequences $\{y_k, u_k\}_{k=0}^{N-1}$ is available for the identification of the assumed stable (unknown) transfer function G(q) describing the system dynamics.

In the model (4.1), $\{\nu_k\}$ is a zero-mean stationary sequence representing some measurement noise which is assumed to have finite variance $\mathbf{E} \{\nu_k^2\} = \sigma_{\nu}^2$, and to be uncorrelated from the input sequence $\{u_k\}$. It is also assumed that the input sequence $\{u_k\}$ is a quasi-stationary process [Lju87], with spectral density $\Phi_u(\omega)$.

Let $\{\mathcal{B}_k(z)\}_{k=0}^{\infty}$ be an orthonormal basis in $H_2(\mathbb{T})$. Now, since the system was assumed to be asymptotically stable, then its transfer function G(z) belongs to $H_2(\mathbb{T})$, and can be uniquely represented by the series expansion

$$G(z) = \sum_{k=0}^{\infty} \theta_k \mathcal{B}_k(z), \qquad (4.2)$$

where θ_k are the 'Generalized Fourier Coefficients' defined as

$$\theta_k \triangleq \langle G, \mathcal{B}_k \rangle \qquad (k = 0, 1, \cdots).$$
 (4.3)

Of course, since the transfer function G(z) is unknown, the coefficients can not be computed as in (4.3). Instead, our objective will be to estimate the parameters of a finite dimensional model

$$G(z,\theta) \triangleq \sum_{k=0}^{p-1} \theta_k \mathcal{B}_k(z), \qquad (4.4)$$

so that the transfer function can be (approximately) identified by using the estimate $\widehat{\theta} \triangleq \left[\widehat{\theta}_0, \widehat{\theta}_1, \cdots, \widehat{\theta}_{p-1}\right]^T$ of the parameter vector $\theta \triangleq \left[\theta_0, \theta_1, \cdots, \theta_{p-1}\right]^T$ as follows

$$G(z,\widehat{\theta}) \triangleq \sum_{k=0}^{p-1} \widehat{\theta}_k \mathcal{B}_k(z).$$
(4.5)

It remains now to choose the parameter estimation method. The model structure (4.4) leads to the linear regressor form

$$y_k = \phi_k^T \ \theta + \nu_k,$$

where the regressor vector ϕ_k is defined as

$$\phi_k^T \triangleq [\mathcal{B}_0(q)u_k, \mathcal{B}_1(q)u_k, \cdots, \mathcal{B}_{p-1}(q)u_k].$$

The obvious choice for the parameter estimation method is the 'least squares' one, since it provides a closed form solution, and leads asymptotically to an efficient estimate.

The least squares estimate $\widehat{\theta}$ of θ is the minimizing argument of the quadratic criterion

$$V_N(\theta) = \frac{1}{N} \sum_{k=0}^{N-1} (y_k - \phi_k^T \theta)^2.$$
 (4.6)

That is

$$\widehat{\theta} \triangleq \operatorname*{arg\,min}_{\theta \in \mathbb{R}^p} \left\{ \frac{1}{N} \sum_{k=0}^{N-1} (y_k - \phi_k^T \theta)^2 \right\}.$$

It is well known that the solution of this minimization problem can be written in closed form as [Lju87, SS89, GP77]

$$\widehat{\theta} = R_p(N)^{-1} \left\{ \frac{1}{N} \sum_{k=0}^{N-1} \phi_k y_k \right\},$$
(4.7)

where

$$R_p(N) \triangleq \frac{1}{N} \sum_{k=0}^{N-1} \phi_k \phi_k^T, \qquad (4.8)$$

with the subscript p indicating the model order (number of terms in the parameterized model (4.4)).

Given the parameter estimate $\hat{\theta}$, the transfer function estimate $G(z, \hat{\theta})$ can then be computed as in equation (4.5).

Remark 4.2.1. For the case of $\{\mathcal{B}_k(z)\}_{k=0}^{\infty}$ being the Orthonormal Bases with fixed poles of Section 3.5, once an estimate $\hat{\theta}$ has been computed, a minimal state-space realization of the estimated model is immediately available by appealing to the results of Subsection 3.5.1. The described identification technique can be easily implemented in software, for instance in a MATLAB environment [Mat94].

We are interested now in analyzing the performance of the proposed system identification scheme. This performance will be evaluated by quantifying the estimation error. In the frequency domain, the estimation error can be written as

$$G(e^{j\omega},\widehat{\theta}) - G(e^{j\omega}) = \underbrace{G(e^{j\omega},\widehat{\theta}) - G(e^{j\omega},\theta_{\star})}_{\text{noise induced}} + \underbrace{G(e^{j\omega},\theta_{\star}) - G(e^{j\omega})}_{\text{undermodelling induced}}$$
(4.9)

where θ_{\star} is the convergence value of the estimate θ when the number of available data points N tends to infinity. We can then recognize two components of the estimation error:

• A component corresponding to the term

$$G(e^{j\omega}, \hat{\theta}) - G(e^{j\omega}, \theta_{\star}),$$

which is due to the noise corruption of the data. Typically, the size of this term is measured as ensemble average as

$$\mathbf{E}\left\{|G(e^{j\omega},\widehat{\theta}) - G(e^{j\omega},\theta_{\star})|^{2}\right\},\$$

and it is also called variance error.

• A component corresponding to the term

$$G(e^{j\omega}, \theta_{\star}) - G(e^{j\omega}),$$

which is due to the fact that the model (4.5) is too simple to represent the real system. We call this term *undermodelling error* or *bias error*.

In this thesis, the study of estimation accuracy will focus on using the Orthonormal Bases with Fixed Poles introduced in Section 3.5 as an effective analysis tool for quantifying bias and variance error. One of the contributions of this thesis will be to show how the bias and variance errors depend on the choice of the poles of the basis functions $\{\mathcal{B}_k(z)\}$.

The undermodelling error is analyzed in Section 4.3, and the noise induced error in Section 4.4.

4.3 Undermodelling Error

The undermodelling induced error arises from the parsimony of the model structure (4.4) (a finite-length series expansion) which cannot completely describe the true dynamics G(z). This error can be quantified in terms of the deviation between the real system and the model

$$|G(e^{j\omega}, \theta_{\star}) - G(e^{j\omega})|.$$

Unfortunately, the convergence value θ_{\star} of the estimate $\hat{\theta}$ is unknown. To quantify the undermodelling error we then compute the error involved in the approximation of G(z) with $\hat{G}_p(z)$ which is the best $H_2(\mathbb{T})$ approximation of G(z) belonging to the subspace spanned by the first p basis functions $\{\mathcal{B}_k(z)\}_{k=0}^{p-1}$.

As already mentioned, when compared to the FIR, Laguerre, Kautz or the more general OBGIF, the OBFP enjoy greater flexibility in the possible choice of the pole location without the need of a cyclic repetition. Unfortunately, there is a price to be paid for this increased flexibility in that these bases do not have the algebraic structure (3.1) so that the analysis of the undermodelling error can not be reduced to that of the FIR case by a simple transformation of the system or change of variables. This is not the case for the Laguerre, Kautz and OBGIF, all of them having the algebraic structure (3.1). As a consequence, the derivation of results quantifying the undermodelling error for the case of the OBFP is considerably more complicated. In [NHG97a], Ninness and co-workers derive an upper bound on the undermodelling error based on the Christoffel-Darboux formula for the reproducing kernel associated with the bases (equation (3.17)). The result is summarized in the following theorem.

Theorem 4.3.1. [NHG97a] Let the transfer function of the system G(z) have partial fraction expansion

$$G(z) = \sum_{\ell=0}^{r-1} \frac{\alpha_{\ell}}{z - \gamma_{\ell}},\tag{4.10}$$

where all the poles satisfy $|\gamma_{\ell}| < 1$. Let $\widehat{G}_p(z)$ denote the best H_2 approximation to G(z) with respect to the p basis functions $\{\mathcal{B}_k(z)\}_{k=0}^{p-1}$ as defined in (3.14), with poles $\{\xi_k\}_{k=0}^{p-1}$, i.e.

$$\widehat{G}_p(z) = \sum_{k=0}^{p-1} \langle G, \mathcal{B}_k \rangle \, \mathcal{B}_k(z).$$

Then

$$\left| G(e^{j\omega}) - \widehat{G}_p(e^{j\omega}) \right| < \sum_{\ell=0}^{r-1} \left| \frac{\alpha_\ell}{e^{j\omega} - \gamma_\ell} \right| \prod_{k=0}^{p-1} \left| \frac{\gamma_\ell - \xi_k}{1 - \overline{\xi_k} \gamma_\ell} \right|$$
(4.11)

Proof: The proof is given in Appendix 4.A.

It is obvious that the theorem also provides an upper bound in the undermodelling error for the FIR and Laguerre bases, since these bases are special cases of the OBFP corresponding to poles $\xi_k = 0, \forall k$, and $\xi_k = \xi \in \mathbb{R}, \forall k$, respectively.

It can be seen from equation (4.11) that if the poles of the system γ_{ℓ} are exactly known, then choosing $\xi_{\ell} = \gamma_{\ell}, \forall \ell$ gives a zero upper bound on the

undermodelling error. The result also implies that the convergence rate of the series expansion can be faster than that of the special cases of FIR, Laguerre, or Kautz basis, if the guesses for the poles ξ_k approach the true poles γ_k . To illustrate this, let us consider the following examples.

Example 4.3.1. Let the transfer function of the true system be

$$G(z) = \frac{0.8}{z+0.8},$$

and let us consider a 10-th order expansion (i.e. p = 10), with guesses for the poles satisfying $|\xi_k| = 0.4, \forall k$. Then, from equation (4.11) the upper bound on the undermodelling error is

$$UB(\omega) = \left| \frac{0.8}{e^{j\omega} + 0.8} \right| \ 0.5882^{10}.$$

The corresponding upper bound using a 10-th order FIR model is

$$UB_{FIR}(\omega) = \left| \frac{0.8}{e^{j\omega} + 0.8} \right| \ 0.8^{10}.$$

We can see that the upper bound using OBFP is $(0.8/0.5882)^{10} = 21.6$ times smaller than the corresponding one using an FIR expansion, even with a 50 % discrepancy between the guesses for the poles and the true poles.

Example 4.3.2. Let us consider now the second order system

$$G(z) = \frac{1}{z+0.8} + \frac{1}{z+0.4},$$

with only two terms in the orthonormal model structure (p = 2). In Figure 4.1, the upper bound on the undermodelling error for different choices of the poles of the basis is plotted as a function of the frequency ω . In that figure, Curve A corresponds to an FIR model structure (that is the guesses for the poles are $\{0,0\}$), Curve B corresponds to a Laguerre structure with poles at $\{-0.2, -0.2\}$, Curve C corresponds to the guesses $\{-0.6, -0.2\}$, and Curve D corresponds to poles at $\{-0.7, -0.3\}$. It can be seen that as the guesses for the poles approach the true poles, the upper bound on the undermodelling error decreases. This also illustrates our claim that when (approximate) 'a priori' information about the dominating dynamics of the system is available, the use of OBFP is preferable over FIR structures since a smaller undermodelling error will be obtained for the same model order.

A result of the same type has been obtained by Van den Hof, Heuberger and co-workers [HbVB95] for the Orthonormal Bases Generated from Inner Functions introduced in Section 3.4. In [HbVB95], the authors show that if the



Figure 4.1: Upper Bound on the Undermodelling Error for different choices of the poles

dynamics of the inner function generating the orthonormal system $G_x(z)$ and the dynamics of the system to be identified G(z) approach each other, then the convergence rate of the series expansion representation of the system becomes very fast. This implies that the number of coefficients to be estimated in order to accurately model the system becomes smaller. An upper bound of this convergence rate is given in Theorem 4.3.2, adapted from [VHB95, HbVB95].

Before presenting this result, we need to introduce some notation. Let $\{\mathcal{B}_{i,k}(z)\}$ be a set of orthonormal bases generated from an inner transfer function $G_x(z)$ as in Theorem 3.4.1. Then any causal, stable system $G(z) \in H_2(\mathbb{T})$ has a unique series representation as in (3.13), i.e.

$$G(z) = \sum_{k=0}^{\infty} \sum_{i=1}^{n_x} \theta_k^i \mathcal{B}_{i,k}(z) = \sum_{k=0}^{\infty} \sum_{i=1}^{n_x} \theta_k^i e_i^T V_k(z)$$
$$= \sum_{k=0}^{\infty} L_k V_k(z)$$

$$= \sum_{k=0}^{p-1} L_k V_k(z) + \sum_{k=p}^{\infty} L_k V_k(z)$$

= $\hat{G}_p(z) + \sum_{k=p}^{\infty} L_k V_k(z),$ (4.12)

where $L_k = [\theta_k^1, \theta_k^2, \dots, \theta_k^{n_x}] \in \ell_2^{1 \times n_x}(\mathbb{N}_0)$, and $\widehat{G}_p(z)$ is the best L_2 approximation to G(z) with respect to the basis functions $\{\mathcal{B}_{i,k}(z)\}$.

Theorem 4.3.2. [VHB95, HbVB95] Let the transfer function G(z) of the system have poles μ_i , $(i = 1, \dots, n_s)$, and let the inner function $G_x(z)$ generating the orthonormal system $\{\mathcal{B}_{i,k}(z)\}$ have poles ρ_j , $(j = 1, \dots, n_x)$. Denote

$$\lambda \triangleq \max_{i} \prod_{j=1}^{n_x} \left| \frac{\mu_i - \rho_j}{1 - \mu_i \rho_j} \right|.$$

Then there exists a constant $c \in \mathbb{R}$ such that, for all $\eta > \lambda$,

$$\sqrt{\sum_{k=p}^{\infty} L_k L_k^T} \le c \ \frac{\eta^{p+1}}{\sqrt{1-\eta^2}}.$$

Proof: See [VHB95, HbVB95]. In the proof, the authors exploit the algebraic structure of the bases

$$\mathcal{B}_{i,m}(z)\mathcal{B}_{i,n}(z) = \mathcal{B}_{i,0}(z)\mathcal{B}_{i,(m+n)}(z)$$
(4.13)

to establish a transformation of signal and systems (the so-called 'Hambo' transform [VHB94a, VHB94b, VHB95, HV96]) so that the system in the transform domain can be obtained by a simple variable transformation from the original system represented with the standard trigonometric bases $\{z^{-n}\}$. The result then follows by using known properties of Fourier series approximation.

The previous theorem implies that when the two sets of poles converge to each other, λ will tend to zero, and the upper bound on the 2-norm of the tail of the series expansion will decrease drastically, reducing in this way the undermodelling error.

4.4 Noise Induced Error

Our interest is now in the quantification of the component of the frequency response estimation error that is induced by the measurement noise (the 'variance error'). Since the analysis for finite data is too complicated, we follow the work of other authors [Lju85, LY85, Wah91b, Wah94b] and consider an asymptotic analysis when the data-length N tends to infinity. Since we are interested here only in the noise induced error, we will also allow the model order p to tend to infinity to avoid the presence of undermodelling error. The quantification of the frequency response noise induced error can be achieved by using known results [Lju87] on the asymptotic statistics of the parameter estimation error and by noting that the frequency response estimate is linearly related to the parameter estimate.

Using the results in [Lju87] (as summarized in Theorems 2.2.1 and 2.2.2) we can draw the following conclusions on the asymptotic (as the number of data N tends to infinity) statistics of the parameter estimate (4.7):

• Asymptotic Estimate: For fixed model order p

$$\widehat{\theta} \xrightarrow{\text{a.s.}} \theta_{\star}$$
 as $N \to \infty$

where

$$\theta_{\star} = \underset{\theta \in \mathbb{R}^{p}}{\operatorname{arg\,min}} \left\{ \frac{1}{2\pi} \int_{-\pi}^{\pi} \left| G(e^{j\omega}) - G_{p}(e^{j\omega}, \theta) \right|^{2} \Phi_{u}(\omega) \, \mathrm{d}\omega \right\}$$
(4.14)

• Asymptotic Distribution of the parameter estimate: For fixed model order p

$$\sqrt{N}(\widehat{\theta} - \theta_{\star}) \xrightarrow{\text{dist}} \mathcal{N}(0, P_p) \quad \text{as} \quad N \longrightarrow \infty$$

where

$$P_{p} \triangleq R_{p}^{-1}Q_{p}R_{p}^{-1},$$

$$R_{p} \triangleq \lim_{N \to \infty} R_{p}(N) = \lim_{N \to \infty} \mathbf{E}\left\{V_{N}''(\theta_{\star})\right\},$$

$$Q_{p} \triangleq \lim_{N \to \infty} N\mathbf{E}\left\{V_{N}'(\theta_{\star})^{T}V_{N}'(\theta_{\star})\right\},$$

Notice now that the transfer function estimate (4.5) can be written as

$$G(z,\widehat{\theta}) \triangleq \sum_{k=0}^{p-1} \widehat{\theta}_k \mathcal{B}_k(z) = \Gamma_p^T(z) \ \widehat{\theta},$$
(4.15)

where we have defined

$$\Gamma_p(z) \triangleq [\mathcal{B}_0(z), \mathcal{B}_1(z), \cdots, \mathcal{B}_{p-1}(z)]^T.$$
(4.16)

This linear relationship between the transfer function estimate and the parameter estimate, together with the above results on the statistics of the parameter estimate, allows us to give the following frequency domain characterization of the asymptotic distribution of the (transfer function) estimate:
• Asymptotic Distribution of the Frequency Response Estimate: For fixed model order p

$$\sqrt{N} \begin{bmatrix} G(e^{j\omega_1}, \widehat{\theta}) - G(e^{j\omega_1}, \theta_\star) \\ G(e^{j\omega_2}, \widehat{\theta}) - G(e^{j\omega_2}, \theta_\star) \end{bmatrix} \xrightarrow{\text{dist}} \mathcal{N}(0, \Lambda'_p(\omega_1, \omega_2))$$

as $N \longrightarrow \infty$, with

$$\Lambda'_p(\omega_1,\omega_2) \triangleq \left[\begin{array}{c} \Gamma_p^T(e^{j\omega_1}) \\ \Gamma_p^T(e^{j\omega_2}) \end{array} \right] \ P_p \ \left[\begin{array}{c} \Gamma_p^T(e^{j\omega_1}) \\ \Gamma_p^T(e^{j\omega_2}) \end{array} \right]^*$$

if $\omega_1 \neq \omega_2$, and where \cdot^* denotes the conjugate transpose.

A measure of the transfer function estimation error induced by the noise is then given by the covariance matrix $\Lambda'_p(\omega_1, \omega_2)$. The exact expression for the covariance matrices P_p and $\Lambda'_p(\omega_1, \omega_2)$ will depend on the particular choice for the orthonormal basis. Unfortunately, these exact expressions for a fixed (finite) model order p are so complicated that they have no practical utility. As mentioned before, the standard approach in the literature (see for instance: [LY85, Lju85] for FIR basis, [Wah91b] for Laguerre basis, [Wah94b] for Kautz basis, [VHB95] for OBGIF, and [NHG97a] for OBFP) has been to provide an approximate quantification of the noise induced error by considering an asymptotic analysis when both the model order p and the number of data N are allowed to tend to infinity.

4.4.1 The FIR case

Historically, the first results regarding the quantification of the noise induced error in this identification setup were obtained by Ljung and Yuan [LY85, Lju85] for the case of FIR model structures. The FIR variance results are summarized in the following theorem.

Theorem 4.4.1. [LY85] Let $G(z, \theta)$ be represented as in (4.4) and let $\{\mathcal{B}_k(z)\}$ be the standard FIR basis. Let $\widetilde{G}(e^{j\omega})$ be defined as

$$\widetilde{G}(e^{\mathbf{j}\omega}) \triangleq G(e^{\mathbf{j}\omega},\widehat{\theta}) - G(e^{\mathbf{j}\omega},\theta_{\star}).$$

Then provided that $p \to \infty$ as $N \to \infty$,

$$\sqrt{\frac{N}{p}} \ \widetilde{G}(e^{j\omega}) \xrightarrow{dist} \mathcal{N}\left(0, \frac{\Phi_{\nu}(\omega)}{\Phi_{u}(\omega)}\right)$$

as $N \to \infty$. In addition

$$\frac{N}{p} \operatorname{Cov} \left\{ \widetilde{G}(e^{j\omega_1}), \widetilde{G}(e^{j\omega_2}) \right\} \to \left\{ \begin{array}{cc} 0 & \text{if} \quad \omega_1 \neq \omega_2 \\ \frac{\Phi_{\nu}(\omega_1)}{\Phi_u(\omega_1)} & \text{if} \quad \omega_1 = \omega_2 \end{array} \right.$$

as $N \to \infty$.

Proof: See [LY85].

The theorem implies that the variance of the transfer function estimate at a particular frequency ω , and for large model order and data-length can be approximated by

$$\operatorname{Var}\left\{G(e^{j\omega},\widehat{\theta})\right\} \approx \frac{p}{N} \ \frac{\Phi_{\nu}(\omega)}{\Phi_{u}(\omega)},\tag{4.17}$$

which is the noise-to-signal ratio with a weighting factor which is the ratio between model order and data-length.

4.4.2 The Fixed Denominator case

Our interest now is to determine if the FIR result in Theorem 4.4.1 can also be applied for the case of a fixed denominator model structure. We can write

$$y_k = G(q, \hat{\theta})u_k + \nu_k = \frac{C(q, \hat{\theta})}{D(q)}u_k + \nu_k, \qquad (4.18)$$

where the poles in D(q) are fixed (and known), and the unknown parameters are the coefficients of the numerator polynomial $C(q, \hat{\theta})$. We can then write

$$y_k = C(q, \hat{\theta}) \left(\frac{1}{D(q)} u_k\right) + \nu_k = C(q, \hat{\theta}) \tilde{u}_k + \nu_k,$$
(4.19)

where

$$\tilde{u}_k \triangleq \frac{1}{D(q)} u_k$$

is a filtered version of u_k . The fixed denominator model structure estimation problem (4.18), being then by (4.19) really an FIR estimation problem with prefiltered input \tilde{u}_k , should be amenable to FIR variance analysis by Theorem 4.4.1 to lead to the conclusion that for a given frequency ω , and for large model order and data-length we can approximate the numerator variance as

$$\operatorname{Var}\left\{C(e^{j\omega},\widehat{\theta})\right\} \approx \frac{p}{N} \; \frac{\Phi_{\nu}(\omega)}{\Phi_{\tilde{u}}(\omega)},$$

and then considering that

$$\Phi_{\tilde{u}}(\omega) = \frac{\Phi_u(\omega)}{|D(e^{\mathsf{j}\omega})|^2},$$

we have

$$\operatorname{Var}\left\{C(e^{\mathrm{j}\omega},\widehat{\theta})\right\} \approx \frac{p|D(e^{\mathrm{j}\omega})|^2}{N} \frac{\Phi_{\nu}(\omega)}{\Phi_u(\omega)}.$$

Therefore, the variability of the full frequency response estimate could be expected by Theorem 4.4.1 to be approximated as

$$\mathsf{Var}\left\{G(e^{\mathsf{j}\omega},\widehat{\theta})\right\} \approx \frac{1}{|D(e^{\mathsf{j}\omega})|^2}\mathsf{Var}\left\{C(e^{\mathsf{j}\omega},\widehat{\theta})\right\}$$

which is (4.17), and therefore that in the general fixed denominator case, the estimation variance does not appear to depend on the location of the poles in the model structure.

This line of reasoning is, however, flawed. The problem is that the filter $D(e^{j\omega})$ is changing as the model order p increases, and therefore the spectrum $\Phi_{\tilde{u}}(\omega)$ is also changing with p, so that we are applying a result (Theorem 4.4.1) which was derived assuming a fixed input spectrum to a case where this spectrum is not fixed.

The remedy for this problem turns out to be to reparameterize the problem in a special orthonormal form which is specifically adapted to the fixed denominator being used. Developing these methods consume the remainder of this chapter.

4.4.3 Variance error using OBFP

Let $\{\mathcal{B}_k(z)\}\$ be a set of orthonormal bases with fixed poles $\{\xi_k\}\$ as defined in (3.14), and let us assume that a system $G(z) \in H_2(\mathbb{T})$ is identified using these bases and least squares techniques as described in Section 4.2. The asymptotic (in model order and data-length) distribution of the transfer function estimate for this case is as follows.

Theorem 4.4.2. Let the input spectral density $\Phi_u(\omega)$ have a finite dimensional spectral factorization, and let the poles $\{\xi_k\}$ be chosen to satisfy the completeness condition

$$\sum_{k=0}^{\infty} (1 - |\xi_k|) = \infty.$$

Then for $N \to \infty$ and $p \to \infty$

$$\sqrt{N} \begin{bmatrix} \gamma_p(\omega_1) & 0 \\ 0 & \gamma_p(\omega_2) \end{bmatrix}^{-1/2} \begin{bmatrix} G(e^{j\omega_1}, \widehat{\theta}) - G(e^{j\omega_1}, \theta_\star) \\ G(e^{j\omega_2}, \widehat{\theta}) - G(e^{j\omega_2}, \theta_\star) \end{bmatrix} \xrightarrow{dist} \mathcal{N}\left(0, \Lambda(\omega_1, \omega_2)\right)$$

where

$$\Lambda(\omega_1, \omega_2) = \begin{bmatrix} \frac{\Phi_{\nu}(\omega_1)}{\Phi_u(\omega_1)} & 0\\ 0 & \frac{\Phi_{\nu}(\omega_2)}{\Phi_u(\omega_2)} \end{bmatrix}$$

if $\omega_1 \neq \omega_2$, and

$$\gamma_p(\omega) \triangleq K_p(\omega, \omega) = \sum_{k=0}^{p-1} |\mathcal{B}_k(e^{j\omega})|^2.$$

Proof: The main difficulty encountered in the proof of these asymptotic results has been that the bases (3.14) employed here do not have the algebraic structure (3.1), so that the problem can not be reduced to the FIR one by a change of variables. Fundamental in the analysis will be the derivation of some results regarding the convergence properties of generalized Toeplitz-like matrices. These results are summarized in Appendix 4.B, and their derivation proceeds based only on the orthonormality of the bases and the Christoffel-Darboux formula for the reproducing kernel associated with them. The proof is given in Appendix 4.A.

As a corollary of the previous theorem we have the following quantification of the noise induced error in the transfer function estimate.

Corollary 4.4.3. Under the same conditions of the previous theorem, but with the strengthened requirement that $E\{e_k^8\} < \infty$ then

$$\lim_{p \to \infty} \lim_{N \to \infty} \frac{N}{\gamma_p(\omega)} \mathbf{E} \left\{ |G(e^{j\omega}, \widehat{\theta}) - G(e^{j\omega}, \theta_\star)|^2 \right\} = \frac{\Phi_\nu(\omega)}{\Phi_u(\omega)}.$$

Proof: Follows along the same lines in the developments in Appendix 9B of [Lju87].

These results imply that the variance of the transfer function estimate at a given frequency ω , and for large N and model order p, can be well approximated by

$$\operatorname{Var}\left\{G(e^{j\omega},\widehat{\theta})\right\} \approx \frac{\gamma_p(\omega)}{N} \frac{\Phi_\nu(\omega)}{\Phi_u(\omega)},\tag{4.20}$$

which is the noise-to-signal ratio weighted with a frequency dependent factor that is determined by the basis functions. This variance expression explicitly shows (through the factor $\gamma_p(\omega) \triangleq K_p(\omega, \omega)$) how the choice of the poles of the basis functions (the poles in the fixed denominator model) affects the noise induced error. In Figure 4.2, the factor $\gamma_p(\omega)$ is plotted for model order p = 4and for various pole choices.

In addition, this result generalizes similar results available for FIR model structures [Lju85, LY85], Laguerre basis [Wah91b], and Kautz basis [Wah94b]. For example, the FIR model structure corresponds to the choice $\xi_k = 0, \forall k$ for the poles of the bases, and in this case the factor $\gamma_p(\omega) = p$, so that the variance expression (4.20) becomes

$$\mathsf{Var}\left\{G(e^{\mathsf{j}\omega},\widehat{\theta})\right\}\approx \frac{p}{N}\frac{\Phi_\nu(\omega)}{\Phi_u(\omega)},$$



Figure 4.2: Reproducing kernel $\gamma_p(\omega) \triangleq K_p(\omega, \omega)$ as a function of frequency ω , and for model order p = 4 and various pole choices.

which is the same result of Ljung and Yuan [Lju85, LY85] as summarized in Theorem 4.4.1. The Laguerre basis instead corresponds to the choice $\xi_k = \xi \in \mathbb{R}, \forall k$ for the poles, so that the variance expression (4.20) becomes

$$\operatorname{Var}\left\{G(e^{j\omega},\widehat{\theta})\right\} \approx \frac{p}{N} \frac{(1-\xi^2)}{|e^{j\omega}-\xi|^2} \frac{\Phi_{\nu}(\omega)}{\Phi_u(\omega)},$$

which is exactly the result obtained originally by Wahlberg [Wah91b].

A result of the same nature as Theorem 4.4.2 has been derived by Van den Hof and co-workers [VHB95] for the orthonormal bases generated from inner functions of Section 3.4. The authors exploit the algebraic structure (4.13) of the bases in order to establish a transformation of systems (the 'Hambo Transform') so that the system in the transformed domain can be obtained by a simple change of variables from the original system represented with the FIR bases $\{z^{-k}\}$. See [VHB95] for the details.

The following theorem summarizes these variance results.

Theorem 4.4.4. [VHB95] Assume that the input spectral density $\Phi_u(\omega)$ is bounded away from zero and sufficiently smooth. Then, for $N, p \longrightarrow \infty$,

$$p^{2}/N \longrightarrow 0$$

$$\frac{N}{pn_{x}} \operatorname{Cov} \left\{ G(e^{j\omega_{1}}, \widehat{\theta}_{N}), G(e^{j\omega_{2}}, \widehat{\theta}_{N}) \right\} \longrightarrow \begin{cases} 0 \quad \text{for} \quad G_{x}(e^{j\omega_{1}}) \neq G_{x}(e^{j\omega_{2}}) \\ V_{0}^{T}(e^{j\omega_{1}})V_{0}(e^{-j\omega_{1}}) \frac{\Phi_{\nu}(\omega_{1})}{\Phi_{u}(\omega_{1})} \\ \text{for} \quad \omega_{1} = \omega_{2} \end{cases}$$

$$(4.21)$$

where $\operatorname{Cov}\left\{G(e^{j\omega_1},\widehat{\theta}_N), G(e^{j\omega_2},\widehat{\theta}_N)\right\}$ is the cross-covariance matrix in the joint distribution of

$$\left[G(e^{j\omega_1},\widehat{\theta}_N) - G(e^{j\omega_1},\theta_\star), G(e^{j\omega_2},\widehat{\theta}_N) - G(e^{j\omega_2},\theta_\star)\right],$$

and where $\Phi_{\nu}(\omega)$ is the measurement noise spectral density.

Proof: See [VHB95]. The key idea of the proof is to exploit the algebraic structure (4.13) of the bases in order to establish a transformation of systems (the 'Hambo Transform') so that the system in the transformed domain can be obtained by a simple change of variables from the original system represented with the standard trigonometric basis $\{z^{-k}\}$. The use of convergence results available for FIR model structures [Lju85, LY85] together with asymptotic properties of Toeplitz matrices [GS58] then gives the result.

As pointed out in [VHB95], the interpretation of the previous theorem is that the variance of the transfer function estimate at a given frequency ω , and for large N and model order p, can be approximated by

$$\operatorname{Var}\left\{G(e^{j\omega},\widehat{\theta})\right\} \approx \frac{pn_x}{N} V_0^T(e^{j\omega}) V_0(e^{-j\omega}) \frac{\Phi_\nu(\omega)}{\Phi_u(\omega)}$$
(4.22)

which is the noise-to-signal ratio weighted with a frequency dependent factor that is determined by the basis functions. This result also generalizes the available FIR, and Laguerre results in [Lju85, LY85], and [Wah91b], respectively. See [VHB95] for the details.

This thesis, via Theorem 4.4.2 has developed a result in a similar vein, but with one very important difference - it is asymptotic in the number of poles, not asymptotic in the number of *repetitions* of poles as the pre-existing result of Theorem 4.4.4 is. This difference is considered very substantial, since the point of results such as Theorem 4.4.4 or our Theorem 4.4.2 is to allow the progression to approximations like (4.22), and this can only be done by assuming that convergence has occurred in results like Theorems 4.4.4 and 4.4.2. If, as commonly occurs in practice, one is using a model with all the poles different (so as to distribute them as much as possible in the hope of minimizing the undermodelling induced error), then there has been no repetition of poles, and it cannot be argued that convergence in Theorem 4.4.4 is likely to have occurred, and hence for (4.22) to be a realistic approximation.

In contrast, in this same scenario of all poles being different, but using new and completely different analysis techniques to those used in deriving Theorem 4.4.4, the new result of Theorem 4.4.2 is relevant since, provided there is a reasonable number of fixed poles (5 or more appears 'reasonable' in our experience), then it can be argued that convergence may have approximately occurred in Theorem 4.4.2 and hence that an approximation like (4.20) can be reasonably argued to be appropriate.

4.5 Bias/Variance Trade-off

The variance expression (4.20) together with the upper bound on the undermodelling error (4.11) provides a complete characterization of the accuracy of the estimates. The results show the well known trade-off that exists in the choice of model order p with regard to the relative size of both error components. Specifically an increase in model order to reduce the undermodelling error will be at the cost of an increase of the variance error.

The results also show how the estimation accuracy is influenced by the choice of the poles of the basis functions. Here, an until now unappreciated phenomenom is manifested, namely, a trade-off in the choice of the poles of the bases regarding the magnitude of the bias and variance errors. More specifically, assuming that the noise induced error and the undermodelling error are uncorrelated we can write

$$\mathbf{E} \left\{ \left| G(e^{j\omega}, \widehat{\theta}) - G(e^{j\omega}) \right|^2 \right\} = \mathbf{E} \left\{ \left| G(e^{j\omega}, \widehat{\theta}) - G(e^{j\omega}, \theta_\star) \right|^2 \right\}$$
$$+ \mathbf{E} \left\{ \left| G(e^{j\omega}, \theta_\star) - G(e^{j\omega}) \right|^2 \right\}$$

Use of Theorems 4.3.1 and 4.4.2 then allows us to upper bound the frequency response estimation error as follows

$$\mathbf{E}\left\{\left|G(e^{j\omega},\widehat{\theta}) - G(e^{j\omega})\right|^{2}\right\} \leq r \sum_{\ell=0}^{r-1} \left|\frac{\alpha_{\ell}}{e^{j\omega} - \gamma_{\ell}}\right|^{2} \prod_{k=0}^{p-1} \left|\frac{\gamma_{\ell} - \xi_{k}}{1 - \overline{\xi_{k}}\gamma_{\ell}}\right|^{2} + \frac{\gamma_{p}(\omega)}{N} \frac{\Phi_{\nu}(\omega)}{\Phi_{u}(\omega)}$$

We can see that if we want to decrease the undermodelling error for a given model order, the poles of the bases have to be chosen close to the true poles, but then the noise induce error cannot be reduced at the frequencies of the poles (due to the presence of the factor $\gamma_p(\omega)$). On the other hand, if we want to reduce the noise induced error for a fixed model order, the poles of the bases have to be chosen well below the frequency at which the noise is dominating, but then if the true poles are not at these frequencies, the undermodelling error will be incremented. The following example illustrates this phenomenom. Example 4.5.1. Let the true system be given by

$$G(q) = \frac{0.1548q + 0.0939}{(q - 0.6065)(q - 0.3679)} = \frac{0.7871}{(q - 0.6065)} - \frac{0.6323}{(q - 0.3679)},$$

and let us consider that N = 1000 samples are available for the estimation in the identification setup of Section 4.2. It is assumed that the measurement noise is a zero mean Gaussian white noise process of variance $\sigma_{\nu}^2 = 0.01$, and that the input is a Gaussian white noise process with variance $\sigma_u^2 = 0.1$.



Figure 4.3: Illustration of Bias/Variance trade-off with respect to the choice of Laguerre pole position.

Figure 4.3 shows the dependence on the fixed pole position of the variance error, the bias error, and the total error at a frequency $\omega = 0.1$ rad/s, when using an 8th order Laguerre model structure.

4.6 Conclusions

In this chapter we have studied the problem of identification, from input-output data in the time domain, of discrete-time single-input, single-output linear systems using rational orthonormal bases and least squares techniques. The focus of the chapter has been the derivation of results concerning the accuracy of the estimation for the case of using orthonormal bases with fixed poles. The results generalize previous works corresponding to FIR, Laguerre and Kautz bases. The way the location of the poles in the orthonormal structure affects the two components of the estimation error has been determined. This shows a new phenomenom of bias/variance trade-off regarding the choice of the pole locations.

APPENDICES

4.A Proofs for Chapter 4

Proof of Theorem 4.3.1 Using the Christoffel-Darboux formula for the reproducing kernel (equation (3.17)), and the partial fraction expansion for G(z) in (4.10) we can write

$$G(\mu) - \widehat{G}_p(\mu) = \langle G(z), K(z, \mu) - K_p(z, \mu) \rangle$$
$$= \left\langle G(z), 1 + \frac{\varphi_p(z)\overline{\varphi_p(\mu)}}{z\overline{\mu} - 1} \right\rangle$$

$$= \langle G(z), 1 \rangle + \left\langle G(z), \frac{\varphi_p(z)\overline{\varphi_p(\mu)}}{z\overline{\mu} - 1} \right\rangle$$
$$= \left\langle G(z), \frac{\varphi_p(z)\overline{\varphi_p(\mu)}}{z\overline{\mu} - 1} \right\rangle$$
$$= -\varphi_p(\mu) \frac{1}{2\pi j} \oint_{\mathbb{T}} \frac{G(z)\overline{\varphi_p(z^{-1})}}{z - \mu} dz$$
$$= -\varphi_p(\mu) \sum_{\ell=0}^{r-1} \alpha_\ell \frac{1}{2\pi j} \oint_{\mathbb{T}} \frac{\overline{\varphi_p(z^{-1})}}{(z - \gamma_\ell)(z - \mu)} dz$$

In passing to the fourth line, use was made of the fact that due to Cauchy Residue Theorem and the analyticity of G(z) outside the unit circle, then

$$\langle G(z), 1 \rangle = \frac{1}{2\pi j} \oint_{\mathbb{T}} G(z^{-1}) \frac{\mathrm{d}z}{z} = 0.$$

Considering now that

$$\overline{\varphi_p(z^{-1})} = \prod_{k=0}^{p-1} \frac{z - \xi_k}{1 - z\overline{\xi_k}},$$

we can write appealing once again to Cauchy Residue Theorem

$$G(\mu) - \widehat{G}_p(\mu) = -\varphi_p(\mu) \sum_{\ell=0}^{r-1} \alpha_\ell \frac{1}{2\pi j} \oint_{\mathbb{T}} \prod_{k=0}^{p-1} \frac{(z-\xi_k)}{(z-\gamma_\ell)(z-\mu)(1-z\overline{\xi_k})} dz$$
$$= -\varphi_p(\mu) \sum_{\ell=0}^{r-1} \frac{\alpha_\ell}{(\gamma_\ell - \mu)} \prod_{k=0}^{p-1} \frac{\gamma_\ell - \xi_k}{1 - \gamma_\ell \overline{\xi_k}}.$$

Now, the result follows by computing the previous equation on the unit circle $(\mu = e^{j\omega})$, taking the module on both sides of the equation, considering that $|\varphi_p(e^{j\omega})| = 1$, and using triangle inequality.

Proof of Theorem 4.4.2 As mentioned at the beginning of Section 4.4, by appealing to the results in [Lju87] it is possible to give the following frequency domain characterization of the asymptotic distribution of the transfer function estimate.

• Asymptotic distribution of the transfer function estimate. For fixed model order p

$$\sqrt{N} \left[\begin{array}{c} G(e^{j\omega_1}, \widehat{\theta}) - G(e^{j\omega_1}, \theta_\star) \\ G(e^{j\omega_2}, \widehat{\theta}) - G(e^{j\omega_2}, \theta_\star) \end{array} \right] \xrightarrow{\text{dist}} \mathcal{N}(0, \Lambda'_p(\omega_1, \omega_2))$$

as $N \longrightarrow \infty$, with

$$\Lambda'_p(\omega_1, \omega_2) \triangleq \left[\begin{array}{c} \Gamma_p^T(e^{j\omega_1}) \\ \Gamma_p^T(e^{j\omega_2}) \end{array} \right] P_p \left[\begin{array}{c} \Gamma_p^T(e^{j\omega_1}) \\ \Gamma_p^T(e^{j\omega_2}) \end{array} \right]^*$$

if $\omega_1 \neq \omega_2$, and where

$$P_{p} \triangleq R_{p}^{-1}Q_{p}R_{p}^{-1},$$

$$R_{p} \triangleq \lim_{N \to \infty} R_{p}(N) = \lim_{N \to \infty} \mathbf{E}\left\{V_{N}''(\theta_{\star})\right\},$$

$$Q_{p} \triangleq \lim_{N \to \infty} N\mathbf{E}\left\{V_{N}'(\theta_{\star})^{T}V_{N}'(\theta_{\star})\right\}.$$

It is then straightforward to conclude that, for fixed model order p

$$\sqrt{N} \begin{bmatrix} \gamma_p(\omega_1) & 0\\ 0 & \gamma_p(\omega_2) \end{bmatrix}^{-1/2} \begin{bmatrix} G(e^{j\omega_1}, \widehat{\theta}) - G(e^{j\omega_1}, \theta_\star)\\ G(e^{j\omega_2}, \widehat{\theta}) - G(e^{j\omega_2}, \theta_\star) \end{bmatrix} \xrightarrow{\text{dist}} \mathcal{N}(0, \Lambda_p(\omega_1, \omega_2))$$
(4.A.1)

as $N \longrightarrow \infty$, where now $\Lambda_p(\omega_1, \omega_2)$ is defined as

$$\Lambda_p(\omega_1,\omega_2) \triangleq \begin{bmatrix} \gamma_p(\omega_1) & 0\\ 0 & \gamma_p(\omega_2) \end{bmatrix}^{-1/2} \Lambda'_p(\omega_1,\omega_2) \begin{bmatrix} \gamma_p(\omega_1) & 0\\ 0 & \gamma_p(\omega_2) \end{bmatrix}^{-1/2}.$$

After some matrix manipulation, $\Lambda_p(\omega_1, \omega_2)$ can be written as

$$\Lambda_{p}(\omega_{1},\omega_{2}) = \begin{bmatrix} \frac{\Gamma_{p}^{\star}(e^{j\omega_{1}})P_{p}\Gamma_{p}(e^{j\omega_{1}})}{\gamma_{p}(\omega_{1})} & \frac{\Gamma_{p}^{\star}(e^{j\omega_{1}})P_{p}\Gamma_{p}(e^{j\omega_{2}})}{\gamma_{p}^{1/2}(\omega_{1})\gamma_{p}^{1/2}(\omega_{2})} \\ \frac{\Gamma_{p}^{\star}(e^{j\omega_{2}})P_{p}\Gamma_{p}(e^{j\omega_{1}})}{\gamma_{p}^{1/2}(\omega_{1})\gamma_{p}^{1/2}(\omega_{2})} & \frac{\Gamma_{p}^{\star}(e^{j\omega_{2}})P_{p}\Gamma_{p}(e^{j\omega_{2}})}{\gamma_{p}(\omega_{2})} \end{bmatrix}.$$
(4.A.2)

It is now clear that in order to analyze the asymptotic distribution (4.A.1) when p tends to infinity, we need only to study the asymptotic behaviour of the term

$$\frac{\Gamma_p^{\star}(e^{j\omega_1})P_p\Gamma_p(e^{j\omega_2})}{\gamma_p^{1/2}(\omega_1)\gamma_p^{1/2}(\omega_2)}.$$
(4.A.3)

To proceed with this analysis, we will provide a frequency domain expression for the matrix P_p .

We introduce first some notation. For any positive function $f : [-\pi, \pi] \to (0, \infty)$ we define the $p \times p$ matrix $M_p(f)$ as

$$M_p(f) \triangleq \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_p(e^{j\omega}) f(\omega) \Gamma_p^{\star}(e^{j\omega}) d\omega.$$
(4.A.4)

We will call $M_p(f)$ a *Toeplitz-like* matrix on account of the fact that for the case of the basis functions being the FIR ones, then $M_p(f)$ has effectively a Toeplitz structure.¹ [GS58].

Considering the expression of $R_p(N)$ in equation (4.8), the matrix R_p can be written as

$$R_p = \lim_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} \mathbf{E} \left\{ \phi_k \phi_k^T \right\}.$$

Using now Parseval's Theorem (in the form given by equation (2.3)), and the definition of $M_p(f)$ in (4.A.4), R_p can be written as

$$R_p = \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_p(e^{j\omega}) \Phi_u(\omega) \Gamma_p^{\star}(e^{j\omega}) d\omega = M_p(\Phi_u).$$

The derivation of a frequency domain expression for the matrix Q_p is more difficult. In [NHG97a] it is shown that Q_p can be written as

$$Q_p = M_p(\Phi_u \Phi_\nu) + \Delta_p,$$

- A Toeplitz matrix is constant along its diagonals.
- A lower (upper) triangular Toeplitz matrix is completely specified by the elements of the first column (row).

¹A matrix is said to be Toeplitz if its i, j-th entry depends only on the difference (i - j) of the indices. Some basic properties of Toeplitz matrices are:

where Δ_p is given by

$$\Delta_p = \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_p(e^{j\omega}) \Phi_u^2(\omega) |\widetilde{G}_p(e^{j\omega})|^2 \Gamma_p^{\star}(e^{j\omega}) d\omega,$$

with

$$\widetilde{G}_p(e^{j\omega}) \triangleq G(e^{j\omega}) - G(e^{j\omega}, \theta_\star).$$

It can be also proved [NHG97a] that²

$$\lim_{p \to \infty} \|\Delta_p\|_2 = 0.$$

Finally, a frequency domain expression of the matrix P_p is then given by

$$P_p = M_p^{-1}(\Phi_u)M_p(\Phi_u\Phi_\nu)M_p^{-1}(\Phi_u) + M_p^{-1}(\Phi_u)\Delta_pM_p^{-1}(\Phi_u).$$

The term (4.A.3) can then be written as

$$\frac{\Gamma_{p}^{\star}(e^{j\omega_{1}})P_{p}\Gamma_{p}(e^{j\omega_{2}})}{\gamma_{p}^{1/2}(\omega_{1})\gamma_{p}^{1/2}(\omega_{2})} = \frac{\Gamma_{p}^{\star}(e^{j\omega_{1}})M_{p}^{-1}(\Phi_{u})M_{p}(\Phi_{u}\Phi_{\nu})M_{p}^{-1}(\Phi_{u})\Gamma_{p}(e^{j\omega_{2}})}{\gamma_{p}^{1/2}(\omega_{1})\gamma_{p}^{1/2}(\omega_{2})} + \frac{\Gamma_{p}^{\star}(e^{j\omega_{1}})M_{p}^{-1}(\Phi_{u})\Delta_{p}M_{p}^{-1}(\Phi_{u})\Gamma_{p}(e^{j\omega_{2}})}{\gamma_{p}^{1/2}(\omega_{1})\gamma_{p}^{1/2}(\omega_{2})} - \frac{\Gamma_{p}^{\star}(e^{j\omega_{1}})M_{p}^{-1}(\Phi_{u})\Delta_{p}M_{p}^{-1}(\Phi_{u})\Gamma_{p}(e^{j\omega_{2}})}{\gamma_{p}^{1/2}(\omega_{1})\gamma_{p}^{1/2}(\omega_{2})} - \frac{\Gamma_{p}^{\star}(e^{j\omega_{1}})M_{p}^{-1}(\Phi_{u})\Delta_{p}M_{p}^{-1}(\Phi_{u})\Gamma_{p}(e^{j\omega_{2}})}{\Gamma \text{erm } 2}.$$
(4.A.5)

Now, the asymptotic analysis when $p \to \infty$ can be carried out by resorting to the results on convergence of Toeplitz-like matrices in Appendix 4.B. It can be proved that for any choices of ω_1 and ω_2 , Term 2 on the right hand side of equation (4.A.5) tends to zero as $p \to \infty$. On the other hand, as $p \to \infty$, Term 1 tends to $\Phi_{\nu}(\omega)/\Phi_u(\omega)$ for $\omega_1 = \omega_2 = \omega$, and tends to zero for $\omega_1 \neq \omega_2$. We refer the reader to [NHG97a, NHG97b] for the remainder details of the proof.

4.B Convergence of Toeplitz-like Matrices

Although for general orthonormal bases the matrix form $M_p(f)$ defined in (4.A.4) will not have a Toeplitz structure, some results on convergence of Toeplitz matrices can be extended to this form. For the orthonormal bases with fixed poles introduced in Section 3.5, these convergence results are as follows.

²Here $\|\cdot\|_2$ stands for the matrix induced 2-norm or spectral norm (i.e. the maximum singular value).

Theorem 4.B.1. Suppose $f(\omega)$ is a real valued and continuous function on $[-\pi,\pi]$. Then provided

$$\sum_{k=0}^{\infty} (1 - |\xi_k|) = \infty,$$

the following limit result holds

$$\lim_{p \to \infty} \frac{\Gamma_p^{\star}(e^{j\omega}) M_p(f) \Gamma_p(e^{j\omega})}{\gamma_p(\omega)} = f(\omega).$$

Proof: [NHG97b]

$$\begin{split} \Gamma_{p}(e^{j\omega})^{\star}M_{p}(f)\Gamma_{p}(e^{j\omega}) &= \\ &= \sum_{m=0}^{p-1}\sum_{n=0}^{p-1}\overline{\mathcal{B}_{m}(e^{j\omega})}\mathcal{B}_{n}(e^{j\omega})[M_{p}(f)]_{m,n}, \\ &= \frac{1}{2\pi}\int_{-\pi}^{\pi}f(\sigma)\sum_{m=0}^{p-1}\sum_{n=0}^{p-1}\overline{\mathcal{B}_{m}(e^{j\omega})}\mathcal{B}_{n}(e^{j\omega})\mathcal{B}_{m}(e^{j\sigma})\overline{\mathcal{B}_{n}(e^{j\sigma})}\,\mathrm{d}\sigma, \\ &= \frac{1}{2\pi}\int_{-\pi}^{\pi}f(\sigma)\left|K_{p}(\omega,\sigma)\right|^{2}\,\mathrm{d}\sigma. \end{split}$$

Therefore, for any $\delta > 0$

$$\frac{1}{2\pi} \left| \frac{\Gamma_p^{\star}(e^{j\omega}) M_p(f) \Gamma_p(e^{j\omega})}{\gamma_p(\omega)} - f(\omega) \right| = \\
= \frac{1}{2\pi \gamma_p(\omega)} \left| \Gamma_p^{\star}(e^{j\omega}) M_p(f) \Gamma_p(e^{j\omega}) - \gamma_p(\omega) f(\omega) \right|, \\
= \frac{1}{2\pi \gamma_p(\omega)} \left| \int_{-\pi}^{\pi} (f(\sigma) - f(\omega)) |K_p(\omega, \sigma)|^2 d\sigma \right|, \\
\leq \frac{1}{2\pi \gamma_p(\omega)} \left| \int_{\sigma \in [\omega - \delta, \omega + \delta]} (f(\sigma) - f(\omega)) |K_p(\omega, \sigma)|^2 d\sigma \right| + \\
+ \frac{1}{2\pi \gamma_p(\omega)} \left| \int_{\sigma \notin [\omega - \delta, \omega + \delta]} (f(\sigma) - f(\omega)) |K_p(\omega, \sigma)|^2 d\sigma \right|.$$

Now, since $f(\omega)$ is continuous, then for δ sufficiently small

$$|f(\sigma) - f(\omega)| \le \epsilon \text{ on } [\omega - \delta, \omega + \delta].$$

and hence

$$\frac{1}{2\pi\gamma_p(\omega)} \left| \int_{\sigma\in[\omega-\delta,\omega+\delta]} \left(f(\sigma) - f(\omega) \right) |K_p(\omega,\sigma)|^2 \,\mathrm{d}\sigma \right| \le \frac{\epsilon}{2\pi\gamma_p(\omega)} \int_{-\pi}^{\pi} |K_p(\omega,\sigma)|^2 \,\mathrm{d}\sigma = \epsilon.$$

Also, since f is continuous on compact $[-\pi,\pi]$ then f is bounded by some $M/2 < \infty$. Therefore

$$\frac{1}{2\pi\gamma_p(\omega)} \left| \int_{\sigma\not\in[\omega-\delta,\omega+\delta]} \left(f(\sigma) - f(\omega) \right) |K_p(\omega,\sigma)|^2 \,\mathrm{d}\sigma \right| \le \frac{M}{2\pi\gamma_p(\omega)} \int_{\sigma\not\in[\omega-\delta,\omega+\delta]} |K_p(\omega,\sigma)|^2 \,\mathrm{d}\sigma.$$

This gives

$$\frac{1}{2\pi} \left| \frac{\Gamma_p^{\star}(e^{j\omega}) M_p(f) \Gamma_p(e^{j\omega})}{\gamma_p(\omega)} - f(\omega) \right| \le \epsilon + \frac{M}{2\pi \gamma_p(\omega)} \int_{\sigma \notin [\omega - \delta, \omega + \delta]} |K_p(\omega, \sigma)|^2 \, \mathrm{d}\sigma$$

Using the result in Lemma 4.C.1 and considering that ϵ is arbitrary then completes the proof.

Theorem 4.B.2. Suppose $f(\omega) \in L_2([-\pi, \pi])$ is positive definite and has finite dimensional spectral factorization. Then provided

$$\sum_{k=0}^{\infty} (1 - |\xi_k|) = \infty$$

the following limit result holds

$$\lim_{p \to \infty} \frac{\Gamma_p(e^{j\omega})^* M_p^{-1}(f) \Gamma_p(e^{j\omega})}{\gamma_p(\omega)} = f^{-1}(\omega)$$

Proof:

$$\frac{\Gamma_p(e^{j\mu})^* M_p(f)^{-1} \Gamma_p(e^{j\omega})}{\gamma_p(\omega)} = \frac{\Gamma_p(e^{j\mu})^* M_p(1/f) \Gamma_p(e^{j\omega})}{\gamma_p(\omega)} + \frac{\Gamma_p(e^{j\mu})^* M_p(f)^{-1} [I - M_p(f) M_p(1/f)] \Gamma_p(e^{j\omega})}{\gamma_p(\omega)}$$

Now by construction, the elements of the vector $\Gamma(e^{j\omega})$ are bounded in magnitude by some finite number K_1 , as defined in Lemma 5.E.2. Similarly, by Lemma 4.C.3 the elements of the vector $M_p(f)^{-1}\Gamma_p(e^{j\omega})$ can also be bounded by some finite number K. In this case, using Lemma 4.C.2 gives that for some $|\eta| < 1$

$$\begin{aligned} |\Gamma_{p}(\omega)^{*}M_{p}(f)^{-1}[I - M_{p}(f)M_{p}(1/f)]\Gamma_{p}(\omega)| &\leq \\ &\leq \sum_{m=0}^{p-1}\sum_{n=0}^{p-1} \left| \left[\Gamma_{p}(\omega)^{*}M_{p}(f)^{-1} \right]_{m} \right| \times \\ &\times \left| \left[M_{p}(f)M_{p}(1/f) \right]_{m,n} - \left[M_{p}(1) \right]_{m,n} \right| \left| \left[\Gamma_{p}(\omega) \right]_{n} \\ &\leq K_{1}K^{2} \sum_{m=0}^{p-1}\sum_{n=0}^{p-1} (\eta^{p-m} + \eta^{m})(\eta^{p-n} + \eta^{n}) \\ &= K_{1}K^{2} \left(\frac{1 - \eta^{p}}{1 - \eta} \right)^{2} (\eta^{p} + \eta)^{2} < \infty. \end{aligned}$$

But by Lemma 5.E.3

$$\frac{1}{2}\sum_{k=0}^{p-1}(1-|\xi_k|) \le \gamma_p(\omega),$$

so that under the conditions of the lemma

$$\lim_{p \to \infty} \frac{\Gamma_p(\omega)^* M_p(f)^{-1} [I - M_p(f) M_p(1/f)] \Gamma_p(\omega)}{\gamma_p(\omega)} = 0.$$

Therefore, by using Theorem 4.B.1

$$\lim_{p \to \infty} \frac{\Gamma_p(\omega)^* M_p(f)^{-1} \Gamma_p(\omega)}{\gamma_p(\omega)} = \lim_{p \to \infty} \frac{\Gamma_p(\omega)^* M_p(1/f) \Gamma_p(\omega)}{\gamma_p(\omega)} = f^{-1}(\omega).$$

4.C Technical Lemmas

Lemma 4.C.1. [NHG97b] Let $K_p(\omega, \sigma)$ denote the reproducing kernel of the OBFP $\{\mathcal{B}_k\}$ introduced in Section 3.5. Then

$$\frac{1}{2\pi} \int_{-\pi}^{\pi} |K_p(\omega, \sigma)|^2 \,\mathrm{d}\sigma = \sum_{m=0}^{p-1} |\mathcal{B}_m(e^{j\omega})|^2 \triangleq \gamma_p(\omega).$$

Furthermore, for any $\delta > 0$, provided

$$\sum_{k=0}^{\infty} (1 - |\xi_k|) = \infty,$$

then

$$\lim_{p \to \infty} \frac{1}{\gamma_p(\omega)} \int_{\sigma \notin [\omega - \delta, \omega + \delta]} |K_p(\omega, \sigma)|^2 \, \mathrm{d}\sigma = 0.$$

Proof: See [NHG97b].

Lemma 4.C.2. Let $f, g \in L_2([-\pi, \pi])$ have finite dimensional spectral factorizations. Then there exists $|\eta| < 1$ and $K < \infty$ such that

$$\left| [M_p(f)M_p(g)]_{m,n} - [M_p(fg)]_{m,n} \right| \le K(\eta^{p-m} + \eta^m)(\eta^{p-n} + \eta^n).$$

Proof: See [NHG96].

Lemma 4.C.3. Suppose $f \in L_2([-\pi, \pi])$ has a finite dimensional spectral factorization. Then $\exists K < \infty$ which is independent of p such that

$$\left| \left[M_p(f) \Gamma_p(\omega) \right]_{\ell} \right| < K, \quad \left| \left[M_p^{-1}(f) \Gamma_p(\omega) \right]_{\ell} \right| < K.$$

Proof: See [NHG96].

$\mathbf{5}$

MIMO Identification using Orthonormal Bases

In this chapter the idea of using orthonormal bases and PEM (least squares techniques) for the identification of discrete-time linear time-invariant systems will be extended from the SISO to the MIMO (Multiple-Input Multiple-Output) setting. It will be shown how the rational orthonormal bases with fixed poles introduced in Section 3.5 (or the corresponding MIMO bases generated as in Section 3.6) can be used to linearly parameterize any multivariable fixed denominator model structure. The use of these bases will allow the incorporation in the identification process of prior knowledge about dominant dynamics of the system, and will facilitate the analysis of the estimation accuracy. As done for the SISO case, the accuracy of the estimation will be quantified by deriving expressions for an upper bound on the undermodelling error and for the asymptotic (in data-length and model order) covariance of the transfer matrix estimate (noise induced error). The asymptotic covariance analysis will be based on the derivation of convergence properties of some block Toeplitz-like matrices. The recently popular Subspace-based State Space System IDentification (4SID) methods for multivariable systems [Vd96] will be briefly reviewed, and through simulation experiments, their performance will be compared with that of the orthonormal basis-based methods proposed here.

5.1 Introduction

To the best of our knowledge, except for the well known special case of FIR multivariable model structure studied in [YL84, Zhu89, Zhu94], the use of more general bases in MIMO system identification has not been studied to date except for the work in [HbVB95] and work by Ninness, the current author and co-workers in [NnGW95, NG96].

Following the SISO paradigm for identification using orthonormal bases and least squares techniques described in the previous chapter, the obvious extension to the MIMO setting would be to construct general orthonormal bases for the space $H_2^{m \times n}(\mathbb{T})$ of $(m \times n)$ transfer matrices whose elements are in $H_2(\mathbb{T})$, and then to parameterize the transfer matrix of the system as a series expansion in terms of these bases. The identification would then be performed by estimating a finite number of expansion coefficients using least squares techniques. However, in contrast to the SISO case, several parameterizations of the system using orthonormal bases are possible in the MIMO context. For instance, in [YL84], Yuan and Ljung use the standard FIR scalar bases $\{z^{-k}\}$ to parameterize the $(m \times n)$ transfer matrix of the system as a linear combination of the bases, where the coefficients are $(m \times n)$ matrices (the impulse response matrices or *Markov (matrix) parameters*).

This idea of using scalar bases in identification of MIMO systems is applied here for the case of the more general orthonormal bases studied in Section 3.5. In addition, it is shown that the parameterization of the MIMO system using scalar bases (with matrix coefficients) is equivalent to the parameterization using matrix bases (generated from the scalar ones via Theorem 3.6.1) with scalar coefficients.

As mentioned in the introduction of the previous chapter, some emphasis has been placed in the literature [Wah91b, Wah94b, Wah91a, Oli95a, Bod95] on the use of orthonormal model structures as an implementational tool with certain numerical properties ensuring the well-posedness of the least squares estimation problem. In this chapter, an upper bound on the condition number of the least squares estimation will be derived. This guarantees a worst case numerical conditioning of the estimation using orthonormal structures. However, it will be shown that since this upper bound is completely specified by the input spectral density and is not affected by the particular bases chosen (as far as they are orthonormal), the numerical conditioning can still be very bad depending on the nature of the input.

In consideration of this, and as has already been mentioned, the approach in this thesis, following [NHG97a], is to consider the orthonormal structure as an analysis tool rather than an implementational tool. We will show that any multivariable fixed denominator model structure can be linearly re-parameterized using rational orthonormal bases with the same fixed poles. The analysis of the estimation accuracy can then be carried out on the orthonormal structure in a more tractable way by exploiting the orthonormality property of the bases.

In this chapter, we extend the single-input, single-output results concerning the accuracy of the estimation presented in [Wah91b, Wah94b] for Laguerre and Kautz models, and in [VHB95, NG97, NHG97a] (as summarized in Sections 4.3 and 4.4) for more general models, to the multivariable setting; and also the multivariable results in [YL84] from the FIR setting to more general model structures which encompass the FIR structure as a special case.

The main contribution of the chapter is the extension of the asymptotic

FIR results of [YL84] to the case of using general orthonormal bases with fixed poles. More specifically, in [YL84] the variance of the FIR transfer function matrix estimate was shown to be approximately (for large data-length N and large model order p) equal to

$$\frac{p}{N} \Phi_u^{-1}(\omega) \otimes \Phi_\nu(\omega)$$

where $\Phi_u(\omega)$ is the input spectral density, $\Phi_\nu(\omega)$ is the output measurement noise spectral density, and \otimes is the Kronecker matrix product (see Appendix A for the definition and properties). In this chapter we show that for the OBFP described in Section 3.5, the above expression should be changed to

$$\frac{\gamma_p(\omega)}{N} \, \Phi_u^{-1}(\omega) \otimes \Phi_\nu(\omega).$$

where $\gamma_p(\omega) \triangleq \sum_{k=0}^{p-1} |\mathcal{B}_k(e^{j\omega})|^2$ (see Figure 4.2). Note that for FIR models $\gamma_p(\omega) = p$ so that the new expression contains the previously known FIR model structure result [YL84] as a special case. The expression is also in formal analogy with the single-input, single-output result of Theorem 4.4.2. The derivation, however, is considerably more complicated. A first difficulty (inherited from the SISO case) is that the OBFP we use in this thesis do not have the algebraic structure that would allow to reduce the problem to the FIR case. A second difficulty is that in the MIMO case several quantities do not commute. Fundamental for the analysis will be the derivation of some results concerning the convergence of block Toeplitz-like matrices.

The rest of the chapter is organized as follows. The identification problem is stated in Section 5.2. In that section we also show how the orthonormal basis with fixed poles introduced in Section 3.5 can be used to linearly reparameterized any multivariable fixed denominator model structure. In Section 5.3, it is shown that the parameterization of the MIMO system using (matrix) orthonormal bases (with scalar coefficients) is equivalent to the parameterization using scalar bases with matrix coefficients. The MIMO identification problem using orthonormal bases and least squares techniques is solved in Section 5.4. The numerical robustness of the identification algorithms is analyzed in Section 5.5. In Section 5.6 we analyze the undermodelling induced error that results from the parsimony of the model structure (due to a finite number of expansion terms), which is too simple to exactly represent the system. In Section 5.7, the main contribution of the chapter is derived. Namely, an expression for the asymptotic (in number of observed data and in model order) covariance of the transfer matrix estimate is obtained. This expression is used to quantify the estimation error induced by the presence of measurement noise. The derived result is consistent with that presented in [YL84] for the particular case of FIR multivariable model structures, and indicates that the noise induced error is asymptotically proportional to the (generalized) noise-to-signal ratio. The analysis is based on new results concerning the convergence of block Toeplitz-like matrices. For the purposes of comparison with the orthonomal basis-based identification method proposed in this chapter, a brief review of Subspace-based State Space System IDentification (4SID) methods for multivariable systems is given in Section 5.9. Finally, some simulation examples are presented in Section 5.10, and some conclusions in Section 5.11.

5.2 **Problem Formulation**

We address the problem of identification of Discrete-Time Linear Time-Invariant MIMO systems from observed input-output data in the time domain. To be more specific, it is assumed that the system has n inputs and m outputs and that N samples of n input sequences $\{u_k^1\}, \{u_k^2\}, \ldots, \{u_k^n\}$, as well as m output sequences $\{y_k^1\}, \{y_k^2\}, \ldots, \{y_k^m\}$ are available for the identification. We assume also that the data are related according to

$$y_k^i = \sum_{j=1}^n G_{ij}(q) u_k^j + \sum_{s=1}^m H_{is}(q) e_k^s,$$
(5.1)

with $i = 1, 2, \dots, m$.

The scalar transfer functions $\{G_{ij}(q)\}$ and $\{H_{ij}(q)\}$ describe respectively the unknown (assumed stable) system dynamics and the disturbance model that are to be identified. A notational simplification is possible by vectorizing:

$$y_{k} \triangleq (y_{k}^{1}, y_{k}^{2}, \dots, y_{k}^{m})^{T},$$

$$u_{k} \triangleq (u_{k}^{1}, u_{k}^{2}, \dots, u_{k}^{n})^{T},$$

$$e_{k} \triangleq (e_{k}^{1}, e_{k}^{2}, \dots, e_{k}^{m})^{T},$$

$$G(q) = \begin{bmatrix} G_{11}(q) & G_{12}(q) & \dots & G_{1n}(q) \\ G_{21}(q) & G_{22}(q) & \dots & G_{2n}(q) \\ \vdots & \vdots & \dots & \vdots \\ G_{m1}(q) & G_{m2}(q) & \dots & G_{mn}(q) \end{bmatrix},$$

$$H(q) = \begin{bmatrix} H_{11}(q) & H_{12}(q) & \dots & H_{1m}(q) \\ H_{21}(q) & H_{22}(q) & \dots & H_{2m}(q) \\ \vdots & \vdots & \dots & \vdots \\ H_{m1}(q) & H_{m2}(q) & \dots & H_{mm}(q) \end{bmatrix},$$

so that (5.1) can be rewritten in matrix form as

$$y_k = G(q) \ u_k + H(q) \ e_k = G(q) \ u_k + \nu_k.$$
(5.2)

Here $\{e_k\}$ is assumed to be a stationary (zero mean) white noise vector processs with covariance matrix $\mathbf{E} \{e_k e_k^T\} = \Lambda$. In this case, the disturbance term $\nu_k = H(q) e_k$ is also a stationary process with spectral density [Lju87]

$$\Phi_{\nu}(\omega) = H(e^{j\omega})\Lambda H^{\star}(e^{j\omega}).$$
(5.3)

A standard approach is to provide the model (5.2) with a finite dimensional parameterization

$$y_k = G(q,\theta) \ u_k + H(q,\theta) \ e_k, \tag{5.4}$$

so that the system can be identified by estimating the (finite dimensional) parameter vector θ .

There are many options available for the estimation of $G(q, \theta)$ and $H(q, \theta)$ within this problem setting. For example, a general prediction error technique (as described in Chapter 2) using a multivariable Box-Jenkins [BJ76] or a state-space model structure could be employed [Lju87] (see [Lju91] for an example of identification of arbitrarily parameterized state-space models within the framework of the System Identification Toolbox [Lju95] for use with MATLAB¹[Mat94]). Unfortunately, as mentioned in Chapter 2, apart from the various difficulties concerning identifiability of these model structures [Lju87, GW74, Gui75, Gui81], this strategy may also result in a numerically intensive iterative (possibly nonlinear and nonconvex) optimization procedure. In addition, the analysis of the (finite data) estimation accuracy becomes very difficult (sometimes intractable) for this case.

Another possibility is to employ one of the recently popular Subspacebased State-Space System IDentification $(4SID)^2$ methods for multivariable systems [Vd96], such as the N4SID [Vd94a, Vd94b, Vd91a, Vib94, Vib95, OV94, VOWL93, VOWL91], MOESP [Ver91, VD91b, VD92], and CVA [Lar90, Lar94] methods. These schemes provide accurate state-space models for multivariable systems directly from input-output data, and have the advantage that no iterative procedures are involved. Instead, they employ reliable numerical algorithms such as Singular Value Decomposition (SVD), and QR-decomposition [GV89]. Work is still progressing on quantifying the estimation error involved with the use of such methods [VWO97, DPS95, DPS94].

In this thesis, our interest is not on the estimation of the model for the additive output noise (that is, on the estimation of $H(q, \theta)$), whose second order statistics will be assumed to be known, but only on the estimation of the transfer matrix $G(q, \theta)$ describing the system dynamics. We concentrate then on the parameterized model

$$y_k = G(q, \theta) \ u_k + \nu_k, \tag{5.5}$$

¹MATLAB is a registered trademark of The MathWorks, Inc.

²Pronounced 'force it'.

rather than on (5.4).

Furthermore, and for the sake of ease of error quantification, lack of identifiability problems and small computational load, we will focus on the study of prediction error methods with a quadratic criterion and with a particular model structure that has fixed poles that are chosen according to prior knowledge about the system. To facilitate the analysis of estimation accuracy we will further re-parameterize this model structure using orthonormal bases with the same fixed poles.

The particular model structure we will consider is given as

$$G(q,\beta) = \mathcal{D}_p^{-1}(q) \sum_{k=0}^{p-1} \beta_k q^k,$$
 (5.6)

$$\mathcal{D}_p(q) = \prod_{\ell=0}^{p-1} (q - \xi_\ell), \tag{5.7}$$

where $\beta_k \in \mathbb{R}^{m \times n}$, $k = 0, \dots, p-1$, are matrices of parameters to be estimated (namely, the coefficients of the numerator polynomials of the individual transfer functions $\{G_{ij}(q)\}$), and $\{\xi_0, \xi_1, \dots, \xi_{p-1}\}$ are the poles chosen by the user to reflect prior knowledge about the true system G(q). The advantage of this model structure is that it allows the input-output relationship to be easily cast in linear regressor form as

$$y_k = \beta^T \psi_k + \nu_k, \tag{5.8}$$

with

$$\beta^{T} \triangleq (\beta_{0}, \beta_{1}, \cdots, \beta_{p-1}), \qquad (5.9)$$
$$\begin{bmatrix} \mathcal{D}_{p}^{-1}(q)I_{n} \end{bmatrix}$$

$$\psi_{k} \triangleq \begin{vmatrix} q \mathcal{D}_{p}^{-1}(q) I_{n} \\ q^{2} \mathcal{D}_{p}^{-1}(q) I_{n} \\ \vdots \\ q^{p-1} \mathcal{D}_{p}^{-1}(q) I_{n} \end{vmatrix} u_{k} = \Upsilon_{p}(q) u_{k}, \qquad (5.10)$$

$$\Upsilon_p(q) \triangleq \left[\mathcal{D}_p^{-1}(q) I_n, q \mathcal{D}_p^{-1}(q) I_n, \cdots, q^{p-1} \mathcal{D}_p^{-1}(q) I_n \right]^T,$$
(5.11)

so that if a quadratic criterion is used, the resulting least squares estimate can be found in a computationally cheap manner. It is well known that the least squares estimate can be written in closed form as

$$\widehat{\beta} = \left(\sum_{k=0}^{N-1} \psi_k^T \psi_k\right)^{-1} \sum_{k=0}^{N-1} \psi_k y_k^T.$$
(5.12)

Our main interest is now to quantify the accuracy of the resulting transfer matrix estimate

$$G(q,\hat{\beta}) = \hat{\beta}^T \Upsilon_p(q), \tag{5.13}$$

and particularly to study how the estimation error is affected by the choice of the poles $\{\xi_k\}$ in $\mathcal{D}_p(q)$.

To undertake this study let us first notice that since via equations (5.6), (5.7), (5.9), (5.11), and (5.12) the estimate $G(q, \hat{\beta})$ is linear in the data, then it is invariant under linear re-parameterization of the model structure (5.6) (See Lemma 5.E.1 in Appendix 5.E). This implies that the same transfer matrix estimate $G(q, \hat{\beta})$ is obtained if one instead chooses the model structure

$$G(q,\Theta) = \sum_{k=0}^{p-1} \theta_k^T \mathcal{B}_k(q) = \Theta^T \ \Gamma_p(q),$$
(5.14)

where the transfer functions $\{\mathcal{B}_{\ell}(q)\}\$ are the orthonormal bases with fixed poles (OBFP) introduced in Section 3.5, with the same poles as in $\mathcal{D}_p(q)$, and where now

$$\Theta \triangleq \left[\theta_0^T, \theta_1^T, \cdots, \theta_{p-1}^T\right]^T,$$

with $\theta_k^T \in \mathbb{R}^{m \times n}$, and $\Gamma_p(q) \triangleq [\mathcal{B}_0(q)I_n, \mathcal{B}_1(q)I_n, \cdots, \mathcal{B}_{p-1}(q)I_n]^T$.

Although the two formulations (5.6)-(5.7) and (5.14) are equivalent in the sense that the obtained tranfer matrix estimates are identical (i.e. $G(q, \hat{\Theta}) = G(q, \hat{\beta})$), the latter structure (5.14) is much preferable from an analytical point of view, since the basis functions $\{\mathcal{B}_{\ell}(q)\}$ are orthonormal, and this property can be exploited in the quantification of the estimation accuracy. Due to this equivalence, it is not difficult to see that the bias and variance error properties of the estimates derived from the two model structures are also identical. However, since the orthonormal structure (5.14) is more tractable, this is the model structure we will employ for the analysis in this chapter, keeping in mind that the results of this analysis can also be applied to the model structure (5.6)-(5.7) or any other linearly equivalent one.

As mentioned in the introduction of the chapter, several parameterizations using orthonormal basis are possible in the MIMO framework. Besides the model structure (5.14) that corresponds to the case of scalar bases $\{\mathcal{B}_{\ell}(z)\}$ (i.e. bases of the space $H_2(\mathbb{T})$) with matrix coefficients $\theta_{\ell}^T \in \mathbb{R}^{m \times n}$, we will also consider an orthonormal structure corresponding to $(m \times n)$ -matrix bases (i.e. bases of the space $H_2^{m \times n}(\mathbb{T})$, as the ones introduced in Section 3.6) with scalar coefficients $\theta_{\ell}^T \in \mathbb{R}$. In the following section we prove that both parameterizations are equivalent.

5.3 An equivalent MIMO parameterization

Let $\{\mathcal{B}_{\ell}^{ij}(z)\}_{\ell=0}^{\infty}$, $(i = 1, \ldots, m; j = 1, \ldots, n)$ be an orthonormal basis for the space $H_2^{m \times n}(\mathbb{T})$ generated from the scalar bases $\{\mathcal{B}_{\ell}(z)\}_{\ell=0}^{\infty}$ as in Theorem 3.6.1. Then any stable causal transfer matrix G(z) in $H_2^{m \times n}(\mathbb{T})$ can be approximated by a linear combination of a finite number of elements of the orthonormal set, that is

$$G(z,\Theta) = \sum_{\ell=0}^{p-1} \sum_{i=1}^{m} \sum_{j=1}^{n} \theta_{\ell}^{ij} \mathcal{B}_{\ell}^{ij}(z).$$
(5.15)

The following Lemma shows that, for the appropriate definition of the parameter matrices θ_{ℓ} , the above expression is equivalent to

$$G(z,\Theta) = \sum_{\ell=0}^{p-1} \theta_{\ell}^{T} \mathcal{B}_{\ell}(z).$$
(5.16)

Lemma 5.3.1. Let $\{\mathcal{B}_{\ell}^{ij}(z)\}_{\ell=0}^{\infty}, (i = 1, \ldots, m; j = 1, \ldots, n)$ be an orthonormal basis on $H_2^{m \times n}(\mathbb{T})$ generated from the scalar basis $\{\mathcal{B}_{\ell}(z)\}_{\ell=0}^{\infty}$ as in Theorem 3.6.1. Then the following identity holds

$$\sum_{\ell=0}^{p-1}\sum_{i=1}^{m}\sum_{j=1}^{n}\theta_{\ell}^{ij}\mathcal{B}_{\ell}^{ij}(z) = \sum_{\ell=0}^{p-1}\theta_{\ell}^{T}\mathcal{B}_{\ell}(z),$$

where the parameter matrices θ_{ℓ} are defined as

$$\theta_{\ell}^{T} \triangleq \begin{bmatrix} \theta_{\ell}^{11} & \dots & \theta_{\ell}^{1n} \\ \vdots & \ddots & \vdots \\ \theta_{\ell}^{m1} & \dots & \theta_{\ell}^{mn} \end{bmatrix}.$$
 (5.17)

Proof: See Appendix 5.A.

Remark 5.3.1. This result means that the two orthonormal structures we have considered thus far (given by equations (5.16) and (5.15)) are completely equivalent, so that we can use any of them indistinctively to represent the transfer matrix of the system. Most of the analysis of the following sections will be carried out for the parameterization with scalar bases and matrix coefficients (5.16). However, when convenient, the parameterization with matrix bases will also be used.

5.4 Parameter Estimation

The representation in (5.14) leads to the convenient linear regressor form

$$y_{k} = G(q, \Theta)u_{k} + \nu_{k},$$

$$= \left(\theta_{0}^{T}, \theta_{1}^{T}, \cdots, \theta_{p-1}^{T}\right) \begin{bmatrix} I_{n}\mathcal{B}_{0}(q) \\ I_{n}\mathcal{B}_{1}(q) \\ \vdots \\ I_{n}\mathcal{B}_{p-1}(q) \end{bmatrix} u_{k} + \nu_{k},$$

$$= \Theta^{T} \Gamma_{p}(q) u_{k} + \nu_{k},$$

$$= \Theta^{T} \Gamma_{p}(q) u_{k} + \nu_{k},$$

$$= \Theta^{T} \phi_{k} + \nu_{k},$$
(5.18)

where we have $defined^3$

$$\Theta^T \triangleq \left(\theta_0^T, \theta_1^T, \cdots, \theta_{p-1}^T\right), \tag{5.19}$$

$$\mathsf{B}_{p}(q) \triangleq \left[\mathcal{B}_{0}(q), \cdots, \mathcal{B}_{p-1}(q)\right]^{T}, \qquad (5.20)$$

$$\Gamma_p(q) \triangleq \mathsf{B}_p(q) \otimes I_n, \tag{5.21}$$

$$\phi_k \triangleq \Gamma_p(q) u_k. \tag{5.22}$$

With the above definitions for Θ and $\Gamma_p(q)$, the transfer matrix is given by

$$G(q,\Theta) = \Theta^T \Gamma_p(q).$$
(5.23)

With the system in linear regressor form (5.18), the most obvious scheme for estimating the parameter matrix Θ is the least squares method. The least squares estimate $\widehat{\Theta}$ of Θ is the minimizing argument of the quadratic criterion

$$V_N(\Theta) = \frac{1}{N} \operatorname{Tr} \left\{ \sum_{k=0}^{N-1} \varepsilon_k(\Theta) \varepsilon_k^T(\Theta) \right\},$$
(5.24)

where $\varepsilon_k(\Theta) = y_k - G(q, \Theta)u_k$. That is

$$\widehat{\Theta} = \underset{\Theta}{\operatorname{arg\,min}} \left\{ V_N(\Theta) \right\} = \underset{\Theta}{\operatorname{arg\,min}} \left\{ \frac{1}{N} \operatorname{Tr} \left\{ \sum_{k=0}^{N-1} \varepsilon_k(\Theta) \varepsilon_k^T(\Theta) \right\} \right\}.$$
(5.25)

It is well known [Lju87] that this optimization problem has an explicit solution given by

$$\widehat{\Theta} = \left(\frac{1}{N}\sum_{k=0}^{N-1}\phi_k\phi_k^T\right)^{-1} \left(\frac{1}{N}\sum_{k=0}^{N-1}\phi_ky_k^T\right) = \widetilde{R}_p^{-1}(N)\left(\frac{1}{N}\sum_{k=0}^{N-1}\phi_ky_k^T\right), \quad (5.26)$$

³Please don't confuse $B_p(q)$ (a column vector of transfer functions), with $\mathcal{B}_p(q)$ (a scalar transfer function).

where we have defined

$$\widetilde{R}_p(N) \triangleq \left(\frac{1}{N} \sum_{k=0}^{N-1} \phi_k \phi_k^T\right).$$
(5.27)

Adopting the vectorized notation

$$Y^{T} = (y_{0}, y_{1}, \cdots, y_{N-1}),$$

$$\Phi^{T} = (\phi_{0}, \phi_{1}, \cdots, \phi_{N-1}),$$

$$V^{T} = (\nu_{0}, \nu_{1}, \cdots, \nu_{N-1}),$$

the model for the N point observed data record can be written as:

$$Y = \Phi \Theta + V, \tag{5.28}$$

so that the estimate $\widehat{\Theta}$ defined in equation (5.25) can be written in a more compact form as

$$\widehat{\Theta} = \Phi^{\dagger} Y, \tag{5.29}$$

where Φ^{\dagger} is the *Moore-Penrose* pseudoinverse of Φ . If there is sufficient input excitation for the indicated inverse to exist this will be given by

$$\Phi^{\dagger} = (\Phi^T \Phi)^{-1} \Phi^T.$$
(5.30)

The frequency response estimate is then given by

$$G(e^{j\omega}, \widehat{\Theta}) = \widehat{\Theta}^T \Gamma_p(e^{j\omega}).$$
(5.31)

Remark 5.4.1. Although the proposed identification scheme can be used with any orthonormal basis $\{\mathcal{B}_{\ell}(z)\} \in H_2(\mathbb{T})$, we will restrict the analysis of estimation accuracy to the generalized orthonormal basis with fixed poles (OBFP) introduced in Section 3.5 (or the equivalent MIMO version generated via Theorem 3.6.1), keeping in mind that these bases allow the representation of a more extensive class of models. In particular, they can be used to represent the multivariable fixed denominator model structure with arbitrary poles in (5.6)-(5.7). This is not the case with other orthonormal bases such as FIR, Laguerre, Kautz, or the more general OBGIF, where the poles are restricted either to be all the same, or to come from the same set which is repeated cyclically as higher model orders are assumed.

In the following sections we analyze the numerical properties of the proposed identification algorithm, as well as the accuracy of the estimation (bias and variance errors).

5.5 Numerical properties

The numerical robustness of identification algorithms is particularly important in the context of multivariable systems where large numbers of parameters need to be estimated [Vd94a, Vib94]. Furthermore, it is well known [GV89] that when least squares procedures are employed, these numerical properties are strongly governed by the condition number of the 'covariance' matrix $\tilde{R}_p(N)$ (defined in (5.27)) in the normal equation (5.26).

For the single-input, single-output case this condition number has been upper bounded in terms of the input spectral density $\Phi_u(\omega)$ when particular orthonormal basis model structures are employed [Wah91b, Wah94b, VHB95, NHG97b, GS58].

This result can be extended to arbitrary orthonormal bases and to the multivariable setting. This is done in the following Lemma, where (as in [Wah91b, Wah94b, NHG97b]) we compute lower and upper bounds on the singular values of the asymptotic (in data-length N) covariance matrix \tilde{R}_p defined as

$$\widetilde{R}_p \triangleq \lim_{N \to \infty} \widetilde{R}_p(N) = \lim_{N \to \infty} \left(\frac{1}{N} \sum_{k=0}^{N-1} \phi_k \phi_k^T \right) = \overline{\mathbf{E}} \left\{ \phi_k \phi_k^T \right\}$$

Lemma 5.5.1. Let the set of singular values of \widetilde{R}_p be denoted as $\sigma[\widetilde{R}_p]$ and let $\underline{\sigma}[\Phi_u(\omega)]$ and $\overline{\sigma}[\Phi_u(\omega)]$ denote the smallest and the largest singular values of the input spectral density $\Phi_u(\omega)$, respectively. Then

$$\sigma[\widetilde{R}_p] \subset \left[\inf_{\omega} \underline{\sigma}[\Phi_u(\omega)], \sup_{\omega} \overline{\sigma}[\Phi_u(\omega)]\right].$$

Proof: See Appendix 5.A.

Using this lemma, the condition number $\kappa[\widetilde{R}_p]$ of \widetilde{R}_p may be straightforwardly upper bounded in terms of the singular values of the input spectral density as

$$1 \le \kappa[\widetilde{R}_p] \le \frac{\sup_{\omega} \overline{\sigma}[\Phi_u(\omega)]}{\inf_{\omega} \underline{\sigma}[\Phi_u(\omega)]}.$$
(5.32)

This upper bound provides a guaranteed worst-case limit for the numerical conditioning. Notice however that since the upper bound depends exclusively on the input excitation, the numerical conditioning can be very bad depending on the nature of this input. For the particular case of white input, the input spectral density $\Phi_u(\omega)$ is constant, so that equation (5.32) implies

$$\kappa[\tilde{R}_p] = 1,$$

which means that the least squares estimation using the orthonormal structure is perfectly numerically conditioned.

Notice that the way the numerical conditioning is affected by an increasing model order (which is the case where the numerical issue is more important [GV89]) or by the particular choices for the poles of the bases, is not reflected by the upper bound we have derived.

As already mentioned, this guarantee of the worst-case numerical conditioning appears to have been one of the main motivations for using orthonormal bases in a system identification setting. However, to the best of our knowledge, there is no theoretical evidence to suggest that an orthonormal structure will necessarily provide a better numerical conditioning when compared with the one obtained with an equivalent non-orthonormal structure. In the following, lower and upper bounds on the ratio between the condition numbers using orthonormal and non-orthonormal model structures are derived.

Let the transfer matrix of the system be represented as a truncated series expansion in terms of the orthonormal basis $\{\mathcal{B}_k(q)\}_{k=0}^{p-1}$. This representation leads to the linear regressor form of equation (5.18), or its vectorized form (5.28). The asymptotic (in data-length) numerical conditioning of the least squares estimation is then given by the condition number of the matrix $\overline{\mathbf{E}} \{\phi_k \phi_k^T\}$.

Similarly, the transfer matrix can be represented as a linear combination of the elements of a (non-orthonormal) set $\{\mathcal{A}_k(q)\}_{k=0}^{p-1}$, spanning the same space as $\{\mathcal{B}_k(q)\}_{k=0}^{p-1}$. For example, for the case of $\{\mathcal{B}_k(q)\}_{k=0}^{p-1}$ been the OBFP introduced in Section 3.5, a possible non-orthonormal set spanning the same space is given by the functions $\mathcal{A}_k(q) = 1/(q - \xi_k)$, with the same poles as in the set $\{\mathcal{B}_k(q)\}$. The representation of the system using the non-orthonormal set also leads to a linear regressor form

$$y_k = G(q, \alpha)u_k + \nu_k = \alpha^T \psi_k + \nu_k,$$

where $\alpha \in \mathbb{R}^{np \times m}$ is the parameter matrix to be estimated, and where

$$\begin{aligned} \psi_k &\triangleq & \Upsilon_p(q) u_k, \\ \Upsilon_p(q) &\triangleq & \mathsf{A}_p(q) \otimes I_n, \\ \mathsf{A}_p(q) &\triangleq & [\mathcal{A}_0(q), \cdots, \mathcal{A}_{p-1}(q)]^T. \end{aligned}$$

In vectorized form we can write

$$Y = \Psi \alpha + V,$$

$$Y^{T} \triangleq (y_{0}, y_{1}, \cdots, y_{N-1}),$$

$$\Psi^{T} \triangleq (\psi_{0}, \psi_{1}, \cdots, \psi_{N-1}),$$

$$V^{T} \triangleq (\nu_{0}, \nu_{1}, \cdots, \nu_{N-1}).$$

In this case the asymptotic (in data-length) numerical conditioning of the least squares estimation of the parameter matrix α is determined by the condition number of the matrix $\overline{\mathbf{E}} \{ \psi_k \psi_k^T \}$.

Our objective is now to relate the matrices $\overline{\mathbf{E}} \left\{ \phi_k \phi_k^T \right\}$ and $\overline{\mathbf{E}} \left\{ \psi_k \psi_k^T \right\}$ in order to compare the numerical conditioning of the estimation using orthonormal and non-orthonormal model structures. To proceed with this, notice that since $\mathcal{A}_k(z) \in \text{Span} \left\{ \{ \mathcal{B}_k(z) \}_{k=0}^{p-1} \right\}$ we can write

$$\mathcal{A}_{k}(z) = \sum_{j=0}^{p-1} \langle \mathcal{A}_{k}(z), \mathcal{B}_{j}(z) \rangle \mathcal{B}_{j}(z),$$

= $[\langle \mathcal{A}_{k}(z), \mathcal{B}_{0}(z) \rangle, \cdots, \langle \mathcal{A}_{k}(z), \mathcal{B}_{p-1}(z) \rangle] \mathsf{B}_{p}(z).$

Then $A_p(z)$ and $B_p(z)$ are related according to

$$\mathsf{A}_p(z) = J^{-1}\mathsf{B}_p(z),$$

where $J \in \mathbb{R}^{p imes p}$ is the nonsingular matrix

$$J = \begin{bmatrix} \langle \mathcal{A}_0(z), \mathcal{B}_0(z) \rangle & \cdots & \langle \mathcal{A}_0(z), \mathcal{B}_{p-1}(z) \rangle \\ \vdots & \vdots & \vdots \\ \langle \mathcal{A}_{p-1}(z), \mathcal{B}_0(z) \rangle & \cdots & \langle \mathcal{A}_{p-1}(z), \mathcal{B}_{p-1}(z) \rangle \end{bmatrix}^{-1}$$

Based on this it is not difficult to show that the regressors ψ_k and ϕ_k are related according to

$$\psi_k = \left(J^{-1} \otimes I_n\right) \phi_k,\tag{5.33}$$

and then

$$\overline{\mathbf{E}}\left\{\psi_{k}\psi_{k}^{T}\right\} = \left(J^{-1}\otimes I_{n}\right)\overline{\mathbf{E}}\left\{\phi_{k}\phi_{k}^{T}\right\}\left(J^{-1}\otimes I_{n}\right)^{T},$$
(5.34)

which is the relation we were looking for. Considering now that for a given norm and matrices A and B of appropriate dimensions [GV89]

$$\kappa[AB] \le \kappa[A]\kappa[B],$$

from equation (5.34) we can write

$$\frac{1}{\kappa[(J \otimes I_n)]^2} \le \frac{\kappa[\overline{\mathbf{E}}\left\{\psi_k \psi_k^T\right\}]}{\kappa[\overline{\mathbf{E}}\left\{\phi_k \phi_k^T\right\}]} \le \kappa[(J \otimes I_n)]^2.$$
(5.35)

The above equation shows that it is not necessarily true that the use of orthonormal model structures represents an improvement on the numerical conditioning of the estimation with respect to the case of using non-orthonormal model structures. The following example illustrates this fact. **Example 5.5.1.** Let us consider the scalar case where the orthonormal functions $\{B_k(z)\}$ are the Laguerre bases

$$\mathcal{B}_k(z) = \left(\frac{\sqrt{1-\xi^2}}{z-\xi}\right) \left(\frac{1-\xi z}{z-\xi}\right)^k \quad ; \quad k \ge 0,$$
(5.36)

and where the non-orthonormal functions $\{A_k(z)\}$ are given by

$$\mathcal{A}_k(z) = \frac{1}{(z-\xi)^{k+1}} \quad ; \quad k \ge 0,$$
(5.37)

with the same fixed pole ξ , and spanning the same space as $\{\mathcal{B}_k(z)\}\$. For the case of model order p = 3, matrix J is given by

$$J = \begin{bmatrix} \sqrt{1-\xi^2} & 0 & 0\\ -\xi\sqrt{1-\xi^2} & (1-\xi^2)\sqrt{1-\xi^2} & 0\\ \xi^2\sqrt{1-\xi^2} & 2\xi(\xi^2-1)\sqrt{1-\xi^2} & \sqrt{1-\xi^2}(1-\xi^4+2\xi^2(\xi^2-1)) \end{bmatrix}$$

Figure 5.1 shows the upper bound $\kappa[(J \otimes I_n)]^2$ in equation (5.35) as a function of the pole location (ξ) for this case.



Figure 5.1: Upper bound on the ratio between the condition numbers for the non-orthonormal and the orthonormal structures $(\kappa[\overline{\mathbf{E}} \{\psi_k \psi_k^T\}]/\kappa[\overline{\mathbf{E}} \{\phi_k \phi_k^T\}])$ as a function of the pole location ξ , and for model order p = 3.

The left hand plot of Figure 5.2 shows the condition numbers $\kappa[\mathbf{\bar{E}} \{\psi_k \psi_k^T\}]$ (dashed line) and $\kappa[\mathbf{\bar{E}} \{\phi_k \phi_k^T\}]$ (solid line) corresponding to the non-orthonormal and the orthonormal structures respectively, as a function of the model order p, for the case $\xi = -0.1$. The right hand plot of the figure shows the ratio $\kappa[\mathbf{\bar{E}} \{\psi_k \psi_k^T\}]/\kappa[\mathbf{\bar{E}} \{\phi_k \phi_k^T\}]$ between these conditions numbers. The simulations were performed with a unit amplitude, square wave input signal of fundamental frequency 0.01 Hz.



Figure 5.2: Left hand figure shows the condition numbers $\kappa[\mathbf{E}\{\psi_k\psi_k^T\}]$ (dashed line) and $\kappa[\mathbf{E}\{\phi_k\phi_k^T\}]$ (solid line) corresponding to the nonorthonormal and the orthonormal structures respectively, as a function of the model order p for the case $\xi = -0.1$. Right hand figure shows the ratio $\kappa[\mathbf{E}\{\psi_k\psi_k^T\}]/\kappa[\mathbf{E}\{\phi_k\phi_k^T\}]$ as a function of p for this case.

The simulations show that for the case p = 3 and $\xi = -0.1$ the upper and lower bounds on the ratio between the condition numbers of the nonorthonormal and orthonormal covariance matrices are

$$\kappa[(J \otimes I_n)]^2 = 1.5688$$
 , $\frac{1}{\kappa[(J \otimes I_n)]^2} = 0.6374$,

while the value of this ratio is

$$\frac{\kappa[\overline{\mathbf{E}}\left\{\psi_k\psi_k^T\right\}]}{\kappa[\overline{\mathbf{E}}\left\{\phi_k\phi_k^T\right\}]} = 0.6698.$$

It is clear from the plots in Figure 5.2 that, in this case, the use of the orthonormal structure (5.36) instead of the more natural non-orthonormal form (5.37) results in a deterioration of the numerical conditioning rather than in an improvement.

5.6 Undermodelling Error

In this section we derive results regarding the quantification of the component of the transfer matrix estimation error that is produced by the restricted complexity of the model used to represent the system. For the single-input, single-output case, the result of Theorem 4.3.1, in the form of an upper bound on the undermodelling error, shows that the flexibility in the choice of the poles of the bases can be exploited in order to provide a more accurate estimation. A similar result can be established for the multivariable case '*mutatis mutandis*' from the single-input, single-output result in Theorem 4.3.1.

Let $\widehat{G}(z)$ denote the best $H_2^{m \times n}(\mathbb{T})$ approximation to G(z) with respect to the pmn basis matrices $\{\mathcal{B}_0^{11}, \cdots, \mathcal{B}_0^{mn}, \cdots, \mathcal{B}_{p-1}^{11}, \cdots, \mathcal{B}_{p-1}^{mn}\}$ defined by equations (3.28)-(3.14). Namely

$$\widehat{G}(z) \triangleq \sum_{k=0}^{p-1} \sum_{i=1}^{m} \sum_{j=1}^{n} \left\langle G, \mathcal{B}_{k}^{ij} \right\rangle \mathcal{B}_{k}^{ij}(z)$$
(5.38)

where $\langle \cdot, \cdot \rangle$ stands for the inner product in the space $H_2^{m \times n}(\mathbb{T})$ as defined in (3.29). Considering the definition of $\mathcal{B}_k^{ij}(z)$ in (3.28), equation (5.38) can be written as

$$\widehat{G}(z) = \sum_{k=0}^{p-1} \begin{bmatrix} \langle G, \mathcal{B}_k^{11} \rangle & \cdots & \langle G, \mathcal{B}_k^{1n} \rangle \\ \vdots & \ddots & \vdots \\ \langle G, \mathcal{B}_k^{m1} \rangle & \cdots & \langle G, \mathcal{B}_k^{mn} \rangle \end{bmatrix} \mathcal{B}_k(z)$$

Then, the *i*, *j*-th component $\widehat{G}_{ij}(z)$ of $\widehat{G}(z)$ is given by

$$\widehat{G}_{ij}(z) = \sum_{k=0}^{p-1} \left\langle G, \mathcal{B}_k^{ij} \right\rangle \mathcal{B}_k(z).$$
(5.39)

Using again the definition of $\mathcal{B}_k^{ij}(z)$, it is straightforward to show that

$$\langle G, \mathcal{B}_k^{ij} \rangle = \langle G_{ij}, \mathcal{B}_k \rangle,$$

where in the RHS, the symbol $\langle \cdot, \cdot \rangle$ stands now for the inner product in the space $H_2(\mathbb{T})$ as defined in (3.27)⁴. Substituting back in equation (5.39), the expression for $\widehat{G}_{ij}(z)$ becomes

$$\widehat{G}_{ij}(z) = \sum_{k=0}^{p-1} \langle G_{ij}, \mathcal{B}_k \rangle \, \mathcal{B}_k(z), \qquad (5.40)$$

⁴The fact that the same symbol is being used to denote both the inner product in $H_2(\mathbb{T})$ and in $H_2^{m \times n}(\mathbb{T})$ should lead to no confusion, since the meaning becomes clear from the context.

which is the best $H_2(\mathbb{T})$ approximation of $G_{ij}(z)$ with respect to the p basis functions $\{\mathcal{B}_0(z), \mathcal{B}_1(z), \dots, \mathcal{B}_{p-1}(z)\}$. In similar way, it can be proved that the i, j-th component $G_{ij}(z)$ of the true transfer matrix G(z) is given by

$$G_{ij}(z) = \sum_{k=0}^{\infty} \langle G_{ij}, \mathcal{B}_k \rangle \, \mathcal{B}_k(z).$$
(5.41)

We are now able to quantify (component-wise) the error between the true transfer function $G_{ij}(z)$ and the restricted complexity approximation $\hat{G}_{ij}(z)$. The result is summarized in the following theorem.

Theorem 5.6.1. Let the *i*, *j*-th component $G_{ij}(z)$ of the transfer matrix G(z) of the system have partial fraction expansion

$$G_{ij}(z) = \sum_{\ell=0}^{r^{ij}-1} \frac{\alpha_{\ell}^{ij}}{z - \gamma_{\ell}^{ij}}$$

where all the poles satisfy $|\gamma_{\ell}^{ij}| < 1$. Let $\widehat{G}_{ij}(z)$ denote the *i*, *j*-th component of best $H_2^{m \times n}(\mathbb{T})$ approximation $\widehat{G}(z)$ to G(z) with respect to the pmn basis matrices $\{\mathcal{B}_0^{11}, \cdots, \mathcal{B}_0^{mn}, \cdots, \mathcal{B}_{p-1}^{11}, \cdots, \mathcal{B}_{p-1}^{mn}\}$ defined by equations (3.28)-(3.14), with fixed poles $\{\xi_k\}(k=0,\cdots,p-1)$. Then

$$\left|G_{ij}(e^{j\omega}) - \widehat{G}_{ij}(e^{j\omega})\right| < \sum_{\ell=0}^{r^{ij}-1} \left|\frac{\alpha_{\ell}^{ij}}{e^{j\omega} - \gamma_{\ell}^{ij}}\right| \prod_{k=0}^{p-1} \left|\frac{\gamma_{\ell}^{ij} - \xi_k}{1 - \overline{\xi_k}\gamma_{\ell}^{ij}}\right|.$$
 (5.42)

Proof: Considering the derived expressions for $G_{ij}(z)$ and $\widehat{G}_{ij}(z)$ in equations (5.41) and (5.40) respectively, it becomes clear that the proof proceeds in an identical way as that of Theorem 4.3.1 for the single-input, single-output case, and therefore it is omitted.

The same comments made on page 60 in relation to the result in Theorem 4.3.1 also apply component-wise to this case. The theorem indicates that the approximation error that results from the use of the restricted complexity model structure (5.38) can be reduced by choosing the poles $\{\xi_k\}$ of the bases close to the true poles $\{\gamma_{\ell}^{ij}\}$ of the system, since this minimizes the product term in the upper bound in (5.42).

Theorem 5.6.1 can also be used to provide a quantification of the undermodelling induced error in the least squares estimate $G(q, \widehat{\Theta})$ by noting that under the given assumptions, by the results in [Lju87] (as summarized in Theorems 2.2.1 and 2.2.2)

$$\widehat{\Theta} \xrightarrow{\text{a.s.}} \Theta_{\star}$$
 as $N \to \infty$,

where

$$\Theta_{\star} = \underset{\Theta \in \mathbb{R}^{m \times np}}{\arg\min} \left\{ \lim_{N \to \infty} \frac{1}{N} \operatorname{Tr} \left\{ \sum_{k=0}^{N-1} \mathbf{E} \left\{ \varepsilon_{k}(\Theta) \varepsilon_{k}^{T}(\Theta) \right\} \right\} \right\}$$

with

$$\varepsilon_k(\Theta) \triangleq y_k - G(q,\Theta)u_k = (G(q) - G(q,\Theta))u_k + \nu_k.$$

Using Parseval's Theorem (in the form given in equation (2.3)), and neglecting terms independent of Θ then allows Θ_{\star} to be re-expressed as

$$\Theta_{\star} = \operatorname*{arg\,min}_{\Theta \in \mathbb{R}^{m \times np}} \left\{ \frac{1}{2\pi} \int_{-\pi}^{\pi} \operatorname{Tr} \left\{ \Phi_{\tilde{u}}(\omega, \Theta) \right\} \, \mathrm{d}\omega \right\},$$
(5.43)

where

$$\Phi_{\tilde{u}}(\omega,\Theta) = \left(G(e^{j\omega}) - G(e^{j\omega},\Theta)\right) \Phi_{u}(\omega) \left(G(e^{j\omega}) - G(e^{j\omega},\Theta)\right)^{\star}$$

In the special case of white input excitation where $\Phi_u = I_n$, this gives the asymptotic estimate as

$$\Theta_{\star} = \underset{\Theta \in \mathbb{R}^{m \times np}}{\operatorname{arg\,min}} \left\{ \frac{1}{2\pi} \sum_{k=1}^{m} \sum_{\ell=1}^{n} \int_{-\pi}^{\pi} |G_{k\ell}(e^{j\omega}) - G_{k\ell}(e^{j\omega}, \Theta)|^2 \, \mathrm{d}\omega \right\}.$$

This criterion is satisfied by the solution

$$\Theta_{\star} = \left[\theta_0^T, \cdots, \theta_{p-1}^T\right],$$

with

$$\theta_k^T = \begin{bmatrix} \langle G, \mathcal{B}_k^{11} \rangle & \cdots & \langle G, \mathcal{B}_k^{1n} \rangle \\ \vdots & \ddots & \vdots \\ \langle G, \mathcal{B}_k^{m1} \rangle & \cdots & \langle G, \mathcal{B}_k^{mn} \rangle \end{bmatrix},$$

so that the result of Theorem 5.6.1 implies

$$\left|G_{ij}(e^{j\omega}) - G_{ij}(e^{j\omega},\Theta_{\star})\right| < \sum_{\ell=0}^{r^{ij}-1} \left|\frac{\alpha_{\ell}^{ij}}{e^{j\omega} - \gamma_{\ell}^{ij}}\right| \prod_{k=0}^{p-1} \left|\frac{\gamma_{\ell}^{ij} - \xi_k}{1 - \overline{\xi_k}\gamma_{\ell}^{ij}}\right|,\tag{5.44}$$

which provides quantification (which applies with probability one) of the asymptotic undermodelling induced error in the case of white input excitation.

The case of non-white input is more difficult. Nevertheless, for such cases it is still possible to derive frequency dependent upper bounds on the undermodelling error similar to that possible for the white input case (5.44), but that are

not as tight or as explicit. To present them, it is necessary to recall that since $\{\mathcal{B}_k^{ij}\}$ is complete in $H_2^{m \times n}(\mathbb{T})$ then any $G(e^{j\omega}) \in H_2^{m \times n}(\mathbb{T})$ may be expanded as

$$G(e^{j\omega},\Theta) = \sum_{\ell=0}^{p-1} \sum_{i=1}^{m} \sum_{j=1}^{n} \theta_{\ell}^{ij} \mathcal{B}_{\ell}^{ij}(e^{j\omega}) + \sum_{\ell=p}^{\infty} \sum_{i=1}^{m} \sum_{j=1}^{n} \theta_{\ell}^{ij} \mathcal{B}_{\ell}^{ij}(e^{j\omega})$$
(5.45)

where $\theta_{\ell}^{ij} \triangleq \langle G, \mathcal{B}_{\ell}^{ij} \rangle$ are the generalized Fourier coefficients. Appealing to Lemma 5.3.1, equation (5.45) can be written as

$$G(e^{j\omega}) = \sum_{\ell=0}^{p-1} \theta_{\ell} \mathcal{B}_{\ell}(e^{j\omega}) + \sum_{\ell=p}^{\infty} \theta_{\ell} \mathcal{B}_{\ell}(e^{j\omega}).$$
(5.46)

where $\theta_{\ell} \in \mathbb{R}^{m \times n}$ are defined in equation (5.17). When G(z) is finite dimensional, a simple argument using Cauchy's Residue Theorem shows that these parameters decay exponentially as $|\theta_k^{ij}| = o(\eta^k)$ as $k \to \infty$ for some $|\eta| < 1$.

In terms of these decaying parameters, and a possibly coloured input spectrum $\Phi_u(\omega)$, a frequency dependent bound on the undermodelling induced estimation error may then be given as follows.

Lemma 5.6.1. Let K_1 be defined as

$$K_1 \triangleq \sqrt{\frac{1 + |\xi_{max}|}{1 - |\xi_{max}|}},$$
 (5.47)

where ξ_{max} denotes the pole with maximum module in the set $\{\xi_k\}$. Then, with probability one:

$$\left| [G(e^{j\omega}) - G(e^{j\omega}, \Theta_0)]_{i,j} \right| \leq \\ \leq K_1 \left(\sum_{\ell=p}^{\infty} \max_{i,j} |\theta_t^{ij}| \right) \left(\sup_{\omega} \gamma_p(\omega) \sqrt{mn} \frac{\sup_{\omega} \overline{\sigma}[\Phi_u(\omega)]}{\inf_{\omega} \underline{\sigma}[\Phi_u(\omega)]} + 1 \right).$$

Proof: See Appendix 5.B.

The chief use for this result is in the derivation of the main result of this chapter (Theorem 5.7.1) where it is used to show that as the model order grows, a particular component of the estimation variance expression decays to zero uniformly in frequency.

A less explicit characterization of the nature of the undermodelling induced estimation error is given by the following Lemma.

Lemma 5.6.2. The frequency response is on average (over frequency) under-estimated in the sense that

$$\int_{-\pi}^{\pi} \operatorname{Tr} \left\{ G(e^{j\omega}, \Theta_{\star}) \Phi_{u}(\omega) G^{\star}(e^{j\omega}, \Theta_{\star}) \right\} \, \mathrm{d}\omega \leq \int_{-\pi}^{\pi} \operatorname{Tr} \left\{ G(e^{j\omega}) \Phi_{u}(\omega) G^{\star}(e^{j\omega}) \right\} \, \mathrm{d}\omega.$$

Proof: The proof is analogous to the one corresponding to the single-input, single-output case given in [Nin96], and can be found in Appendix 5.B.

5.7 Noise Induced Error

In this section, the main result (Theorem 5.7.1) of the chapter, regarding the quantification of the component of the frequency response estimation error that is induced by the measurement noise, is derived. The result is the extension to general multivariable model structures with fixed poles of several asymptotic variance expressions that have been derived in the literature in the single-input, single-output context [LY85, Wah91b, Wah94b, VHB95, NHG97a], or in the multivariable framework but only for FIR model structures [YL84]. The result is built on earlier work that applies only to single-input, single-output systems [NHG97a], as summarized in Section 4.4.

Here again, as in the SISO case, a quantification of the noise induced error is given by the covariance matrix of the transfer matrix estimate $G(e^{j\omega}, \widehat{\Theta})$. However, since it is desiderable to be able to quantify the error in each individual component $G_{ij}(e^{j\omega}, \Theta)$ of the matrix $G(e^{j\omega}, \Theta)$ it is more convenient to use the methods in [YL84, Zhu89, Zhu90] and deal with a vectorized form of $G(e^{j\omega}, \Theta)$. This vectorized form is obtained by stacking the columns of $G(e^{j\omega}, \Theta)$ on top of each other, that is by applying the vec-operator (see Appendix A for the definition and some properties) to the matrix $G(e^{j\omega}, \Theta)$ to obtain

$$g(e^{j\omega}, \Theta) \triangleq \operatorname{vec} G(e^{j\omega}, \Theta).$$

Analyzing the distribution of the estimate $g(e^{j\omega}, \widehat{\Theta})$ then provides information about the noise induced error in each component of $G(e^{j\omega}, \Theta)$.

We start the analysis by re-defining the parameter matrix in such a way that the vectorized transfer matrix is still a linear function of the parameters, so that the linear regressor form of the model (5.18) is preserved and therefore a closed form solution for the least squares estimation can also be obtained for this case. Considering the linear regressor model in equation (5.18), and using properties of the Kronecker products and vec-operator (see Appendix A) we can write

$$y_{k} = \Theta^{T} \phi_{k} + \nu_{k},$$

$$= \operatorname{vec} \{\Theta^{T} \phi_{k}\} + \nu_{k},$$

$$= (\phi_{k}^{T} \otimes I_{m}) \operatorname{vec} \Theta^{T} + \nu_{k},$$

$$= \psi_{k}^{T} \eta + \nu_{k},$$
(5.48)

where

$$\eta \triangleq \operatorname{vec} \Theta^T \tag{5.49}$$

is the new parameter vector, and

$$\psi_k \triangleq \phi_k \otimes I_m \tag{5.50}$$
is the new regressor matrix. It is clear that the new parameter vector η linearly parameterizes the vectorized transfer matrix $g(e^{j\omega}, \Theta)$, since

$$g(e^{j\omega}, \Theta) = \operatorname{vec} \{\Theta^T \Gamma_p(e^{j\omega})\}, = (\Gamma_p^T(e^{j\omega}) \otimes I_m) \operatorname{vec} \Theta^T, = \widetilde{\Gamma}_p^T(e^{j\omega}) \eta,$$
(5.51)

where the definition

$$\widetilde{\Gamma}_p(e^{\mathbf{j}\omega}) \triangleq \Gamma_p(e^{\mathbf{j}\omega}) \otimes I_m$$

has been made.

It is not difficult to prove (see Lemma 5.C.1 in Appendix 5.C) that the least squares estimate $\hat{\eta}$ of η , given by the well known expression

$$\widehat{\eta} = \left(\frac{1}{N}\sum_{k=0}^{N-1}\psi_k\psi_k^T\right)^{-1} \left(\frac{1}{N}\sum_{k=0}^{N-1}\psi_ky_k\right),$$
(5.52)

and the least squares estimate $\widehat{\Theta}$ of Θ in equation (5.26), are related according to

$$\widehat{\eta} = \operatorname{vec} \widehat{\Theta}^T.$$

The vectorized transfer matrix estimate is then given by

$$g(e^{\mathsf{j}\omega},\widehat{\eta}) = \widetilde{\Gamma}_p^T(e^{\mathsf{j}\omega}) \ \widehat{\eta}.$$

We undertake now the derivation of the main result of the chapter regarding the asymptotic (in data-length and model order) distribution of the (vectorized) transfer matrix estimate. Fundamental in this analysis will be some results concerning the asymptotic properties of what we will call *Block Toeplitz-like matrices.* These results are presented in the following subsection.

5.7.1 Convergence of Block Toeplitz-like Matrices

In this subsection and the following ones we consider positive definite real matrices $M_p(F, W)$ of dimensions $nmp \times nmp$, defined by two real symmetric positive definite matrix valued functions $F(\omega)$ and $W(\omega)$ (of dimensions $n \times n$ and $m \times m$ respectively) as

$$M_p(F,W) \triangleq \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_p(\omega) F(\omega) \Gamma_p^{\star}(\omega) \otimes W(\omega) \,\mathrm{d}\omega.$$
 (5.53)

Here, for the sake of notational simplicity, we have used the notation $\Gamma_p(\omega)$ as a shorthand for $\Gamma_p(e^{j\omega})$ as defined in (5.21), viz

$$\Gamma_p(\omega) \equiv \Gamma_p(e^{j\omega}) \triangleq \left[I_n \mathcal{B}_0(e^{j\omega}), I_n \mathcal{B}_1(e^{j\omega}), \cdots, I_n \mathcal{B}_{p-1}(e^{j\omega}) \right]^T,$$

where $\{\mathcal{B}_k(z)\}$ are orthonormal basis functions in $H_2(\mathbb{T})$.

We refer to $M_p(F, W)$ in (5.53) as a *Block Toeplitz-like Matrix*, since for the case of $\{\mathcal{B}_k(z)\}$ being the standard FIR (or trigonometric) bases $\{\mathcal{B}_k(z) = z^{-k}\}$, $M_p(F, W)$ has effectively a block Toeplitz structure⁵ [GS58]. To see this, let us consider the explicit expression of $M_p(F, W)$ for this case. Using properties of the Kronecker product (see Appendix A) we can write

$$M_{p}(F,W) \triangleq \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_{p}(\omega) F(\omega) \Gamma_{p}^{\star}(\omega) \otimes W(\omega) d\omega,$$

$$= \frac{1}{2\pi} \int_{-\pi}^{\pi} \left(\mathsf{B}_{p}(e^{j\omega}) \otimes I_{n} \right) F(\omega) \left(\mathsf{B}_{p}^{\star}(e^{j\omega}) \otimes I_{n} \right) \otimes W(\omega) d\omega,$$

$$= \frac{1}{2\pi} \int_{-\pi}^{\pi} \mathsf{B}_{p}(e^{j\omega}) \mathsf{B}_{p}^{\star}(e^{j\omega}) \otimes (F(\omega) \otimes W(\omega)) d\omega.$$

That $M_p(F, W)$ has a block Toeplitz structure now follows from the fact that for the case of FIR basis, the matrix $(B_p(e^{j\omega})B_p^*(e^{j\omega}))$ has itself a Toeplitz structure, since it is given by

$$\mathsf{B}_{p}(e^{\mathsf{j}\omega})\mathsf{B}_{p}^{\star}(e^{\mathsf{j}\omega}) = \begin{bmatrix} 1 & e^{\mathsf{j}\omega} & e^{\mathsf{j}^{2}\omega} & \cdots & e^{\mathsf{j}(p-1)\omega} \\ e^{-\mathsf{j}\omega} & 1 & e^{\mathsf{j}\omega} & \cdots & e^{\mathsf{j}(p-2)\omega} \\ e^{-\mathsf{j}^{2}\omega} & e^{-\mathsf{j}\omega} & 1 & \cdots & e^{\mathsf{j}(p-3)\omega} \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ e^{-\mathsf{j}(p-1)\omega} & e^{-\mathsf{j}(p-2)\omega} & e^{-\mathsf{j}(p-3)\omega} & \cdots & 1 \end{bmatrix}$$

For arbitrary orthonormal bases $\{\mathcal{B}_k(z)\}\$ the matrix form (5.53) will in general not have a Toeplitz structure. However, some asymptotic properties of Toeplitz matrices [GS58, HN77, HW89] will also hold for the form (5.53), what justifies the name *Toeplitz-like* we are using. In particular, the following Lemma provides a result concerning the convergence of a quadratic form of the block Toeplitz-like matrix (5.53) as the model order p tends to infinity, for the case of $\{\mathcal{B}_k(z)\}\$ being the OBFP introduced in Section 3.5.

Lemma 5.7.1. Let $M_p(F, W)$ be the $(mnp \times mnp)$ block Toeplitz-like matrix defined in equation (5.53), where $F(\omega)$, and $W(\omega)$ (of dimensions $(n \times n)$ and $(m \times m)$ respectively) are assumed to be positive definite, Lipschitz continuous in $[-\pi, \pi]$, and with finite dimensional spectral factorizations.

⁵A matrix is said to be block Toeplitz if its i, j-th block entry depends only on the difference (i - j) of the indices.

It is also assumed that all the poles $\{\xi_k\}$ of the basis functions are chosen in the open unit disc \mathbb{D} . Then the following limit result holds

$$\lim_{p \to \infty} \frac{(\Gamma_p^{\star}(\mu) \otimes I_m) M_p(F, W)(\Gamma_p(\omega) \otimes I_m)}{\gamma_p^{1/2}(\omega) \gamma_p^{1/2}(\mu)} = \begin{cases} F(\omega) \otimes W(\omega) & ; \ \mu = \omega \\ 0 & ; \ \mu \neq \omega \end{cases}$$

where

$$\gamma_p(\omega) \triangleq K_p(\omega, \omega) = \sum_{k=0}^{p-1} \left| \mathcal{B}_k(e^{j\omega}) \right|^2.$$

Proof: See Appendix 5.D.

Lemma 5.7.2. Let $M_p(\cdot, \cdot)$ be the $(mnp \times mnp)$ block Toeplitz-like matrix defined in equation (5.53), and let $W(\omega) > 0, X(\omega) > 0, Z(\omega) > 0$ and $U(\omega) > 0$ be spectral densities (of dimensions $(n \times n), (m \times m), (n \times n)$ and $(m \times m)$, respectively) with finite dimensional spectral factorizations. Then there exists $|\eta| < 1$ and $K < \infty$ such that

$$\left| [M_p(W, X)M_p(Z, U)]_{smn+y, tmn+x} - [M_p(WZ, XU)]_{smn+y, tmn+x} \right| \le mnK^2(\eta^{p-s} + \eta^s)(\eta^{p-t} + \eta^t)$$

Proof: See Appendix 5.D.

5.7.2 Asymptotic Distribution of the (Vectorized) Tranfer Matrix Estimate

Based on the convergence results of the block Toeplitz-like matrices of the previous subsection, we are now able to analyze the asymptotic (in data-length and model order) distribution of the vectorized transfer matrix estimate. The result is given in the following theorem.

Theorem 5.7.1. Let $\Phi_u(\omega)$ and $\Phi_\nu(\omega)$ be the input and measurement noise spectral densities, respectively, and let $\Phi_u(\omega)$ have a finite dimensional spectral factorisation. If the poles $\{\xi_k\}$ of the bases (3.14) are all chosen within the open unit disc \mathbb{D} , then as $N \to \infty$ and $p \to \infty$

$$\left(\sqrt{N} \left[\begin{array}{cc} \gamma_p(\omega_1) & 0\\ 0 & \gamma_p(\omega_2) \end{array}\right]^{-1/2} \otimes I_{mn} \right) \left[\begin{array}{cc} g(e^{j\omega_1}, \widehat{\eta}) - g(e^{j\omega_1}, \eta_0)\\ g(e^{j\omega_2}, \widehat{\eta}) - g(e^{j\omega_2}, \eta_0) \end{array}\right] \xrightarrow{dist} \frac{dist}{2} \mathcal{N}\left(0, \Lambda(\omega_1, \omega_2)\right)$$

where

$$\Lambda(\omega_1,\omega_2) = \left[\begin{array}{cc} \Phi_u^{-1}(\omega_1) \otimes \Phi_\nu(\omega_1) & 0\\ 0 & \Phi_u^{-1}(\omega_2) \otimes \Phi_\nu(\omega_2) \end{array} \right]$$

if $\omega_1 \neq \omega_2$, and

$$\gamma_p(\omega) \triangleq K_p(\omega, \omega) = \sum_{k=0}^{p-1} |\mathcal{B}_k(e^{j\omega})|^2.$$

Proof: By appealing to the results in [Lju87] as summarized in Theorems 2.2.1 and 2.2.2, it is possible to conclude that under the given assumptions

$$\sqrt{N}\left(\widehat{\eta}-\eta_{\star}
ight) \xrightarrow{\text{dist}} \mathcal{N}\left(0,P_{p}
ight) \qquad \text{as} \quad N \to \infty,$$

where

$$P_{p} \triangleq R_{p}^{-1}Q_{p}R_{p}^{-1},$$

$$R_{p} \triangleq \mathbf{E}\left\{V_{N}''(\eta_{\star})\right\},$$

$$Q_{p} \triangleq \lim_{N \to \infty} N\mathbf{E}\left\{V_{N}'(\eta_{\star})\left(V_{N}'(\eta_{\star})\right)^{T}\right\},$$

$$V_{N}(\eta) \triangleq \frac{1}{N}\sum_{k=0}^{N-1} \operatorname{Tr}\left\{(y_{k} - (\phi_{k}^{T} \otimes I_{m})\eta)(y_{k} - (\phi_{k}^{T} \otimes I_{m})\eta)^{T}\right\}.$$

Therefore, since

$$g(e^{\mathbf{j}\omega},\widehat{\eta}) = \widetilde{\Gamma}_p^T(e^{\mathbf{j}\omega})\widehat{\eta},$$

where $\widetilde{\Gamma}_p(\omega) \triangleq \Gamma_p(\omega) \otimes I_m$, it is possible to conclude that, as $N \to \infty$,

$$\begin{pmatrix} \sqrt{N} \begin{bmatrix} \gamma_p(\omega_1) & 0 \\ 0 & \gamma_p(\omega_2) \end{bmatrix}^{-1/2} \otimes I_{mn} \end{pmatrix} \begin{bmatrix} g(e^{j\omega_1}, \widehat{\eta}) - g(e^{j\omega_1}, \eta_\star) \\ g(e^{j\omega_2}, \widehat{\eta}) - g(e^{j\omega_2}, \eta_\star) \end{bmatrix} \xrightarrow{\text{dist}} \\ \xrightarrow{\text{dist}} \mathcal{N} \left(0, \Lambda_p(\omega_1, \omega_2) \right),$$

where

$$\begin{split} \Lambda_p(\omega_1,\omega_2) &\triangleq \left(\left[\begin{array}{c} \gamma_p(\omega_1) & 0 \\ 0 & \gamma_p(\omega_2) \end{array} \right]^{-1/2} \otimes I_{mn} \right) \left[\begin{array}{c} \widetilde{\Gamma}_p^T(\omega_1) \\ \widetilde{\Gamma}_p^T(\omega_2) \end{array} \right] \times \\ &\times P_p \left[\begin{array}{c} \widetilde{\Gamma}_p^T(\omega_1) \\ \widetilde{\Gamma}_p^T(\omega_2) \end{array} \right]^{\star} \left(\left[\begin{array}{c} \gamma_p(\omega_1) & 0 \\ 0 & \gamma_p(\omega_2) \end{array} \right]^{-1/2} \otimes I_{mn} \right). \end{split}$$

After some matrix manipulation, $\Lambda_p(\omega_1,\omega_2)$ can be written as

$$\Lambda_p(\omega_1, \omega_2) = \begin{bmatrix} \frac{\widetilde{\Gamma}_p^{\star}(\omega_1) P_p \widetilde{\Gamma}_p(\omega_1)}{\gamma_p(\omega_1)} & \frac{\widetilde{\Gamma}_p^{\star}(\omega_1) P_p \widetilde{\Gamma}_p(\omega_2)}{\gamma_p^{1/2}(\omega_1) \gamma_p^{1/2}(\omega_2)} \\ \frac{\widetilde{\Gamma}_p^{\star}(\omega_2) P_p \widetilde{\Gamma}_p(\omega_1)}{\gamma_p^{1/2}(\omega_1) \gamma_p^{1/2}(\omega_2)} & \frac{\widetilde{\Gamma}_p^{\star}(\omega_2) P_p \widetilde{\Gamma}_p(\omega_2)}{\gamma_p(\omega_2)} \end{bmatrix}$$

,

so that it becomes clear that in order to analyze its asymptotic behavior when the model order p tends to infinity, we need only to study the asymptotic behavior of the term

$$\frac{\widetilde{\Gamma}_{p}^{\star}(\omega_{1})P_{p}\widetilde{\Gamma}_{p}(\omega_{2})}{\gamma_{p}^{1/2}(\omega_{1})\gamma_{p}^{1/2}(\omega_{2})}.$$
(5.54)

We first provide a frequency domain expression for the covariance matrix P_p . Considering the definition of the matrix R_p , simple algebra gives

$$R_p = \lim_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} \mathbf{E} \left\{ \phi_k \phi_k^T \otimes I_m \right\}.$$

Using now Parseval's Theorem, and the definition in (5.53), the following frequency domain expression for R_p is obtained

$$R_p = \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_p(e^{j\omega}) \Phi_u(\omega) \Gamma_p^{\star}(e^{j\omega}) \otimes I_m \,\mathrm{d}\omega = M_p(\Phi_u, I_m).$$
(5.55)

The derivation of a frequency domain expression for Q_p is more difficult. By Lemma 5.E.4 and the definition (5.53) Q_p may be expressed as

$$Q_p = \underbrace{\frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_p(\omega) \Phi_u(\omega) \Gamma_p^{\star}(\omega) \otimes \Phi_\nu(\omega) \, \mathrm{d}\omega}_{=M_p(\Phi_u, \Phi_\nu)} + \Delta_p = M_p(\Phi_u, \Phi_\nu) + \Delta_p,$$

where

$$\Delta_p \triangleq \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_p(\omega) \Phi_u(\omega) \Gamma_p^{\star}(\omega) \otimes \widetilde{G}_p(\omega) \Phi_u(\omega) \widetilde{G}_p^{\star}(\omega) \, \mathrm{d}\omega,$$
$$\widetilde{G}_p(\omega) \triangleq G(e^{j\omega}) - G(e^{j\omega}, \eta_{\star}).$$

Finally, matrix P_p can then be written as

$$P_{p} = M_{p}^{-1}(\Phi_{u}, I_{m}) \left(M_{p}(\Phi_{u}, \Phi_{\nu}) + \Delta_{p} \right) M_{p}^{-1}(\Phi_{u}, I_{m}),$$

$$= M_{p}^{-1}(\Phi_{u}, I_{m}) M_{p}(\Phi_{u}, \Phi_{\nu}) M_{p}^{-1}(\Phi_{u}, I_{m}) + M_{p}^{-1}(\Phi_{u}, I_{m}) \Delta_{p} M_{p}^{-1}(\Phi_{u}, I_{m}).$$

With this expression for P_p the term in (5.54) becomes

$$\frac{\widetilde{\Gamma}_{p}^{\star}(\omega_{2})P_{p}\widetilde{\Gamma}_{p}(\omega_{1})}{\gamma_{p}^{1/2}(\omega_{1})\gamma_{p}^{1/2}(\omega_{2})} = \frac{\widetilde{\Gamma}_{p}^{\star}(\omega_{1})M_{p}^{-1}(\Phi_{u},I_{m})M_{p}(\Phi_{u},\Phi_{\nu})M_{p}^{-1}(\Phi_{u},I_{m})\widetilde{\Gamma}_{p}(\omega_{2})}{\gamma_{p}^{1/2}(\omega_{1})\gamma_{p}^{1/2}(\omega_{2})} + \frac{\widetilde{\Gamma}_{p}^{\star}(\omega_{1})M_{p}^{-1}(\Phi_{u},I_{m})\Delta_{p}M_{p}^{-1}(\Phi_{u},I_{m})\widetilde{\Gamma}_{p}(\omega_{2})}{\gamma_{p}^{1/2}(\omega_{1})\gamma_{p}^{1/2}(\omega_{2})}.$$
(5.56)

We analyze first the asymptotic behavior of the second term on the RHS of equation (5.56). For the i, j-th component we have⁶

$$\left\| \left[\frac{\widetilde{\Gamma}_{p}^{\star}(\omega_{1})M_{p}^{-1}(\Phi_{u},I_{m})\Delta_{p}M_{p}^{-1}(\Phi_{u},I_{m})\widetilde{\Gamma}_{p}(\omega_{2})}{\gamma_{p}^{1/2}(\omega_{1})\gamma_{p}^{1/2}(\omega_{2})} \right]_{i,j} \right\| \leq \\ \leq \left\| \frac{\widetilde{\Gamma}_{p}^{\star}(\omega_{1})M_{p}^{-1}(\Phi_{u},I_{m})\Delta_{p}M_{p}^{-1}(\Phi_{u},I_{m})\widetilde{\Gamma}_{p}(\omega_{2})}{\gamma_{p}^{1/2}(\omega_{1})\gamma_{p}^{1/2}(\omega_{2})} \right\|_{2} \\ \leq \left\| M_{p}^{-1}(\Phi_{u},I_{m}) \right\|_{2}^{2} \left\| \Delta_{p} \right\|_{2}$$

where in passing to the second line use was made of the fact [GV89] that for a given matrix $A = (a_{i,j})$

$$\max_{i,j} |a_{i,j}| \le ||A||_2, \tag{5.57}$$

and in passing to the last line of the fact that, by Lemma 5.E.6,

$$\|\widetilde{\Gamma}_p(\omega)\|_2 = \sqrt{\gamma_p(\omega)}.$$

Considering now that by Lemma 5.E.8, $\|M_p^{-1}(\Phi_u, I_m)\|_2$ is bounded above, and that $\|\Delta_p\|_2 \to 0$ as $p \to \infty$ (see Lemma 5.E.9), we can conclude that

$$\lim_{p \to \infty} \frac{\widetilde{\Gamma}_{p}^{\star}(\omega_{1})M_{p}^{-1}(\Phi_{u}, I_{m})\Delta_{p}M_{p}^{-1}(\Phi_{u}, I_{m})\widetilde{\Gamma}_{p}(\omega_{2})}{\gamma_{p}^{1/2}(\omega_{1})\gamma_{p}^{1/2}(\omega_{2})} = 0.$$
 (5.58)

Let us analyze now the asymptotic behavior of the first term on the RHS of equation (5.56). We can write

$$\frac{\widetilde{\Gamma}_{p}^{\star}(\omega_{1})M_{p}^{-1}(\Phi_{u}, I_{m})M_{p}(\Phi_{u}, \Phi_{\nu})M_{p}^{-1}(\Phi_{u}, I_{m})\widetilde{\Gamma}_{p}(\omega_{2})}{\gamma_{p}^{1/2}(\omega_{1})\gamma_{p}^{1/2}(\omega_{2})} = \frac{\widetilde{\Gamma}_{p}^{\star}(\omega_{1})M_{p}(\Phi_{u}^{-1}, \Phi_{\nu})\widetilde{\Gamma}_{p}(\omega_{2})}{\gamma_{p}^{1/2}(\omega_{1})\gamma_{p}^{1/2}(\omega_{2})}$$

$$= \frac{\widetilde{\Gamma}_{n}^{\star}(\omega_{1})M_{n}^{-1}(\Phi_{u}, I_{m})\widetilde{\Gamma}_{p}(\omega_{2})}{\widetilde{\Gamma}_{n}^{\star}(\omega_{1})M_{n}^{-1}(\Phi_{u}, I_{m})\left[M_{n}(I_{n}, \Phi_{\nu}) - M_{n}(\Phi_{u}, I_{m})M_{n}(\Phi_{u}^{-1}, \Phi_{\nu})\right]\widetilde{\Gamma}_{n}(\omega_{2})$$
(5.59)

⁶Here the symbol $\|\cdot\|_2$ stands for the matrix induced 2-norm or spectral norm (i.e. the maximum singular value), not to be confussed with the H_2 -norm induced by the inner product denoted by $\|\cdot\|$ or $\|\cdot\|_{H_2}$.

We will prove that the last two terms on the RHS of the above equation (terms (5.60) and (5.61)) both tend to zero as p tends to infinity, and that the first term (5.59) tends, for $\omega_1 = \omega_2 = \omega$, to $\Phi_u^{-1}(\omega) \otimes \Phi_\nu(\omega)$ when p tends to infinity.

To proceed with this, let us consider first the term (5.60). By construction, the elements of the matrix $\tilde{\Gamma}_p(\omega)$ are bounded in magnitude by some finite number K_1 (as defined in equation (5.47)), i.e.

$$\left| \left[\widetilde{\Gamma}_p(\omega) \right]_{i,j} \right| \le K_1.$$

To see this, recall that the elements of this matrix are either equal to zero, or equal to $\mathcal{B}_k(e^{j\omega})$, and that by Lemma 5.E.2, $|\mathcal{B}_k(e^{j\omega})| \leq K_1$. Furthermore, by appealing to the relation (5.57), Lemma 5.E.6, and Lemma 5.E.8 we can write

$$\left\| \left[\frac{\widetilde{\Gamma}_{p}^{\star}(\omega_{1})M_{p}^{-1}(\Phi_{u}, I_{m})}{\gamma_{p}^{1/2}(\omega_{1})} \right]_{i,j} \right\| \leq \left\| \frac{\widetilde{\Gamma}_{p}^{\star}(\omega_{1})M_{p}^{-1}(\Phi_{u}, I_{m})}{\gamma_{p}^{1/2}(\omega_{1})} \right\|_{2}$$
$$\leq \left\| M_{p}^{-1}(\Phi_{u}, I_{m}) \right\|_{2}$$
$$\leq \left(\inf_{\omega} \underline{\sigma}(\Phi_{u}(\omega)) \right)^{-1} \triangleq K_{2}.$$

In this case, using Lemma 5.7.2 gives that for some $|\eta| < 1$

$$\begin{split} \left| \left[\frac{\widetilde{\Gamma}_{p}^{\star}(\omega_{1})M_{p}^{-1}(\Phi_{u},I_{m})}{\gamma_{p}^{1/2}(\omega_{1})} [M_{p}(I_{n},\Phi_{\nu}) - M_{p}(\Phi_{u},I_{m})M_{p}(\Phi_{u}^{-1},\Phi_{\nu})]\widetilde{\Gamma}_{p}(\omega_{2}) \right]_{i,j} \right| \leq \\ \leq & \sum_{s=0}^{p-1} \sum_{t=0}^{p-1} \sum_{x=1}^{mn} \sum_{y=1}^{mn} \left| \left[\frac{\widetilde{\Gamma}_{p}^{\star}(\omega_{1})M_{p}(\Phi_{u},I_{m})^{-1}}{\gamma_{p}^{1/2}(\omega_{1})} \right]_{i,tmn+x} \right| \times \\ & \times \left| [M_{p}(I_{n},\Phi_{\nu})]_{smn+y,tmn+x} - \left[M_{p}(\Phi_{u},I_{m})M_{p}(\Phi_{u}^{-1},\Phi_{\nu}) \right]_{smn+y,tmn+x} \right| \times \\ & \times \left| \left[\widetilde{\Gamma}_{p}(\omega_{2}) \right]_{smn+y,j} \right| \\ \leq & K^{2}K_{1}K_{2} \sum_{x=1}^{mn} \sum_{y=1}^{mn} \sum_{s=0}^{p-1} \sum_{t=0}^{p-1} mn(\eta^{p-s}+\eta^{s})(\eta^{p-t}+\eta^{t}) \\ = & K^{2}K_{1}K_{2}(mn)^{3} \left(\frac{1+\eta}{1-\eta} \right)^{2} (1-\eta^{p})^{2} < \infty. \end{split}$$

But by Lemma 5.E.3 we have

$$\gamma_p(\omega) \ge \frac{1}{2} \sum_{k=0}^{p-1} (1 - |\xi_k|)$$

so that under the conditions of the theorem we can conclude that for any choice of ω_1 and ω_2

$$\lim_{p \to \infty} \frac{\widetilde{\Gamma}_p^{\star}(\omega_1) M_p^{-1}(\Phi_u, I_m) [M_p(I_n, \Phi_\nu) - M_p(\Phi_u, I_m) M_p(\Phi_u^{-1}, \Phi_\nu)] \widetilde{\Gamma}_p(\omega_2)}{\gamma_p^{1/2}(\omega_1) \gamma_p^{1/2}(\omega_2)} = 0$$

Using an identical argument for the term (5.61) we can conclude that

$$\lim_{p \to \infty} \frac{\Gamma_p^{\star}(\omega_1) M_p^{-1}(\Phi_u, I_m) \left[M_p(\Phi_u, \Phi_\nu) - M_p(I_n, \Phi_\nu) M_p(\Phi_u, I_m) \right] M_p^{-1}(\Phi_u, I_m) \Gamma_p(\omega_2)}{\gamma_p^{1/2}(\omega_1) \gamma_p^{1/2}(\omega_2)} = 0$$

Finally, by applying Lemma 5.7.1 to the term (5.59) we have

$$\lim_{p \to \infty} \frac{\widetilde{\Gamma}_p^{\star}(\omega_1) M_p(\Phi_u^{-1}, \Phi_\nu) \widetilde{\Gamma}_p(\omega_2)}{\gamma_p^{1/2}(\omega_1) \gamma_p^{1/2}(\omega_2)} = \begin{cases} \Phi_u^{-1}(\omega_1) \otimes \Phi_\nu(\omega_1) & ; & \omega_1 = \omega_2 \\ 0 & ; & \omega_1 \neq \omega_2 \end{cases}$$

what completes the proof.

As a direct consequence of the previous theorem we have the following corollary.

Corollary 5.7.2. Under the same conditions of the previous theorem, but with the strengthened requirement on the sequence $\{e_k\}$ of having bounded eighth moments, then

$$\lim_{p \to \infty} \lim_{N \to \infty} \frac{N}{\gamma_p(\omega)} \mathsf{Cov}\left\{g(e^{j\omega}, \widehat{\eta})\right\} = \Phi_u^{-1}(\omega) \otimes \Phi_\nu(\omega).$$

Proof: The result follows by appealing to the results in Chapter 9 of [Lju87] (as summarized in Theorems 2.2.1 and 2.2.2) and using the same arguments as in the proof of Theorem 5.7.1.

Theorem 5.7.1 and its corollary provide a generalization to the multivariable and arbitrary fixed pole setting of the asymptotic results available for singleinput, single-output systems [LY85, Wah91b, Wah94b, VHB95, NHG96], and for multivariable systems with FIR model structures [YL84].

Apart from this generalizing aspect, the previous theorem and corollary also have significant utility in providing an approximate expression for the covariance matrix of the (vectorized) transfer matrix estimate. For large model order and data-length, this covariance matrix can be approximated by

$$\operatorname{Cov}\left\{g(e^{j\omega},\widehat{\eta})\right\} \approx \frac{\gamma_p(\omega)}{N} \, \Phi_u^{-1}(\omega) \otimes \Phi_\nu(\omega). \tag{5.62}$$

This approximate expression implies the following component-wise quantification of the noise induced error in the frequency response estimate

$$\mathbf{E}\left\{|G_{i\ell}(e^{j\omega}) - G_{i\ell}(e^{j\omega},\eta_{\star})|^{2}\right\} \approx \frac{\gamma_{p}(\omega)}{N} \left[\Phi_{u}^{-1}(\omega)\right]_{\ell\ell} \left[\Phi_{\nu}(\omega)\right]_{ii}.$$

The generalizing aspect above mentioned can be demonstrated explicitly by noting that for the particular case of FIR model structures the factor $\gamma_p(\omega)$ is given by

$$\gamma_p(\omega) \triangleq \sum_{k=0}^{p-1} |\mathcal{B}_k(e^{j\omega})|^2 = \sum_{k=0}^{p-1} |e^{j\omega k}|^2 = p,$$

so that the covariance expression reduces to

$$\mathsf{Cov}\left\{g(e^{j\omega},\widehat{\eta})\right\} \approx \frac{p}{N} \ \Phi_u^{-1}(\omega) \otimes \Phi_\nu(\omega),$$

which is the same expression as that derived in [YL84] where only the FIR model structure case is studied.

A significant aspect of the approximate covariance expression (5.62) is that it explicitly shows how the choice of the fixed pole locations affects the noise induced estimation error. Specifically, equation (5.62) expresses the already known principle that the noise induce error is proportional to the 'noise-tosignal' ratio and the model order, and inversely proportional to the number of observed data. In addition, it also shows a new phenomenom, viz, that the choice of the poles affects the noise induced error via the frequency dependent factor $\gamma_p(\omega)$ (See figure 4.2, where the factor $\gamma_p(\omega)$ is plotted for model order p = 4 and for various pole choices. It can be noted from this diagram that for the case of all the poles at the origin, which corresponds to an FIR model structure, then $\gamma_p(\omega) = p = 4$.).

5.8 Bias/Variance Trade-off

The results in Theorems 5.6.1 and 5.7.1 provide a complete characterization of the estimation accuracy and show, as in the SISO case, two different phenomena of bias/variance trade-off: the well known trade-off with respect to model order selection, and an until now unrecognized phenomenom of trade-off with respect to the choice of the poles of the basis functions. The comments we made in Section 4.5 apply also here component-wise.

5.9 Intermezzo: Brief Overview of Subspace-Based Identification Methods

For the purposes of comparison with the multivariable identification technique we propose in this chapter, a brief overview of a class of methods that also provide MIMO estimates in closed form is presented in this section. The methods are known as Subspace Identification Methods.

5.9.1 Introduction

State-space models of the type introduced in Subsubsection 2.1.1.3 are very convenient to represent multivariable systems for the purposes of control design and simulation. This is so since many control problems can be solved more elegantly in the state-space domain, while the solution to these problems is sometimes more involved, or not so elegant with other representations (such as transfer functions or matrix fraction descriptions (MFD) [Kai80]). State-space models have been successfully used to provide accurate descriptions of many industrial processes. In addition, or rather as a consequence of the above comments, most of the available CACSD⁷ software packages (e.g., the *Control System Toolbox* [GLLT92] for use with MATLAB ⁸) can handle this type of models, although most of them also allow some other system representations such as transfer function models.

Even though traditional PEM can be used for the estimation of state-space models, typically the parameters in this model structure appear in a nonlinear fashion so that the minimum of the cost function cannot be computed analytically. Instead, the estimation will usually involve a computationally intensive (possibly nonlinear and non convex) iterative optimization procedure, with the inherent problems associated with these techniques, such as existence of local minima of the cost function, initialization problems and no guaranteed convergence. Another difficulty of the use of prediction error methods for the estimation of state space models is the large number of parameters that need to be estimated (for example consider the case of a fully parameterized state space model [McK94, McK95]). The number of parameters could be reduced by resorting to a canonical parameterization (see [Kai80]); however, as pointed out in [Vib94, Vd94a], the problem of finding a numerically reliable canonical parameterization for multivariable systems is not trivial and is, to a large extent, unsolved.

In recent years considerable amount of research has been devoted to the development of new identification methods that are able to deliver reliable statespace models of multivariable dynamical systems directly from input-output data, and that require only a modest computational complexity without the need of iterative optimization procedures. These techniques have become collectively known as Subspace-based State-Space System IDentification (4SID) methods (see for instance the recent book [Vd96] for a unified description of the different subspace algorithms, and the survey paper [Vib95]), and have their origin in state-space realization theory as developed in the sixties [HK66, Kun78]. The main computational tools employed by subspace methods are QR and singular-value decompositions [GV89].

All 4SID methods involve at some stage the computation of the extended

⁷CACSD stands for Computer Aided Control System Design.

⁸MATLAB is a registered trademark of The MathWorks, Inc.

observability matrix from input-output data. In [Vib94, Vib95], subspace methods are grouped into the following two sub-classes:

- Realization-based 4SID Methods: in which an explicit estimate of the impulse response coefficients is obtained as a previous step of the estimation of the extended observability matrix [HK66, ZM74, Kun78, LS91, Liu92].
- Direct 4SID Methods: in which the estimate of the extended observability matrix in obtained directly from the input-output data [DVM⁺88, Ver91, VD91b, VD92, VOWL93, MR93, Ver94, OV94, Vd91a, Vd93, Vd94a, Vd94b, Vd95a, Vd95b, CV97, Lar90, Lar94].

The various algorithms, with particular names attached to them, are then classified according to the way the observability matrix is estimated from inputoutput data, and how it is used to compute the system matrices.

In the following Subsection the fundamental concepts of *realization theory* will be reviewed and the basic realization-based 4SID methods will be presented.

5.9.2 Realization-based 4SID Methods

We consider the state-space description introduced in Subsubsection 2.1.1.3 (equations (2.8) and (2.9)), which is repeated here for convenience

$$x_{k+1} = Ax_k + Bu_k, (5.63)$$

$$y_k = Cx_k + Du_k, (5.64)$$

where $x_k \in \mathbb{R}^{n_x}$, $y_k \in \mathbb{R}^m$, $u_k \in \mathbb{R}^n$ are respectively the state, the output and the input vectors at time k, and $A \in \mathbb{R}^{n_x \times n_x}$, $B \in \mathbb{R}^{n_x \times n}$, $C \in \mathbb{R}^{m \times n_x}$, and $D \in \mathbb{R}^{m \times n}$ are the system matrices. It is assumed that the pair (A, B) is controllable and the pair (C, A) is observable (in the sense given by Definitions 2.1.4 and 2.1.5, respectively), which implies that the realization in minimal with McMillan degree n_x .

As already mentioned in Chapter 2, the realization (A, B, C, D) uniquely defines the input-output properties of the system via

$$y_k = \sum_{\ell=0}^{\infty} g(\ell) \, u_{k-\ell},$$

where

$$g(\ell) = \begin{cases} D, & \ell = 0\\ CA^{\ell-1}B, & \ell > 0 \end{cases}$$

are the impulse response coefficients.

Classical realization theory [HK66] deals with the estimation of the system matrices (A, B, C, D) from the impulse response coefficients $\{g(\ell)\}$. Most of the realization algorithms are based on the fact that the impulse response block Hankel matrix \mathcal{H}_{ij} (constructed from the impulse response coefficients as in equation (2.12)), can be factorized as

$$\mathcal{H}_{ij} = \mathcal{O}_i \mathcal{C}_j, \tag{5.65}$$

where \mathcal{O}_i and \mathcal{C}_j are respectively the extended observability and controllability matrices, and that these matrices always have rank n_x . Then, if any full rank factorization of \mathcal{H}_{ij} of the form

$$\mathcal{H}_{ij} = \widehat{\mathcal{O}}_i \widehat{\mathcal{C}}_j, \tag{5.66}$$

is available, the matrices $\widehat{\mathcal{O}}_i$ and $\widehat{\mathcal{C}}_j$ in this factorization can be interpreted as the extended observability and controllability matrices for some realization $(\widehat{A}, \widehat{B}, \widehat{C}, \widehat{D})$ of the system (which will be equivalent to the realization (A, B, C, D) in (5.63)-(5.64), modulo a similarity transformation). The fact that \mathcal{H}_{ij} has rank n_x can be used to infer the unknown system order. The problem then is the computation of the system matrices from the factorization (5.66) of the Hankel matrix. Considering the definitions of the extended observability and controllability matrices in equations (2.10) and (2.11) respectively, it is clear that the matrices \widehat{C} and \widehat{B} can be recovered by taking the first block row of $\widehat{\mathcal{O}}_i$, and the first block column of $\widehat{\mathcal{C}}_j$, respectively. Furthermore, the matrix \widehat{D} is readily available from the impulse response coefficient g(0).

Regarding the matrix \widehat{A} , there are several ways to compute it from the factorization (5.66) of \mathcal{H}_{ij} . Before presenting them we introduce some notation. For a given block matrix $\mathcal{H}, \overline{\mathcal{H}}$ will denote the matrix obtained by deleting the first block row of \mathcal{H} , while $\underline{\mathcal{H}}$ will denote the matrix obtained by deleting the last block row of \mathcal{H} .

One of the methods for computing \widehat{A} , proposed by Zeiger and McEwen in [ZM74], follows from the observation that

$$\overline{\overline{\mathcal{H}_{ij}}} = \widehat{\mathcal{O}}_i \widehat{A} \, \widehat{\mathcal{C}}_j,$$

so that \widehat{A} can be computed as

$$\widehat{A} = \widehat{\mathcal{O}}_i^{\dagger} \overline{\overline{\mathcal{H}_{ij}}} \widehat{\mathcal{C}}_j^{\ddagger},$$

where $(\cdot)^{\ddagger}$ denotes the right pseudo-inverse⁹.

⁹The right pseudo-inverse of the (non square) matrix A is defined as $A^{\ddagger} \triangleq A^T (AA^T)^{-1}$. Not to be confused with the left (Moore-Penrose) pseudo-inverse defined as $A^{\dagger} \triangleq (A^T A)^{-1} A^T$.

Another possibility, proposed by Kung in [Kun78] (see also [JP85]), is to note that the observability matrix \widehat{O}_i enjoys the so-called *shift-invariant* property, reflected by the identity

$$\overline{\widehat{\mathcal{O}}_i} = \underline{\widehat{\mathcal{O}}_i} \widehat{A}$$

The matrix A can then be computed by solving (in the least squares sense) this equation to obtain

$$\widehat{A} = \underline{\widehat{\mathcal{O}}_i}^{\dagger} \ \overline{\widehat{\mathcal{O}}_i}.$$

A dual result can be obtained by exploiting the shift invariant property of the controllability matrix \widehat{C}_i .

This completes the basic deterministic realization-based algorithm. In practice, however, it is more desiderable to start the algorithm directly from inputoutput measurements. This raises the question of how the impulse response coefficients can be estimated from input-output data. Several methods have been proposed in the literature for achieving this. For instance, in [LS92] a method is proposed where the response of the system to impulse input signals is directly measured. Another possibility is to use correlation analysis [Lju87, SS89] to estimate the input-output cross-covariance function from which a finite number of impulse response coefficients can be estimated.

A fundamental aspect of the realization algorithms not considered yet is the way the factorization (5.66) of the (estimated) impulse response Hankel matrix is performed, as well as the rank determination needed to estimate the system order. This issue is important for the case in which the impulse response is corrupted by noise (or it has been estimated from noisy data) because the rank determination of a matrix is very sensitive to perturbations on its entries [GV89]. Both the Zeiger-McEwen [ZM74] and the Kung [Kun78] algorithms employ the singular-value decomposition (SVD) [GV89] to factorize the impulse response Hankel matrix.

The SVD of \mathcal{H}_{ij} is defined as:

$$\mathcal{H}_{ij} = \begin{bmatrix} U_1 U_2 \end{bmatrix} \begin{bmatrix} \Sigma_1 & 0\\ 0 & \Sigma_2 \end{bmatrix} \begin{bmatrix} V_1^T\\ V_2^T \end{bmatrix}, \qquad (5.67)$$

where $U = [U_1U_2]$ and $V = [V_1V_2]$ are orthogonal matrices whose columns are respectively the *left* and *right singular vectors* of \mathcal{H}_{ij} , and where

$$\Sigma = \begin{bmatrix} \Sigma_1 & 0\\ 0 & \Sigma_2 \end{bmatrix}$$

is a diagonal matrix containing the singular values of \mathcal{H}_{ij} in non-increasing order, with Σ_1 containing the n_x largest, and Σ_2 the $(\min(i, j) - n_x)$ smallest.

For the case in which the impulse response is not corrupted by noise, \mathcal{H}_{ij} has rank n_x and then $\Sigma_2 = 0$. In this case equation (5.67) can be written as

$$\mathcal{H}_{ij} = U_1 \Sigma_1 V_1^T = \left(U_1 \Sigma_1^{1/2} \right) \left(\Sigma_1^{1/2} V_1^T \right),$$

which provides the factorization (5.66) by choosing

$$\widehat{\mathcal{O}}_i = U_1 \Sigma_1^{1/2}, \widehat{\mathcal{C}}_j = \Sigma_1^{1/2} V_1^T.$$

When the available measurements of the impulse response are noisy, the matrix \mathcal{H}_{ij} will in general have full rank and therefore a rank reduction step will be necessary. This rank reduction is provided by the SVD, but the user must decide which is the number of significant singular values or equivalently which is the system order. Once the system order is given, a low-rank approximation of \mathcal{H}_{ij} can be computed as

$$\widehat{\mathcal{H}}_{ij} = \left(U_1 \Sigma_1^{1/2} \right) \left(\Sigma_1^{1/2} V_1^T \right),\,$$

but now $\widehat{\mathcal{H}}_{ij}$ will in general not be a Hankel matrix, neither the estimates of $\widehat{\mathcal{O}}_i$ and $\widehat{\mathcal{C}}_j$ will have the shift invariant property. As a consequence, the algorithms will provide only approximate estimates of the system matrices. Some results concerning the quantification of the error induced by the noise corruption of the impulse response data are given in [Kun78] and [DD87].

5.9.3 Direct 4SID Algorithms

In the direct 4SID algorithms, the impulse response Hankel matrix is not required for the estimation of the extended observability matrix that can be computed directly from the input-output data. This represents an advantange with respect to the realization-based algorithms, since in practice the impulse response coefficients are difficult to measure.

We describe the basic direct 4SID algorithms corresponding to two different cases, namely:

- Deterministic Case, where the unknown system is assumed to have only deterministic inputs (u_k) .
- Combined Deterministic-Stochastic Case, where the unknown system is assumed to have both deterministic (u_k) and stochastic $(w_k$ and $v_k)$ inputs.

5.9.3.1 Deterministic Case

We consider that the unknown system has the state-space representation given in (5.63)-(5.64). The identification objective is to estimate the model order n_x , and the system matrices A, B, C and D in (5.63)-(5.64) from N samples of the inputs u_k and the outputs y_k . Before presenting the basic deterministic 4SID algorithm we introduce some notation adapted from [Vd96]. The input block Hankel matrix is defined as

$$U_{2i,j}^{0} \triangleq \frac{\begin{bmatrix} u_{0} & u_{1} & \cdots & u_{j-1} \\ u_{1} & u_{2} & \cdots & u_{j} \\ \vdots & \vdots & \cdots & \vdots \\ u_{i-1} & u_{i} & \cdots & u_{i+j-2} \\ \hline u_{i} & u_{i+1} & \cdots & u_{i+j-1} \\ u_{i+1} & u_{i+2} & \cdots & u_{i+j} \\ \vdots & \vdots & \cdots & \vdots \\ u_{2i-1} & u_{2i} & \cdots & u_{2i+j-2} \end{bmatrix}} \triangleq \frac{\begin{bmatrix} U_{0} \\ U_{i,j} \end{bmatrix}}{\begin{bmatrix} U_{j} \\ U_{i,j} \end{bmatrix}} \triangleq \begin{bmatrix} U_{p} \\ U_{f} \end{bmatrix}$$
(5.68)

where:

- The superscript in the matrices $U_{2i,j}^0, U_{i,j}^0$ and $U_{i,j}^i$ denotes the subscript of the first element of the first row, the first subscript denotes the number of block rows, and the second subscript, the number of (block) columns of the block Hankel matrix.
- The input Hankel matrix $U_{2i,j}^0$ is partitioned off into two parts of *i* block rows each. Somewhat arbitrarily, the upper part $U_p \triangleq U_{i,j}^0$ will be called the *past inputs* (with the subscript 'p' standing for 'past'), while the bottom part $U_f \triangleq U_{i,j}^i$ will be called the *future inputs* (with the subscript 'f' standing for 'future').
- The number of block rows *i* has to be larger than the expected order of the system to be identified.
- Typically, the number of columns j is such that all the available data is used. That is, if N is the total number of data, then j = N 2i + 2.

The output block Hankel matrices $Y_{2i,j}^0$, Y_p and Y_f are defined in similar way. The state sequence matrix is defined as:

$$X_j^0 \triangleq [x_0, x_1, \cdots, x_{j-1}],$$
 (5.69)

where the superscript denotes the subscript of the first element of the state sequence, and the subscript denotes the number of columns. To be consistent with the definitions of the past and future inputs and outputs we define the past state sequence as

$$X_p \triangleq X_i^0,$$

and the future state sequence as

$$X_f \triangleq X_j^i.$$

The block lower triangular Toeplitz matrix H_i is defined as

$$H_{i} \triangleq \begin{bmatrix} D & 0 & 0 & \cdots & 0 \\ CB & D & 0 & \cdots & 0 \\ CAB & CB & D & \cdots & 0 \\ \vdots & \vdots & \vdots & \ddots & 0 \\ CA^{i-2}B & CA^{i-3}B & CA^{i-4}B & \cdots & D \end{bmatrix},$$

where the subscript indicates the number of block rows of the Toeplitz matrix.

With these definitions, the following matrix input-output relations can be derived from the state-space representation (5.63)-(5.64)

$$Y_{2i,j}^{0} = \mathcal{O}_{2i}X_{j}^{0} + H_{2i}U_{2i,j}^{0}, \qquad (5.70)$$

$$Y_p = \mathcal{O}_i X_p + H_i U_p, \tag{5.71}$$

$$Y_f = \mathcal{O}_i X_f + H_i U_f. \tag{5.72}$$

Furthermore, the following 'state equation' relating the past and future state sequences can also be derived from (5.63)-(5.64)

$$X_f = A^i X_p + \mathcal{C}_i^{rd} U_p, \tag{5.73}$$

where C_i^{rd} is the reversed extended (deterministic) controllability matrix defined as

$$\mathcal{C}_i^{rd} \triangleq \left[A^{i-1}B, A^{i-2}B, \cdots, AB, B \right],$$

with the subscript i denoting the number of block columns, the superscript r standing for 'reversed' and the superscript d standing for 'deterministic'.

From equations (5.73) and (5.71), the future state sequence matrix X_f can be written as a linear combination of the past input U_p and past output Y_p Hankel matrices. To see this, we compute X_p from (5.71) and substitute into (5.73), to obtain

$$X_{f} = A^{i}\mathcal{O}_{i}^{\dagger}(Y_{p} - H_{i}U_{p}) + \mathcal{C}_{i}^{rd}U_{p},$$

$$= \left(\mathcal{C}_{i}^{rd} - A^{i}\mathcal{O}_{i}^{\dagger}H_{i}\right)U_{p} + A^{i}\mathcal{O}_{i}^{\dagger}Y_{p},$$

$$= \left[\left(\mathcal{C}_{i}^{rd} - A^{i}\mathcal{O}_{i}^{\dagger}H_{i}\right) \quad A^{i}\mathcal{O}_{i}^{\dagger}\right]\begin{bmatrix}U_{p}\\Y_{p}\end{bmatrix},$$

$$= L_{p \to f}W_{p},$$
(5.74)

where in passing to the last line we have defined the block Hankel matrix of past inputs and outputs $W_p \triangleq W_{i,j}^0 \triangleq \begin{bmatrix} U_p^T & Y_p^T \end{bmatrix}^T$ (with the subscripts and superscript in $W_{i,j}^0$ having the same meaning as in $U_{i,j}^0$), and the matrix $L_{p\to f}$ relating the past inputs and outputs with the future states

$$L_{p \to f} \triangleq \left[\begin{pmatrix} \mathcal{C}_i^{rd} - A^i \mathcal{O}_i^{\dagger} H_i \end{pmatrix} \quad A^i \mathcal{O}_i^{\dagger} \right].$$

From these equations it is possible to compute the extended observability matrix \mathcal{O}_i and the future state sequence matrix X_f based only on the inputoutput data. The main tool employed is the singular value decomposition (SVD). The basic deterministic algorithm proceeds as follows.

Algorithm 5.9.1 (Deterministic Algorithm).

1. Perform the full SVD of the Hankel matrix of future inputs U_f

$$U_f = \Phi \begin{bmatrix} \Sigma & 0 \end{bmatrix} \begin{bmatrix} \Psi_1^T \\ \Psi_2^T \end{bmatrix} = \Phi \Sigma \Psi_1^T.$$

Now notice that since $\Psi = \begin{bmatrix} \Psi_1 & \Psi_2 \end{bmatrix}$ is an orthogonal matrix, then $\Psi^T \Psi = I$, so that

$$\Psi_1^T \Psi_2 = 0,$$

and then

$$U_f \Psi_2 = 0.$$

We can then define the orthogonal complement of U_f as

$$U_f^{\perp} \triangleq \Psi_2.$$

Multiplying the input-output equation (5.72) from the right by U_f^{\perp} then gives

$$Y_f U_f^{\perp} = \mathcal{O}_i X_f U_f^{\perp}. \tag{5.75}$$

Substituting the expression for X_f in (5.74), in the previous equation we obtain

$$Y_f U_f^{\perp} = \mathcal{O}_i L_{p \to f} W_p U_f^{\perp},$$

so that we can write

$$\mathcal{O}_i L_{p \to f} = Y_f U_f^{\perp} \left(W_p U_f^{\perp} \right)^{\ddagger}.$$

Finally, multiplying both sides of the above equation from the right by W_p yields

$$\mathcal{O}_i X_f = Y_f U_f^{\perp} \left(W_p U_f^{\perp} \right)^{\ddagger} W_p.$$
(5.76)

Now the matrices \mathcal{O}_i and X_f can be obtained by performing the SVD of the matrix on the right hand side of the equation above. Note that this matrix depends only on the available input-output data. It can be proved that the matrix on the right-hand side of equation (5.76) is equivalent to the oblique projection of the row space¹⁰ of the future outputs Y_f along the row space of the future inputs U_f onto the row space of the past inputs and outputs W_p , defined as

$$\mathcal{P}_{i} \triangleq \left(Y_{f} - Y_{f}U_{f}^{T}\left(U_{f}U_{f}^{T}\right)^{\dagger}U_{f}\right)\left(W_{p} - W_{p}U_{f}^{T}\left(U_{f}U_{f}^{T}\right)^{\dagger}U_{f}\right)^{\dagger}W_{p}.$$

2. Perform the SVD of the matrix $\mathcal{P}_i = Y_f U_f^{\perp} \left(W_p U_f^{\perp} \right)^{\ddagger} W_p$, to obtain

$$\mathcal{P}_i = \mathcal{O}_i X_f = \begin{bmatrix} U_1 U_2 \end{bmatrix} \begin{bmatrix} \Sigma_1 & 0\\ 0 & \Sigma_2 \end{bmatrix} \begin{bmatrix} V_1^T\\ V_2^T \end{bmatrix}.$$
(5.77)

In the noise free case, $\Sigma_2 = 0$, and then

$$\mathcal{P}_i = Y_f U_f^{\perp} \left(W_p U_f^{\perp} \right)^{\ddagger} W_p = \mathcal{O}_i X_f = U_1 \Sigma_1 V_1^T = \left(U_1 \Sigma_1^{1/2} \right) \left(\Sigma_1^{1/2} V_1^T \right).$$

- 3. The order n_x of the system (5.63)-(5.64) is given by the number of singular values different from zero in (5.77), i.e. by $n_x = \operatorname{rank}(\Sigma_1)$.
- 4. The extended observability matrix is given by

$$\mathcal{O}_i = \left(U_1 \Sigma_1^{1/2} \right).$$

5. The future state sequence is given by

$$X_f = \left(\Sigma_1^{1/2} V_1^T\right).$$

6. System Matrices: The system matrices can be computed basically in two different ways. One of the methods uses the extended observability matrix \mathcal{O}_i , and the other one uses the future state sequence X_f . We describe here the first method.

¹⁰The row (column) space of a matrix A is the space spanned by the rows (columns) of A.

• Matrix A can be computed by exploiting the shift-invariant property of \mathcal{O}_i , as in the realization algorithm by Kung [Kun78], as described in Subsection 5.9.2, i.e.

$$A = \underline{\mathcal{O}_i}^{\dagger} \ \overline{\mathcal{O}_i}$$

- Matrix C can be computed by taking the first block row of \mathcal{O}_i .
- Matrices B and D. With matrices A and C known, matrices B and D can be computed from the input-output equation (5.72) as follows. Compute a full row rank matrix O[⊥]_i such that O[⊥]_iO_i = 0. A natural candidate for O[⊥]_i is the matrix U^T₂ in (5.77). This is so since matrix U = [U₁ U₂] is an orthogonal matrix and then U^T₂U₁ = 0, which implies U^T₂O_i = 0.

Multiplying the input-output equation (5.72) from the left by \mathcal{O}_i^{\perp} , and from the right by U_f^{\dagger} , we obtain

$$\mathcal{O}_i^{\perp} Y_f U_f^{\dagger} = \mathcal{O}_i^{\perp} H_i, \qquad (5.78)$$

which is a set of linear equations in the matrices B and D (provided that matrices A and C are known). The system is typically overdetermined (for the case of noisy data) and it can be solved for B and D in the least squares sense (see [DVM⁺88], or [Vd96] for details).

This concludes the basic deterministic 4SID algorithm. At several points in the algorithm some matrices are assumed to have full rank or to be nonsingular. In order for this to be ensured, the following assumptions have to be made [Vd96]

- i. The input u_k is persistently exciting of order 2i, in the sense given by Definition 2.1.3.
- ii. The intersection of the row space of U_f and the row space of X_p is empty.

5.9.3.2 Combined Deterministic-Stochastic Case

We consider now that the unknown system has both stochastic (w_k and v_k) and deterministic (u_k) inputs. A state-space representation of the combined system was introduced in Subsubsection 2.1.1.3, in equations (2.5), (2.6) and (2.7). For the reader's convenience, we rewrite these equations here.

$$x_{k+1} = Ax_k + Bu_k + w_k, (5.79)$$

$$y_k = Cx_k + Du_k + v_k, (5.80)$$

$$\mathbf{E}\left\{ \left(\begin{array}{c} w_k \\ v_k \end{array}\right) \left(w_s^T v_s^T\right) \right\} = \left(\begin{array}{c} Q & S \\ S^T & R \end{array}\right) \delta_{ks}$$
(5.81)

The identification objective for this case is to estimate the system order n_x , the system matrices A, B, C and D, and the second order statistics of the process and measurement noise defined by the covariance matrices Q, S and R, from N samples of the inputs u_k and outputs y_k .

Before presenting the algorithm we need to introduce some notation. The system (5.79)-(5.80) can be regarded as the superposition of two subsystems: one reflecting the influence of the deterministic input (u_k) , which will be called the 'deterministic subsystem', and the other one reflecting the influence of the stochastic inputs (the noise sequences w_k and v_k), which will be called the 'stochastic subsystem'. The state variable x_k and the output y_k can accordingly be written as the superposition of a deterministic and a stochastic component, as follows

$$x_k = x_k^d + x_k^s, (5.82)$$

$$y_k = y_k^d + y_k^s, (5.83)$$

where the superscript d denotes 'deterministic' and the superscript s denotes 'stochastic'. The deterministic subsystem is then given by

$$\begin{aligned} x_{k+1}^d &= A x_k^d + B u_k, \\ y_k^d &= C x_k^d + D u_k, \end{aligned}$$

while the stochastic subsystem by

$$\begin{aligned} x_{k+1}^s &= A x_k^s + w_k, \\ y_k^s &= C x_k^s + v_k. \end{aligned}$$

It is assumed that the pair (A, C) is observable, and that the pair $(A, \begin{bmatrix} B & Q^{1/2} \end{bmatrix})$ is controllable, which implies that all the modes of the system are excited either by the deterministic input u_k or by the process noise w_k .

The processes w_k and v_k are assumed to be zero mean white noise vector sequences independent of x_k^s , which implies that

It is also assumed that the process $\{x_k^s\}$ is stationary, so that

$$\begin{array}{rcl} \mathbf{E} \left\{ x_k^s \right\} &=& 0, \\ \mathbf{E} \left\{ x_k^s (x_k^s)^T \right\} &\triangleq& \Sigma_x, \end{array}$$

where the stochastic state covariance matrix Σ_x is independent of the time k. It is not difficult to show that Σ_x satisfies the following Lyapunov equation [Vd94a, Vd96]

$$\Sigma_x = A \Sigma_x A^T + Q. \tag{5.84}$$

The stochastic output covariance matrices are defined as

$$\Lambda_i \triangleq \mathbf{E}\left\{y_{k+i}^s (y_k^s)^T\right\},\tag{5.85}$$

and the cross-covariance matrix between stochastic outputs and states as

$$G \triangleq \mathbf{E} \left\{ x_{k+1}^s (y_k^s)^T \right\}.$$
(5.86)

It is straightforward to show that the following identities hold:

$$\Lambda_0 \triangleq \mathbf{E}\left\{y_k^s(y_k^s)^T\right\} = C\Sigma_x C^T + R, \qquad (5.87)$$

$$G = A\Sigma_x C^T + S, (5.88)$$

and, for $i = 1, 2, \cdots$,

$$\Lambda_i = CA^{i-1}G, \tag{5.89}$$

$$\Lambda_{-i} = G^T (A^{i-1})^T C^T.$$
(5.90)

The reversed extended stochastic controllability matrix C_i^{rs} is defined as

$$\mathcal{C}_i^{rs} \triangleq \begin{bmatrix} A^{i-1}G, & A^{i-2}G, & \cdots, & AG, & G \end{bmatrix},$$

where the subscript *i* denotes the number of block columns. The following block Toeplitz matrices can be constructed from the output covariance matrices Λ_i

$$M_{i} \triangleq \begin{bmatrix} \Lambda_{i} & \Lambda_{i-1} & \Lambda_{i-2} & \cdots & \Lambda_{1} \\ \Lambda_{i+1} & \Lambda_{i} & \Lambda_{i-1} & \cdots & \Lambda_{2} \\ \vdots & \vdots & \vdots & \vdots & \vdots \\ \Lambda_{2i-1} & \Lambda_{2i-2} & \Lambda_{2i-3} & \cdots & \Lambda_{i} \end{bmatrix} = \mathcal{O}_{i} \mathcal{C}_{i}^{rs}, \qquad (5.91)$$
$$N_{i} \triangleq \begin{bmatrix} \Lambda_{0} & \Lambda_{-1} & \Lambda_{-2} & \cdots & \Lambda_{1-i} \\ \Lambda_{1} & \Lambda_{0} & \Lambda_{-1} & \cdots & \Lambda_{2-i} \\ \vdots & \vdots & \vdots & \vdots & \vdots \\ \Lambda_{i-1} & \Lambda_{i-2} & \Lambda_{i-3} & \cdots & \Lambda_{0} \end{bmatrix}. \qquad (5.92)$$

Consistently with equation (5.82) are defined the deterministic $X_j^{i^d}$ and the stochastic $X_j^{i^s}$ state sequence matrices as

$$X_j^{i^d} \triangleq \begin{bmatrix} x_i^d, & x_{i+1}^d, & \cdots, & x_{i+j-1}^d \end{bmatrix}, X_j^{i^s} \triangleq \begin{bmatrix} x_i^s, & x_{i+1}^s, & \cdots, & x_{i+j-1}^s \end{bmatrix}.$$

It is obvious that these matrices satisfy

$$X_j^i = X_j^{i^d} + X_j^{i^s}.$$

Similarly as we did for the deterministic case in Subsubsection 5.9.3.1, we define the past and future deterministic and stochastic state sequences as

$$\begin{aligned} X_p^d &\triangleq X_j^{0^d} \quad , \quad X_f^d &\triangleq X_j^{i^d}, \\ X_p^s &\triangleq X_j^{0^s} \quad , \quad X_f^s &\triangleq X_j^{i^s}, \end{aligned}$$

and the past and future deterministic and stochastic block Hankel output matrices as

$$\begin{array}{ll} Y_p^d \triangleq Y_{i,j}^{0^d} &, \quad Y_f^d \triangleq Y_{i,j}^{i^d}, \\ Y_p^s \triangleq Y_{i,j}^{0^s} &, \quad Y_f^s \triangleq Y_{i,j}^{i^s}. \end{array}$$

The matrix input-output equations for this case, analogous to equations (5.71) to (5.73) in the deterministic problem, become

$$Y_p = \mathcal{O}_i X_p^d + H_i U_p + Y_p^s,$$

$$Y_f = \mathcal{O}_i X_f^d + H_i U_f + Y_f^s,$$

$$X_f^d = A^i X_p^d + \mathcal{C}_i^{rd} U_p.$$

The algorithm for this combined deterministic-stochastic case is based on the computation of the extended observability matrix \mathcal{O}_i and a non-steady state Kalman filter estimate of the state sequences (we will denote this estimate by \widehat{X}_j^i) from input-output data. The system matrices are then recovered from \mathcal{O}_i and \widehat{X}_i^i .

The non-steady state Kalman filter estimate \hat{x}_k of the state x_k , given the input and output measurements u_0, u_1, \dots, u_{k-1} and y_0, y_1, \dots, y_{k-1} , is given by the following recursive expressions (see [Vd94a, Vd96] for the proof of this result)

$$\widehat{x}_{k} = A\widehat{x}_{k-1} + K_{k-1} \left(y_{k-1} - C\widehat{x}_{k-1} - Du_{k-1} \right), \qquad (5.93)$$

$$K_{k-1} = (G - AP_{k-1}C^{T}) (\Lambda_{0} - CP_{k-1}C^{T})^{-1}, \qquad (5.94)$$

$$P_{k} = AP_{k-1}A^{T} + (G - AP_{k-1}C^{T}) (\Lambda_{0} - CP_{k-1}C^{T})^{-1} (G - AP_{k-1}C^{T})^{T},$$
(5.95)

with initial state estimate $\hat{x}_0 = 0$, and initial covariance of the state estimate P_0 . The Kalman filter estimate can be explicitly written as

$$\widehat{x}_{k} = \begin{bmatrix} A^{k} - \Omega_{k}\mathcal{O}_{k} & | \mathcal{C}_{k}^{rd} - \Omega_{k}H_{k} & | \Omega_{k} \end{bmatrix} \begin{bmatrix} \widehat{x}_{0} \\ u_{0} \\ \vdots \\ u_{k-1} \\ y_{0} \\ \vdots \\ y_{k-1} \end{bmatrix}, \quad (5.96)$$

where the matrix Ω_k is defined as

$$\Omega_k \triangleq \left(\mathcal{C}_k^{rs} - A^k P_0 \mathcal{O}_k^T \right) \left(N_k - \mathcal{O}_k P_0 \mathcal{O}_k^T \right)^{-1}.$$

Equation (5.96) shows that the Kalman filter state estimate \hat{x}_k is a linear combination of the past inputs and outputs $u_0, u_1, \dots, u_{k-1}, y_0, y_1, \dots, y_{k-1}$, and the initial state estimate \hat{x}_0 . We can then compute the Kalman filter state estimate sequence matrix as

$$\widehat{X}_{j}^{i} \triangleq [\widehat{x}_{i}, \widehat{x}_{i+1}, \cdots, \widehat{x}_{i+j-1}],$$

$$= [A^{i} - \Omega_{i}\mathcal{O}_{i} \mid [\mathcal{C}_{i}^{rd} - \Omega_{i}H_{i} \mid \Omega_{i}]] \begin{bmatrix} \widehat{X}_{0} \\ W_{p} \end{bmatrix},$$

where \widehat{X}_0 is the sequence of initial states. This equation can be interpreted as a bank of Kalman filters working in parallel in each of the columns of the block Hankel matrix of past inputs and outputs W_p .

We are now able to present the basic combined deterministic-stochastic algorithm. The algorithm proceeds as follows:

Algorithm 5.9.2 (Combined Deterministic-Stochastic Algorithm).

1. From the input-output data compute the *oblique projection* of the row space of Y_f along the row space of U_f on the row space of W_p

$$\mathcal{P}_{i} \triangleq \left(Y_{f} - Y_{f}U_{f}^{T}\left(U_{f}U_{f}^{T}\right)^{\dagger}U_{f}\right)\left(W_{p} - W_{p}U_{f}^{T}\left(U_{f}U_{f}^{T}\right)^{\dagger}U_{f}\right)^{\dagger}W_{p}.$$

In [Vd94a, Vd96] it is proved that in the limit when the number of measurements $N \to \infty$ (and consequently $j \to \infty$), the matrix \mathcal{P}_i equals the product of the extended observability matrix \mathcal{O}_i and the Kalman filter state sequence estimate \widehat{X}_j^i , i.e.

$$\mathcal{P}_i = \mathcal{O}_i \widehat{X}_j^i,$$

where the Kalman filter is run from particular values of the initial estimates \hat{X}_0 and P_0 , which are computed from the input-output data. The interested reader is referred to [Vd94a, Vd96] for the details.

2. Perform the SVD of the matrix \mathcal{P}_i to obtain

$$\mathcal{P}_i = \mathcal{O}_i \widehat{X}_j^i = \begin{bmatrix} U_1 U_2 \end{bmatrix} \begin{bmatrix} \Sigma_1 & 0 \\ 0 & 0 \end{bmatrix} \begin{bmatrix} V_1^T \\ V_2^T \end{bmatrix} = \left(U_1 \Sigma_1^{1/2} \right) \left(\Sigma_1^{1/2} V_1^T \right).$$
(5.97)

3. The system order n_x is equal to the number of singular values different from zero in (5.97). That is $n_x = \operatorname{rank}(\Sigma_1)$.

4. The extended observability matrix is given by

$$\mathcal{O}_i = U_1 \Sigma_1^{1/2}.$$

5. The Kalman filter state estimate sequence matrix is given by

$$\widehat{X}_j^i = \Sigma_1^{1/2} V_1^T.$$

6. System matrices: As in the purely deterministic case, there are several ways to compute the system matrices from the extended observability matrix and the Kalman filter state estimate sequence matrix. We don't analyze this topic here and refer the interested reader to [Vd96].

The N4SID¹¹ algorithms [Vd94a], MOESP¹² algorithms [Ver94], and CVA¹³ algorithms [Lar90, Lar94] are variations of Algorithm 5.9.2, where the matrix \mathcal{P}_i is weighted with different matrices for each case, before the SVD is performed (see [Vd94a, Vd96] for details).

5.10 Simulation Examples

In this section, the proposed method for identification of multivariable systems using OBFP is illustrated with a brief simulation study.

An underlying continuous time multivariable system, with m = 2 outputs and n = 2 inputs and with transfer matrix representation given by:

$$G(s) = \begin{bmatrix} \frac{0.1e^{-s}}{(s+1)(s+0.1)} & \frac{1}{(s+2)(s+0.5)} \\ \frac{0.21}{(s+0.7)(s+0.3)} & \frac{0.32e^{-s}}{(s+0.4)(s+0.8)} \end{bmatrix}$$
(5.98)

is considered. It is assumed that there is available an observation of N = 2000samples spaced 1 second apart of the outputs $\{y_k^1\}, \{y_k^2\}$ of G(s) when the inputs $\{u_k^1\}, \{u_k^2\}$ are unit amplitude square waves of fundamental frequencies 0.02 Hz and 0.05 Hz respectively. The output data is corrupted with stationary and white Gaussian distributed noise of variances $\sigma_{e^1}^2 = \sigma_{e^2}^2 = 0.001$ and with colouring filter H(q) = I. Based on this observed data, the identification objective is to estimate the dynamics of the zero order hold equivalent [ÅW84]

¹¹N4SID stands for "Numerical algorithms for Subspace State Space System IDentification".
¹²MOESP stands for "Multivariable Output-Error State sPace".

¹³CVA stands for "Canonical Variate Analysis".

discrete time system

$$G(z) = \text{ZOH} \{G(s)\} = \begin{bmatrix} \frac{(0.0355z + 0.0247)}{z(z - 0.9048)(z - 0.3679)} & \frac{(0.2364z + 0.1038)}{(z - 0.6065)(z - 0.1353)} \\ \frac{(0.0760z + 0.0545)}{(z - 0.7408)(z - 0.4966)} & \frac{(0.1087z + 0.0729)}{z(z - 0.6703)(z - 0.4493)} \end{bmatrix}.$$
(5.99)



Figure 5.3: Left hand figure shows the true (solid line) and estimated (dashdotted line) frequency response. The estimate was obtained using a fixed denominator structure and a least squares criterion with N = 2000 samples. The four plots shown are the four possible responses from the 2 inputs to the 2 outputs. The right hand figure shows the true poles as crosses and the a-priori guesses as circles.

In the left hand diagram of Figure 5.3 we show the results as true and estimated Nyquist plots (for each of the four scalar entries of the estimated transfer function matrix) when using the least squares estimation methods of Section 5.4, and the fixed denominator model structure of (5.16)-(3.14). Five estimates based on five different noise realizations are shown as dash-dotted lines while the true frequency responses are shown as solid lines. The poles in the structure (5.16)-(3.14) where chosen at

 $\{\xi_k\} = \{0.2231, 0.8187, 0.0498, 0.3329, 0.2466, 0.5488, 0.5220, 0.2019\}.$

These choices correspond to an eighth order model and 32 scalar parameters being estimated. The choice of the pole locations is of limited accuracy as illustrated in the right hand diagram of Figure 5.3 where the true pole locations are shown as crosses, and the above choices are shown as circles.

Another estimation approach was also investigated whereby a state space model structure was found using a standard prediction error identification



Figure 5.4: Left hand figure shows estimation results over five different noise realizations when using an 8th order state-space model structure and a prediction error estimation method. The right hand figure shows the results using the same five noisy data records and the same model structure, but using an N4SID method to obtain the estimates. In both cases the true response is the solid line and the estimates are the dash-dot lines. (N = 2000 samples)

algorithm, implemented with The Mathworks's System Identification Toolbox [Lju95] along the lines suggested in [Lju91]. In this case, an eighth order model was used to capture the eight different poles in (5.98). This implies (after a suitable canonical form for the state space model structure is selected) the estimation of 48 parameters. The same data used to provide the estimates shown in Figure 5.3 were used to derive the prediction error method estimates.

The results are shown in the left hand diagram of Figure 5.4. As can be seen, the results are more accurate than for the fixed denominator structure, but as reference to Table 5.1 shows, they involve two orders of magnitude more computation. On the other hand, the derivation of the state-space model structure estimate does not require approximate prior knowledge of the location of system poles.

Finally, a sixth order state-space model was estimated using the Subspace State Space System Identification (N4SID) algorithm proposed in [Vd94a] (see the combined deterministic-stochastic algorithm in Section 5.9). The results are shown in the left hand diagram of Figure 5.4. The results are as accurate as for the prediction error method estimates, but as reference to Table 5.1 shows, they involve an order of magnitude less computation.

To complete the simulation example, this set of identification experiments was repeated with the only change being that the amount of observed data was decreased by 80% to 400 samples. The results are shown in Figures 5.5-5.6. The key point to notice is that the variability of the parameter estimates increases only slightly (as compared to the 2000 sample case) when using the

Method	# of Parameters	Flops	Order	Addit. Inform.
OBFP	32	4.59×10^{6}	8	Guesses for Poles
PEM	48	1.42×10^{8}	8	Obs. Indices: [4 4]
N4SID	72 (D=0; X0=0)	3.92×10^7	6	Aux Order: 10

Table 5.1: Computational Load of different Identification Methods (N=2000 samples)



Figure 5.5: Same experiment as shown in left hand diagrams of Figures 5.3 and 5.4 but with only 20% the amount of observed data (400 samples). Fixed denominator model structure on the left and state-space structure estimated using prediction error method on the right.

fixed denominator structure, but increases very markedly when estimating a state-space model structure with a general prediction error method or with a subspace method. This deterioration of performance for relatively small amount of observed data comes to no surprise for the case of subspace methods which provide accurate estimates only when the number of data is large (when $N \rightarrow \infty$).

Together, these examples suggest that if prior knowledge of pole positions is available, and specially if only short data records are available or if computational load is a serious concern, then estimation using a fixed denominator model structure (re-parameterized using Orthonormal Bases with Fixed Poles) can yield improved results when compared to state-space structure estimates using prediction error or subspace methods. However, if no prior information is available, (and if the calculation of confidence regions valid for finite data is not required) then it is likely that Subspace Methods will yield better results.



Figure 5.6: Same experiment as shown in right hand diagram of Figure 5.4 but with only 20% the amount of observed data (400 samples). State-space model structure estimated by N4SID methods.

5.11 Conclusions

In this chapter we have studied the problem of identification, from input-output data in the time domain, of discrete-time linear multivariable systems using fixed denominator model structures and least squares techniques. The estimation accuracy has been analyzed by quantifying the undermodelling error and the noise induced error. Fundamental in this analysis has been the reparameterization of the fixed denominator model structure using rational orthonormal bases with the same fixed poles. The main contributions of the chapter have been the extension to the multivariable setting, and to the case of using general orthonormal bases with fixed poles, of some single-input, singleoutput FIR results concerning the asymptotic (in model order and data-length) distribution of the frequency domain estimation error. Fundamental in the derivations were some results on convergence properties of block Toeplitz-like matrices that were also established in the chapter. As in the SISO case, a new phenomenom of bias/variance trade-off with respect to the choice of the poles in the model structure could also be recognized in the multivariable setting of this chapter.

The numerical robustness of the proposed identification scheme was also analyzed, and it was shown that the estimation with an orthonormal structure does not necessarily guarantee a better numerical conditioning when compared with the estimation using an equivalent (with the same fixed poles) non-orthonormal structure, the exception being the case of white input in which the estimation with an orthonormal structure is perfectly numerically conditioned.

For the purposes of comparison with the orthonormal basis-based estimation method proposed here, a brief overview of Subspace Methods, which also provide closed form estimates, was presented in this chapter. The simulation results showed that both methods have a similar performance, specially when prior information about the system dynamics is available and a relatively small amount of data is observed.

APPENDICES

5.A Proofs for Lemmas 5.3.1 and 5.5.1

Proof of Lemma 5.3.1 From the definition of the bases $\{\mathcal{B}_{\ell}^{ij}(z)\}$ in equation (3.28) we can write

$$\mathcal{B}_{\ell}^{ij}(z) = \mathcal{B}_{\ell}(z) \begin{bmatrix} 0 & \dots & 0 & \dots & 0 \\ \vdots & \ddots & \vdots & \ddots & \vdots \\ 0 & \dots & 1 & \dots & 0 \\ \vdots & \ddots & \vdots & \ddots & \vdots \\ 0 & \dots & 0 & \dots & 0 \end{bmatrix} \leftarrow i$$

$$\uparrow$$

so that

$$\theta_{\ell}^{ij} \mathcal{B}_{\ell}^{ij}(z) = \mathcal{B}_{\ell}(z) \begin{bmatrix} 0 & \dots & 0 & \dots & 0 \\ \vdots & \ddots & \vdots & \ddots & \vdots \\ 0 & \dots & \theta_{\ell}^{ij} & \dots & 0 \\ \vdots & \ddots & \vdots & \ddots & \vdots \\ 0 & \dots & 0 & \dots & 0 \end{bmatrix} \leftarrow i$$

$$\uparrow$$

Then

$$\sum_{i=1}^{m} \sum_{j=1}^{n} \theta_{\ell}^{ij} \mathcal{B}_{\ell}^{ij}(z) = \begin{bmatrix} \theta_{\ell}^{11} & \dots & \theta_{\ell}^{1n} \\ \vdots & \ddots & \vdots \\ \theta_{\ell}^{m1} & \dots & \theta_{\ell}^{mn} \end{bmatrix} \mathcal{B}_{\ell}(z) = \theta_{\ell}^{T} \mathcal{B}_{\ell}(z)$$

and the result follows straightforwardly.

Proof of Lemma 5.5.1 Using Parseval's Theorem, \widetilde{R}_p can be given the following frequency domain representation

$$\widetilde{R}_p = \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_p(e^{j\omega}) \Phi_u(\omega) \Gamma_p^{\star}(e^{j\omega}) \,\mathrm{d}\omega, \qquad (5.A.1)$$

where $\Gamma_p(e^{j\omega})$ is defined in equations (5.20) and (5.21). Then, appealing to the orthonormality of the bases we can write

$$\widetilde{R}_{p} \geq \inf_{\omega} \underline{\sigma}[\Phi_{u}(\omega)] \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_{p}(e^{j\omega}) \Gamma_{p}^{\star}(e^{j\omega}) d\omega$$
$$= \inf_{\omega} \underline{\sigma}[\Phi_{u}(\omega)] I_{np}.$$

Using a similar argument for the upper bound we obtain

$$\widetilde{R}_{p} \leq \sup_{\omega} \overline{\sigma}[\Phi_{u}(\omega)] \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_{p}(e^{j\omega}) \Gamma_{p}^{\star}(e^{j\omega}) d\omega$$
$$= \sup_{\omega} \overline{\sigma}[\Phi_{u}(\omega)] I_{np}.$$

The result now follows straightforwardly.

5.B Proofs for Lemmas 5.6.1 and 5.6.2

Proof of Lemma 5.6.1 Let the true system be represented by

$$y_k = G(q) \, u_k + \nu_k,$$

where

$$G(q) = \sum_{k=0}^{\infty} \theta_k \mathcal{B}_k(q),$$

and let $\Theta_0 \triangleq [\theta_0^T, \theta_1^T, \cdots, \theta_{p-1}^T]$ be the 'true' truncated parameter matrix, and $\widehat{\Theta}$ the least squares estimate as given in equation (5.26). Then the error between the parameter estimate and the true truncated parameter matrix is given by

$$\widetilde{\Theta} \triangleq \widehat{\Theta} - \Theta_0.$$

Considering equation (5.26) in the limit when N tends to infinity, we can write

$$\overline{\mathbf{E}}\left\{\phi_{k}\phi_{k}^{T}\right\}\widehat{\Theta}=\overline{\mathbf{E}}\left\{\phi_{k}y_{k}^{T}\right\},$$
(5.B.2)

where the notation

$$\overline{\mathbf{\mathsf{E}}}\left\{x_{k}\right\} \triangleq \lim_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} \mathbf{\mathsf{E}}\left\{x_{k}\right\}$$

due to Ljung [Lju87] has been used. Now, since

$$y_k = \Theta_0^T \phi_k + \sum_{\ell=p}^{\infty} \theta_\ell^T \mathcal{B}_\ell(q) u_k = \Theta_0^T \phi_k + z_k,$$

where $z_k \triangleq [G(q) - G(q, \Theta_0)]u_k = \sum_{\ell=p}^{\infty} \theta_{\ell}^T \mathcal{B}_{\ell}(q)u_k$, then

$$\overline{\mathbf{\mathsf{E}}}\left\{\phi_{k}y_{k}^{T}\right\} = \overline{\mathbf{\mathsf{E}}}\left\{\phi_{k}\phi_{k}^{T}\right\}\Theta_{0} + \overline{\mathbf{\mathsf{E}}}\left\{\phi_{k}z_{k}^{T}\right\}.$$

Equation (5.B.2) can then be written as

$$\overline{\mathbf{E}}\left\{\phi_{k}\phi_{k}^{T}\right\}\widehat{\Theta}=\overline{\mathbf{E}}\left\{\phi_{k}\phi_{k}^{T}\right\}\Theta_{0}+\overline{\mathbf{E}}\left\{\phi_{k}z_{k}^{T}\right\},$$

so that the error between the parameter estimate and the true truncated parameter matrix is given by

$$\widetilde{\Theta} = \widetilde{R}_p^{-1} \overline{\mathbf{E}} \left\{ \phi_k z_k^T \right\}, \qquad (5.B.3)$$

where the definition $\widetilde{R}_p \triangleq \overline{\mathbf{E}} \left\{ \phi_k \phi_k^T \right\}$ has been used.

Therefore, the error in the transfer matrix estimate is given by

$$G(e^{j\omega}) - G(e^{j\omega}, \widehat{\Theta}) = \overline{\mathbf{E}} \left\{ z_k \phi_k^T \right\} \widetilde{R}_p^{-1} \Gamma_p(e^{j\omega}) + \sum_{\ell=p}^{\infty} \theta_\ell^T \mathcal{B}_\ell(e^{j\omega}),$$

and then

$$\begin{aligned} \left| \left[G(e^{j\omega}) - G(e^{j\omega}, \widehat{\Theta}) \right]_{i,j} \right| &= \\ &= \left| \left[\mathbf{\overline{E}} \left\{ z_k \phi_k^T \right\} \widetilde{R}_p^{-1} \Gamma_p(e^{j\omega}) + \sum_{\ell=p}^{\infty} \theta_\ell^T \mathcal{B}_\ell(e^{j\omega}) \right]_{i,j} \right|, \\ &\leq \left| \left[\mathbf{\overline{E}} \left\{ z_k \phi_k^T \right\} \widetilde{R}_p^{-1} \Gamma_p(e^{j\omega}) \right]_{i,j} \right| + \sum_{\ell=p}^{\infty} \left| \theta_\ell^{ij} \right| \left| \mathcal{B}_\ell(e^{j\omega}) \right|, \\ &\leq \left\| \mathbf{\overline{E}} \left\{ z_k \phi_k^T \right\} \right\|_2 \left\| \widetilde{R}_p^{-1} \right\|_2 \left\| \Gamma_p(e^{j\omega}) \right\|_2 + \sum_{\ell=p}^{\infty} \left| \theta_\ell^{ij} \right| \left| \mathcal{B}_\ell(e^{j\omega}) \right|, \\ &\leq \frac{\sqrt{\gamma_p(\omega)} \left\| \mathbf{\overline{E}} \left\{ z_k \phi_k^T \right\} \right\|_2}{\inf_{\omega} \underline{\sigma}(\Phi_u(\omega))} + \sum_{\ell=p}^{\infty} \left| \theta_\ell^{ij} \right| \left| \mathcal{B}_\ell(e^{j\omega}) \right|. \end{aligned}$$
(5.B.4)

Now, by Parseval's Theorem, we can write

$$\overline{\mathbf{E}}\left\{z_k\phi_k^T\right\} = \frac{1}{2\pi} \int_{-\pi}^{\pi} \sum_{\ell=p}^{\infty} \theta_\ell^T \mathcal{B}_\ell(e^{j\omega}) \Phi_u(\omega) \Gamma_p^\star((e^{j\omega}) \mathrm{d}\omega)$$

so that

$$\begin{split} \left\| \overline{\mathbf{\mathsf{E}}} \left\{ z_{k} \phi_{k}^{T} \right\} \right\|_{2} &\leq \sum_{\ell=p}^{\infty} \left\| \theta_{\ell}^{T} \right\|_{2} \left\| \frac{1}{2\pi} \int_{-\pi}^{\pi} \mathcal{B}_{\ell}(e^{j\omega}) \Phi_{u}(\omega) \Gamma_{p}^{\star}((e^{j\omega}) \mathrm{d}\omega \right\|_{2}, \\ &\leq \sum_{\ell=p}^{\infty} \left\| \theta_{\ell}^{T} \right\|_{2} \frac{1}{2\pi} \int_{-\pi}^{\pi} \left| \mathcal{B}_{\ell}(e^{j\omega}) \right| \left\| \Phi_{u}(\omega) \right\|_{2} \left\| \Gamma_{p}^{\star}((e^{j\omega}) \right\|_{2} \mathrm{d}\omega, \\ &\leq K_{1} \sup_{\omega} \overline{\sigma}(\Phi_{u}(\omega)) \sum_{\ell=p}^{\infty} \left\| \theta_{\ell}^{T} \right\|_{2} \frac{1}{2\pi} \int_{-\pi}^{\pi} \sqrt{\gamma_{p}(\omega)} \mathrm{d}\omega, \\ &\leq K_{1} \sqrt{mn} \sup_{\omega} \sqrt{\gamma_{p}(\omega)} \sup_{\omega} \overline{\sigma}(\Phi_{u}(\omega)) \sum_{\ell=p}^{\infty} \max_{i,j} \left| \theta_{\ell}^{ij} \right|, \end{split}$$

with the constant K_1 defined in the statement of the lemma. Substituting back in equation (5.B.4) then yields the result.

Proof of Lemma 5.6.2 Equation (5.43) can be interpreted as the norm minimization problem

$$\Theta_{\star} = \operatorname*{arg\,min}_{\Theta \in \mathbb{R}^{np \times m}} \left\{ \left\| G(e^{j\omega}) - G(e^{j\omega}, \Theta) \right\|_{u} \right\}$$

where $\|\cdot\|_u$ stands for the norm on the Hilbert space $H^{m\times n}_2(\mathbb{T})$ induced by the inner product

$$\langle F, W \rangle_u = \frac{1}{2\pi} \int_{-\pi}^{\pi} \operatorname{Tr} \left\{ F(e^{j\omega}) \Phi_u(\omega) W^{\star}(e^{j\omega}) \right\} d\omega.$$

Now, since $G(e^{j\omega}, \Theta)$ is in the subspace of $(H_2^{m \times n}(\mathbb{T}), \|\cdot\|_u)$ spanned by the basis functions $\{\mathcal{B}_{\ell}(e^{j\omega})\}^{-14}$, then by the Projection Theorem

$$\left\langle G(e^{j\omega}) - G(e^{j\omega}, \Theta_{\star}), G(e^{j\omega}, \Theta_{\star}) \right\rangle_{u} = 0$$

so that

$$\left\langle G(e^{\mathrm{j}\omega}),G(e^{\mathrm{j}\omega},\Theta_{\star})\right\rangle _{u}=\left\|G(e^{\mathrm{j}\omega},\Theta_{\star})\right\|_{u}^{2}.$$

Use of Cauchy-Schwarz inequality then gives the result.

5.C Relation between the estimates $\widehat{\eta}$ and $\widehat{\Theta}$

Lemma 5.C.1. Let Θ be defined as in equation (5.19) and η as in equation (5.49), then the least squares estimate $\hat{\eta}$ of η is related to the least squares estimate $\hat{\Theta}$ of Θ according to

$$\widehat{\eta} = \mathsf{vec} \ \widehat{\Theta}^T$$

¹⁴To be precise, by the matrix bases $\{\mathcal{B}_{\ell}^{ij}(e^{j\omega})\}$.

Proof: The least squares estimate $\widehat{\eta}$ of η is given by

$$\widehat{\eta} = \left(\sum_{k=0}^{N-1} \psi_k \psi_k^T\right)^{-1} \sum_{k=0}^{N-1} \psi_k y_k$$
(5.C.5)

Considering that $\psi_k riangleq (\phi_k \otimes I_m)$ we have

$$\psi_k \psi_k^T = (\phi_k \otimes I_m)(\phi_k^T \otimes I_m) = \phi_k \phi_k^T \otimes I_m$$

and then

$$\sum_{k=0}^{N-1} \psi_k \psi_k^T = \sum_{k=0}^{N-1} (\phi_k \phi_k^T \otimes I_m) = \left(\sum_{k=0}^{N-1} \phi_k \phi_k^T\right) \otimes I_m$$

so that

$$\left(\sum_{k=0}^{N-1}\psi_k\psi_k^T\right)^{-1} = \left(\sum_{k=0}^{N-1}\phi_k\phi_k^T\right)^{-1} \otimes I_m$$

Furthermore

$$\sum_{k=0}^{N-1} \psi_k y_k = \sum_{k=0}^{N-1} (\phi_k \otimes I_m) y_k$$

Equation (5.C.5) can then be written as

$$\widehat{\eta} = \left(\left(\sum_{k=0}^{N-1} \phi_k \phi_k^T \right)^{-1} \otimes I_m \right) \sum_{k=0}^{N-1} (\phi_k \otimes I_m) y_k$$
(5.C.6)

On the other hand, the least squares estimate $\widehat{\Theta}$ of Θ is given by

$$\widehat{\Theta}^T = \left(\sum_{k=0}^{N-1} \phi_k y_k^T\right)^T \left(\sum_{k=0}^{N-1} \phi_k \phi_k^T\right)^{-1},$$

so that

vec
$$\widehat{\Theta}^T = \left(\left(\sum_{k=0}^{N-1} \phi_k \phi_k^T \right)^{-1} \otimes I_m \right)$$
 vec $\left\{ \sum_{k=0}^{N-1} y_k \phi_k^T \right\}$ (5.C.7)

 But

$$\operatorname{vec} \left\{ \sum_{k=0}^{N-1} y_k \phi_k^T \right\} = \sum_{k=0}^{N-1} \operatorname{vec} \left\{ y_k \phi_k^T \right\},$$
$$= \sum_{k=0}^{N-1} (\phi_k \otimes I_m) \operatorname{vec} y_k,$$
$$= \sum_{k=0}^{N-1} (\phi_k \otimes I_m) y_k.$$

In this case equation (5.C.7) can be written as

vec
$$\widehat{\Theta}^T = \left(\left(\sum_{k=0}^{N-1} \phi_k \phi_k^T \right)^{-1} \otimes I_m \right) \sum_{k=0}^{N-1} (\phi_k \otimes I_m) y_k.$$
 (5.C.8)

The result then follows by noting that the right hand sides of equations (5.C.6) and (5.C.8) are equal.

5.D Proof of Convergence of Block Toeplitz-like Matrices

Proof of Lemma 5.7.1 Let $\mathsf{B}_p(\omega)$ denote

$$\mathsf{B}_{p}(\omega) \equiv \mathsf{B}_{p}(e^{j\omega}) \triangleq \left[\mathcal{B}_{0}(e^{j\omega}), \mathcal{B}_{1}(e^{j\omega}), \cdots, \mathcal{B}_{p-1}(e^{j\omega})\right]^{T}.$$
(5.D.9)

Then

$$\begin{split} &\Gamma_p(\omega) &= \mathsf{B}_p(\omega) \otimes I_n, \\ &\widetilde{\Gamma}_p(\omega) &= \mathsf{B}_p(\omega) \otimes I_n \otimes I_m = \mathsf{B}_p(\omega) \otimes I_{nm}, \end{split}$$

so that

$$\begin{split} \Gamma_p^{\star}(\mu)\Gamma_p(\sigma) &= (\mathsf{B}_p^{\star}(\mu)\otimes I_n)(\mathsf{B}_p(\sigma)\otimes I_n), \\ &= \mathsf{B}_p^{\star}(\mu)\mathsf{B}_p(\sigma)\otimes I_n, \\ &= \sum_{k=0}^{p-1}\overline{\mathcal{B}_k(e^{j\mu})}\mathcal{B}_k(e^{j\sigma})\otimes I_n, \\ &= K_p(\sigma,\mu)\otimes I_n = K_p(\sigma,\mu)I_n, \end{split}$$

and similarly

$$\Gamma_p^{\star}(\sigma)\Gamma_p(\omega) = K_p(\omega,\sigma)I_n,$$

where $K_p(\cdot, \cdot)$ is the Reproducing Kernel associated with the orthonormal system $\{\mathcal{B}_k(z)\}$. Now we can write

$$\widetilde{\Gamma}_{p}^{\star}(\mu)M_{p}(F,W)\widetilde{\Gamma}_{p}(\omega) = = \frac{1}{2\pi} \int_{-\pi}^{\pi} (\Gamma_{p}^{\star}(\mu) \otimes I_{m}) \left(\Gamma_{p}(\sigma)F(\sigma)\Gamma_{p}^{\star}(\sigma) \otimes W(\sigma)\right) (\Gamma_{p}(\omega) \otimes I_{m}) d\sigma = \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_{p}^{\star}(\mu)\Gamma_{p}(\sigma)F(\sigma)\Gamma_{p}^{\star}(\sigma)\Gamma_{p}(\omega) \otimes W(\sigma) d\sigma = \frac{1}{2\pi} \int_{-\pi}^{\pi} K_{p}(\sigma,\mu)K_{p}(\omega,\sigma) (F(\sigma) \otimes W(\sigma)) d\sigma$$
(5.D.10)

Considering the case $\mu=\omega$ the previous expression becomes

$$\widetilde{\Gamma}_{p}^{\star}(\mu)M_{p}(F,W)\widetilde{\Gamma}_{p}(\omega) = \frac{1}{2\pi}\int_{-\pi}^{\pi}|K_{p}(\omega,\sigma)|^{2}\left(F(\sigma)\otimes W(\sigma)\right)\mathrm{d}\sigma$$

Therefore, using Lemma 4.C.1, we have

$$\frac{\widetilde{\Gamma}_{p}^{\star}(\mu)M_{p}(F,W)\widetilde{\Gamma}_{p}(\omega)}{\gamma_{p}(\omega)} - F(\omega) \otimes W(\omega) =
= \frac{1}{\gamma_{p}(\omega)} \left(\widetilde{\Gamma}_{p}^{\star}(\mu)M_{p}(F,W)\widetilde{\Gamma}_{p}(\omega) - \gamma_{p}(\omega)F(\omega) \otimes W(\omega)\right)
= \frac{1}{2\pi\gamma_{p}(\omega)} \int_{-\pi}^{\pi} |K_{p}(\omega,\sigma)|^{2} \left(F(\sigma) \otimes W(\sigma) - F(\omega) \otimes W(\omega)\right) d\sigma$$

Considering now the i, j-th component of the above matrix difference, we have

$$\left| \left[\frac{\widetilde{\Gamma}_{p}^{\star}(\mu)M_{p}(F,W)\widetilde{\Gamma}_{p}(\omega)}{\gamma_{p}(\omega)} - F(\omega) \otimes W(\omega) \right]_{i,j} \right| \leq \\ \leq \frac{1}{2\pi\gamma_{p}(\omega)} \int_{-\pi}^{\pi} |K_{p}(\omega,\sigma)|^{2} \left| [F(\sigma) \otimes W(\sigma)]_{i,j} - [F(\omega) \otimes W(\omega)]_{i,j} \right| d\sigma \\ = \frac{1}{2\pi\gamma_{p}(\omega)} \int_{\sigma\in\Omega} |K_{p}(\omega,\sigma)|^{2} \left| [F(\sigma) \otimes W(\sigma)]_{i,j} - [F(\omega) \otimes W(\omega)]_{i,j} \right| d\sigma + \\ + \frac{1}{2\pi\gamma_{p}(\omega)} \int_{\sigma\notin\Omega} |K_{p}(\omega,\sigma)|^{2} \left| [F(\sigma) \otimes W(\sigma)]_{i,j} - [F(\omega) \otimes W(\omega)]_{i,j} \right| d\sigma$$

$$(5.D.11)$$

where $\Omega = [\omega - \delta, \omega + \delta]$, for some $\delta > 0$. Now, since $F(\omega)$ and $W(\omega)$ are assumed (component-wise) continuous, then for an arbitrary $\varepsilon > 0$ there is a δ sufficiently small such that

$$\left| \left[F(\sigma) \otimes W(\sigma) \right]_{i,j} - \left[F(\omega) \otimes W(\omega) \right]_{i,j} \right| < \varepsilon$$

for $\sigma \in \Omega$. Therefore, using this and Lemma 4.C.1 the first term on the RHS of equation (5.D.11) can be bounded as

$$\frac{1}{2\pi\gamma_p(\omega)} \int_{\sigma\in\Omega} |K_p(\omega,\sigma)|^2 \left| [F(\sigma)\otimes W(\sigma)]_{i,j} - [F(\omega)\otimes W(\omega)]_{i,j} \right| \mathrm{d}\sigma \leq \frac{\varepsilon}{2\pi\gamma_p(\omega)} \int_{-\pi}^{\pi} |K_p(\omega,\sigma)|^2 \,\mathrm{d}\sigma = \varepsilon$$

On the other hand, since $F(\omega)$ and $W(\omega)$ are continuous on compact $[-\pi, \pi]$, then $[F(\omega) \otimes W(\omega)]_{i,j}$ is bounded by some constant $M/2 < \infty$. Therefore, the

second term in the RHS of equation (5.D.11) can be bounded as

$$\frac{1}{2\pi\gamma_p(\omega)} \int_{\sigma\notin\Omega} |K_p(\omega,\sigma)|^2 \left| [F(\sigma)\otimes W(\sigma)]_{i,j} - [F(\omega)\otimes W(\omega)]_{i,j} \right| d\sigma \leq \frac{M}{2\pi\gamma_p(\omega)} \int_{\sigma\notin\Omega} |K_p(\omega,\sigma)|^2 d\sigma$$

We finally have

$$\left| \left[\frac{\widetilde{\Gamma}_{p}^{\star}(\mu)M_{p}(F,W)\widetilde{\Gamma}_{p}(\omega)}{\gamma_{p}(\omega)} - F(\omega) \otimes W(\omega) \right]_{i,j} \right| \leq \\ \leq \varepsilon + \frac{M}{2\pi\gamma_{p}(\omega)} \int_{\sigma\notin\Omega} |K_{p}(\omega,\sigma)|^{2} d\sigma$$

Using Lemma 4.C.1 and the fact that ε is arbitrary then gives the result for $\mu=\omega.$

Proof of Lemma 5.7.2 The (smn + y, tmn + x)-th element of matrix $M_p(W, X)$ is given by

$$[M_p(W,X)]_{smn+y,tmn+x} = \frac{1}{2\pi} \int_{-\pi}^{\pi} \mathcal{B}_s \overline{\mathcal{B}}_t [W(\omega) \otimes X(\omega)]_{y,x} d\omega$$
$$= \langle \mathcal{B}_s, \mathcal{B}_t [W \otimes X]_{y,x} \rangle.$$

Therefore

$$[M_p(W, X)M_p(Z, U)]_{smn+y,tmn+x} =$$

$$= \sum_{k=0}^{p-1} \sum_{r=1}^{mn} [M_p(W, X)]_{smn+y,kmn+r} [M_p(Z, U)]_{kmn+r,tmn+x}$$

$$= \sum_{k=0}^{p-1} \sum_{r=1}^{mn} \langle \mathcal{B}_s[W \otimes X]_{y,r}, \mathcal{B}_k \rangle \overline{\langle \mathcal{B}_t[Z \otimes U]_{r,x}, \mathcal{B}_k \rangle}$$
(5.D.12)

On the other hand

$$[M_p(WZ, XU)]_{smn+y,tmn+x} = \frac{1}{2\pi} \int_{-\pi}^{\pi} \mathcal{B}_s \overline{\mathcal{B}}_t [WZ \otimes XU]_{y,x} d\omega$$
$$= \frac{1}{2\pi} \int_{-\pi}^{\pi} \mathcal{B}_s \overline{\mathcal{B}}_t [(W \otimes X)(Z \otimes U)]_{y,x} d\omega$$

Considering that

$$[(W \otimes X)(Z \otimes U)]_{y,x} = \sum_{r=1}^{mn} [W \otimes X]_{y,r} [Z \otimes U]_{r,x}$$
we can write

$$[M_p(WZ, XU)]_{smn+y,tmn+x} = \sum_{r=1}^{mn} \frac{1}{2\pi} \int_{-\pi}^{\pi} \mathcal{B}_s \overline{\mathcal{B}_t}[W \otimes X]_{y,r}[Z \otimes U]_{r,x} d\omega$$
$$= \sum_{r=1}^{mn} \langle \mathcal{B}_s[W \otimes X]_{y,r}, \mathcal{B}_t[Z \otimes U]_{r,x} \rangle \quad (5.D.13)$$

Now, from equations (5.D.12) and (5.D.13), and Lemma 5.E.10 we have

$$\begin{split} \left| [M_{p}(W, X)M_{p}(Z, U)]_{smn+y,tmn+x} - [M_{p}(WZ, XU)]_{smn+y,tmn+x} \right| = \\ &= \left| \sum_{r=1}^{mn} \left[\sum_{k=0}^{p-1} \langle \mathcal{B}_{s}[W \otimes X]_{y,r}, \mathcal{B}_{k} \rangle \overline{\langle \mathcal{B}_{t}[Z \otimes U]_{r,x}, \mathcal{B}_{k} \rangle} - \langle \mathcal{B}_{s}[W \otimes X]_{y,r}, \mathcal{B}_{t}[Z \otimes U]_{r,x} \rangle \right] \right| \\ &\leq \sum_{r=1}^{mn} \left| \sum_{k=0}^{p-1} \langle \mathcal{B}_{s}[W \otimes X]_{y,r}, \mathcal{B}_{k} \rangle \overline{\langle \mathcal{B}_{t}[Z \otimes U]_{r,x}, \mathcal{B}_{k} \rangle} - \langle \mathcal{B}_{s}[W \otimes X]_{y,r}, \mathcal{B}_{t}[Z \otimes U]_{r,x} \rangle \right| \\ &\leq \sum_{r=1}^{mn} \left\| \mathcal{B}_{s}[W \otimes X]_{y,r} - \widehat{V}_{y,r} \right\|_{H_{2}} \left\| \mathcal{B}_{t}[Z \otimes U]_{r,x} - \widehat{J}_{r,x} \right\|_{H_{2}} \end{split}$$
(5.D.14)

where

$$\widehat{V}_{y,r}(\omega) = \sum_{k=0}^{p-1} \left\langle \mathcal{B}_s[W \otimes X]_{y,r}, \mathcal{B}_k \right\rangle \mathcal{B}_k(\omega)$$

and

$$\widehat{J}_{r,x}(\omega) = \sum_{k=0}^{p-1} \left\langle \mathcal{B}_t[Z \otimes U]_{r,x}, \mathcal{B}_k \right\rangle \mathcal{B}_k(\omega)$$

But by Lemma 5.E.11, for some $K < \infty$ and $|\eta| < 1$ we have

$$\begin{aligned} \left\| \mathcal{B}_{s}[W \otimes X]_{y,r} - \widehat{V}_{y,r} \right\|_{H_{2}} &\leq K(\eta^{p-s} + \eta^{s}) \\ \left\| \mathcal{B}_{t}[Z \otimes U]_{r,x} - \widehat{J}_{r,x} \right\|_{H_{2}} &\leq K(\eta^{p-t} + \eta^{t}). \end{aligned}$$

Substituting this into (5.D.14) then gives the result.

5.E Technical Lemmas

Lemma 5.E.1. Invariance of LSE under linear re-parameterization. Let the parameterized model be given by

$$y_k = G(q, \Theta)u_k + \nu_k = \Theta^T \phi_k + \nu_k \tag{5.E.15}$$

where

$$G(q,\Theta) \triangleq \Theta^T(\mathsf{B}_p(q) \otimes I_n)$$
(5.E.16)

$$\mathsf{B}_{p}(q) \triangleq [\mathcal{B}_{0}(q), \mathcal{B}_{1}(q), \cdots, \mathcal{B}_{p-1}(q)]^{T}$$
(5.E.17)

$$\phi_k = (\mathsf{B}_p(q) \otimes I_n) u_k. \tag{5.E.18}$$

Let

$$G(q,\beta) \triangleq \beta^{T}(\mathsf{A}_{p}(q) \otimes I_{n})$$
(5.E.19)

$$\mathsf{A}_p(q) \triangleq J\mathsf{B}_p(q), \tag{5.E.20}$$

be a linear re-parameterization of the transfer matrix of the system, where J is a nonsingular matrix of appropriate dimensions. Then, the least squares estimates of the parameter matrices Θ and β are related according to

$$\widehat{\beta} = (J^T \otimes I_n)^{-1} \widehat{\Theta},$$

so that the corresponding transfer matrix estimates are the same, i.e.

$$G(q,\widehat{\beta}) = G(q,\widehat{\Theta}).$$

Proof: The re-parameterization (5.E.19)-(5.E.20) leads to the linear regressor form

$$y_k = G(q, \beta)u_k + \nu_k = \beta^T \psi_k + \nu_k, \qquad (5.E.21)$$

where the new regressor matrix ψ_k is related to the original one, $\phi_k,$ according to

$$\psi_k = (J \otimes I_n)\phi_k. \tag{5.E.22}$$

Adopting the vectorized notation

$$\begin{array}{lll} Y^{T} &=& \left(y_{0}, y_{1}, \cdots, y_{N-1}\right), \\ V^{T} &=& \left(\nu_{0}, \nu_{1}, \cdots, \nu_{N-1}\right), \\ \Phi^{T} &=& \left(\phi_{0}, \phi_{1}, \cdots, \phi_{N-1}\right), \\ \Psi^{T} &=& \left(\psi_{0}, \psi_{1}, \cdots, \psi_{N-1}\right), \end{array}$$

the model for the N point observed data record can be written as:

$$Y = \Phi \Theta + V, \tag{5.E.23}$$

or in re-parameterized form

$$Y = \Psi \beta + V. \tag{5.E.24}$$

Now, the least squares estimate $\widehat{\beta}$ of the parameter matrix β is given by

$$\widehat{\beta} = \left(\Psi^T \Psi\right)^{-1} \Psi^T Y.$$

Based on equation (5.E.22), simple algebra gives

$$\Psi = \Phi(J^T \otimes I_n),$$

so that $\widehat{\beta}$ can be written as

$$\widehat{\beta} = (\Psi^T \Psi)^{-1} \Psi^T Y,$$

$$= ((J \otimes I_n) \Phi^T \Phi (J \otimes I_n)^T)^{-1} (J \otimes I_n) \Phi^T Y,$$

$$= (J \otimes I_n)^{-T} (\Phi^T \Phi)^{-1} (J \otimes I_n)^{-1} (J \otimes I_n) \Phi^T Y,$$

$$= (J \otimes I_n)^{-T} (\Phi^T \Phi)^{-1} \Phi^T Y,$$

$$= (J \otimes I_n)^{-T} \widehat{\Theta}.$$

It remains now to prove that the transfer matrix estimates with both parameterizations are the same. We have

$$G(q, \widehat{\beta}) = \widehat{\beta}^{T}(\mathsf{A}_{p}(q) \otimes I_{n}),$$

$$= \widehat{\Theta}^{T}(J \otimes I_{n})^{-1}(J \otimes I_{n})(\mathsf{B}_{p}(q) \otimes I_{n}),$$

$$= \widehat{\Theta}^{T}(\mathsf{B}_{p}(q) \otimes I_{n}),$$

$$= G(q, \widehat{\Theta}).$$

Lemma 5.E.2. Let $\{B_k\}$ be a set of OBFP defined as in equation (3.14) from the set of poles $\{\xi_k\}$, and let ξ_{max} denote the pole with maximum module in this set. Then

$$\sqrt{\frac{1-|\xi_{max}|}{1+|\xi_{max}|}} \le |\mathcal{B}_k(e^{j\omega})| \le \sqrt{\frac{1+|\xi_{max}|}{1-|\xi_{max}|}} \triangleq K_1 < \infty$$

Proof:

$$\begin{aligned} \left| \mathcal{B}_{k}(e^{j\omega}) \right| &= \left| \left(\frac{\sqrt{1 - |\xi_{k}|^{2}}}{e^{j\omega} - \xi_{k}} \right) \prod_{i=0}^{k-1} \left(\frac{1 - \overline{\xi_{i}}e^{j\omega}}{e^{j\omega} - \xi_{i}} \right) \right| \\ &= \left| \left(\frac{\sqrt{1 - |\xi_{k}|^{2}}}{e^{j\omega} - \xi_{k}} \right) \right| = \frac{\sqrt{1 - |\xi_{k}|^{2}}}{|e^{j\omega} - \xi_{k}|} \end{aligned}$$

Now, since by assumption is $|\xi_k| \leq |\xi_{max}| < 1$, we have

$$(1 - |\xi_{max}|) \le (1 - |\xi_k|) \le \left| e^{j\omega} - \xi_k \right| \le (1 + |\xi_k|) \le (1 + |\xi_{max}|)$$

and the result follows straightforwardly.

Lemma 5.E.3. Let $\gamma_p(\omega)$ be defined as

$$\gamma_p(\omega) \triangleq K_p(\omega, \omega) = \sum_{k=0}^{p-1} \left| \mathcal{B}_k(e^{j\omega}) \right|^2$$

and let ξ_{max} denote the pole with maximum module in the set $\{\xi_k\}$ of poles of the basis functions $\{\mathcal{B}_k\}$. Then the following inequalities hold

$$\frac{p}{2}\left(1 - |\xi_{max}|\right) \le \frac{1}{2} \sum_{k=0}^{p-1} (1 - |\xi_k|) \le \gamma_p(\omega) \le p \frac{1 + |\xi_{max}|}{1 - |\xi_{max}|}$$

Proof: The result is a direct consequence of Lemma 5.E.2. Lemma 5.E.4. Let Q_p and $\widetilde{G}_p(\omega)$ be defined as

$$\begin{aligned} Q_p &\triangleq \lim_{N \to \infty} N \mathbf{E} \left\{ V_N'(\theta_0) \left(V_N'(\theta_0) \right)^T \right\}, \\ \widetilde{G}_p(\omega) &\triangleq G(e^{j\omega}) - G(e^{j\omega}, \theta_0). \end{aligned}$$

Then

$$Q_p = \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_p(\omega) \Phi_u(\omega) \Gamma_p^{\star}(\omega) \otimes \Phi_{\nu}(\omega) \,\mathrm{d}\omega + \Delta_p$$

where

$$\Delta_p \triangleq \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_p(\omega) \Phi_u(\omega) \Gamma_p^{\star}(\omega) \otimes \widetilde{G}_p(\omega) \Phi_u(\omega) \widetilde{G}_p^{\star}(\omega) \,\mathrm{d}\omega.$$

Proof: By the definition of Q_p

$$Q_p = \lim_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} \sum_{\ell=0}^{N-1} (\phi_k \otimes I_m) \mathbf{E} \left\{ \varepsilon_k \varepsilon_\ell^T \right\} (\phi_\ell^T \otimes I_m)$$

where ϕ_k is defined in (5.22) and

$$\varepsilon_k = \underbrace{[G(q) - G_p(q, \theta_0)]u_k}_{Z_k} + \nu_k.$$

Therefore, using Lemma 5.E.5

$$Q_{p} = \lim_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} \sum_{\ell=0}^{N-1} (\phi_{k} \otimes I_{m}) \mathbf{E} \{ \nu_{k} \nu_{\ell}^{T} \} (\phi_{\ell}^{T} \otimes I_{m}) + \lim_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} \sum_{\ell=0}^{N-1} (\phi_{k} \otimes I_{m}) \mathbf{E} \{ z_{k} z_{\ell}^{T} \} (\phi_{\ell}^{T} \otimes I_{m})$$

$$= \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_{p}(\omega) \Phi_{u}(\omega) \Gamma_{p}^{\star}(\omega) \otimes \Phi_{\nu}(\omega) d\omega + \lim_{\ell \to \pi} \int_{-\pi}^{\pi} \Gamma_{p}(\omega) \Phi_{u}(\omega) \Gamma_{p}^{\star}(\omega) \otimes \widetilde{G}_{p}(\omega) \Phi_{u}(\omega) \widetilde{G}_{p}^{\star}(\omega) d\omega .$$

Lemma 5.E.5.

$$\lim_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} \sum_{\ell=0}^{N-1} (\phi_k \otimes I_m) \mathbf{E} \left\{ \nu_k \nu_\ell^T \right\} (\phi_\ell^T \otimes I_m) = \\ = \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_p(\omega) \Phi_u(\omega) \Gamma_p^{\star}(\omega) \otimes \Phi_\nu(\omega) \, \mathrm{d}\omega.$$

Proof: Since $\{\nu_k\}$ is a stationary process with an associated spectral density $\Phi_{\nu}(\omega)$ then

$$\mathbf{E}\left\{\nu_{k}\nu_{\ell}^{T}\right\} \triangleq R_{\nu}(k-\ell) = \frac{1}{2\pi}\int_{-\pi}^{\pi} \Phi_{\nu}(\omega)e^{-\mathrm{j}\omega(k-\ell)}\,\mathrm{d}\omega$$

Therefore,

$$\frac{1}{N} \sum_{k=0}^{N-1} \sum_{\ell=0}^{N-1} (\phi_k \otimes I_m) \mathbf{E} \left\{ \nu_k \nu_\ell^T \right\} (\phi_\ell^T \otimes I_m) =$$

$$= \frac{1}{2\pi N} \int_{-\pi}^{\pi} \sum_{k=0}^{N-1} \sum_{\ell=0}^{N-1} (\phi_k \otimes I_m) \Phi_\nu (\phi_\ell^T \otimes I_m) e^{-j\omega(k-\ell)} d\omega$$

$$= \frac{1}{2\pi N} \int_{-\pi}^{\pi} \sum_{k=0}^{N-1} \sum_{\ell=0}^{N-1} (\phi_k \phi_\ell^T \otimes \Phi_\nu(\omega)) e^{-j\omega(k-\ell)} d\omega$$

$$= \frac{1}{2\pi N} \int_{-\pi}^{\pi} \left(\sum_{k=0}^{N-1} \sum_{\ell=0}^{N-1} \phi_k \phi_\ell^T e^{-j\omega(k-\ell)} \right) \otimes \Phi_\nu(\omega) d\omega$$

$$= \frac{1}{2\pi} \int_{-\pi}^{\pi} \sum_{\tau=1-N}^{N-1} \underbrace{\left(\frac{1}{N} \sum_{k=0}^{N-1} \phi_k \phi_{k-\tau}^T \right)}_{R_{\phi_N}(\tau)} e^{-j\omega\tau} \otimes \Phi_\nu(\omega) d\omega$$
(5.E.25)

where use has been made of the change of variable $k - \ell = \tau$ and it has been assumed that $u_k = 0$ for k < 0. Since u_k is a quasi-stationary signal, then also $\phi_k \triangleq \Gamma_p(q)u_k$ is quasi-stationary, and then its covariance matrix is given by

$$R_{\phi}(\tau) \triangleq \lim_{N \to \infty} \frac{1}{N} \sum_{k=0}^{N-1} \phi_k \phi_{k-\tau}^T = \lim_{N \to \infty} R_{\phi_N}(\tau)$$

and the spectral density by

$$\Phi_{\phi}(\omega) \triangleq \sum_{\tau=-\infty}^{\infty} R_{\phi}(\tau) e^{-\mathbf{j}\omega\tau}$$

Now, the result follows by taking the limits when N tends to infinity in equation (5.E.25), and considering that

$$\Phi_{\phi}(\omega) = \Gamma_{p}(\omega)\Phi_{u}(\omega)\Gamma_{p}^{\star}(\omega)$$

Lemma 5.E.6. Let $\Gamma_p(\omega)$ and $\widetilde{\Gamma}_p(\omega)$ be defined as

$$\Gamma_p(\omega) \triangleq \left[\mathcal{B}_0(e^{j\omega}), \mathcal{B}_1(e^{j\omega}), \cdots, \mathcal{B}_{p-1}(e^{j\omega}) \right]^T \otimes I_n$$

$$\widetilde{\Gamma}_p(\omega) \triangleq \Gamma_p(\omega) \otimes I_m$$

respectively. Then

$$\|\Gamma_p(\omega)\|_2 = \|\widetilde{\Gamma}_p(\omega)\|_2 = \sqrt{\gamma_p(\omega)}$$

Proof: Let $B_p(\omega)$ be defined as in equation (5.D.9), so that $\Gamma_p(\omega)$ and $\widetilde{\Gamma}_p(\omega)$ can be written as

$$\begin{split} &\Gamma_p(\omega) &= &\mathsf{B}_p(\omega)\otimes I_n \\ &\widetilde{\Gamma}_p(\omega) &= &\mathsf{B}_p(\omega)\otimes I_n\otimes I_m = \mathsf{B}_p(\omega)\otimes I_{nm} \end{split}$$

respectively. Then

$$\begin{aligned} \|\Gamma_{p}(\omega)\|_{2} &\triangleq \sup_{x \neq 0} \frac{\|\Gamma_{p}(\omega)x\|_{2}}{\|x\|_{2}} \\ &= \sup_{x \neq 0} \frac{\sqrt{x^{*}\Gamma_{p}^{*}(\omega)\Gamma_{p}(\omega)x}}{\sqrt{x^{*}x}} \\ &= \sup_{x \neq 0} \frac{\sqrt{x^{*}(\mathsf{B}_{p}^{*}(\omega) \otimes I_{n})(\mathsf{B}_{p}(\omega) \otimes I_{n})x}}{\sqrt{x^{*}x}} \\ &= \sup_{x \neq 0} \frac{\sqrt{x^{*}(\mathsf{B}_{p}^{*}(\omega)\mathsf{B}_{p}(\omega) \otimes I_{n})x}}{\sqrt{x^{*}x}} = \sqrt{\gamma_{p}(\omega)} \end{aligned}$$

where the last equality follows considering that $\mathsf{B}_p^{\star}(\omega)\mathsf{B}_p(\omega) = \gamma_p(\omega)$. In similar way it can be proved that $\|\widetilde{\Gamma}_p(\omega)\|_2 = \sqrt{\gamma_p(\omega)}$.

Lemma 5.E.7. Let R_p be defined as in equation (5.55), and let $\lambda(R_p)$ denote the eigenvalues of R_p , and $\underline{\sigma}(\Phi_u(\omega))$ and $\overline{\sigma}(\Phi_u(\omega))$ denote the minimum and maximum singular values of $\Phi_u(\omega)$, respectively. Then

$$\inf_{\omega} \underline{\sigma}(\Phi_u(\omega)) \le \lambda(R_p) \le \sup_{\omega} \overline{\sigma}(\Phi_u(\omega))$$
(5.E.26)

and

$$\left(\sup_{\omega} \overline{\sigma}(\Phi_u(\omega))\right)^{-1} \le \lambda(R_p^{-1}) \le \left(\inf_{\omega} \underline{\sigma}(\Phi_u(\omega))\right)^{-1}.$$
(5.E.27)

Proof: For any μ , R_p can be written as

$$R_{p} = \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_{p}(e^{j\omega}) \left(\mu I_{n} - (\mu I_{n} - \Phi_{u}(\omega))\right) \Gamma_{p}^{\star}(e^{j\omega}) \otimes I_{m} d\omega$$
$$= \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_{p}(e^{j\omega}) \mu I_{n} \Gamma_{p}^{\star}(e^{j\omega}) \otimes I_{m} d\omega -$$
$$- \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_{p}(e^{j\omega}) \left(\mu I_{n} - \Phi_{u}(\omega)\right) \Gamma_{p}^{\star}(e^{j\omega}) \otimes I_{m} d\omega$$

Now choose $\mu = \sup_{\omega} \overline{\sigma}(\Phi_u(\omega))$. Since $\Phi_u(\omega) \ge 0$, then $(\mu I_n - \Phi_u(\omega)) \ge 0$, and then

$$\frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_p(e^{j\omega}) \left(\mu I_n - \Phi_u(\omega)\right) \Gamma_p^{\star}(e^{j\omega}) \otimes I_m \,\mathrm{d}\omega \ge 0$$

so that we can write

$$R_{p} \leq \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_{p}(e^{j\omega}) \mu I_{n} \Gamma_{p}^{\star}(e^{j\omega}) \otimes I_{m} d\omega$$

$$= \sup_{\omega} \overline{\sigma}(\Phi_{u}(\omega)) \left(\frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_{p}(e^{j\omega}) \Gamma_{p}^{\star}(e^{j\omega}) d\omega\right) \otimes I_{m}$$

$$= \sup_{\omega} \overline{\sigma}(\Phi_{u}(\omega)) I_{mnp}$$

where in passing to the last line the following identity was used

$$\frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_p(e^{j\omega}) \Gamma_p^{\star}(e^{j\omega}) \,\mathrm{d}\omega = I_{np}$$

On the other hand, R_p can be written as

$$R_{p} = \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_{p}(e^{j\omega}) \left(\mu I_{n} + \left(\Phi_{u}(\omega) - \mu I_{n}\right)\right) \Gamma_{p}^{\star}(e^{j\omega}) \otimes I_{m} d\omega$$
$$= \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_{p}(e^{j\omega}) \mu I_{n} \Gamma_{p}^{\star}(e^{j\omega}) \otimes I_{m} d\omega +$$
$$+ \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_{p}(e^{j\omega}) \left(\Phi_{u}(\omega) - \mu I_{n}\right) \Gamma_{p}^{\star}(e^{j\omega}) \otimes I_{m} d\omega$$

Now choose $\mu = \inf_{\omega} \underline{\sigma}(\Phi_u(\omega))$. Since $\Phi_u(\omega) \ge 0$, then $(\Phi_u(\omega) - \mu I_n) \ge 0$, and then

$$\frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_p(e^{j\omega}) \left(\Phi_u(\omega) - \mu I_n \right) \Gamma_p^{\star}(e^{j\omega}) \otimes I_m \, \mathrm{d}\omega \ge 0$$

so that we can write

$$R_{p} \geq \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_{p}(e^{j\omega}) \mu I_{n} \Gamma_{p}^{\star}(e^{j\omega}) \otimes I_{m} d\omega$$

$$= \inf_{\omega} \underline{\sigma}(\Phi_{u}(\omega)) \left(\frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_{p}(e^{j\omega}) \Gamma_{p}^{\star}(e^{j\omega}) d\omega\right) \otimes I_{m}$$

$$= \inf_{\omega} \underline{\sigma}(\Phi_{u}(\omega)) I_{mnp}$$

Finally, we have

$$\inf_{\omega} \underline{\sigma}(\Phi_u(\omega)) \ I_{mnp} \le R_p \le \sup_{\omega} \overline{\sigma}(\Phi_u(\omega)) \ I_{mnp}$$
(5.E.28)

so that

$$\left(\sup_{\omega} \overline{\sigma}(\Phi_u(\omega))\right)^{-1} I_{mnp} \le R_p^{-1} \le \left(\inf_{\omega} \underline{\sigma}(\Phi_u(\omega))\right)^{-1} I_{mnp}$$
(5.E.29)

and the result follows straightforwardly.

Lemma 5.E.8. Let R_p be defined as in equation (5.55). Then

$$\left\|R_p^{-1}\right\|_2 = \left\|M_p^{-1}(\Phi_u, I_m)\right\|_2 \le \left(\inf_{\omega} \underline{\sigma}(\Phi_u(\omega))\right)^{-1}$$

Proof: The result follows directly from Lemma 5.E.7 considering that the 2-norm equals the maximum singular value.

Lemma 5.E.9. Let Δ_p be defined as

$$\Delta_p \triangleq \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_p(\omega) \Phi_u(\omega) \Gamma_p^{\star}(\omega) \otimes \widetilde{G}_p(\omega) \Phi_u(\omega) \widetilde{G}_p^{\star}(\omega) \,\mathrm{d}\omega$$

Then

$$\lim_{p \to \infty} \left\| \Delta_p \right\|_2 = 0$$

Proof:

$$\begin{split} \Delta_{p} \|_{2} &\leq \frac{1}{2\pi} \int_{-\pi}^{\pi} \left\| \Gamma_{p}(\omega) \Phi_{u}(\omega) \Gamma_{p}^{\star}(\omega) \otimes \widetilde{G}_{p}(\omega) \Phi_{u}(\omega) \widetilde{G}_{p}^{\star}(\omega) \right\|_{2} \, \mathrm{d}\omega \\ &= \frac{1}{2\pi} \int_{-\pi}^{\pi} \left\| \Gamma_{p}(\omega) \Phi_{u}(\omega) \Gamma_{p}^{\star}(\omega) \right\|_{2} \left\| \widetilde{G}_{p}(\omega) \Phi_{u}(\omega) \widetilde{G}_{p}^{\star}(\omega) \right\|_{2} \, \mathrm{d}\omega \\ &\leq \frac{1}{2\pi} \int_{-\pi}^{\pi} \left\| \Gamma_{p}(\omega) \right\|_{2}^{2} \left\| \Phi_{u}(\omega) \right\|_{2}^{2} \left\| \widetilde{G}_{p}(\omega) \right\|_{2}^{2} \, \mathrm{d}\omega \\ &\leq \frac{1}{2\pi} \int_{-\pi}^{\pi} \gamma_{p}(\omega) \left\| \Phi_{u}(\omega) \right\|_{2}^{2} \left\| \widetilde{G}_{p}(\omega) \right\|_{F}^{2} \, \mathrm{d}\omega \\ &\leq \left(\sup_{\omega} \overline{\sigma}(\Phi_{u}(\omega)) \right)^{2} \left(\sup_{\omega} \gamma_{p}(\omega) \right) \left\| \frac{1}{2\pi} \int_{-\pi}^{\pi} \left\| \widetilde{G}_{p}(\omega) \right\|_{F}^{2} \, \mathrm{d}\omega \\ &= \left(\sup_{\omega} \overline{\sigma}(\Phi_{u}(\omega)) \right)^{2} \left(\sup_{\omega} \gamma_{p}(\omega) \right) \left\| \widetilde{G}_{p}(\omega) \right\|_{H_{2}}^{2} \end{split}$$

Here, $\|\cdot\|_F$ stands for the Frobenius¹⁵ matrix norm, while $\|\cdot\|_{H_2}$ stands for the H_2 -norm¹⁶ induced by the inner product. Finally, the result follows under the assumption that $\|\widetilde{G}_p(\omega)\|_{H_2}^2$ decays to zero faster than $1/\gamma_p(\omega)$, so that

$$\lim_{p \to \infty} \left(\sup_{\omega} \gamma_p(\omega) \right) \left\| \widetilde{G}_p(\omega) \right\|_{H_2}^2 = 0.$$

Lemma 5.E.10. [NHG96] For $f,g \in L_2(\mathbb{T})$, let \widehat{f} and \widehat{g} be defined as

$$\widehat{f} \triangleq \sum_{k=0}^{p-1} \langle f, \mathcal{B}_k \rangle \mathcal{B}_k$$
$$\widehat{g} \triangleq \sum_{k=0}^{p-1} \langle g, \mathcal{B}_k \rangle \mathcal{B}_k$$

respectively. Then

$$\left|\sum_{k=0}^{p-1} \langle f, \mathcal{B}_k \rangle \overline{\langle g, \mathcal{B}_k \rangle} - \langle f, g \rangle \right| \leq \left\| f - \widehat{f} \right\|_{L_2} \left\| g - \widehat{g} \right\|_{L_2}.$$

Proof: See Lemma C.3 in [NHG96].

¹⁵The Frobenius norm of the matrix $A = (a_{ij})_{m \times n}$ is defined as $||A||_F^2 \triangleq \sum_{i=1}^m \sum_{j=1}^n |a_{ij}|^2$. ¹⁶The H_2 -norm induced by the inner product is given by $||G(\omega)||_{H_2}^2 \triangleq \langle G, G \rangle = \frac{1}{2\pi} \int_{-\pi}^{\pi} \text{Tr} \left\{ G(e^{j\omega}) G^*(e^{j\omega}) \right\} \, d\omega = \frac{1}{2\pi} \int_{-\pi}^{\pi} ||G(e^{j\omega})||_F^2 \, d\omega$.

Lemma 5.E.11. [NHG96] For $f \in L_2(\mathbb{T})$, let \widehat{h} be defined as

$$\widehat{h} \triangleq \sum_{k=0}^{p-1} \left\langle \mathcal{B}_m f, \mathcal{B}_k \right\rangle \mathcal{B}_k.$$

Then if f has a finite dimensional spectral factorization there exists $K<\infty, \ |\eta|<1$ such that for $m\leq p$

$$\left\| \mathcal{B}_m f - \hat{h} \right\|_{L_2} \le K(\eta^{p-m} + \eta^m)$$

Proof: See Lemma C.4 in [NHG96].

6

Adaptive Tracking Performance Analysis using Orthonormal Bases

A fundamental problem in control and signal processing is to track the timevarying parameters of a system or the properties of a signal as they vary with time. In recent years, considerable effort has been devoted to the derivation and analysis of a variety of adaptation algorithms for their use in recursive identification, adaptive filtering and control. In this chapter, a frequency domain analysis of the tracking performance of several adaptive algorithms for the recursive identification of time-varying linear systems will be carried out for the case in which the system is represented by a fixed denominator model structure. As done in the preceding chapters, to facilitate the analysis these model structures will be re-parameterized using the orthonormal bases with fixed poles introduced in Section 3.5. We concentrate on the study of the trade-off between disturbance rejection and tracking ability. This trade-off will be illustrated by showing how the quality of the adaptive estimation is influenced by such things as input and noise spectral densities, step size of the adaptive algorithms, and the choice of the fixed pole locations in the model structure.

6.1 Introduction

The derivation of a mathematical model for a system from observation data is the topic of System Identification. When dealing with time-invariant systems, the identification can be carried out off-line, which means that the parameter estimation is performed once the whole set of data is collected from the system. This cannot be done when the system properties vary with time. In this case the identification has to be performed on-line, which basically means that the estimate of the unknown parameters is updated each time a new data becomes available. This leads us to the field of Recursive Identification, where a consirable amount of research has been carried out in recent years (see for instance the books [You84, GS84, WS85, CG91]). In this context, identification algorithms have to adapt themselves to track the changes in the system, and for this reason they are called adaptive algorithms or (adaptive) tracking algorithms.

At this point, a question that arises is what characteristics of the algorithms should be considered to evaluate their performance. In general, one is interested in the following aspects of the algorithm

- Stability: exponential stability of the algorithm is necesary to guarantee boundedness of the tracking error, and is a pre-requisite for the practical applications and the analysis. The general conditions for stability of the algorithms we consider here have been established by Guo and Ljung in [GL95a], and therefore this issue is not studied in this chapter.
- Transient Response: the dominating dynamics governing the behaviour of the adjusted parameters when there is an abrupt change in the true system, or after an initialization.
- Steady-State Response: the variability of the estimated parameters with respect to the 'true' parameters, after a long period of time, when the system is assumed to be time-invariant (or slowly time-varying).

We concentrate in this chapter only in the analysis of the last two issues, viz. the transient response and the steady-state response of the algorithms.

A second question that arises is how to quantify the quality of the estimation. For instance, Guo and Ljung in [GL95b, GL95a] consider a general family of tracking algorithms, and quantify the quality of the estimation in the parameter space. The quantification is done in terms of the covariance matrix of the parameter tracking error (the so-called Mean Squares Error of the parameter estimate). Since the exact expression for this error is, in general, very complicated, the authors propose a simple expression that provides a good approximation when the adaptation rate of the algorithm is small. In this chapter, we use these results but, following the suggestions in [EJLW92, RJL96, Joh95, Joh93] about the utility of analysis of adaptive algorithms in the frequency domain, the quality of the estimate is analyzed here in terms of the covariance matrix of the frequency response estimation error rather than in terms of the covariance matrix of the parameter error. These frequency domain expressions are still too complicated to be of any practical utility. Drawing inspiration from the work in [GL89, LG90, LY85, Gun88] we derive more tractable expressions by considering high model orders.

Recent work in the areas of Adaptive Filtering [Wil95, Wil93a, WZ96] and Recursive Identification [GW90], suggesting the use of new model structures -as an alternative to the popular FIR structures- in adaptive algorithms has inspired us to use the fixed denominator model structures we consider in this chapter. Specifically, the fixed poles adaptive filters (FPAF) proposed in [Wil93a], the Gamma filters in [Pdd93], the adaptive IIR filters in [WZ96], and the recursively identified Laguerre models in [GW90], can all be regarded as fixed denominator model structures. All these model structures can be represented in a unified way by re-parameterizing them using the OBFP introduced in Chapter 3 (Section 3.5). As mentioned several times in this thesis, these fixed denominator model structures can be regarded as generalizations of the FIR structure, where the poles need not all be fixed at the origin. As pointed out in [Wil95, WZ96, GW90], the use of these structures can lead to many improvements in terms of estimation accuracy, and implementation and computational complexity, while still retaining the desirable convergence properties enjoyed by adaptive FIR schemes.

This chapter provides a frequency domain analysis of the tracking performance of some standard adaptation algorithms when used to track time-varying systems represented by these 'generalized FIR' structures. Both a transient, and a steady state analysis of the behavior of the frequency response estimation error are performed. The analysis shows that there is an important design compromise between tracking ability and noise sensitivity of the algorithms, and it makes explicit how this trade-off is influenced by input and noise spectral densities, choice of step size (or adaptation rate) of the algorithm, and (what is a main focus here) the choice of the fixed pole positions in the model structure.

The material in this chapter is closely related with the work of Gunnarsson and Ljung [GL89, LG90, Gun88] who studied adaptive FIR algorithms in the frequency domain. Specifically, we borrow from [GL89, Gun88] the main idea of this chapter which is to simplify error expressions by considering large model order, and small adaptation rate. The results of this chapter specialise to some of those in [GL89, Gun88] when all the poles of the orthonormal bases are chosen at the origin.

These ideas, together with the use of recent results by Guo and Ljung [GL95b] that provide approximations to the parameter covariance for a general class of adaptive algorithms under mild assumptions, constitute the base upon which our results are constructed. An important tool employed in the derivation of these results is the re-parameterization of the fixed denominator model structure into the orthonormal form studied in Chapter 3 (Section 3.5), in order to facilitate the theoretical analysis.

6.2 Motivation

A result derived by Gunnarsson and Ljung [GL89, Gun88] that has proved to be of great utility in the intuitive understanding and design [EJLW92, RJL96, Joh93, Joh95] of certain adaptive tracking schemes, is that the variability of a recursively computed p-th order FIR transfer function estimate $G(e^{j\omega}, \hat{\theta})$ may be approximated (for time invariant systems) by the simple expression

$$\operatorname{Var}\{G(e^{j\omega},\widehat{\theta})\} \approx p \frac{\mu \sigma_{\nu}^{2} \kappa}{[\Phi_{u}(\omega)]^{r}}$$
(6.1)

where κ is some constant, μ is the step size, σ_{ν}^2 is the white measurement noise variance, $\Phi_u(\omega)$ is the input excitation spectral density, and r = 1 for Recursive Least Squares (RLS), r = 1/2 for Kalman Filtering and r = 0 for the Least Mean Square (LMS) algorithm.

In assessing the utility of the idea of implementing adaptive filters with arbitrary fixed poles, a natural question arises as to how (6.1) should be modified from the FIR case so as to describe $\operatorname{Var}\{G(e^{j\omega}, \widehat{\theta})\}$. The most obvious course is to conclude that these new 'fixed denominator' model structures are really just the old FIR ones with an input $\{u_k\}$ pre-filtered by an all-pole filter F(q) = 1/D(q) where $D(q) = \prod_{n=0}^{p-1} (q - \xi_n)$ with $\{\xi_n\}$ being the user chosen guesses as to the true pole locations. This would imply that the variability of the FIR 'numerator' part is then given by the expression (6.1) with the substitution

$$\Phi_u(\omega) \mapsto |F(e^{j\omega})|^2 \Phi_u(e^{j\omega}) = \frac{\Phi_u(\omega)}{|D(e^{j\omega})|^2}$$
(6.2)

made. The frequency domain variability of the whole model structure, being the FIR numerator part divided by the frequency response of the fixed denominator part, should then be (6.1) with the substitution (6.2) and then divided by $|D(e^{j\omega})|^2$. Clearly the $|D(e^{j\omega})|^2$ terms will cancel, and the conclusion will ensue that the variability of $\operatorname{Var}\{G(e^{j\omega},\widehat{\theta})\}$ is invariant to the choice of fixed pole location.

This can be tested on a simple example wherein the true system is

$$G(q) = \frac{0.1548q + 0.0939}{(q - 0.6065)(q - 0.3679)}$$

and a p = 10'th order model is fitted using RLS when the input $\{u_k\}$ has spectral density $\Phi_u(\omega) = 10(1.25 - \cos \omega)^{-1}$, the output measurements $\{y_k\}$ are corrupted by white noise of variance $\sigma_{\nu}^2 = 0.01$, and the algorithm is run for N = 2000 data samples. In this case, the true variability can be estimated by the sample average over 200 Monte-Carlo simulations with different input and noise realizations. This can then be compared to the approximation (6.1). For the case of all the $\{\xi_n\}$ being at the origin (so that a true FIR structure is employed), then the results of such a comparison are shown in the top plot of figure 6.1 - the approximation (6.1) being the dash-dotted line, and the Monte-Carlo estimate of true variability being the solid line. The agreement is excellent.



Figure 6.1: Comparison of (solid line) true variability to (dash-dotted line) FIR based theoretical approximation (6.1). Top plot is case of all poles at origin (FIR case), bottom plot is case of all poles away from origin. The dashed line is the improved new approximation presented in this chapter. In the top plot, the dash-dotted line showing the pre-existing approximation (6.1) obscures the new approximation because the new and old approximations are identical for the FIR case.

However, if all the poles are chosen away from the origin, five at $\xi_k = 0.2$ and five at $\xi_k = 0.8$ then when examining the theoretical prediction (6.1) and the true variability as shown in the bottom plot of figure 6.1, the agreement between the two has disappeared.

Clearly, the previous heuristic reasoning that tried to adapt the FIR result (6.1) to a situation for which it is prima-facie applicable is flawed, and this indicates the need for custom analysis of fixed denominator model structure estimation.

Such is the purpose of this chapter, and pertaining to this note in the bottom diagram of figure 6.1 the good agreement between the true variability and the dashed line, which is a plot of the new 'extended' approximation presented in this chapter. This new approximation is the old one (6.1) with the model order term p replaced with a frequency dependent function $\gamma_p(\omega)$. However, in order to develop the ideas that lead to this and other enhanced approximations, a more formal problem definition is required.

6.3 **Problem Formulation**

We consider the (recursive) identification of linear slowly time-varying systems from observed input-output data. We assume that the underlying model relat-

ing the observed input $\{u_k\}$ and output $\{y_k\}$ sequences is given by

$$y_k = G_k(q)u_k + \nu_k \tag{6.3}$$

where $\{\nu_k\}$ is a zero mean white noise process with variance $\mathbf{E}\{\nu_k^2\} = \sigma_{\nu}^2 < \infty$ and where $G_k(q)$ represents the transfer function of the time-varying linear system at time k

$$G_k(q) = \sum_{n=0}^{\infty} g_k(n) q^{-n}$$

with square summable impulse response sequence $\{g_k(n)\} \in \ell_2$. In order for this representation to make sense we have of course to assume that the time variation is slow so that $G_k(q)$ can be given the usual interpretation of transfer function describing the input-output properties of the system.

It is assumed that the input signal $\{u_k\}$ is a bounded, deterministic, and quasi-stationary sequence in the sense of [Lju87], as summarized in Chapter 2 (page 6).

We also assume that the input sequence $\{u_k\}$ is weakly uncorrelated with the noise $\{\nu_k\}$ in the sense that $|\mathbf{E} \{u_k \nu_{k-\tau}\}| \to 0$ as $\tau \to \infty$, and that the input spectral density has a finite dimensional spectral factorization and it is positive definite $\Phi_u(\omega) > 0$ for almost all ω , what guarantees that the input signal is persistently exciting of any order [Reg95] (c.f. Proposition 2.1.1).

Our interest is on the (recursive) estimation of the (assumed unknown) time varying dynamics $G_k(q)$ by means of the observations of inputs $\{u_k\}$ and outputs $\{y_k\}$. Of the many approaches available in the literature to solve this estimation problem [SK95, GS84, Lju87, CG91], a common choice that facilitates the analysis of the adaptive algorithms, is to express the model (6.3) in linear regression form

$$y_k = \phi_k^T \theta_k + \nu_k \tag{6.4}$$

where the 'regression vector' ϕ_k depends in general on past outputs and inputs, and $\theta_k \in \mathbb{R}^p$ is a vector of p parameters in a model structure $G(q, \theta_k)$ that attempts to describe the true dynamics $G_k(q)$. An estimate of $G_k(q)$ can then be computed as $G(q, \hat{\theta}_k)$ where the parameter estimate $\hat{\theta}_k$ is obtained recursively. In this context, several well-known adaptive algorithms such as Least Mean Squares (LMS) (see [WS85] and the references therein), Recursive Least Squares (RLS) [GS84, You84], and the Kalman Filter (KF) [Kal60, KB61, Guo90] can be used for the estimation.

We restrict the analysis in this chapter to the three above mentioned adaptation algorithms, namely the LMS, RLS, and KF algorithms, which can be represented in a unified way by the general adaptation law [GL89]

$$\theta_{k+1} = \theta_k + L_k (y_k - \phi_k^T \theta_k), \tag{6.5}$$

where L_k is a gain vector that is computed in different ways, according to the particular algorithm. For instance, when the gain vector is given by

$$L_k = \mu \phi_k, \quad \mu \in (0, 1),$$
 (6.6)

then the update law (6.5) is the 'gradient-type' algorithm known as *Least Mean* Squares [WS85]. The case

$$L_k = P_k \phi_k \tag{6.7}$$

where P_k satisfies

$$P_{k} = \frac{1}{\lambda} \left\{ P_{k-1} - \frac{P_{k-1}\phi_{k}\phi_{k}^{T}P_{k-1}}{\lambda + \phi_{k}^{T}P_{k-1}\phi_{k}} \right\}; \quad \lambda \in (0, 1)$$
(6.8)

initialized with some positive definite matrix P_0 , corresponds to the *Recursive* Least Squares algorithm, where the constant λ is known as the 'forgetting factor'. Finally, the case

$$L_{k} = \frac{\mu P_{k-1} \phi_{k}}{\sigma^{2} + \mu \phi_{k}^{T} P_{k-1} \phi_{k}}$$
(6.9)

where P_k satisfies the Riccati equation

$$P_{k} = P_{k-1} - \mu \frac{P_{k-1}\phi_{k}\phi_{k}^{T}P_{k-1}}{\sigma^{2} + \mu\phi_{k}^{T}P_{k-1}\phi_{k}} + \mu\Sigma$$
(6.10)

with $\Sigma > 0$ and symmetric matrix, corresponds to the Kalman Filter algorithm. The KF gives an estimate of the parameter vector θ_k , when its time variation is modeled via a random walk as

$$\theta_{k+1} = \theta_k + \rho \, w_k \tag{6.11}$$

where w_k is a stationary zero mean vector white noise process with covariance matrix $\mathbf{E} \{w_k w_k^T\} = Q$. The estimate is optimal in a mean-square sense under Gaussian assumptions if $\mu = \rho, \sigma^2 = \sigma_{\nu}^2$ and $\Sigma = Q$. In the sequel it is assumed that $\{w_k\}$ is weakly uncorrelated with $\{u_k\}$ in the sense that $|\mathbf{E} \{u_k w_{k-\tau}\}| \to 0$ as $\tau \to \infty$.

The design variables of the three algorithms are the gain μ for the LMS algorithm, the forgetting factor λ for the RLS algorithm, and the matrices σ^2 and Σ , and the gain μ for the Kalman Filter.

A central question when using these algorithms is the accuracy of the estimate. The most common way to quantify this accuracy is to compute the so-called Mean Square Error (MSE) of the transfer function estimate

Mean Square Error
$$\triangleq \mathbf{E} \left\{ |G_k(q) - G(q, \widehat{\theta}_k)|^2 \right\},$$
 (6.12)

as a measure of how well the estimate $G(q, \hat{\theta}_k)$ approximates the 'true' transfer function $G_k(q)$. To do that, we need first to examine the accuracy of the parameter estimate $\hat{\theta}_k$ itself [SK95, GS84]. This may be achieved by defining the parameter estimation error $\tilde{\theta}_k$ as

$$\widetilde{\theta}_k \triangleq \theta_k - \widehat{\theta}_k. \tag{6.13}$$

where θ_k represents the true parameter vector that allows the model structure to exactly describe the underlying time varying dynamics as $G(q, \theta_k) = G_k(q)$. Of course this can only be done if the system belongs to the model set. In the general case, the model structure $G(q, \theta_k)$ will be too simple to exactly describe the true dynamics $G_k(q)$ for any value of θ_k , and so there will be no 'true' parameter θ_k . However, as pointed out in [GL89, LY85], we still can define θ_k as the 'best' approximation (for example in L_2 sense) of a given order, and use this value to calculate $\tilde{\theta}_k$. In this case, the true transfer function $G_k(q)$ can be expressed as

$$G_k(q) = G(q, \theta_k) + \Delta_p(q)$$

where $\Delta_p(q)$ represents the undermodelling error. As the model order increases to infinity this error tends to zero $(\Delta_p(q) \to 0 \text{ as } p \to \infty)$. It is reasonable then (following [GL89, LY85]) to perform an analysis asymptotic in model order. Since this analysis will involve deriving expressions that are asymptotic in p, there is no point in continually include the term $\Delta_p(q)$ (that tends to zero as p goes to infinity) throughout the derivations, and so we will ignore this term from the beginning. The validity of this strategy is confirmed by simulation example in Section 6.7. For the case in which $G_k(q)$ is represented using the orthonormal bases with fixed poles of Section 3.5, the rate at which $\Delta_p(e^{j\omega})$ tends to zero with increasing model order p is established via Theorem 4.3.1.

Substituting (6.13) into the general update equation (6.5), and considering that the time-variation of the parameters is given by the random walk (6.11), gives that the parameter error satisfies the following difference equation [GL95b]

$$\widetilde{\theta}_{k+1} = \left(I - L_k \phi_k^T\right) \widetilde{\theta}_k + \rho \,\omega_k - L_k \nu_k.$$
(6.14)

Equation (6.14) could now be used to calculate the covariance matrix of the parameter error $\mathbf{E}\left\{\widetilde{\theta}_{k}\widetilde{\theta}_{k}^{T}\right\}$ as a preliminary step for the computation of the MSE of the transfer function estimate, which ultimately is the magnitude we have chosen to measure the quality of the estimation. Unfortunately, as pointed out in [GL89, GL95b, Gun88, LG90], the exact expression for this covariance will be very complicated except in very special circumstances. To obtain simple expressions approximating this covariance matrix we resort to the main result of [GL95b] which establishes that, under certain assumptions on the regressors

 $\{\phi_k\}$ (the so-called ϕ -mixing condition), and on the noise ν_k and parameter drift ω_k , the covariance $\mathbf{E}\left\{\widetilde{\theta}_k\widetilde{\theta}_k^T\right\}$ may be approximated by the matrix Π_k given by the linear deterministic difference equation

$$\Pi_{k+1} = (I - \mu S_k R) \Pi_k (I - \mu S_k R)^T + \mu^2 \sigma_\nu^2 S_k R S_k + \rho^2 Q$$
(6.15)

where $R \triangleq \mathbf{E} \left\{ \phi_k \phi_k^T \right\}$ and S_k is defined as

LMS:

$$S_k = I, \tag{6.16}$$

RLS:

$$S_k = (1+\mu)S_{k-1} - \mu S_{k-1}RS_{k-1}; \quad S_0 = P_0,$$
(6.17)

Kalman Filter:

$$S_{k} = S_{k-1} - \mu S_{k-1} R S_{k-1} + \frac{\mu}{\sigma^{2}} \Sigma; \quad S_{0} = \frac{1}{\sigma^{2}} P_{0},$$
(6.18)

The degree of the approximation is quantified in [GL95b] by a result of the form¹

$$\left\| \mathbf{E} \left\{ \widetilde{\theta}_k \widetilde{\theta}_k^T \right\} - \Pi_k \right\| \le \kappa(\mu)$$

where $\kappa(\mu)$ is a bounded function that tends to zero as μ tends to zero, where μ is a measure of the step size of the algorithm.

However, as mentioned earlier in this section, our interest is not in the quantification of the accuracy in the parameter space, but in quantifying how close the estimated model $G(q, \hat{\theta}_k)$ is to the true system $G_k(q)$ in terms of the error

$$\widetilde{G}_k(e^{\mathbf{j}\omega}) \triangleq G(e^{\mathbf{j}\omega},\widehat{\theta}_k) - G_k(e^{\mathbf{j}\omega})$$

in the estimated frequency response, and ultimately in terms of the MSE of the transfer function estimate (6.12).

In order to be able to compute this frequency response estimation error, we need to relate it with the error in the parameter space. Since we are assuming a linear regression model for the system, we are restricted to considering model structures $G(q, \theta_k)$ for which the estimated frequency response depends linearly on the estimated parameters, i.e. of the form

$$G(e^{j\omega}, \widehat{\theta}_k) = \Gamma_p^T(e^{j\omega})\widehat{\theta}_k, \qquad (6.19)$$

¹Here $\|\cdot\|$ stands for the 2-norm of matrices or operator norm

where

$$\Gamma_p(q) \triangleq [\mathcal{B}_0(q), \mathcal{B}_1(q), \cdots, \mathcal{B}_{p-1}(q)]^T$$
(6.20)

is a vector of p rational-stable-causal transfer functions $\mathcal{B}_n(q)$. For example, $\mathcal{B}_n(q) = q^{-n}$ corresponds to an FIR model structure.

Using equations (6.15), (6.19) and (6.20), an approximate expression for the Mean Square Error (MSE) of the transfer function estimate can then be computed as

$$\mathbf{E}\left\{|\widetilde{G}_{k}(e^{j\omega})|^{2}\right\} = \Gamma_{p}^{\star}(e^{j\omega})\mathbf{E}\left\{\widetilde{\theta}_{k}\widetilde{\theta}_{k}^{T}\right\}\Gamma_{p}(e^{j\omega}) \approx \Gamma_{p}^{\star}(e^{j\omega})\Pi_{k}\Gamma_{p}(e^{j\omega})$$
(6.21)

This MSE of the transfer function estimate can then be used as a frequency domain quantification of the performance of the adaptive algorithms. Unfortunately again, this expression will in general be too complicated to give any practical insight. Following the ideas in [GL89, LY85, Gun88], we will derive simple approximations for (6.21) that are increasingly accurate for increasing model order p. Through simulation experiments we will show that the simplified expressions are good approximations even for relatively small model order.

To be able to proceed with this asymptotic analysis we need to be more explicit about the expression for $\Gamma_p(e^{j\omega})$ in (6.21). That is, to be more specific on the exact formulation of the transfer functions $\{\mathcal{B}_n(q)\}$ determining the model structure $G_k(q, \theta)$.

6.4 Model Structures

In recent years, in an adaptive filtering context, Williamson and co-workers have proposed and studied, in a series of papers [Wil95, Wil93b, Wil93a, WZ96], a new class of Infinite Impulse Response (IIR) adaptive filters that have been termed 'Fixed Pole Adaptive Filters' (FPAF), or more generically 'Vector Space Adaptive Filters' (VSAF). For the case of real poles, these adaptive filters can be formulated as

$$G(q, \theta'_k) = \sum_{n=0}^{p-1} \theta'_k(n) \prod_{i=0}^n \frac{1}{(q-\xi_i)} = \sum_{n=0}^{p-1} \theta'_k(n) \mathcal{A}_n(q),$$
(6.22)

where the 'fixed poles' $\{\xi_n\}_{n=0}^{p-1}$ are chosen to reflect the available prior information about the likely pole positions of the true time varying system $G_k(q)$, and where $\theta'_k(n)$ are adjustable filter parameters. In passing to the last equality we have defined

$$\mathcal{A}_n(q) \triangleq \prod_{i=0}^n \frac{1}{(q-\xi_i)}.$$

To contemplate the case of complex conjugate poles, the formulation is modified to

$$G(q,\theta'_k) = \sum_{n=0}^{p-1} \left(\theta'_k(n)q + \theta''_k(n)\right) \prod_{i=0}^n \frac{1}{(q-\xi_i)(q-\overline{\xi_i})}.$$
(6.23)

A special case of the structure (6.22) arises when all the poles $\{\xi_n\}$ are chosen at the origin in which case (6.22) is an FIR model structure.

Although IIR filters have several advantages over FIR filters when used to model very long impulse responses, FIR models have been preferred almost exclusively for the adaptive applications [WS85], mainly due to the fact that the global convergence of the adaptive algorithms can be ensured, in contrast to IIR filters for which theoretical guarantee of global convergence is difficult to provide due to the possibility of existence of multimodal error surfaces² [WS85, Reg95].

Williamson and co-workers [WZ96] show that the FPAF structure (6.22)-(6.23), though being an IIR model structure, preserves the global convergence characteristics of FIR filters. They also provide some simulation results showing that significant improvements in the estimation accuracy can be achieved if the prior information about the dominant modes of the system is used to chose the fixed poles in the model structure close to these dominant poles. Later in this chapter, we will give some theoretical support to this observations by resorting to the results on the undermodelling error for the OBFP we presented in Theorem 4.3.1.

In spite of the above mentioned advantages of the model structure (6.22), its generality (as compared to an FIR structure) makes the frequency domain analysis of the adaptive algorithms much more difficult. To overcome this difficulty and make the analysis more tractable, we replace the model structure (6.22) with an orthonormal re-parameterization of the form

$$G(q,\theta_k) = \sum_{n=0}^{p-1} \theta_k(n) \mathcal{B}_n(q)$$
(6.24)

where $\{\mathcal{B}_n(q)\}\$ are the Orthonormal Bases with Fixed Poles introduced in Section 3.5, whose formulation we rewrite here for convenience,

$$\mathcal{B}_n(q) = \left(\frac{\sqrt{1-|\xi_n|^2}}{q-\xi_n}\right) \prod_{k=0}^{n-1} \left(\frac{1-\overline{\xi_k}q}{q-\xi_k}\right).$$
(6.25)

As shown for the time-invariant case in Chapter 4, the model structure

²This basically means that the surface describing the mean square tracking error in the (adaptive) parameter space may have multiple local minima.

(6.24) can be written in the linear regression form (6.4) by defining

$$\phi_k \triangleq \Gamma_p^T(q)u_k, \tag{6.26}$$

$$\Gamma_p(q) \triangleq [\mathcal{B}_0(q), \mathcal{B}_1(q), \cdots, \mathcal{B}_{p-1}(q)]^T, \qquad (6.27)$$

$$\theta_k \triangleq [\theta_k(0), \theta_k(1), \cdots, \theta_k(p-1)]^T.$$
(6.28)

Before going further with the analysis, a natural question we should answer is the following: How do the estimates with the orthonormal re-parameterization (6.24) relate to the estimates with the original model structure (6.22), when using the different adaptive algorithms?

To answer this question, let us notice first that since the poles of the model structures (6.24) and (6.22) are identical, we can always find a nonsingular (constant) matrix $J \in \mathbb{R}^{p \times p}$ such that the matrices $\Gamma_p(q)$ and $\Gamma'_p(q)$ associated with each parameterization are related according to

$$\Gamma_p(q) = J^T \Gamma'_p(q), \tag{6.29}$$

with $\Gamma_p(q)$ defined in (6.27), and where $\Gamma'_p(q)$ is defined analogously as

$$\Gamma'_p(q) \triangleq [\mathcal{A}_0(q), \mathcal{A}_1(q), \cdots, \mathcal{A}_{p-1}(q)]^T$$

It is then clear that the regressor vectors ϕ_k and ϕ'_k associated respectively to the model structures (6.24) and (6.22), are related according to

$$\phi_k = J^T \phi'_k \tag{6.30}$$

Based on this observations, we are now able to prove the following result showing that the estimations using RLS and KF algorithms are invariant under the linear re-parameterization (6.29), while the estimation using LMS algorithm is not.

Lemma 6.4.1. The RLS and KF algorithms are invariant under the linear re-parameterization (6.29), which means that the estimates corresponding to the model structures (6.24) and (6.22) are related according to

$$\widehat{\theta}'_k = J\widehat{\theta}_k \tag{6.31}$$

provided that the initialization is consistent with the linear re-parameterization, i.e.

$$P_0' = J P_0 J^T$$

for the RLS algorithm, and

$$\Sigma' = J\Sigma J^T$$

for the KF. On the other hand, the LMS algorithm is not invariant under linear re-parameterization and consequently equation (6.31) doesn't hold for this case.

Proof: See Appendix 6.A.

It is clear now that, since the RLS and KF algorithms are frequency domain invariant to linear model re-parameterization, the analysis using the orthonormal structure will provide results that are valid for any fixed denominator model structure such as (6.22). Unfortunately, the LMS algorithm is not invariant under linear re-parameterisations [EJLW92, Wil95], and so any subsequent results for this algorithm will only pertain to the orthonormal model structure (6.24). This is not considered a significant limitation, since for the LMS algorithm the orthonormal structure (6.24) is attractive from the viewpoint of numerical robustness and enhanced convergence rate under white noise excitation.

6.5 Transient analysis

In this section, we study the transient behavior of the frequency response estimation error for the Least Mean Squares algorithm.

The dynamic behaviour of the frequency response estimation error can be studied by using the approximation (6.21) substituted into (6.15) to obtain

$$\mathbf{E}\left\{|\widetilde{G}_{k+1}(e^{j\omega})|^{2}\right\} \approx \Gamma_{p}^{\star}(e^{j\omega})(I-\mu S_{k}R)\Pi_{k}(I-\mu S_{k}R)^{T}\Gamma_{p}(e^{j\omega}) + \mu^{2}\sigma_{\nu}^{2}\Gamma_{p}^{\star}(e^{j\omega})S_{k}RS_{k}\Gamma_{p}(e^{j\omega}) + \rho^{2}\Gamma_{n}^{\star}(e^{j\omega})Q\Gamma_{p}(e^{j\omega}).$$
(6.32)

As the reader can see, this is a very complicated expression from which it is difficult to extract useful design insights. However, if the model order p is assumed to be large, then equation (6.32) can be simplified considerably.

Here again, an important rôle in the analysis will be played by the reproducing kernel $\gamma_p(\omega) \triangleq K_p(\omega, \omega)$ associated to the orthonormal bases with fixed poles $\{\mathcal{B}_n(q)\}$ in (6.25). We recall that $\gamma_p(\omega)$ can be computed as

$$\gamma_p(\omega) = \sum_{n=0}^{p-1} |\mathcal{B}_n(e^{j\omega})|^2.$$
 (6.33)

For the case of FIR model structures, it is well known that the error in the estimated models induced by the measurement noise is inversely proportional to the model order p. This has been rigorously proved by Ljung and Yuan in [LY85] for off-line identification in the time-invariant case, and by Gunnarsson and Ljung in [GL89] for the case of recursive identification. The frequency dependent quantity $\gamma_p(\omega)$ will serve to capture how this phenomenon generalizes to the fixed-denominator model structures (6.22),(6.24). As pointed out in Chapter 4, for the case $\xi_n = 0$ that corresponds to (6.24) being an FIR model structure, the factor $\gamma_p(\omega)$ equals p.

It is also necessary to define a quantity which reflects the time varying nature of the system $G_k(q)$ in the frequency domain, and is also commensurate with the parameter space model for this time variation (6.11). Following [GL89], this may be achieved by using the linear relationship (6.19) in combination with (6.11) to obtain the model for the time variation as

$$G_{k+1}(e^{j\omega}) = G_k(e^{j\omega}) + \rho \Gamma_n^T(e^{j\omega}) w_k$$

so that

160

$$\mathsf{E}\left\{|G_{k+1}(e^{j\omega}) - G_k(e^{j\omega})|^2\right\} = \rho^2 \Gamma_p^{\star}(e^{j\omega}) Q \Gamma_p(e^{j\omega}) = \rho^2 \delta_p(\omega).$$
(6.34)

where we have defined

$$\delta_p(\omega) \triangleq \Gamma_p^{\star}(e^{j\omega}) Q \Gamma_p(e^{j\omega}). \tag{6.35}$$

In the sequel, simplifications for equation (6.32) will be obtained by considering increasing model order p. To facilitate this analysis it is necessary (as in [GL89]) to assume that as p grows, the covariance matrix Q is extended accordingly in such a way that it is always positive definite and of bounded norm. In this case, the limit

$$\delta(\omega) \triangleq \lim_{p \to \infty} \frac{\delta_p(\omega)}{\gamma_p(\omega)}$$
(6.36)

exists and is non-zero. Given these definitions, the following theorem provides a simple frequency domain characterisation of the tracking characteristics of the LMS adaptation scheme when using the general fixed denominator model structure (6.24).

Theorem 6.5.1. For the LMS algorithm and the model structure (6.24), then using the approximation (6.15) for the covariance matrix of the parameter estimation error, the following limit result can be established

$$\lim_{p \to \infty} \left| \frac{1}{\gamma_p(\omega)} \mathbf{E} \left\{ |\widetilde{G}_{k+1}(e^{j\omega})|^2 \right\} - \left(\frac{[1 - \mu \Phi_u(\omega)]^2}{\gamma_p(\omega)} \mathbf{E} \left\{ |\widetilde{G}_k(e^{j\omega})|^2 \right\} + \mu^2 \sigma_\nu^2 \Phi_u(\omega) + \rho^2 \delta(\omega) \right) \right| = 0$$

Proof: See Appendix 6.A.

The interpretation of this result is that for large model order p, the equation governing the behaviour of the MSE of the transfer function estimate for the

LMS algorithm and fixed denominator model structures can be approximated by the following first order difference equation

$$\mathsf{E}\left\{|\widetilde{G}_{k+1}(e^{j\omega})|^{2}\right\} \approx \approx [1 - \mu \Phi_{u}(\omega)]^{2} \mathsf{E}\left\{|\widetilde{G}_{k}(e^{j\omega})|^{2}\right\} + \mu^{2} \sigma_{\nu}^{2} \Phi_{u}(\omega) \gamma_{p}(\omega) + \rho^{2} \delta(\omega) \gamma_{p}(\omega).$$

$$(6.37)$$

The transient response is then governed by the homogeneous part of this difference equation. That is

$$\mathbf{E}\left\{|\widetilde{G}_{k+1}(e^{\mathbf{j}\omega})|^2\right\} \approx [1 - \mu \Phi_u(\omega)]^2 \mathbf{E}\left\{|\widetilde{G}_k(e^{\mathbf{j}\omega})|^2\right\},\tag{6.38}$$

which for small μ can be approximated by

$$\mathbf{E}\left\{|\widetilde{G}_{k+1}(e^{j\omega})|^2\right\} \approx [1 - 2\mu\Phi_u(\omega)]\mathbf{E}\left\{|\widetilde{G}_k(e^{j\omega})|^2\right\},\tag{6.39}$$

and it is then characterized by the (frequency dependent) time constant for the decay of the MSE

$$\tau(\omega) \triangleq [1 - 2\mu \Phi_u(\omega)].$$

From equation (6.39) it is clear that the condition on the stepsize μ for stability of the algorithm for all ω is

$$\mu < \frac{1}{\sup_{\omega} \Phi_u(\omega)}.$$
(6.40)

These results are similar to the ones presented in [GL89, Gun88] for the FIR case. A substantial difference between the MSE expression (6.37) and those in [GL89, Gun88, EJLW92], is the presence of the factor $\gamma_p(\omega)$ in the driving terms, making explicit how the tracking error is affected by the choice of poles in the model structure (6.24).

Similarly to the results in [GL89, Gun88], equation (6.37) shows the design trade-off (for the LMS algorithm) between the tracking ability and the noise rejection properties of the algorithm. To be more specific, from (6.37) it can be seen that the tracking error decays like $[1 - \mu \Phi_u(\omega)]^{2k} \approx [1 - 2\mu \Phi_u(\omega)]^k$, so that tracking would be better at frequencies where $\mu \Phi_u(\omega)$ is large, but at these frequencies the 'noise driving' term $\sigma_{\nu}^2 \mu^2 \Phi_u(\omega) \gamma_p(\omega)$ would also be large. However, unlike the expressions in [GL89, Gun88], the presence of the modulation factor $\gamma_p(\omega)$ in the driving terms in (6.37) suggests that the tracking performance could be improved, but not necessarily at the expense of a deterioration in the noise rejection properties of the algorithm, by an appropriate choice of the poles in the model structure.

In order to be able to apply these results –which are asymptotic in model order– in practical problems we need to evaluate the accuracy of the approximation (6.37) for the low model orders likely to be used in these cases. This is examined in Section 6.7 where for small enough step size μ , (6.37) is shown to faithfully predict the frequency domain convergence properties of the LMS algorithm – see figure 6.7.

6.6 Steady State Analysis

In this section we are interested in the quantification of the steady state behavior of the frequency response estimation error. That is, we are interested in the error $\mathbf{E}\left\{|\tilde{G}_k(e^{j\omega})|^2\right\}$ for large values of k. In order to compute this steady-state error we need first to analyze the limiting behavior of the solution Π_k of (6.15) for large k; that is

$$\lim_{k\to\infty}\Pi_k\triangleq\Pi.$$

Of course, for this limit to exist (and indeed for the approximation (6.32) to hold) it is necessary that the adaptive algorithm (6.5) be stable. As mentioned in the introduction to this chapter, sufficient conditions for the exponential stability of the general adaptation algorithm (6.5), for any of the choices (6.6),(6.7)-(6.8), or (6.9)-(6.10) have been established by Guo and Ljung in [GL95a]. These conditions are very technical but can be summarized as follows:

- Persistency of excitation: the sum of the covariance matrices of the regressors E {φ_kφ_h^T} over a finite time span of arbitrary length has full rank. In general this condition is satisfied if the input spectral density Φ_u(ω) is positive definite for almost all frequencies ω. That is, if the input sequence is persistently exciting of any order.
- Weak dependance of the regressors: the dependance between the regressors φ_k and {φ_i, ν_{i-1}, w_i} decays to zero as the time distance (k i) tends to infinity. This is the so-called 'φ-mixing condition' [GL95a, GL95b].
- The measurement noise $\{\nu_k\}$ and the parameter drift $\{w_k\}$ are zero mean white noise sequences.

Once the stability of the algorithm has been ensured, the steady-state parameter error Π may be evaluated by determining the steady-state solutions

$$S \triangleq \lim_{k \to \infty} S_k$$

of (6.16)–(6.18) and then substituting them into (6.32) before examining its own steady state solution with terms of order $\mu^2 \Pi$ discarded [GL95b]. The results of this strategy for the three adaptive algorithms are as follows.

LMS: Here S = I so that Π is the solution of the Lyapunov equation

$$\Pi R + R\Pi = \mu \sigma_{\nu}^{2} R + \frac{\rho^{2}}{\mu} Q.$$
 (6.41)

RLS: Here $S = R^{-1}$ so that Π is given by

$$\Pi = \frac{\mu \sigma_{\nu}^2}{2} R^{-1} + \frac{\rho^2}{2\mu} Q.$$
 (6.42)

Kalman Filter: This case is more difficult. S is the solution of

$$\sigma^2 SRS = \Sigma,$$

which may be expressed as

$$S = \frac{1}{\sqrt{\sigma^2}} R^{-1/2} \left(R^{1/2} \Sigma R^{1/2} \right)^{1/2} R^{-1/2}, \tag{6.43}$$

so that Π is given by the solution of

$$SR\Pi + \Pi RS = \frac{\mu \sigma_{\nu}^2}{\sigma^2} \Sigma + \frac{\rho^2}{\mu} Q.$$
 (6.44)

For the special case of $\Sigma = Q$ this system has solution

$$\Pi = \frac{\sigma^2}{2} \left(\mu \frac{\sigma_{\nu}^2}{\sigma^2} + \frac{\rho^2}{\mu} \right) S.$$
(6.45)

with S given in (6.43).

Based on these expressions for the steady-state parameter error Π together with equation (6.21) it is now possible to quantify the steady state estimation error in the frequency domain as

$$\mathbf{E}\left\{|\widetilde{G}(e^{j\omega})|^{2}\right\} \triangleq \lim_{k \to \infty} \mathbf{E}\left\{|\widetilde{G}_{k}(e^{j\omega})|^{2}\right\} \approx \Gamma_{p}^{\star}(e^{j\omega})\Pi\Gamma_{p}(e^{j\omega}).$$
(6.46)

However, the resulting expression is so complicated that it is difficult to extract useful design insight from it. As in the previous section, we will perform an asymptotic analysis with increasing model order p in order to simplify (6.46). The resulting expressions will be more tractable, and will provide some insight on how the steady-state estimation error is affected by factors such as the step size, the measurement noise energy, the input spectral density, and perhaps most interestingly, the choice of fixed pole position. The effect of this choice of poles will be quantified by the term $\gamma_p(\omega)$.

The simplified error equations for the LMS, RLS and KF algorithms are presented in Theorems 6.6.1, 6.6.2 and 6.6.3 respectively.

• LMS Algorithm: In this case, the simplified error quantification is as follows.

Theorem 6.6.1. For the LMS algorithm and the model structure (6.24), the following limit result holds,

$$\lim_{p \to \infty} \frac{1}{\gamma_p(\omega)} \mathbf{E} \left\{ |\tilde{G}(e^{\mathbf{j}\omega})|^2 \right\} = \frac{1}{2} \left[\mu \sigma_{\nu}^2 + \frac{\rho^2 \delta(\omega)}{\mu \Phi_u(\omega)} \right]$$

Proof: See Appendix 6.A.

The interpretation of this theorem is that for large model order p, and after the algorithm has converged (large k), the steady-state MSE can be approximated by

$$\mathbf{E}\left\{|\widetilde{G}_{k}(e^{j\omega})|^{2}\right\} \approx \frac{\gamma_{p}(\omega)}{2} \left[\mu\sigma_{\nu}^{2} + \frac{\rho^{2}\delta(\omega)}{\mu\Phi_{u}(\omega)}\right].$$
(6.47)

For the case of all the poles $\{\xi_n\}$ in the model structure (6.24) chosen at the origin, then $\gamma_p(\omega) = p$ and the above expression specializes to that derived in [GL89, Gun88] (although in [GL89], only the case $\rho = 1$ is considered). However, an important difference with the results for the FIR case in [GL89, Gun88], is the presence of the frequency dependent factor $\gamma_p(\omega)$ in the expression of the MSE (6.47) showing how pole choices other than FIR influence the frequency domain estimation error. As well, (6.47) illustrates that for time-invariant systems (corresponding to $\rho = 0$), then the error is proportional to the step size μ and the measurement noise variance σ_{ν}^2 , while for time varying systems ($\rho \neq 0$), another error component arises due to the parameter drift, which is inversely proportional to step size, and is also inversely proportional to input spectral density $\Phi_u(\omega)$. In this latter case, we have a compromise in the choice of the step size μ , since a small value of μ will reduce the contribution of the measurement noise to the MSE, but simultaneously will increase the contribution of the parameter drift to that error.

A fundamental question that now arises is one concerning the reliability of using the approximation (6.47) -that has been derived considering large model order- for the relatively small model orders used in practical applications. The most suitable way to deal with this issue would be to quantify the convergence rate in Theorem 6.6.1. This appears to be extremely difficult. Instead, the approach used in [GL89] is taken wherein the validity of (6.47) for finite p is examined via a simulation study. This is done in Section 6.7, where it is shown (see Figures 6.3 and 6.5) that for a tenth order model, (6.47) is quite an accurate approximation; in fact, as shown in Figure 6.4, this holds even for as low as a fourth order model. • **RLS Algorithm:** The same sort of analysis can also be performed in this case, with the results as follows.

Theorem 6.6.2. For the RLS algorithm and the model structure (6.22) or (6.24), the following limit result holds

$$\lim_{p \to \infty} \frac{1}{\gamma_p(\omega)} \mathbf{E} \left\{ |\tilde{G}(e^{j\omega})|^2 \right\} = \frac{1}{2} \left[\frac{\mu \sigma_{\nu}^2}{\Phi_u(\omega)} + \frac{\rho^2}{\mu} \delta(\omega) \right] \quad ; \mu = 1 - \lambda$$

Proof: See Appendix 6.A.

A similar interpretation to the previous theorem can be given also in this case. Namely, this theorem means that for large model order p, and after the algorithm has converged (large k), the steady state MSE can be approximated by

$$\mathbf{E}\left\{|\widetilde{G}_{k}(e^{j\omega})|^{2}\right\} \approx \frac{\gamma_{p}(\omega)}{2} \left[\frac{\mu\sigma_{\nu}^{2}}{\Phi_{u}(\omega)} + \frac{\rho^{2}}{\mu}\delta(\omega)\right].$$
(6.48)

Comparing this approximation to (6.47) illustrates a fundamental difference between the steady state behavior of LMS and RLS in terms of how the input spectral density affects the noise and tracking performance. Specifically, (6.48) shows that for stationary systems ($\rho = 0$), the RLS estimation error is inversely proportional to the input spectral density $\Phi_u(\omega)$, while (6.47) shows that the LMS estimation error is invariant to the size of this spectral density; see Figure 6.5 for a simulation study validation of this phenomenon. Conversely when $\rho \neq 0$, (6.47) shows that for LMS the tracking ability increases with increasing input spectral density, while for RLS the tracking ability is invariant to this factor, and only depends on step size.

These observations, in the context of FIR model structures have already been made in [GL89, Gun88], and as per the LMS case, when all the poles are chosen at the origin and hence $\gamma_p(\omega) = p$, then (6.48) is identical to the expressions for RLS steady state error presented in [GL89, Gun88]. However, again as per the LMS case, the inclusion of the frequency dependent factor $\gamma_p(\omega)$ in (6.48) shows how the choice of fixed denominator pole position in the model structure $G(q, \theta)$ affects the estimation error in the frequency domain (Figure 4.2).

The question of the validity of using an asymptotic result as a finite data and model order approximation in (6.48) again arises, and again this is dealt with in Section 6.7 via a simulation study. For example in Figure 6.4, (6.48) is shown to be quite accurate even for only a fourth order model and 300 data points.

 Kalman Filter Algorithm: The same strategy of considering large model order can be used in this case. However, as already mentioned, there are particular difficulties in solving for the steady state parameter covariance Π, and this leads to the treatment of only a specialised case in which Σ = Q.

Theorem 6.6.3. For the Kalman Filter algorithm, the model structure (6.22) or (6.24) and under the assumption that $\Sigma = Q$, then

$$\lim_{p \to \infty} \frac{1}{\gamma_p(\omega)} \mathbf{E} \left\{ |\widetilde{G}(e^{j\omega})|^2 \right\} = \frac{1}{2} \left(\mu \frac{\sigma_{\nu}^2}{\sigma^2} + \frac{\rho^2}{\mu} \right) \sqrt{\frac{\sigma^2 \delta(\omega)}{\Phi_u(\omega)}}$$

Proof: See Appendix 6.A.

Similarly to the previous results, the interpretation of this theorem is that for large model order p and after the algorithm has converged (large k) then the MSE can be approximated by

$$\mathbf{E}\left\{|\widetilde{G}_{k}(e^{\mathbf{j}\omega})|^{2}\right\} \approx \frac{\gamma_{p}(\omega)}{2} \left(\mu \frac{\sigma_{\nu}^{2}}{\sigma^{2}} + \frac{\rho^{2}}{\mu}\right) \sqrt{\frac{\sigma^{2}\delta(\omega)}{\Phi_{u}(\omega)}}.$$
(6.49)

In terms of how input spectral density affects noise sensitivity and tracking ability, (6.49) shows that Kalman Filter based algorithms sit between the LMS and RLS algorithms in that instead of being affected separately, both noise and tracking performance are affected equally (but to a lesser extent due to the $\sqrt{\cdot}$ operation-see Figure 6.5) by the size of the input spectral density $\Phi_u(\omega)$. Again, the accuracy of the approximation (6.49) is validated experimentally in Section 6.7 to show that in fact it is meaningful for low model orders; see Figure 6.4.

Note that the approximations (6.47), (6.48) and (6.49) can also be used to calculate the optimal step size μ^{opt} which will minimise the MSE at a particular frequency, with the results being

$$\mu^{\mathsf{opt}} = \frac{\rho}{\sigma_{\nu}} \sqrt{\frac{\delta(\omega)}{\Phi_u(\omega)}}$$

for the LMS and RLS algorithms, and

$$\mu^{\mathsf{opt}} = \frac{\rho\sigma}{\sigma_{\nu}}$$

for the Kalman Filter algorithm. It is clear from these expressions that for the LMS, and RLS algorithms a value μ^{opt} that minimizes the MSE for all

frequencies can not be obtained. On the other hand, for the Kalman Filter case the value of μ^{opt} is independent of frequency and so it will minimize the MSE at all frequencies. For all of these cases, the minimum MSE is

$$\mathbf{E}\left\{|\widetilde{G}_k(e^{j\omega})|^2\right\}_{\min} = \rho \,\sigma_\nu \gamma_p(\omega) \sqrt{\frac{\delta(\omega)}{\Phi_u(\omega)}}.$$

The presence of the term $\gamma_p(\omega)$ in all these error quantifications shows that the orthonormal parameterisation (6.24), (6.25) is more than just an essential tool for the analysis of general fixed denominator model structures. Instead, the orthonormal 'basis functions' $\{\mathcal{B}_n(q)\}$ appear as an intrinsic part of adaptive estimation with any fixed denominator model structure $G(q, \theta)$.

For example, for RLS and Kalman Filtering schemes, then in steady-state whether or not the fixed denominator model structure is ab-initio cast in the orthonormal form (6.24), the complete contribution of the fixed pole choice to the frequency domain error properties is captured by the term $\gamma_p(\omega)$ which via (6.33) is itself completely described by the orthonormal basis $\{\mathcal{B}_n(q)\}$.

In other words, for any model structure with fixed denominator such as (6.22) and (6.24), the RLS and Kalman Filter frequency domain error quantification depends only on the location of the poles $\{\xi_n\}$ and is quantified via the factor $\gamma_p(\omega)$ associated with the orthonormal basis functions $\{\mathcal{B}_n(q)\}$.

6.7 Simulation Examples

In this section, the utility of the previous theoretical analysis will be demonstrated via several simulation studies. In all cases, it is assumed that there is an underlying continuous time system with transfer function

$$G(s) = \frac{1}{(s+1)(10s+1)}$$

from which input-output data is collected by sampling every one second. We begin the study by considering the case of stationary systems, but later on, time variations away from G(s) will also be analyzed. It is assumed that the input $\{u_k\}$ is stationary and Gaussian with spectral density

$$\Phi_u(\omega) = \frac{10}{1.25 - \cos\omega}$$

and that the observed output $\{y_k\}$ is corrupted by a white Gaussian noise sequence $\{\nu_k\}$ of variance $\sigma_{\nu}^2 = 0.01$. Based on this observed data, the identification objective is to estimate the zero order hold equivalent [ÅW84] discrete time system

$$G(q) = \text{ZOH}\left\{\frac{1}{(s+1)(10s+1)}\right\} = \frac{0.0355q + 0.0247}{(q-0.9048)(q-0.3679)}$$
(6.50)

via the model structure (6.24) with poles $\{\xi_n\}$ chosen to correspond to continuous time guesses of 0.2 and 0.25 radians per second. Note that these poles, being far from either of the true poles at 0.1 and 1 rad/s, are particularly bad guesses. They have been chosen to dispel any suspicion in the sequel that the high accuracy of the approximations (6.47), (6.48) and (6.49) illustrated in Figures 6.3 to 6.7 derives from unreasonable prior knowledge or idealized conditions.

All three algorithms, the LMS with $\mu = 0.001$, RLS with $\lambda = 0.999$ and $P_0 = I$, and the Kalman Filter with $\mu = 0.001$, $\Sigma = 0.1$, $P_0 = I$ and $\sigma^2 = 0.01$ were employed with a tenth order model structure (p = 10). The parameter space convergence results are shown in the plots of Figure 6.2, the fast convergence illustrating that these examples do not represent a case of unreasonably slow adaptation. Again, this choice is made to illustrate the robustness of the theoretical analysis to the violation of certain assumptions (small μ) that it is performed under. These estimation experiments were performed five hundred



Figure 6.2: Parameter space convergence for LMS, RLS and Kalman Filter.

times with different realizations for the input and measurement noise. This allowed the true frequency domain estimation error $\mathbf{E}\left\{|\widetilde{G}_k(e^{j\omega})|^2\right\}$ to be estimated by calculating its sample value as an average over the 500 realizations. This is plotted as the solid line in Figures 6.3 to 6.5. The dash-dotted lines in these figures are the approximations (6.47), (6.48) and (6.49) derived from Theorems 6.6.1, 6.6.2 and 6.6.3, respectively.

To be more specific, in the left hand diagram of Figure 6.3, the LMS approxi-



Figure 6.3: Comparison of Sample Variability (over 500 experiments) vs theoretically derived approximations. On the left is the LMS algorithm vs the approximation (6.47). On the right is the RLS algorithm vs the approximation (6.48). In all cases a 10th order model and 800 data samples were used.

mation (6.47) is profiled against the true average error distribution with respect to frequency, and appears to be highly accurate. This is in spite of the approximation (6.47) being derived from the asymptotic in p result in Theorem 6.6.1, but being applied in this simulation to only a p = 10-th order model.

Similarly, in the right hand diagram of Figure 6.3, the RLS approximation (6.48) is profiled against the true error distribution, and is again quite accurate. Finally, in the left hand diagram of Figure 6.4, the same validity of (6.49) as an approximant for the error distribution of the Kalman Filter algorithm is demonstrated.

To illustrate just how robust the approximations can be to the use of a low model order, the case for the RLS algorithm and a model order of only p = 4 and a data length of only 300 samples is shown in the right hand diagram of Figure 6.4. The approximation (dash-dotted) line still appears to be a highly informative indication of the true variability (solid line).

In Figure 6.5, all these tenth order Monte-Carlo simulation results are compared to their theoretical approximants (6.47), (6.48) and (6.49) in one diagram. The top plots are for the LMS algorithm, the middle ones are for the Kalman filter and the bottom ones are for RLS. The fact that the RLS variability is approximately an order of magnitude lower than that for LMS is supported theoretically by examining the approximants (6.47) and (6.48). Specifically,





Figure 6.4: Comparison of Sample Variability (over 500 experiments) vs theoretically derived approximations. On the left is the Kalman Filter algorithm vs the approximation (6.49). On the right is the RLS algorithm vs the approximation (6.48). For the Kalman Filter on the left, 10th order model and 800 data samples were used. For RLS on the right, only 300 data samples and only a 4th order model were used.

(6.47) indicates that for $\rho = 0$, the LMS steady state error is unaffected by input spectral density $\Phi_u(\omega)$, while (6.48) illustrates that for RLS, this same steady state error is inversely proportional to $\Phi_u(\omega)$. Therefore, for stationary systems, the ratio of LMS error to RLS error should be equal to $\Phi_u(\omega)$, which is observed in Figure 6.5.

As well, a comparison of the theoretical expressions (6.49) and (6.48) indicates that the Kalman Filter variability should be larger than the RLS variability by a factor $\sqrt{\Phi_u(\omega)\delta(\omega)}/\sigma$, and this is also supported by the larger observed variability for the Kalman filter in Figure 6.5. Notice also the slower roll-off at high frequencies of the Kalman filter error as compared to the RLS or LMS error. This is due to the presence of the $\sqrt{\cdot}$ term in the Kalman filter error expression (6.49) which is not present in the LMS or RLS expressions (6.47) or (6.48).

Turning away from steady-state (large k) performance, the LMS (with μ decreased to $\mu = 0.0003$ and $\Phi_u(\omega) = (1.57 - 1.75 \cos \omega)^{-1}$) transient analysis approximation in equation (6.37) was also tested via simulation with the results shown in Figure 6.7. In this figure, the variance $\mathbf{E}\left\{|\tilde{G}_k(e^{j\omega})|^2\right\}$ was estimated as the sample variance calculated over 500 simulations with different input and

170



Figure 6.5: Comparison of observed variability (solid line) vs theory (dashdotted line) for LMS (top), Kalman Filter (middle) and RLS (bottom).

measurement noise realisations. This estimate is shown as the solid lines. The top plot with the faster decay rate is the error at $\omega = 0.1$ rad/s, and the bottom plot with the slower decay rate is the error at $\omega = 0.4$ rad/s. The dash-dotted lines in these two diagrams are the estimates for this transient response derived from (6.37). Again, the agreement between the observed error and the theoretical prediction is quite close in spite of the fact that the approximation (6.37) is derived from an asymptotic in p result and then applied to a small (p = 10) model order. In particular, note that (6.37) is able to clearly explain and predict the slower convergence at the higher frequency where the input spectral density is smaller.

Finally, the validity of the approximations (6.47), (6.48) and (6.49) for the case of non-stationary plants ($\rho \neq 0$) was tested by starting with the system (6.50) and then perturbing it according to the random walk model (6.34) with Q chosen so as to imply a $\rho^2 \delta_p(\omega)$ shown in the right hand diagram of Figure 6.6. In the left hand diagram of that figure is shown (solid line) the sample mean square variability of the RLS estimate in steady state (k = 800) versus that predicted (dash-dotted line) via the theoretically derived approximation (6.48). As in previous simulations, a tenth order model was used, and also as in previous simulations the agreement between sample observation and theoretical approximation is good.

If fact, counter-intuitively the agreement between observation and theory appears better in the time varying plant case than in the previous time invariant case. This can be explained by noting that in the time invariant case, the poles $\{\xi_k\}$ in the model were deliberately chosen to be far from the true plant poles in order to test the robustness of the approximations (6.47), (6.48) and (6.49). In the time varying case under the model (6.50) this is not possible, so that there is no under-modeling component in the results shown in Figure 6.6.



Figure 6.6: Comparison of Sample Variability (over 500 experiments) vs theoretically derived approximation for RLS algorithm and non-stationary plant. On the left is the sample estimation error after k = 800 iterations (solid line) compared to the theoretical approximation (6.48) as a dashdotted line. On the right is the non-stationarity modeled in the frequency domain by $\rho^2 \delta_p(\omega)/\mu$.

6.8 Conclusions

In this chapter, a frequency domain analysis of the tracking performance of several adaptive estimation schemes has been carried out for the case in which the system is represented by fixed denominator model structures. The transient, as well as the steady-state behaviour of the frequency response tracking error was analyzed. The main contribution of the chapter was to extend known results for FIR model structures, where the poles are fixed at the origin, to more general model structures where the poles may be placed arbitrarily. The key tool in this analysis was to re-parameterize the system using orthonormal bases with fixed poles that generalize the classical FIR bases. The analysis


Figure 6.7: Transient behavior of LMS at two different frequencies - observed (Monte Carlo Average) vs Theoretical prediction via approximation (6.37). Top plot is error $\mathbf{E}\left\{|\widetilde{G}_k(e^{j\omega})|^2\right\}$ at $\omega = 0.1$ rad/s. Bottom plot is error $\mathbf{E}\left\{|\widetilde{G}_k(e^{j\omega})|^2\right\}$ at $\omega = 0.4$ rad/s.

showed how the choice of the poles in the orthonormal structure affects the tracking and noise rejection abilities of the algorithms. A fundamental rôle in this frequency domain error characterization was shown to be played by the reproducing kernel $\gamma_p(\omega)$ associated with the orthonormal bases. The validity of using the results derived for infinite model order in a finite model setting was examined via simulation. The simulation experiments showed that the results, which are exact for infinite model order were shown to provide good approximations even for relatively small model orders.

APPENDICES

Throughout the Appendices in this chapter we will use the Toeplitz-like matrix form $M_p(f)$ introduced in Chapter 4, as defined in equation (4.A.4).

6.A Proof for Chapter 6

Proof of Lemma 6.4.1

• RLS algorithm: The parameter update law is given by

$$\widehat{\theta}_{k+1} = \widehat{\theta}_k + P_k \phi_k (y_k - \phi_k^T \widehat{\theta}_k), \qquad (6.A.1)$$

$$P_{k} = \frac{1}{\lambda} \left\{ P_{k-1} - \frac{P_{k-1}\phi_{k}\phi_{k}^{T}P_{k-1}}{\lambda + \phi_{k}^{T}P_{k-1}\phi_{k}} \right\}; \quad \lambda \in (0, 1).$$
(6.A.2)

Considering that $\phi_k = J^T \phi'_k$, equations (6.A.1) and (6.A.2) can be written as

$$\widehat{\theta}_{k+1} = \widehat{\theta}_k + P_k J^T \phi'_k (y_k - \phi'^T_k J \widehat{\theta}_k), \qquad (6.A.3)$$

$$P_{k} = \frac{1}{\lambda} \left\{ P_{k-1} - \frac{P_{k-1}J^{T}\phi_{k}\phi_{k}^{T}JP_{k-1}}{\lambda + \phi_{k}^{T}JP_{k-1}J^{T}\phi_{k}^{'}} \right\}.$$
 (6.A.4)

Multiplying equation (6.A.3) on the left by J, and equation (6.A.4) on the left by J and on the right by J^T , we obtain

$$J\widehat{\theta}_{k+1} = J\widehat{\theta}_k + JP_k J^T \phi'_k (y_k - \phi'^T_k J\widehat{\theta}_k), \qquad (6.A.5)$$

$$JP_{k}J^{T} = \frac{1}{\lambda} \left\{ JP_{k-1}J^{T} - \frac{JP_{k-1}J^{T}\phi_{k}'\phi_{k}^{T}JP_{k-1}J^{T}}{\lambda + \phi_{k}'^{T}JP_{k-1}J^{T}\phi_{k}'} \right\}.$$
 (6.A.6)

Now, the result for the RLS algorithm follows by defining $P_k' \triangleq J P_k J^T$.

• KF algorithm: The parameter update law in this case is given by

$$\widehat{\theta}_{k+1} = \widehat{\theta}_k + \frac{\mu P_{k-1} \phi_k}{\sigma^2 + \mu \phi_k^T P_{k-1} \phi_k} (y_k - \phi_k^T \widehat{\theta}_k),$$

$$P_k = P_{k-1} - \mu \frac{P_{k-1} \phi_k \phi_k^T P_{k-1}}{\sigma^2 + \mu \phi_k^T P_{k-1} \phi_k} + \mu \Sigma.$$

Considering that $\phi_k = J^T \phi'_k$, these equations can be written as

$$\widehat{\theta}_{k+1} = \widehat{\theta}_k + \frac{\mu P_{k-1} J^T \phi'_k}{\sigma^2 + \mu \phi'^T_k J P_{k-1} J^T \phi'_k} (y_k - \phi'^T_k J \widehat{\theta}_k), \qquad (6.A.7)$$

$$P_{k} = P_{k-1} - \mu \frac{P_{k-1}J^{T}\phi_{k}'\phi_{k}'^{T}JP_{k-1}}{\sigma^{2} + \mu\phi_{k}'^{T}JP_{k-1}J^{T}\phi_{k}'} + \mu\Sigma.$$
(6.A.8)

Multiplying equation (6.A.7) on the left by J, and equation (6.A.8) on the left by J and on the right by J^T , we obtain

$$J\widehat{\theta}_{k+1} = J\widehat{\theta}_{k} + \frac{\mu J P_{k-1} J^{T} \phi_{k}'}{\sigma^{2} + \mu \phi_{k}'^{T} J P_{k-1} J^{T} \phi_{k}'} (y_{k} - \phi_{k}'^{T} J \widehat{\theta}_{k}),$$

$$J P_{k} J^{T} = J P_{k-1} J^{T} - \mu \frac{J P_{k-1} J^{T} \phi_{k}' \phi_{k}'^{T} J P_{k-1} J^{T}}{\sigma^{2} + \mu \phi_{k}'^{T} J P_{k-1} J^{T} \phi_{k}'} + \mu J \Sigma J^{T}.$$

Now, the result for the KF algorithm follows by defining $P'_k \triangleq JP_k J^T$ and $\Sigma' \triangleq J\Sigma J^T$.

• LMS algorithm: In this case the parameter update law is given by

$$\widehat{\theta}_{k+1} = \widehat{\theta}_k + \mu \phi_k (y_k - \phi_k^T \widehat{\theta}_k).$$

Considering that $\phi_k = J^T \phi'_k$, we can write

$$\widehat{\theta}_{k+1} = \widehat{\theta}_k + \mu J^T \phi'_k (y_k - \phi'^T_k J \widehat{\theta}_k).$$

From this equation we can easily conclude that the LMS estimate is not invariant under the linear re-parameterization $\phi_k = J^T \phi'_k$.

Proof of Theorem 6.5.1 Considering equation (6.32) with $S_k = I$, and the definition (6.35), we can write

$$\frac{1}{\gamma_{p}(\omega)} \mathbf{E} \left\{ |\widetilde{G}_{k+1}(e^{j\omega})|^{2} \right\} =
= \frac{1}{\gamma_{p}(\omega)} \Gamma_{p}^{\star}(e^{j\omega}) (I - \mu R) \Pi_{k} (I - \mu R)^{T} \Gamma_{p}(e^{j\omega})
+ \mu^{2} \sigma_{\nu}^{2} \frac{1}{\gamma_{p}(\omega)} \Gamma_{p}^{\star}(e^{j\omega}) R \Gamma_{p}(e^{j\omega}) + \rho^{2} \frac{\delta_{p}(\omega)}{\gamma_{p}(\omega)}.$$
(6.A.9)

By Parseval's Theorem matrix ${\cal R}$ can be given the following spectral representation

$$R \triangleq \mathbf{E}\left\{\phi_k \phi_k^T\right\} = \frac{1}{2\pi} \int_{-\pi}^{\pi} \Gamma_p(e^{j\omega}) \Phi_u(\omega) \Gamma_p^{\star}(e^{j\omega}) d\omega$$
(6.A.10)

Now, using the notation (4.A.4), equation (6.A.9) can be written as

$$\frac{1}{\gamma_{p}(\omega)} \mathbf{E} \left\{ |\widetilde{G}_{k+1}(e^{j\omega})|^{2} \right\} =
= \frac{1}{\gamma_{p}(\omega)} \Gamma_{p}^{\star}(e^{j\omega}) M_{p}(1-\mu\Phi_{u}) \Pi_{k} M_{p}(1-\mu\Phi_{u}) \Gamma_{p}(e^{j\omega})
+ \mu^{2} \sigma_{\nu}^{2} \frac{1}{\gamma_{p}(\omega)} \Gamma_{p}^{\star}(e^{j\omega}) M_{p}(\Phi_{u}) \Gamma_{p}(e^{j\omega}) + \rho^{2} \frac{\delta_{p}(\omega)}{\gamma_{p}(\omega)}.$$
(6.A.11)

Finally, taking the limit as $p \to \infty$, and applying Theorem 4.B.1, Lemma 6.B.1, and the definition of $\delta(\omega)$ in (6.36), then gives the result.

Proof of Theorem 6.6.1 Using the formulation (6.A.10) together with the notation (4.A.4) in (6.41) gives that in the limit as $k \to \infty$

$$\Pi M_p(\Phi_u) + M_p(\Phi_u)\Pi = \mu \sigma_{\nu}^2 M_p(\Phi_u) + \frac{\rho^2}{\mu} Q$$

so that

$$\frac{\Gamma_p^{\star}(e^{j\omega})\Pi M_p(\Phi_u)\Gamma_p(e^{j\omega})}{\gamma_p(\omega)} + \frac{\Gamma_p^{\star}(e^{j\omega})M_p(\Phi_u)\Pi\Gamma_p(e^{j\omega})}{\gamma_p(\omega)} = \frac{\mu\sigma_\nu^2\Gamma_p^{\star}(e^{j\omega})M_p(\Phi_u)\Gamma_p(e^{j\omega})}{\gamma_p(\omega)} + \frac{\rho^2\delta_p(\omega)}{\mu\gamma_p(\omega)}$$

Now the result follows by taking the limit of both sides as $p \to \infty$ while using Theorem 4.B.1, Lemma 6.B.2 and the definition of $\delta(\omega)$ in (6.36).

Proof of Theorem 6.6.2 Using the formulation (6.A.10) together with the notation (4.A.4) in (6.42) gives that in the limit as $k \to \infty$

$$\frac{1}{\gamma_p(\omega)} \mathbf{E} \left\{ |\widetilde{G}(e^{j\omega})|^2 \right\} = \frac{\Gamma_p^{\star}(e^{j\omega})\Pi\Gamma_p(e^{j\omega})}{\gamma_p(\omega)} = \\ = \frac{\mu \sigma_{\nu}^2}{2\gamma_p(\omega)} \Gamma_p^{\star}(e^{j\omega}) M_p^{-1}(\Phi_u)\Gamma_p(e^{j\omega}) + \frac{\rho^2 \delta_p(\omega)}{2\mu \gamma_p(\omega)}$$

Taking the limit of both sides as $p \to \infty$ while using Theorem 4.B.2 and the definition of $\delta(\omega)$ in (6.36) then gives the result.

Proof of Theorem 6.6.3 Substituting the value of S from (6.45) into (6.44) with $\Sigma = Q$ gives that Π is the solution of

$$\Pi R \Pi = \frac{\sigma^2}{4} \left(\mu \frac{\sigma_{\nu}^2}{\sigma^2} + \frac{\rho^2}{\mu} \right)^2 Q.$$
 (6.A.12)

Define two positive definite, $p \times p$ dimensional real matrices A_p and B_p to be asymptotically equivalent $A_p \sim B_p$ if they leave quadratic forms with $\Gamma_p(e^{j\omega})$ invariant:

$$A_p \sim B_p \Leftrightarrow \lim_{p \to \infty} \frac{1}{\gamma_p(\omega)} \Gamma_p^{\star}(e^{j\omega}) (A_p - B_p) \Gamma_p(e^{j\omega}) = 0.$$

Using the definition

$$\alpha^2 \triangleq \frac{\sigma^2}{4} \left(\frac{\mu \sigma_{\nu}^2}{\sigma^2} + \frac{\rho^2}{\mu}\right)^2$$

and the notation (4.A.4), the matrix

$$\alpha M_p\left(\sqrt{\frac{\delta(\omega)}{\Phi_u(\omega)}}\right)$$

is an asymptotically equivalent solution to Π given by (6.A.12) since using the representation (6.A.10)

$$\frac{1}{\gamma_{p}(\omega)}\Gamma_{p}^{\star}(e^{j\omega})\left(\Pi R\Pi - \alpha^{2}M_{p}(\sqrt{\delta/\Phi_{u}})M_{p}(\Phi_{u})M_{p}(\sqrt{\delta/\Phi_{u}})\right)\Gamma_{p}(e^{j\omega}) = \\
= \frac{\alpha^{2}}{\gamma_{p}(\omega)}\Gamma_{p}^{\star}(e^{j\omega})\left(M_{p}(\delta) - M_{p}(\sqrt{\delta/\Phi_{u}})M_{p}(\Phi_{u})M_{p}(\sqrt{\delta/\Phi_{u}})\right)\Gamma_{p}(e^{j\omega}) + \\$$
(6.A.13)

+
$$\frac{\alpha^2}{\gamma_p(\omega)} \Gamma_p^*(e^{j\omega}) \left(Q - M(\delta)\right) \Gamma_p(e^{j\omega}).$$
 (6.A.14)

Now, considering the term (6.A.13), by Theorem 4.B.1 and Lemma 6.B.3

$$\lim_{p \to \infty} \frac{1}{\gamma_p(\omega)} \Gamma_p^{\star}(e^{j\omega}) \left(M_p(\delta) - M_p(\sqrt{\delta/\Phi_u}) M_p(\Phi_u) M_p(\sqrt{\delta/\Phi_u}) \right) \Gamma_p(e^{j\omega}) = \\ = \delta(\omega) - \sqrt{\frac{\delta(\omega)}{\Phi_u(\omega)}} \Phi_u(\omega) \sqrt{\frac{\delta(\omega)}{\Phi_u(\omega)}} = 0.$$

Considering the term (6.A.14), and using the definition of $\delta(\omega)$ and Theorem 4.B.1

$$\lim_{p \to \infty} \frac{1}{\gamma_p(\omega)} \Gamma_p^{\star}(e^{j\omega}) \left(Q - M(\delta)\right) \Gamma_p(e^{j\omega}) = \delta(\omega) - \delta(\omega) = 0.$$

Therefore, since $\Pi \sim \alpha M_p(\sqrt{\delta(\omega)/\Phi_u(\omega)})$, then again by Theorem 4.B.1

$$\lim_{p \to \infty} \frac{1}{\gamma_p(\omega)} \mathbf{E} \left\{ |\tilde{G}(e^{j\omega})|^2 \right\} = \lim_{p \to \infty} \frac{1}{\gamma_p(\omega)} \Gamma_p^{\star}(e^{j\omega}) \Pi \Gamma_p(e^{j\omega})$$
$$= \lim_{p \to \infty} \frac{\alpha}{\gamma_p(\omega)} \Gamma_p^{\star}(e^{j\omega}) M_p(\sqrt{\delta/\Phi_u}) \Gamma_p(e^{j\omega})$$
$$= \alpha \sqrt{\frac{\delta(\omega)}{\Phi_u(\omega)}}.$$

6.B Technical Results

Lemma 6.B.1. Let $Q_p \in \mathbb{R}^{p \times p}$ be a symmetric, positive definite matrix with $\|Q_p\|_2 < \infty$ for all p, and let $f(\omega)$ be a real valued function, continuous on $[-\pi,\pi]$ and having a finite dimensional spectral factorization. Suppose that

$$\sum_{k=0}^{\infty} (1 - |\xi_k|) = \infty.$$

Then

$$\lim_{p \to \infty} \frac{1}{\gamma_p(\omega)} \Gamma_p^{\star}(\omega) M_p(f) Q_p M_p(f) \Gamma_p(\omega) = f^2(\omega) \lim_{p \to \infty} \frac{1}{\gamma_p(\omega)} \Gamma_p^{\star}(\omega) Q_p \Gamma_p(\omega).$$

Proof: For simplicity of notation, define the function $g(\sigma) \triangleq f(\sigma) - f(\omega)$. Then

$$\frac{1}{\gamma_{p}(\omega)} \left| \Gamma_{p}^{\star}(\omega) M_{p}(f) Q_{p} M_{p}(f) \Gamma_{p}(\omega) - f^{2}(\omega) \Gamma_{p}^{\star}(\omega) Q_{p} \Gamma_{p}(\omega) \right| \leq
\leq \frac{1}{\gamma_{p}(\omega)} \left| \Gamma_{p}^{\star}(\omega) M_{p}(g) Q_{p} M_{p}(g) \Gamma_{p}(\omega) \right| +
+ \frac{|f(\omega)|}{\gamma_{p}(\omega)} \left| \Gamma_{p}^{\star}(\omega) [Q_{p} M_{p}(g) + M_{p}(g) Q_{p}] \Gamma_{p}(\omega) \right|.$$
(6.B.15)

Considering the first term in this upper bound, we can write

$$\frac{1}{\gamma_{p}(\omega)} \left| \Gamma_{p}^{\star}(\omega) M_{p}(g) Q_{p} M_{p}(g) \Gamma_{p}(\omega) \right| \leq \\
\leq \frac{\|Q_{p}\|_{2}}{\gamma_{p}(\omega)} \left| \Gamma_{p}^{\star}(\omega) M_{p}^{2}(g) \Gamma_{p}(\omega) \right| \\
\leq \frac{\|Q_{p}\|_{2}}{\gamma_{p}(\omega)} \left| \Gamma_{p}^{\star}(\omega) M_{p}(g^{2}) \Gamma_{p}(\omega) \right| +$$
(6.B.16)

+
$$\frac{\|Q_p\|_2}{\gamma_p(\omega)} \left| \Gamma_p^{\star}(\omega) [M_p^2(g) - M_p(g^2)] \Gamma_p(\omega) \right|.$$
 (6.B.17)

Now, by Theorem 4.B.1, the term (6.B.16) tends to zero as p tends to infinity, since

$$\lim_{p \to \infty} \frac{1}{\gamma_p(\omega)} \Gamma_p^{\star}(\omega) M_p(g^2) \Gamma_p(\omega) = g^2(\omega) = 0.$$

For the term (6.B.17) we have that since by construction the elements of $\Gamma_p(\omega)$ are bounded in magnitude by some finite number K_1 (as defined in Lemma 5.E.2), use of Lemma 4.C.2 give us that for some $|\eta| < 1$

$$\begin{aligned} \left| \Gamma_{p}^{\star}(\omega) [M_{p}^{2}(g) - M_{p}(g^{2})] \Gamma_{p}(\omega) \right| &\leq \\ &\leq \sum_{m=0}^{p-1} \sum_{n=0}^{p-1} \left| [\Gamma_{p}(\omega)^{\star}]_{m} \right| \times \\ &\times \left| [M_{p}^{2}(g)]_{m,n} - [M_{p}(g^{2})]_{m,n} \right| \left| [\Gamma_{p}(\omega)]_{n} \right| \\ &\leq K_{1} K^{2} \sum_{m=0}^{p-1} \sum_{n=0}^{p-1} (\eta^{p-m} + \eta^{m}) (\eta^{p-n} + \eta^{n}) \\ &= K_{1} K^{2} \left(\frac{1 - \eta^{p}}{1 - \eta} \right)^{2} (\eta^{p} + \eta)^{2} < \infty. \end{aligned}$$

Considering that

$$\frac{1}{2}\sum_{k=0}^{p-1}(1-|\xi_k|) \le \gamma_p(\omega),$$

then, under the conditions of the theorem, also the term (6.B.17) tends to zero as p tends to infinity, i.e.

$$\lim_{p \to \infty} \frac{\|Q_p\|_2}{\gamma_p(\omega)} \left| \Gamma_p^{\star}(\omega) [M_p^2(g) - M_p(g^2)] \Gamma_p(\omega) \right| = 0.$$

Using Lemma 6.B.2 to deal with the remaining term in the upper bound (6.B.15) then completes the proof.

Lemma 6.B.2. Let $Q_p \in \mathbb{R}^{p \times p}$ be a symmetric, positive definite matrix with $\|Q_p\|_2 < \infty$ for all p, and let $f(\omega)$ be a real valued function, continuous on $[-\pi,\pi]$. Suppose that

$$\sum_{k=0}^{\infty} (1 - |\xi_k|) = \infty.$$

Then

$$\lim_{p \to \infty} \frac{1}{\gamma_p(\omega)} \Gamma_p^{\star}(\omega) M_p(f) Q_p \Gamma_p(\omega) =$$
$$= \lim_{p \to \infty} \frac{1}{\gamma_p(\omega)} \Gamma_p^{\star}(\omega) Q_p M_p(f) \Gamma_p(\omega) =$$
$$= f(\omega) \lim_{p \to \infty} \frac{1}{\gamma_p(\omega)} \Gamma_p^{\star}(\omega) Q_p \Gamma_p(\omega).$$

Proof: For simplicity of notation, define the function $g(\sigma) \triangleq f(\sigma) - f(\omega)$. Then

$$\frac{1}{\gamma_p(\omega)} \left| \Gamma_p^{\star}(\omega) M_p(f) Q_p \Gamma_p(\omega) - f(\omega) \Gamma_p^{\star}(\omega) Q_p \Gamma_p(\omega) \right| = \frac{1}{\gamma_p(\omega)} \left| \Gamma_p^{\star}(\omega) M_p(g) Q_p \Gamma_p(\omega) \right|.$$

Now define the functions $g^+(\sigma) \triangleq \max[g(\sigma), 0]$ and $g^-(\sigma) \triangleq \min[g(\sigma), 0]$ so that $g(\sigma) = g^+(\sigma) + g^-(\sigma)$ and hence

$$\frac{1}{\gamma_{p}(\omega)} \left| \Gamma_{p}^{\star}(\omega)Q_{p}M_{p}(g)\Gamma_{p}(\omega) \right| \leq \\
\leq \frac{1}{\gamma_{p}(\omega)} \left| \Gamma_{p}^{\star}(\omega)Q_{p}M_{p}(g^{+})\Gamma_{p}(\omega) \right| + \\
+ \frac{1}{\gamma_{p}(\omega)} \left| \Gamma_{p}^{\star}(\omega)Q_{p}M_{p}(g^{-})\Gamma_{p}(\omega) \right|.$$
(6.B.18)

Considering only the first term in this upper bound, note that provided $f(\omega)$ is not equal to a constant (in which case the Lemma is trivial since $M_p(1) = I$), then for $x \in \mathbb{R}^p$ arbitrary

$$x^{T} M_{p}(g^{+}) x = \frac{1}{2\pi} \int_{-\pi}^{\pi} |x^{T} \Gamma_{p}(\sigma)|^{2} g^{+}(\sigma) \, \mathrm{d}\sigma > 0$$

so that $M_p(g^+)$ is positive definite and hence

$$\frac{1}{\gamma_p(\omega)} \left| \Gamma_p^{\star}(\omega) Q_p M_p(g^+) \Gamma_p(\omega) \right| =
= \frac{1}{\gamma_p(\omega)} \left| \Gamma_p^{\star}(\omega) M_p^{1/2}(g^+) [M_p^{-1/2}(g^+) Q_p M_p^{1/2}(g^+)] M_p^{1/2}(g^+) \Gamma_p(\omega) \right|
\leq \frac{||Q_p||_2}{\gamma_p(\omega)} \left| \Gamma_p^{\star}(\omega) M_p(g^+) \Gamma_p(\omega) \right|.$$

Finally, using Theorem 4.B.1

$$\lim_{p \to \infty} \frac{1}{\gamma_p(\omega)} \Gamma_p^{\star}(\omega) M_p(g^+) \Gamma_p(\omega) = g^+(\omega) = 0.$$

Using the same argument for the remaining term in the upper bound (6.B.18) then completes the proof.

Lemma 6.B.3. Let $f, g \in L_2([-\pi, \pi])$ have finite dimensional spectral factorizations. Then provided $\sum_{k=0}^{\infty} (1 - |\xi_k|) = \infty$,

$$M_p(f)M_p(g)M_p(f) \sim M_p(f^2g)$$
 as $p \to \infty$.

with the notation ' \sim ' as introduced in the proof of Theorem 6.6.3.

Proof: Using Lemmas 4.C.2 and 4.C.3

$$\begin{split} \Gamma_{p}^{\star}(\omega) \left[M_{p}(f) M_{p}(g) M_{p}(f) - M_{p}(f^{2}g) \right] \Gamma_{p}(\omega) &= \\ &= \Gamma_{p}^{\star}(\omega) M_{p}(f) \left[M_{p}(g) M_{p}(f) - M_{p}(fg) \right] \Gamma_{p}(\omega) + \\ &+ \Gamma_{p}^{\star}(\omega) \left[M_{p}(f) M_{p}(fg) - M_{p}(f^{2}g) \right] \Gamma_{p}(\omega) \\ &\leq \sum_{m=0}^{p-1} \sum_{n=0}^{p-1} \left| [\Gamma_{p}(\omega)^{\star} M_{p}(f)]_{m} \right| \left| [M_{p}(g) M_{p}(f)]_{m,n} - [M_{p}(fg)]_{m,n} \right| \left| [\Gamma_{p}(\omega)]_{n} \right| + \\ &+ \sum_{m=0}^{p-1} \sum_{n=0}^{p-1} \left| [\Gamma_{p}(\omega)^{\star}]_{m} \right| \left| [M_{p}(f) M_{p}(fg)]_{m,n} - [M_{p}(f^{2}g)]_{m,n} \right| \left| [\Gamma_{p}(\omega)]_{n} \right| \\ &\leq K^{3} \sum_{m=0}^{p-1} \sum_{n=0}^{p-1} (\eta^{p-m} + \eta^{m})(\eta^{p-n} + \eta^{n}), \\ &= K^{3} \left(\frac{1 - \eta^{p}}{1 - \eta} \right)^{2} (\eta^{p} + \eta)^{2} < \infty. \end{split}$$

Noting that $2\gamma_p(\omega) \ge \sum_{k=0}^{p-1} (1-|\xi|)$ then completes the proof.

7

Conclusions

In this thesis we have analyzed several aspects regarding the use of rational orthonormal bases in identification of discrete-time linear systems from inputoutput data in the time domain.

We concentrated on the use of orthonormal bases with fixed poles that generalize the more common FIR, Laguerre and Kautz bases, and that allows the incorporation in the identification process of prior information about dominant dynamics of the system by an appropriate choice of the poles in the orthonormal structure.

Apart from the well known properties of orthonormal bases leading to linear regressor forms and guaranteeing worst case numerical conditioning of the least squares estimation, an aspect we have emphasized in this thesis is the use of orthonormal bases with fixed poles as an analysis tool that facilitates the quantification of the estimation accuracy of any (multivariable) fixed denominator model structure. The numerical robustness of the least squares estimation using orthonormal structures was also analyzed and compared to the case of using an equivalent non-orthonormal structure.

One of the main contributions of this thesis has been the extension to the multivariable setting and to general fixed denominator model structures of existent single-input single-output FIR results quantifying the undermodelling error and the asymptotic distribution of the transfer matrix estimate (the variance error). The variance error result establishes that the variance of the transfer matrix estimate at a given frequency, and for large model order and data-length, can be approximated by the generalized noise to signal ratio $(\Phi_u^{-1}(\omega) \otimes \Phi_\nu(\omega))$ weighted with a frequency dependent factor $(\gamma_p(\omega)/N)$ that depends on the particular bases. These results explicitly show how the choice of the poles of the bases influences the bias and variance errors. In addition, they illustrate an until now unknown phenomenom of bias/variance trade-off with respect to the location of the poles of the bases.

Another contribution of this thesis, in the framework of recursive identification, is the performance analysis of general adaptive algorithms using orthonormal bases with fixed poles. Here again, the analysis illustrates how several factors, such as step size of the algorithm, excitation properties of the input signal and measurement noise, and choice of the fixed poles in the model structure, influence the disturbance rejection properties and the tracking ability of the algorithms.

A

Kronecker Product and vec-Operator

In this Appendix, some basic properties of Kronecker products and the vecoperator are summarized. A more detailed treatment of these topics can be found in [Bre78].

• Kronecker Product: Let $A = (a_{ij})$ and $B = (b_{ij})$ be $m \times n$ and $\ell \times p$ sized matrices. Then the Kronecker product of A and B, denoted $A \otimes B$, is defined as the $m\ell \times np$ sized matrix

$$A \otimes B = \begin{pmatrix} a_{11}B & a_{12}B & \cdots & a_{1n}B \\ a_{21}B & a_{22}B & \cdots & a_{2n}B \\ \vdots & & \vdots \\ a_{m1}B & a_{m2}B & \cdots & a_{mn}B \end{pmatrix}$$

• vec -Operator: is an operator which turns an $m \times n$ matrix A into a vector by stacking the columns of A on top of one another:

$$\operatorname{vec} A = \begin{pmatrix} A_1 \\ A_2 \\ \vdots \\ A_n \end{pmatrix} \quad (m \times 1) \ \operatorname{vector}$$

where A_k is the k-th column of A.

- Operations with Kronecker Products: The following identities hold
 - 1. $(A \otimes B)(C \otimes D) = AC \otimes BD$
 - 2. $(A \otimes B)^T = A^T \otimes B^T$
 - **3**. $(A \otimes B)^{-1} = A^{-1} \otimes B^{-1}$

Finally a useful property of the vec-operator is that when A is $m\times n$ and B is $n\times \ell$ then

$$\operatorname{vec} AB = (I_\ell \otimes A) \operatorname{vec} B = (B^T \otimes I_m) \operatorname{vec} A.$$

Lemma A.0.4. The eigenvalues of $A \otimes B$ are $\lambda_i \mu_j$ where λ_i are the eigenvalues of A and μ_j are the eigenvalues of B.

Proof: See [Bel60], Theorem 3, page 235.

Lemma A.0.5. Let σ_i^A and σ_j^B denote the singular values of the matrices A and B respectively. Then the matrix $A \otimes B$ has singular values $\sigma_i^A \sigma_j^B$.

Proof: By definition, the singular values of $A \otimes B$ are the square root of the eigenvalues of the matrix $(A \otimes B)(A \otimes B)^*$. But

$$(A \otimes B)(A \otimes B)^{\star} = AA^{\star} \otimes BB^{\star}$$

so that the result follows by applying Lemma A.0.4 to the RHS of the previous equation.

Lemma A.0.6. The following identity holds

$$||A \otimes B||_2 = ||A||_2 ||B||_2$$

Proof: The results follows directly by applying Lemma A.0.5 and considering that the 2-norm of a matrix equals its maximum singular value.

B

Notation

q	forward shift operator .
\mathbb{D}	the open unit disk in the complex plane $\{z\in\mathbb{C}: z <1\}$.
T	the unit circle $\{z: z =1\}$.
\mathbb{E}	the open region outside the unit disk $\{z: z >1\}$.
\mathbb{C}	complex numbers.
\mathbb{R}	real numbers.
Z	set of integer numbers .
\mathbb{N}_0	set of non-negative integer numbers.
$\ell_2(\mathbb{Z})$	Hilbert space of (two-sided) square summable sequences with support in $\mathbb Z$.
$\ell_2(\mathbb{N}_0)$	Hilbert space of (one-sided) square summable sequences with support in \mathbb{N}_0 .
$L_2(\mathbb{T})$	Hilbert space of Lebesgue square-integrable functions on the unit circle.
$H_2(\mathbb{T})$	Hardy space of Lebesgue square-integrable functions on the unit circle $\mathbb T$ that are analytic outside the unit circle (i.e., analytic on $\mathbb E.$
$H^{m imes n}_2(\mathbb{T})$	Hardy space of $(m \times n)$ transfer matrices whose entries are in $H_2(\mathbb{T})$.
$\{\mathcal{B}_k(z)\}_{k=0}^{p-1}$	set of orthonormal functions.
$K_p(z,\mu)$	Reproducing Kernel associated with an orthonormal set.

$\langle \cdot, \cdot angle$	inner product (in a Hilbert space).
$\langle \cdot, \cdot angle_{H}$	inner product in Hilbert space H .
·	norm induced by the inner product.
$\ \cdot\ _2$	2-norm of matrices or spectral norm.
$\ \cdot\ _F$	Frobenius norm of matrices.
A^T, A^\star	Transpose and conjugate transpose of matrix A .
Acronyms	
FIR	Finite Impulse Response (filter or model structure).
IIR	Infinite Impulse Response (filter or model structure).
SISO	Single-Input Single-Output (system).
MIMO	Multiple-Input Multiple-Output (system).
DT	Discrete Time (system).
LTI	Linear Time-Invariant (system).
OBFP	Orthonormal Bases with Fixed Poles.
OBGIF	Orthonormal Bases Generated fron Inner Functions.
PEM	Prediction Error Methods.
rkHs	reproducing kernel Hilbert space.
DFT	Discrete Fourier Transform.
BIBO	Bounded-Input Bounded-Output (stability).
LMS	Least Mean Squares.
RLS	Recursive Least Squares.
KF	Kalman Filter.
4SID	Subspace-based State-Space System IDentification methods.
N4SID	Numerical algorithms for Subspace State-Space System IDentification.
MOESP	Multivariable Output Error State sPace.
CVA	Canonical Variate Analysis.
SVD	Singular Value Decomposition.

Bibliography

- [And82] B. D. O. Anderson. Exponential convergence and persistent excitation. In Proceedings of the 21st. IEEE Conference on Decision and Control, Orlando, 1982.
- [Aro50] N. Aronszajn. Theory of reproducing kernels. Transactions American Mathematical Society, 68:337–404, 1950.
- [ÅW84] K.J. Åstrom and B. Wittenmark. Computer Controlled Systems: Theory and Design. Prentice-Hall, 1984.
- [Bel60] Richard Bellman. Introduction to Matrix Analysis. McGraw-Hill Book Company, Inc., New York, 1960.
- [BJ76] G. E. Box and G. M. Jenkins. Time Series Analysis, forecasting and control. Holden-Day Series in time series analysis and digital processing. Holden-Day, Oakland, revised edit. edition, 1976.
- [Bod95] Per Bodin. On wavelets and orthonormal bases in system identification. Technical Licentiate Thesis, Automatic Control, Department of Signals, Sensors and Systems, Royal Institute of Technology (KTH), Stockholm, Sweden, May 1995.
- [Bre78] J. W. Brewer. Kronecker products and matrix calculus in system theory. *IEEE Transactions on Circuits and Systems*, CAS-25(9):772-781, September 1978.
- [Bro65] Paul W. Broome. Discrete orthonormal sequences. Journal of the Association for Computing Machinery, 12(2):151-168, April 1965.
- [CG91] Han-Fu Chen and Lei Guo. Identification and Stochastic Adaptive Control. Systems & Control: Foundations & Applications. Birkhäuser, Boston, 1991.
- [Cle63] Preston R. Clement. On completeness of basis functions used for signal analysis. SIAM Review, 5(2):131–139, 1963.

[CV97]	C. T. Chou and M. Verhaegen. Subspace algorithms for the identi- fication of multivariable dynamic errors-in-variables models. <i>Auto-</i> <i>matica</i> , 33(10):1857-1869, 1997.
[Dav75]	P.J. Davis. Interpolation and Approximation. Dover Publication, Inc., New York, 1975.
[DD81]	Patrick DeWilde and Harry Dym. Schur recursions, error formu- las, and convergence of rational estimators for stationary stochas- tic sequences. <i>IEEE Transactions on Information Theory</i> , IT- 27(4):446-461, July 1981.
[DD87]	P. De Groen and B. De Moor. The fit of a sum of exponentials to noisy data. <i>Journal of Computational and Applied Mathematics</i> , 20:175–187, 1987.
[den93a]	A. C. den Brinker. Adaptive modified Laguerre filters. Signal Processing, 31:69–79, 1993.
[den93b]	A. C. den Brinker. Adaptive orthonormal filters. In <i>Proceedings of the 12th IFAC World Congress</i> , volume 5, pages 287–292, Sydney, Australia, July 1993.
[den94]	A. C. den Brinker. Laguerre-domain adaptive filters. <i>IEEE Trans-</i> actions on Signal Processing, 42(4):953-956, April 1994.
[den96]	A. C. den Brinker. Optimality conditions for a specific class of trun- cated kautz series. <i>IEEE Transactions on Circuits and Systems-</i> <i>II: Analog and Digital Processing</i> , 43(8):597–600, August 1996.
[Doo53]	J.L. Doob. <i>Stochastic Processes</i> . John Wiley and Sons, Inc., London, 1953.
[DPS94]	M. Deistler, K. Peternell, and W. Scherrer. The relative efficiency of different subspace methods. In M. Blanke and T. Söderström, editors, <i>Proceedings of the 10th IFAC Symposium on System</i> <i>Identification SYSID'94</i> , volume 2, pages 158–163, Copenhagen Denmark, July 1994. IFAC, Danish Automation Society.
[DPS95]	M. Deistler, K. Peternell, and W. Scherrer. Consistency and rela- tive efficiency of subspace methods. <i>Automatica</i> , 31(12):1865–1875, 1995.
[DVM ⁺ 88]	B. De Moor, J. Vandewalle, M. Moonen, L. Vandenberghe, and P. Van Mieghem. A geometrical strategy for the identification of state space models of linear multivariable systems with singular

value decomposition. In Proceedings of the IFAC Symposium on Identification and System Parameter Estimation, pages 700-704, Beijing, China, August 1988.

- [EJLW92] B. Egardt, C.R. Johnson Jr., L. Ljung, and G.A. Williamson. Adaptive Systems Performance in the Frequency Domain. In Proceedings of the 4th IFAC Symposium on Adaptive Systems, Control and Signal Processing, pages 33-39, Grenoble, France, 1992. IFAC.
- [FD93] Y. Fu and G.A. Dumont. An optimum time scale for discrete Laguerre network. *IEEE Transactions on Automatic Control*, 38(6):934-938, June 1993.
- [GKB89] G. Gu, P.P. Khargonekar, and E. Bruce Lee. Approximation of infinite-dimensional systems. *IEEE Transactions on Automatic* Control, 34(6):610-618, June 1989.
- [GL89] Svante Gunnarsson and Lennart Ljung. Frequency domain tracking characteristics of adaptive algorithms. IEEE Transactions on Acoustics, Speech, and Signal Processing, 37(7):1072-1089, July 1989.
- [GL95a] Lei Guo and Lennart Ljung. Exponential stability of general tracking algorithms. IEEE Transactions on Automatic Control, 40(8):1376-1387, August 1995.
- [GL95b] Lei Guo and Lennart Ljung. Performance analysis of general tracking algorithms. IEEE Transactions on Automatic Control, 40(8):1388-1402, August 1995.
- [GLLT92] A. Grace, A. J. Laub, J. N. Little, and C. M. Thompson. Control System Toolbox User's Guide. The MathWorks, Inc., 24 Prime Park Way, Natick, MA 01760-1500, 1992.
- [Glo84] K. Glover. All optimal Hankel norm approximations of linear multivariable systems and their L^{∞} error bounds. International Journal of Control, 39(6):1115–1193, 1984.
- [GP77] G.C. Goodwin and R.L. Payne. Dynamic System Identification. Academic Press, 1977.
- [GS58] U. Grenander and G. Szegö. Toeplitz Forms and their Applications. University of California Press, Berkeley, 1958.
- [GS84] G.C. Goodwin and K.W. Sin. Adaptive Filtering Prediction and Control. Prentice-Hall, Inc., New Jersey, 1984.

[Gui75]	R. Guidorzi. Canonical structures in the identification of multivariable systems. Automatica, $11(4)$:361–374, 1975.
[Gui81]	R. Guidorzi. Invariants and canonical forms for systems structural and parametric identification. <i>Automatica</i> , 17(1):117–133, 1981.
[Gun88]	S. Gunnarsson. Frequency Domain Aspects of Modeling and Con- trol in Adaptive Systems. PhD thesis, Department of Electrical En- gineering, Linköping University, S-581 83 Linköping, Sweden, 1988.
[Guo90]	Lei Guo. Estimating time-varying parameters by the kalman filter based algorithm: Stability and convergence. <i>IEEE Transactions on Automatic Control</i> , 35(2):141–147, February 1990.
[GV89]	Gene H. Golub and Charles F. Van Loan. <i>Matrix Computations</i> . The John Hopkins University Press, Baltimore and London, 2nd edition, 1989.
[GW74]	K. Glover and J. C. Willems. Parametrizations of linear dynamical systems: Canonical forms and identifiability. <i>IEEE Transactions opn Automatic Control</i> , 19(6):640-645, 1974.
[GW90]	S. Gunnarsson and B. Wahlberg. Some asymptotic results in re- cursive identification using Laguerre models. In <i>Proceedings of the</i> 29th IEEE Conference on Decision and Control, volume 2, pages 1068–1073, Honolulu, Hawaii, December 1990.
[GW91]	S. Gunnarsson and B. Wahlberg. Some asymptotic results in recursive identification using Laguerre models. <i>International Journal of Adaptive Control and Signal Processing</i> , 5:313–333, 1991.
[HB90]	P.S.C. Heuberger and O.H. Bosgra. Approximate system identifica- tion using system based orthonormal functions. In <i>Proceedings of</i> <i>the 29th Conference on Decision and Control</i> , pages 1086–1092, Honolulu, Hawaii, December 1990.
[HbVB95]	P.S.C. Heuberger, P.M.J. Van den Hof, and O.H. Bosgra. A general- ized orthonormal basis for linear dynamical systems. <i>IEEE Trans-</i> <i>actions on Automatic Control</i> , 40(3):451-465, March 1995.
[HK66]	B. L. Ho and R. E. Kalman. Effective construction of linear state- variable models from input-output functions. $Regelungstechnik$, $14(12):545-592$, 1966.
[HN77]	E. J. Hannan and D. F. Nicholls. The estimation of the prediction error variance. <i>Journal of the American Statistical Association</i> , 72(360):834-840, December 1977.

- [HV96] P.S.C. Heuberger and P.M.J. Van den Hof. The Hambo transform: A signal and system transform induced by generalized orthonormal basis functions. In *Proceedings of the 13th IFAC World Congress*, volume I, pages 357–362, San Francisco, California, June 30-July 5 1996.
- [HVB92] P.S.C. Heuberger, P.M.J. Van den Hof, and O.H. Bosgra. On orthogonal basis functions that contain system dynamics. Selected Topics in Identification Modelling and Control, 5:37-44, December 1992.
- [HVB93] P.S.C. Heuberger, P.M.J. Van den Hof, and O.H. Bosgra. A generalized orthonormal basis for linear dynamical systems. In Proceedings of the 32nd Conference on Decision and Control, pages 2850-2855, San Antonio, Texas, December 1993.
- [HW89] E. J. Hannan and B. Wahlberg. Covergence rates of inverse Toeplitz matrix forms. *Journal of Multivariate Analysis*, 31:127–135, 1989.
- [Joh93] C. R. Johnson Jr. Adaptive Filtering in Communication Systems. School of Electrical Engineering, Cornell University, Ithaca, New York, 1993.
- [Joh95] C. R. Johnson Jr. On the interaction of adaptive filtering, identification and control. *IEEE Signal Processing Magazine*, (12), 1995.
- [JP85] J. Juang and R. Pappa. An eigensystem realization algorithm for modal parameter identification and model reduction. Journal of Guidance, Control and Dynamics, 8:620-627, 1985.
- [Kai80] T. Kailath. *Linear Systems*. Prentice Hall, New Jersey, 1980.
- [Kal60] R. E. Kalman. A new approach to linear filtering and prediction problems. Journal of Basic Engineering, Transactions ASME, 82, Ser. D(1):35-45, 1960.
- [Kau52] William H. Kautz. Network synthesis for specified transient response. Technical Report 209, Massachusetts Institute of Technology, Research Laboratory of Electronics, 1952.
- [KB61] R. E. Kalman and R. S. Bucy. New results in linear filtering and prediction theory. Journal of Basic Engineering, Transactions ASME, 83, Ser. D(3):95-108, 1961.
- [KP77] R.E. King and P.N. Paraskevopoulos. Digital laguerre filters. Circuit Theory and Applications, 5:81-91, 1977.

[KP79]	R.E. King and P.N. Paraskevopoulos. Parametric identification of discrete time SISO systems. International Journal of Control, 30(6):1023-1029, 1979.
[Kun78]	S. Y. Kung. A new identification method and model reduction al- gorithm via singular value decomposition. In <i>Proceedings of the</i> 12th. Asilomar Conference on Circuits, Systems and Comput- ers, pages 705-714, Pacific Grove, CA, 1978.
[Lar90]	Wallace E. Larimore. Canonical variate analysis in identification, filtering, and adaptive control. In <i>Proceedings of the 29th Conference on Decision and Control</i> , pages 596–604, Honolulu, Hawaii, December 1990. IEEE.
[Lar94]	Wallace E. Larimore. The optimality of canonical variate identifi- cation by example. In <i>Proceedings of SYSID'94</i> , volume 2, pages 151–156, Copenhagen, Denmark, July 1994. IFAC.
[Lee60]	Yuk Wing Lee. <i>Statistical Theory of Communication</i> . John Wiley & Sons, Inc., New York, 1960.
[LG90]	Lennart Ljung and Svante Gunnarsson. Adaptation and tracking in system identification – A survey. <i>Automatica</i> , 26(1):7–21, 1990.
[Liu92]	K. Liu. Identification of Multi-Input and Multi-Output systems by observability range space extraction. In <i>Proceedings of the 31st.</i> <i>Conference on Decision and Control</i> , pages 915–920, Tucson, Ari- zona, USA, 1992.
[Lju85]	L. Ljung. Asymptotic variance expressions for identified black-box transfer function models. <i>IEEE Transactions on Automatic Control</i> , AC-30(9):834-844, 1985.
[Lju87]	L. Ljung. System Identification: Theory for the User. Prentice-Hall, Inc., New Jersey, 1987.
[Lju91]	L. Ljung. Identifying state space models with Mathwork's System Identification Toolbox. In Proceedings of 9th IFAC-IFORS Sym- posium on Identification and System Parameter Estimation, pages 1582–1584, Budapest, Hungary, 1991. IFAC.
[Lju95]	L. Ljung. System Identification Toolbox User's Guide. The Math- Works, Inc., 24 Prime Park Way, Natick, MA 01760-1500, 1995.
[LS91]	K. Liu and R. E. Skelton. Identification and control of NASA's ACES structure. In <i>Proceedings of the American Control Con-</i> ference, pages 3000–3006, Boston, MA, USA, 1991.

- [LS92] K. Liu and R. E. Skelton. Identification of linear systems from their impulse responses. In Proceedings of the American Control Conference, pages 1243-1247, 1992.
- [LW93] P. Lindskog and B. Wahlberg. Application of Kautz models in system identification. In *Proceedings of the 12th IFAC World Congress*, volume 5, pages 308-312, Sydney, Australia., July 1993.
- [LY85] L. Ljung and Z. D. Yuan. Asymptotic properties of black-box identification of transfer functions. *IEEE Transactions on Automatic Control*, AC-30(6):514-530, 1985.
- [Mäk90a] P.M. Mäkilä. Approximation of stable systems by Laguerre filters. Automatica, 26(2):333-345, 1990.
- [Mäk90b] P.M. Mäkilä. Laguerre series approximation of infinite dimensional systems. *Automatica*, 26(6):985–995, 1990.
- $[M\"ak91] P.M. M\"akil\"a. Laguerre methods and <math>H^{\infty}$ identification of continuous-time systems. International Journal of Control, 53(3):689-707, 1991.
- [Mat94] The MathWorks, Inc., 24 Prime Park Way, Natick, Mass. 01760-1500. MATLAB- High-Performance Numeric Computation and Visualization Software, User's Guide, 1994.
- [McK94] Tomas McKelvey. On state-space models in system identification. Technical Licenciate Thesis, Department of Electrical Engineering, Linköping University, S-581 83 Linköping, Sweden, May 1994.
- [McK95] Tomas McKelvey. Identification of State-Space Models from Time and Frequency Data. PhD thesis, Department of Electrical Engineering, Linköping University, S-581 83 Linköping, Sweden, 1995.
- [MJM89] D. F. Marshall, W. K. Jenkings, and J. J. Murphy. The use of orthogonal transforms for improving performance of adaptive filters. *IEEE Transactions on Circuits and Systems*, 36(4):474-484, April 1989.
- [MR93] M. Moonen and J. Ramos. A subspace algorithm for balance state space system identification. *IEEE Transactions on Automatic* Control, 38:1727-1729, 1993.
- [NG94a] Brett Ninness and Fredrik Gustafsson. A unifying construction of orthonormal bases for system identification. In *Proceedings of the*

33rd IEEE Conference on Decision and Control, pages 3388–3393, December 1994.

- [NG94b] Brett Ninness and Fredrik Gustafsson. A unifying construction of orthonormal bases for system identification. Technical Report EE9432, Department of Electrical and Computer Engineering at the University of Newcastle, August 1994.
- [NG96] B.M. Ninness and J.C. Gómez. Asymptotic analysis of MIMO system estimates by the use of orthonormal bases. In Proceedings of the 13th IFAC World Congress, volume I, pages 363-368, San Francisco, California, June 30-July 5 1996.
- [NG97] Brett Ninness and Fredrik Gustafsson. A unifying construction of orthonormal bases for system identification. *IEEE Transactions* on Automatic Control, 42(April):515-521, 1997.
- [NHG96] B. Ninness, H. Hjalmarsson, and F. Gustafsson. Bias and variance analysis of estimation using fixed denominator model structures. Technical Report EE9617, Department of Electrical and Computer Engineering, The University of Newcastle, Callaghan, NSW 2308, Australia, 1996.
- [NHG97a] B. Ninness, H. Hjalmarsson, and F. Gustafsson. The fundamental role of general orthonormal bases in system identification. Technical Report EE9739, Department of Electrical and Computer Engineering, The University of Newcastle, Callaghan, NSW 2308, Australia, 1997. Submitted to IEEE Transactions on Automatic Control.
- [NHG97b] B. Ninness, H. Hjalmarsson, and F. Gustafsson. Generalised Fourier and Toeplitz results for rational orthonormal bases. Technical Report EE9740, Department of Electrical and Computer Engineering, The University of Newcastle, Callaghan, NSW 2308, Australia, 1997. Submitted to SIAM Journal on Control and Optimization.
- [Nin96] Brett Ninness. Integral constraints in the accuracy of least squares estimation. *Automatica*, 32(3):391–397, 1996.
- [NnGW95] B. Ninness, J.C. Gómez, and S. Weller. MIMO system identification using orthonormal basis functions. In Proceedings of the 34th IEEE Conference on Decision and Control, pages 703-708, New Orleans, December 1995.
- [Nur87] Yu. Nurges. Laguerre models in problems of approximation and identification of discrete systems. *Automation and Remote Con*-

trol, pages 346–352, 1987. Pulished by Plenum Publishing, Translated from Avtomatica i Telemekhanika (3) pp 88-96, March 1987.

- [Oli94a] T. Oliveira e Silva. Kautz Filters. Technical report, INESC /Aveiro, Universidade de Aveiro, 3800 Aveiro, Portugal, September 1994.
 English translation of a work written in Portuguese for the "Prémio Científico IBM 94".
- [Oli94b] T. Oliveira e Silva. Optimality conditions for truncated Laguerre networks. *IEEE Transactions on Signal Processing*, 42(9):2528– 2530, September 1994.
- [Oli95a] T. Oliveira e Silva. Laguerre filters An introduction. Revista do DETUA, 1(3):237-248, January 1995.
- [Oli95b] T. Oliveira e Silva. On the determination of the optimal pole position of Laguerre filters. *IEEE Transactions on Signal Processing*, 43(9):2079-2087, September 1995.
- [Oli95c] T. Oliveira e Silva. Optimality conditions for Laguerre Lattice Filters. *IEEE Signal Processing Letters*, 2(5):97–98, May 1995.
- [Oli95d] T. Oliveira e Silva. Optimality conditions for truncated Kautz networks with two periodically repeating complex conjugate poles. *IEEE Transactions on Automatic Control*, 40(2):342-346, February 1995.
- [Oli95e] T. Oliveira e Silva. Rational orthonormal functions on the unit circle and on the imaginary axis, with applications in system identification. Technical Report ver. 1.0a, Automatic Control Group, Royal Institute of Technology (KTH), Stockholm, Sweden, October 1995.
- [OS89] A.V. Oppenheim and R.W. Schafer. Discrete Time Signal Processing. Prentice Hall, New Jersey, 1989.
- [OV94] B. Ottersten and M. Viberg. A subspace based instrumental variable method for state-space system identification. In M. Blanke and T. Söderström, editors, Proceedings of the 10th IFAC Symposium on System Identification SYSID'94, volume 2, pages 139-144, Copenhagen Denmark, July 1994. IFAC, Danish Automation Society.
- [Pap84] A. Papoulis. Probability, Random Variables, and Stochastic Processes. McGraw Hill, London, 1984.
- [Par91] Jonathan R. Partington. Approximation of delay systems by Fourier-Laguerre series. Automatica, 27(3):569-572, 1991.

- [Pdd93] Jose C. Principe, Bert de Vries, and Pedro G. de Oliveira. The gamma filter – A new class of adaptive IIR filters with restricted feedback. IEEE Transactions on Signal Processing, 41(2):649– 656, February 1993.
- [PM92] J. G. Proakis and D. G. Manolakis. Digital Signal Processing: Principles, Algorithms, and Applications. Macmillan Publishing Company, New York, 2nd edition, 1992.
- [PS82] Lars Pernebo and Leonard Silverman. Model reduction via balanced state space representations. *IEEE Transactions on Automatic* Control, AC-27(2):382-387, April 1982.
- [PT91] Hector Perez and Shigeo Tsujii. A system identification algorithm using orthogonal functions. *IEEE Transactions on Signal Pro*cessing, 39(3):752-755, March 1991.
- [Reg95] P. A. Regalia. Adaptive IIR Filtering in Signal Processing and Control, volume 90 of Electrical Engineering and Electronics. Marcel Dekker, Inc., New York, 1995.
- [RJL96] Z. Reznic, C. R. Johnson Jr., and F. López de Victoria. Frequency domain interpretation of LMS convergence of a fractionally spaced equalizer. *IEEE Signal Processing Letters*, (3):206-208, 1996.
- [RM87] R. A. Roberts and C. T. Mullis. Digital Signal Processing. Addisson-Wesley Publishing Company, Reading, Massachusetts, 1987.
- [Ros85] S. M. Ross. Introduction to Probability Models. Academic Press, Orlando, Florida, 3rd. edition, 1985.
- [Rud74] W. Rudin. Real and Complex Analysis. McGraw Hill, New York, 1974.
- [SK95] V. Solo and X. Kong. Adaptive Signal Processing Algorithms. Prentice-Hall Inc., New Jersey, 1995.
- [SS89] T. Söderström and P. Stoica. System Identification. Prentice Hall, New York, 1989.
- [Sze59] G. Szegö. Orthogonal Polynomials, volume 23 of Colloquium Publications. American Mathematical Society, New York, 1959.
- [Vd91a] Peter Van Overschee and Bart de Moor. Subspace algorithms for the stochastic identification problem. In *Proceedings of the 30-th*

IEEE Conference on Decision and Control, Brighton, England, pages 1321–1326, December 1991.

- [VD91b] M. Verhaegen and E. Deprettere. A fast, recursive MIMO state space identification algorithm. In *Proceedings of the 30th Conference* on Decision and Control, pages 1349–1354, Brighton, England, December 1991.
- [VD92] M. Verhaegen and P. Dewilde. Subespace model identification, Part
 I: the output error state space model identification class of algorithms. International Journal of Control, 56:1187-1210, 1992.
- [Vd93] Peter Van Overschee and Bart de Moor. N4SID: Numerical algorithms for state space subspace system identification. In Proceedings of the 12-th IFAC World Congress, volume 7, pages 361-364, Sydney, Australia, 1993.
- [Vd94a] Peter Van Overschee and Bart de Moor. N4SID: Subspace algorithms for the identification of combined deterministic-stochastic systems. *Automatica*, 30(1):75–93, 1994.
- [Vd94b] Peter Van Overschee and Bart de Moor. A unifying theorem for subspace identification algorithms and its interpretation. In M. Blanke and T. Söderström, editors, Proceedings of the 10th IFAC Symposium on System Identification SYSID'94, volume 2, pages 145– 150, Copenhagen Denmark, July 1994. IFAC, Danish Automation Society.
- [Vd95a] Peter Van Overschee and Bart de Moor. Choice os state-space basis in combined deterministic-stochastic subspace identification. Automatica, 31(12):1877-1883, 1995.
- [Vd95b] Peter Van Overschee and Bart de Moor. A unifying theorem for three subspace system identification algorithms. Automatica, 31(12):1853-1864, 1995.
- [Vd96] Peter Van Overschee and Bart de Moor. Subspace Identification for Linear Systems: Theory, Implementation, Applications. Kluwer Academic Publishers, Boston, 1996.
- [Ver91] M. Verhaegen. A novel non-iterative MIMO state space model identification technique. In Proceedings of the 9th IFAC-IFORS Symposium on Identification and System Parameter Estimation, pages 1453-1458, Budapest, Hungary, 1991.

- [Ver94] M. Verhaegen. Identification of the deterministic part of mimo state space models given in innovations form from input-ouput data. Automatica, Special Issue on Statistical Signal Processing and Control, 30(1):61-74, 1994.
- [VHB94a] P.M.J. Van den Hof, P.S.C. Heuberger, and J. Bokor. Identification with generalized orthonormal basis functions - statistical analysis and error bounds. In M. Blanke and T. Söderström, editors, Proceedings of the 10th IFAC Symposium on System Identification SYSID'94, volume 3, pages 207-212, Copenhagen Denmark, July 1994. IFAC, Danish Automation Society.
- [VHB94b] P.M.J. Van den Hof, P.S.C. Heuberger, and J. Bokor. System identification with generalized orthonormal basis functions. In Proceedings of the 33rd IEEE Conference on Decision and Control, pages 3382-3387, December 1994.
- [VHB95] P.M.J. Van den Hof, P.S.C. Heuberger, and J. Bokor. System identification with generalized orthonormal basis functions. Automatica, 31(12):1821-1834, December 1995.
- [Vib94] Mats Viberg. Subspace methods in system identification. In M. Blanke and T. Söderström, editors, Proceedings of the 10th IFAC Symposium on System Identification SYSID'94, volume 1, pages 1-12, Copenhagen Denmark, July 1994. IFAC, Danish Automation Society.
- [Vib95] Mats Viberg. Subspace-based methods for the identification of linear time-invariant systems. *Automatica*, 31(12):1835–1851, 1995.
- [VOWL91] M. Viberg, B. Ottersten, B. Wahlberg, and L. Ljung. A statistical perspective on state-space modeling using subspace methods. In Proceedings of the 30-th IEEE Conference on Decision and Control, Brighton, England, pages 1337-1342, December 1991.
- [VOWL93] M. Viberg, B. Ottersten, B. Wahlberg, and L. Ljung. Performance of subspace based state space system identification methods. In *Proceedings of the IFAC International Symposium*, volume 7, pages 369-372, 1993.
- [VWO97] M. Viberg, B. Wahlberg, and B. Ottersten. Analysis of state space system identification methods based on instrumental variables and subspace fitting. Automatica, 33(9):1603-1616, 1997.

[WAH]

- [Wah91a] B. Wahlberg. Identification of resonant systems using Kautz filters. In Proceedings of the 30th Conference on Decision and Control, pages 2005-2010, Brighton, England, December 1991.
- [Wah91b] B. Wahlberg. System identification using Laguerre models. *IEEE* Transactions on Automatic Control, AC-36(5):551-562, 1991.
- [Wah94a] B. Wahlberg. Laguerre and Kautz models. In M. Blanke and T. Söderström, editors, Proceedings of the 10th IFAC Symposium on System Identification SYSID'94, volume 3, pages 1-12, Copenhagen Denmark, July 1994. IFAC, Danish Automation Society.
- [Wah94b] B. Wahlberg. System identification using Kautz models. IEEE Transactions on Automatic Control, AC-39(6):1276-1282, June 1994.
- [Wal32] J. L. Walsh. On interpolation and approximation by rational functions with preassigned poles. *Transactions of the American Mathematical Society*, 34:22-74, 1932.
- [Wal35] J. L. Walsh. Interpolation and Approximation by Rational Functions in the Complex Domain, volume XX of Colloquium Publications. American Mathematical Society, 1935.
- [WH93] B. Wahlberg and E. J. Hannan. Parametric signal modelling using Laguerre filters. The Annals of Applied Probabililty, 3(2):467-496, 1993.
- [Wil93a] Geoffrey A. Williamson. Globally convergent adaptive filters with infinite impulse responses. In Proceedings of the 1993 IEEE International Conference on Acoustic, Speech and Signal Processing, volume III, pages 543-546, Minneapolis, April 1993.
- [Wil93b] Geoffrey A. Williamson. Linear in the parameters identification for classes of systems. In Proceedings of the 32nd Conference on Decision and Control, pages 2607-2612, San Antonio, Texas, December 1993.
- [Wil95] Geoffrey A. Williamson. Tracking random walk systems with vector space adaptive filters. IEEE Transactions on Circuits and Systems II: Analog and Digital Signal Processing, 42(8):543-547, August 1995.
- [WS85] Bernard Widrow and Samuel D. Stearns. Adaptive Signal Processing. Signal Processing Series. Prentice-Hall, Inc., Englewood Cliffs, NJ, 1985.

[WZ96]	G. A. Williamson and S. Zimmermann. Globally convergent adap- tive IIR filters based on fixed pole locations. <i>IEEE Transactions</i> on Signal Processing, 44(6):1418-1427, June 1996.
[YH62]	T. Y. Young and W. H. Huggings. Discrete orthonormal exponen- tials. In <i>Proceedings of the National Electronic Conference</i> , vol- ume 18, pages 10–18, Chicago, October 1962.
[YL84]	Z.D. Yuan and L. Ljung. Black box identification of multivariable transfer functions: Asymptotic properties and optimal input design. <i>International Journal of Control</i> , 40:233-256, 1984.
[You84]	Peter Young. <i>Recursive Estimation and Time-Series Analysis.</i> Springer-Verlag, Berlin, 1984.
[You88]	N. Young. An Introduction to Hilbert Space. Cambridge University Press, Cambridge, 1988.
[ZBD88]	C. Zervos, P. R. Bélanger, and G. A. Dumont. On PID con- troller tuning using orthonormal series identification. <i>Automatica</i> , 24(2):165-175, 1988.
[ZD88]	C.C. Zervos and G.A. Dumont. Deterministic adaptive control based on Laguerre series representation. <i>International Journal of Con-</i> <i>trol</i> , 48(6):2333–2359, 1988.
[ZDP90]	C. C. Zervos, G. A. Dumont, and G. L. Pageau. Laguerre-based adaptive control of pH in an industrial bleach plant extraction stage. <i>Automatica</i> , 26(4):781–787, 1990.
[Zhu89]	Y.C. Zhu. Black box identification of MIMO transfer functions: Asymptotic properties of prediction error models. International Journal of Adaptive Control and Signal Processing, 3:357-373, 1989.
[Zhu90]	Y.C. Zhu. Identification and Control of MIMO Industrial Pro- cesses: An Integration Approach. PhD thesis, Technische Univer- siteit Eindhoven, The Netherlands, February 1990.
[Zhu94]	Y.C. Zhu. A frequency domain criterion for MIMO model or- der/structure selection. In M. Blanke and T. Söderström, editors, <i>Proceedings of the 10th IFAC Symposium on System Identifica-</i> <i>tion SYSID'94</i> , volume 2, pages 516-521, Copenhagen Denmark, July 1994. IFAC, Danish Automation Society.

BIBLIOGRAPHY

[ZM74] H. Zeiger and A. McEwen. Approximate linear realizations of given dimension via Ho's algorithm. IEEE Transactions on Automatic Control, 19:153, 1974.